

Di Gessa, Giorgio (2011) 'Active ageing' and health: an exploration of longitudinal data for four European countries. PhD thesis, London School of Hygiene & Tropical Medicine. DOI: https://doi.org/10.17037/PUBS.00682446

Downloaded from: http://researchonline.lshtm.ac.uk/682446/

DOI: 10.17037/PUBS.00682446

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$ 

Available under license: http://creativecommons.org/licenses/by-nc-nd/2.5/

# 'Active ageing' and health: An exploration of longitudinal data for four European countries

Giorgio Di Gessa Department for Population Studies London School of Hygiene & Tropical Medicine December 2011 Academic Year 2010-2011



Ph.D. Thesis, University of London Supervisors: Prof. Emily Grundy Dr Anne Jamieson

## ACKNOWLEDGMENTS

I would like to thank the Bloomsbury Consortium for providing me with funding to carry out this research. I hope that my work will prove useful and will prompt helpful discussion in the future. I would also like to thank my supervisors Professor Emily Grundy and Dr Anne Jamieson for their help and guidance. A special thanks to my advisory committee member Professor Bianca De Stavola, whose encouragement, invaluable support, advice and patience in clarifying my statistics-related problems were crucial in the production of this thesis. I would also like to thank Bianca for her help on a personal level in difficult times. Your support was much appreciated.

Finally I would like to thank both my family and my friends for trusting me just the way I am. Thanks for pushing me with an enormous dose of impudent optimism and uncontainable passion. You all know how to regenerate and rejuvenate me. Thanks for your disarming patience and both your cynicism and encouragement that keep my feet 'almost' on the ground. You all taught me that there is not much of a hurry to become 'mature and wise'. "There's a lot to be said about the freshness of youth ... Great strength is often born of vulnerability, new vision from unwearied eyes". Words cannot express my gratitude to my friends (you know who you are, no need to mention names!): you contributed your time, energy, vision, passion and support. Chatting over coffees and beers (mainly about anything but this research study) has kept my sanity. Or at least I like to believe so. Thanks for keeping my spirits up and pushing me to learn the extent of my wings! I love you all. You are all family to me!

# ABSTRACT

'Active Ageing' has been promoted by the World Health Organisation (WHO) as a strategy for promoting the health and well-being of older people. Keeping active and involved in a range of activities not restricted to those associated with labour market participation may, it has been suggested, be beneficial for older people. In this research three domains of 'engagement' were considered: paid work, formal involvement (i.e. activities such as voluntary work, attendance at training courses and participation in political organisations) and informal involvement (i.e. activities such as providing care and help to family, and looking after grandchildren). Using the first two waves of the Survey of Health, Ageing and Retirement in Europe (SHARE) and the English Longitudinal Study of Ageing (ELSA), this thesis investigated both the cross-sectional association between socio-economic, demographic and health-related variables and engagement at baseline, and the longitudinal association between engagement at baseline and self-rated health (SRH) and depressive symptoms at follow-up (controlling for baseline measures of health). The analysis was based on sample members aged 50-69 at baseline in Denmark, France, Italy and England, countries selected to represent different welfare regimes.

Cross-sectional findings showed that levels of engagement in paid work and formal activities varied across countries, whereas socio-economic, demographic and health-related characteristics were similarly associated with engagement in all countries under study. This suggested that country-specific factors, such as retirement policies, might play an important role in determining older people's level of engagement in paid work. Cross-sectional results also suggested that work and formal engagement were associated with good health, whereas –among certain subpopulations –informal activities were associated with bad health. Longitudinal analyses showed that, in all countries, respondents in paid work at baseline were more likely to improve their SRH and less likely to become depressed than those who were 'inactive'. Formal and informal engagement were not significantly associated with health at follow-up. Longitudinal results and associations found, however, might have been biased by the high rates of attrition, as multiple imputation techniques and sensitivity analyses suggested.

The current research study confirms that engagement in work is an important pathway to health in late life. More attention, however, should be paid to people's working lives, the quality of work and work conditions as these may influence participation in, and withdrawal from, the labour market.

## **STATEMENT OF AUTHORSHIP**

The thesis titled "Active Ageing and Health: an exploration of longitudinal data for four European countries" has been submitted for the degree of Doctor of Philosophy. I, the undersigned, hereby declare that:

- I am the sole author of this thesis.
- I am solely responsible for all quantitative data management, including checks on internal consistency (within and between waves), data quality (validating data with external sources), item non-response, file merging and the development of new variables and indicators to capture longitudinal changes. I also carried out both the cross-sectional and the longitudinal analyses.
- I have read and understood the London School of Hygiene & Tropical Medicine definitions of plagiarism given in the Assessment Irregularity Procedures. I have fully and appropriately acknowledged and referenced the ideas and work of others, whether published or unpublished, in my thesis.
- I have prepared my thesis specifically for the degree of Doctor of Philosophy while under supervision by Professor Emily Grundy (London School of Hygiene and Tropical Medicine) and Anne Jamieson (Reader at Birkbeck, University of London). My two supervisors defined the broad topic area and secured funding for my work from the Bloomsbury Consortium.
- My thesis does not contain work extracted from a thesis, dissertation or research paper previously presented for another degree or diploma at this or any other university.

Signatures Candidate

Gene

Supervisors

# **Table of Contents**

Ackn	owled	lgments	i
Absti	ract		ii
State	ment	of Authorship	<b>ii</b> i
1. II	NTRO	DDUCTION	1
1.1	Aim	and Context	1
1.2	Stru	icture of Thesis	2
1.3	Ethi	ical Clearance	3
2. T	`he 'A	ctive Ageing' Manifesto	4
2.1	<b>'Ac</b>	tive ageing' policies – a focus on the WHO definition	4
2.2	The	'active ageing' concept: its historical evolution and intertwining with ag	eing
theo	ories		7
2.3	Wh	y engagement and health are connected	13
2.	.3.1	Empirical evidence	15
2.4	Lim	itations of previous studies	22
2.5	The	current study	23
2.	.5.1	Aim and objectives of the current study	25
3. V	Velfai	e regimes and their influence on policies in four Europea	in
coun	tries.		28
3.1	The	• 'historical' approach to country differences	28
3.2	Esp	ing-Andersen's Three worlds of Welfare Capitalism	30
3.3	Cri	tiques of Esping-Andersen's typology	32
3.4	We	lfare regimes: still a valuable instrument to explain country differences	35
3.	.4.1	A sensitive way to make use of a welfare regimes typology	36
3.5	Pen	sions and Retirement policies	42
3.	.5.1	DENMARK	42
3.	.5.2	FRANCE	43
3.	.5.3	ITALY	45
3.	.5.4	UK	47
3.6	Сог	iclusions	48
4. F	Popula	ation Ageing in Europe: an overview	50

4.1	Age	eing as a global phenomenon	50
4.2	'Ch	allenges' related to an ageing population	51
4.3	Age	eing in Italy, France, Denmark and England	52
4.	3.1	Differences within similar trends	52
4.	3.2	Current populations	55
4.	3.3	Future trends	56
4.4	Co	nclusions	61
5. E	LSA	and SHARE questionnaires: an overview	63
5.1	Sor	ne background rationale	63
5.2	Bri	ef history of both ELSA and SHARE questionnaires	64
5.3	Ins	trument Design and Implementation	65
5.	.3.1	ELSA	65
5.	.3.2	SHARE	69
5.4	Lo	ngitudinal Survey Participation for ELSA and SHARE	71
5.5	Str	ucture and content of the ELSA and SHARE questionnaire	74
5.	.5.1	ELSA questionnaire	74
5.	.5.2	SHARE questionnaire	76
5.6	We	ighting Strategies	78
5.7	Str	engths and weaknesses of ELSA and SHARE	79
6. E	ELSA	and SHARE: data quality and some descriptive analyses	83
6.1	Ini	tial response rate, nonresponse bias and data quality	83
6.2	De	scriptive Statistics and assessment of data quality	86
6.3	So	cio-Demographic characteristics	86
6.	.3.1	Age and sex composition	<b>8</b> 6
6	.3.2	Education	89
6	.3.3	Marital status	<b>9</b> 0
6	.3.4	Household composition and other characteristics	92
6.4	En	ployment and Socio-Economic Position	94
6	.4.1	Employment Status	94
6	.4.2	Household Income and Wealth	99
6.5	He	alth status and its measurement	100
6	5.1	Self-Rated Health (SRH)	102
6	5.5.2	Chronic Diseases	105

6.5.3	Physical Functioning	108
6.5.4	Mental Health	114
6.6	Formal, Informal and Family Involvement	116
6.6.1	Activities in SHARE	117
6.6.2	Activities in ELSA	122
6.7	Conclusion	124
7. Ho	usehold and Personal Characteristics of older people who	
engage	in work, formal, informal and family activities	. 126
7.1	Activities and their determinants	126
7.1.1	Labour market participation	127
7.1.2	Formal, Informal and Family participation	128
7.1.3	Accumulation or competition between participation in formal or inform	nal
activ	vities and employment status	129
7.2	Methods	129
7.2.1	Sample	129
7.2.2	2. Measures	130
7.2.3	Statistical Modelling	137
7.3	Results	139
7.3.1	Logistic regression: analysis of proportions in paid work	139
7.3.2	2 Logistic regressions for formal activities	146
7.3.3	<b>B</b> Logistic regressions for being involved in informal and family activities	151
7.4	Conclusions	159
8. Att	rition in ELSA and SHARE: who is missing at wave 2	. 162
8.1	Attrition in Longitudinal Study. An introduction	162
<b>8.1.</b> ]	l Potential reasons	162
8.1.2	2 Missingness mechanisms	163
8.2	Attrition in ELSA and SHARE	164
8.3	Attrition Rates in ELSA and SHARE	164
8.4	Determinants of Attrition	165
8.4. Biva	Respondent characteristics and differential non-response in wave 2; ariate analyses	165
8.4.2 Mu	2 Respondent characteristics and differential non-response in wave 2; ltivariate model of response to main interview	171

8.5	Conclusions	
9. Lo	ngitudinal Association between Baseline engagement, an	nd self-
percei	ved health and depression at follow-up	175
9.1	Health and its determinants	176
9.1.	1 Self-rated Health	176
9.1.	2 Depression	177
9.2	Methods	177
9.2	1 Sample	177
9.2	2 Measures	177
9.2	3 Statistical Modelling	
9.3	Results from the multivariate analyses for the "complete record"	
9.3	1 Self-rated health at follow up controlling for baseline Self-rated	h <b>ealth</b> 189
9.3	2 Depression at follow up controlling for baseline depression	194
9.3	3 SRH and Depression <i>changes</i> over time	
9.4 MAR	Results from the multivariate analyses of the multiple 'imputed datas and NMAR.	ets' under 
9.4	1 Self-Rated Health	
9.4	2 Depression	
9.4	3 Sensitivity Analysis	
9.5	Conclusions	
10. Co	onclusions and Discussion	
10.1	Main findings	211
10. eng	1.1 Cross-sectional associations of Older People's characteristics wire agement, And Relationships between different forms of activities	<b>th</b> 211
10.	1.2 Longitudinal Association between Baseline Engagement and He	alth at
fol	ow-up	215
10.	1.3 Sample Attrition and Its Impact	216
10.2	Strength of study	
10.3	Limitations	
10.4	Implications of findings	
10.5	Future research	
Biblio	graphy	
Аррен	ndix	

# List of Tables

Table 3.1 Percentage of people in naid work, by age-group and sex $= 2005$ 39
Table 3.2 Percentage of women in paid work by country and specific groups – 2005
Table 3.3 Percentage of workers working part-time, by country sex and age-group – 2005. 40
Table 3.4 Total Social Protection expenditures as % of GDP by country 2005
Table 3.5 Social Cohesion Indicators by country 2010 41
Table 3.6 Percentage of people older than 65 who receive long-term care in institutions and
percentage of 65+ who live alone, by country and gender
Table 4.1 Life Expectancy at Birth and at 65 (in years). Comparisons between periods, selected
countries
Table 4.2 Population by broad age-groups (% distribution), 2006
Table 4.3 Size and Age structure of the older population in selected countries, 2006
Table 4.4 Age-Sex Distribution of the older population, 2006
Table 4.5 Projection of the proportion of the population aged under 15 (%)
Table 4.6 Projection of the proportion of the population aged over 65 (%)
Table 4.7 Completed Total Fertility Rate by birth cohorts, selected years
Table 4.8 Percentage of women childless at age 45 by birth cohort, various countries
Table 5.1 Achieved sample by age and sex, ELSA (wave 1)
Table 5.2 Distribution of Respondents by country, sex and age. Individual and household
response rates (RR)71
Table 5.3 Distribution of wave 1 SHARE respondents successfully re-interviewed at wave 2, by
country
Table 5.4 Overview of Content of ELSA Questionnaire (wave 1)75
Table 5.5 Overview of Content of SHARE Questionnaire (wave 1)77
Table 6.1 Response Rate of SHARE and other European Surveys85
Table 6.2 Distribution of Respondents by country, sex and age
Table 6.3 Distribution of the ELSA and SHARE respondents by age group and sex (with and
without weights); comparisons with the census data
Table 6.4 Level of Education (ISCED-97) by gender, age-group and country
Table 6.5 Distribution of legal marital status, by gender, age-group and country91
Table 6.6 Distribution of living arrangements, by age, sex and country
Table 6.7 Distribution of respondents living with at least one child in the same household (%)
by age group, gender and country
Table 6.8 Distribution of self-reported current economic status, by sex and country95
Table 6.9 Percentages of the SHARE and ELSA respondents 'in paid work', by gender and age
(<70). Comparisons with OECD statistics
Table 6.10 Percentage of mortgage-free house ownership, by age and sex and country 100
Table 6.11 Distribution of Self-Reported Health, by sex, age-group and country104
Table 6.12 Percentage distribution of number of chronic illnesses, by sex age-group and country
Table 6.13 Difficulty with mobility and upper-limb functions, by age-group, sex and country
(%)

Table 6.14 Difficulty with Activities of daily living (ADLs) by sex, age-group and country (%) Table 6.15 Distribution of number of difficulties with Instrumental Activities of daily living Table 6.16 Percentage distribution of INSEE and SHARE respondents who reported at least Table 6.17 Percentage distribution of ISTAT and SHARE respondents who reported at least one Table 6.18 Percentage of respondents reporting three or more depressive symptoms, by gender, Table 6.22 Activities and Organisational memberships, by age-group and sex –England...... 123 Table 7.1 Percentage of people aged 50-69 in paid work, by age-group, sex and country ......134 Table 7.2 Percentage of people aged 50-69 involved in formal activities, by age-group, sex and Table 7.3 Percentage of people aged 50-69 involved in informal and family-related activities, by age-group, sex and country......135 Table 7.4 Percentage distribution of the engagement in the combination of the three forms of Table 7.5 Participation in Paid Work -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions with legal marital status, by country and sex ...... 142 Table 7.6 Participation in Paid Work –Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions, by country and sex. Selected results obtained when presence Table 7.7 Participation in Paid Work -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions, by country and sex. Selected results obtained when Table 7.9 Participation in Paid Work - Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions with and without education and considering the interaction term between education and country -all SHARE countries considered all together Table 7.10 Result Summary in terms of odds ratios of being in paid work considering the interaction terms between country and level of education, by gender - SHARE respondents .. 146 Table 7.11 Participation in Formal Activities --Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions, by country and sex ......149 Table 7.12 Participation in Formal Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions with the 'living alone' indicator and dummy Table 7.13 Participation in Formal Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions when all SHARE countries are grouped  Table 7.14 Participation in Informal and Family-related Activities --Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions, by country and sex 155 Table 7.15 Participation in Informal and Family-related Activities --Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions, by country and sex. Table 7.16 Participation in Informal and Family-related Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions, by country and sex. Selected results obtained when the variable indicating whether the respondent lived alone was Table 7.17 Participation in Informal and Family-related Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions with the inclusion of dummy indicators of engagement in the other two forms of activity, by country and sex......157 Table 7.18 Participation in Informal and Family-related Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted logistic Regressions when all SHARE Table 8.3 Attrition Rates for Wave 1 respondents by a 'living alone' indicator, marital status and Table 8.4 Attrition Rates for Wave 1 respondents by selected health indicators, country and Table 8.5 Attrition Rates for Wave 1 respondents by 'participation' in the three forms of Table 8.6 Participation in wave 2 of the survey -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions analysis including 'living alone' indicator and Table 9.1 Percentage distribution of respondents by self reported health at baseline and change Table 9.3 Percentage distribution of changes in 'depression' between wave 1 and 2, by sex and Table 9.4 Percentage distribution of changes in 'paid work status' between wave 1 and 2, by sex Table 9.5 Percentage distribution of changes in 'formal engagement' between wave 1 and 2, by sex and country, by sex and country ......185 Table 9.6 Distribution of changes in 'informal and family-related engagement' between wave 1 Table 9.7 Association between engagement at baseline and SRH at follow-up, controlling for baseline SRH. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Table 9.8 Association between changes in levels of engagement over time and SRH at followup, controlling for baseline SRH. Log Odds Ratios and 95% Confidence Intervals obtained from 

Table 9.9 Association between engagement at baseline and depression at follow-up, controlling for baseline depression. Log Odds Ratios and 95% Confidence Intervals obtained from fully Table 9.10 Association between changes in levels of engagement over time and depression at follow-up, controlling for baseline depression. Log Odds Ratios and 95% Confidence Intervals Table 9.11 Association between engagement at baseline and changes (for the better) in SRH over time. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted ordinal Table 9.12 Association between engagement at baseline and changes in depression over time. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted ordinal logistic Table 9.13 Association between engagement at baseline and changes (for the better) in SRH Table 9.14 Association between engagement at baseline and depression at follow-up, controlling Table 9.15 Association between engagement at baseline and depression at follow-up, controlling Table 9.16 Association between engagement at baseline and depression at follow-up, controlling for baseline depression. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted logistic regression with imputed datasets under MNAR (assuming depression among 

# **List of Figures**

Figure 4.1 Total Fertility Rates, 1950-2006 – selected countries	53
Figure 4.2 Projections of Age-Sex Pyramids in 2020 and 2035	57
Figure 5.1. ELSA sample frame definition tree	67
Figure 5.2 Response tree diagram	68
Figure 5.3 Responses across ELSA waves 1-3 (only Cohort 1 core members)	72
Figure 5.4 Attrition Rates for Wave 1 respondents	73
Figure 6.1 Self-Reported Economic Activity by age and country	97
Figure 8.1 Attrition Rates for Wave 1 Respondents (%), by gender and country	. 166
Figure 8.2 Attrition Rates for Wave 1 Respondents by education, gender and country (%)	. 167
Figure 8.3 Attrition Rates for Wave 1 Respondents by area of residence, gender and country	168

# 1. INTRODUCTION

### **1.1 AIM AND CONTEXT**

This study intends to determine the socio-economic and demographic characteristics associated with engagement of older people in four European countries and to assess whether activities older people engage in are complementary or whether they substitute for each other. It also aims to assess whether their level of engagement varies by country. This cross-national comparative individual- level data approach, however, focuses mainly on whether older people engage differently across countries because of differences in their personal and household characteristics, without studying in details specific policies or programmes. Moreover, this thesis analyses whether engagement at baseline is associated with health at follow-up (once baseline health is controlled for), helping us determine whether similar forms of engagement play a similar role in different countries. Finally, this study investigates whether such longitudinal associations might be biased by sample attrition.

Consistently low fertility levels combined with extended longevity have resulted in demographic ageing of the European population. Population ageing raises concerns about the social and economic changes associated with it, yet older people can also make positive contributions to society. The World Health Organization (WHO) adopted in 2002 the term 'active ageing' to express the process for achieving the vision of longer life accompanied by continuing opportunities for health and participation. 'Active ageing' has nowadays achieved widespread currency, largely thanks to the efforts of the WHO: this emphasises the connection between a broader range of activities than those normally associated with production and the labour market and emphasises the participation and inclusion of older people as full citizens. Extending participation beyond the labour market has the potential to help ensure that more people continue to play a full and active role in the society, with a strong emphasis on mental and physical well-being. WHO suggests activity as a strategy for the preservation of health as people age (Walker, 2002, WHO, 2002).

Several cross-sectional studies suggest that older people who are active have better health than those less engaged. However, health status may affect participation in various activities and this, rather than beneficial effects of activity, may account for reported associations between activity and health. Moreover, since 'active ageing' is a relative newcomer in Europe, comparative studies are lacking in order to assess both whether the level of engagement in various activities and factors associated with it vary across countries and whether activity is indeed longitudinally positively associated with health equally across these different settings.

### **1.2 STRUCTURE OF THESIS**

The following chapter (Two) provides an overview of the study's conceptual framework and assesses arguments and evidence around the association between activity and health. Even though European countries are facing similar demographic challenges (as described in chapter Four), chapter Three will discuss how retirement and social policies continue to differ in different European welfare state regimes, comparing the four countries that are the focus of this study, namely Denmark, France, Italy and England.

This study used data drawn from the first two waves of two parallel European longitudinal surveys, the Surveys of Health and Retirement in Europe (SHARE) and the English Longitudinal Study of Ageing (ELSA). The rationale, designs and implementations of both surveys are briefly described in chapter Five. Chapter Six presents some descriptive statistics and quality checks on socio-economic, demographic and health characteristics of the surveys' respondents, focusing on the four European countries under study.

There are three quantitative results chapters. Chapter Seven presents baseline crosssectional findings of level of engagement among cohort participants aged 50-69 at baseline in each of the countries under study, separately for men and women. Three domains of engagement were considered: paid work, formal involvement (i.e. activities such as voluntary work, attendance at training courses and participation in political or religious organisations) and informal involvement (i.e. activities such as providing care and help to family, friends and neighbours and looking after grandchildren). Chapter Eight explores attrition and analyses patterns of missing data using simple logistic regression. Chapter Nine presents longitudinal analyses in order to better understand whether engagement is beneficial for older people's health. This chapter uses the first two waves of both ELSA and SHARE, focusing on self-rated health and symptoms of depression. Chapter Nine also explores multiple imputations techniques to take account of attrition in the surveys.

The discussion chapter (Ten) seeks to bring together the findings from the crosssectional and longitudinal analyses, and assesses the strengths and limitations of this work, the implications of the study findings and the possible influence of policy context.

### **1.3 ETHICAL CLEARANCE**

Ethical approval for this work was provided by the London School of Hygiene & Tropical Medicine Ethics Committee.

# 2. THE 'ACTIVE AGEING' MANIFESTO

Being actively engaged with wider society has been linked with well-being and good health in later life. This chapter introduces the recent concept of 'active ageing' which has broadened the idea of engagement by including a wide range of activities other than just paid work; the theoretical background explaining this association is also explored. Evidence shows that a positive association between 'engagement' in different forms of activities and health exists. Most previous studies, however, are not fully comparable with each other: very specific samples of older people have been investigated and the lack of common measurements of activities and health makes the comparisons of results problematic. Moreover, very few studies tend to consider participation in its most inclusive and holistic definition and are limited to specific activities. Further investigation of the longitudinal association between engagement and health which is truly comparable between nationally-representative samples is therefore necessary to further knowledge on the importance of remaining active and to assess whether different contexts might affect either participation in various forms of activities or the direction of their association with health.

### 2.1 'ACTIVE AGEING' POLICIES – A FOCUS ON THE WHO DEFINITION

The concept of 'active ageing' has recently achieved widespread acknowledgement in Europe becoming a policy priority in the European Union (Commission of the EC, 1999a, b) and being promoted by a number of bodies such as the Organisation for Economic Co-operation and Development (OECD, 2000) and the World Health Organisation (WHO, 2002).

In particular, the World Health Organisation (WHO) identifies participation as one of the key pillars of the active ageing policy together with health and security (WHO, 2001a, 2002). Proponents believe that ageing well can be achieved by keeping fit, active and involved in both work and leisure activities in their own homes and communities. Extending participation beyond the labour market has the potential to help ensure that more people continue to play a full and active role in the society, especially in light of the fact that work participation is low among people over 65 (Eurostat, 2008). People who retire from work as well as those who are ill or have disabilities can still actively contribute to their families, peers and communities. The WHO active ageing policy framework broadens most of the pre-existing approaches, including all meaningful activities which are common among older people (Hutchison et al., 2006). "The word 'active' refers to continuing participation in social, economic, cultural, spiritual and civic affairs, not just the ability to be physically active or to participate in the labour force" (WHO, 2002)(pg. 12). Moreover, WHO explicitly accounts for the heterogeneity of older people and their individual life experiences, shifting the focus of ageing research from older people to the process of human development over the life-course. The importance of earlier life experiences as well as the complex interaction of a wide variety of factors such as education, employment and family life, material security and health are all acknowledged under the WHO approach (Ney, 2005). Such a holistic approach avoids the narrower previous terms for ageing well such as 'positive', 'productive', 'healthy' and 'successful' (Hutchison et al., 2006). Despite the fact that these approaches all consider ageing as a lifelong process rather than a static period of time, they mainly focus their attention on specific facets of such processes, such as the maintenance of health, the preservation of autonomy, engagement in activities which generate goods or services or the counterbalance of negative perceptions of ageing and retirement (Davey, 2002).

The WHO 'active ageing' framework should not be confused with the OECD and EU ones. By the early 2000s, the latter international bodies were also promoting policies labelled under the same terminology even though, de facto, they both adopted a 'productivist' approach (Davey, 2002, Ervik, 2006, Hutchison et al., 2006). OECD, in fact, defines 'active ageing' as "the capacity of people, as they grow older, to lead productive lives in society and the economy. This means that people can make flexible choices in the way they spend time over life – learning, working, and partaking in leisure activities and giving care" (OECD, 2000; pg. 126). According to the European commission, "active ageing is about adjusting our life practices to the fact that we live longer and are more resourceful and in better health than ever before, and about seizing the opportunities offered by these improvements. In practice it means adopting healthy

life styles, working longer, retiring later and being active after retirement" (Commission of the EC, 1999b, a). Both bodies, despite using the same terminology as WHO, stress particularly the promotion of the maintenance of a productive life as people grow older (Ervik, 2006, Hutchison et al., 2006) and approach ageing as a challenge rather than a conquest and an opportunity for all people, as the WHO clearly states (WHO, 2002). Moreover, the OECD narrows the core of attention down to the critical transition from work to retirement and to older people as a specific group instead of adopting a life-course based approach as the EU does (Avramov and Maskova, 2003, Ervik, 2006, Kasneci, 2007). Finally, both the OECD and the EU assume that older people are in the position of being able to make the 'right' choices to optimise their working lifespan and increase their level of activity, somehow also prescribing what good life is (Ervik, 2006, Hutchison et al., 2006).

The WHO, on the contrary, recognises that older people can find personal satisfaction and fulfilment through a variety of activities such as work and volunteering; taking care of their grandchildren and other family members and having positive relationships with friends or simply stimulating their minds through hobbies, travels or reading books. People can enjoy and experience an active life throughout their lives, given their personal capabilities, interests and preferences. WHO also promotes a more inclusive society without intergenerational conflicts, shifting the focus of policy from older people as a separate group to the entire society. Older people are an integral part of society and, as such, they have the same rights as the rest of the society to meet their needs for independence, equality, participation and security. Potential `activation' throughout life is argued to be beneficial not only for personal health, autonomy and self fulfilment but also contributes to the well being of other individuals and the larger society, promoting community participation and solidarity.

The WHO 'active ageing' umbrella is also not exempt from critiques. As with most terminology, faults in the semantics of the term have been pointed out. Just as successful implies a contest in which there are winners and losers –and therefore unsuccessful people (Strawbridge et al., 2002) –, the holistic and multifaceted 'active' term may further marginalise people who do not conform to images of vigorous, healthy and independent agers almost denying ageing to a certain extent (Westerhof and Tulle,

2007), whereas limits to 'activeness' do exist (Avramov and Maskova, 2003, Ervik, 2006). This new political discourse, moreover, still stresses too much the role that personal choices play, making older people the main actors responsible for their own health and life. Certain opportunities, such as access to education or medical care, depend largely upon the society rather than on individuals. Similarly, gender, economic and social differences exist and should not be ignored when framing policies and strategies (Ervik, 2006). Finally, if on the one hand WHO acknowledges the fact that not all older people will necessarily be active in the same way (activities may differ by country, culture or socio-economic characteristics), on the other keeping the definition of activity deliberately vague renders the concept –and in particular its operationalisation –more complicated and difficult to study.

## 2.2 THE 'ACTIVE AGEING' CONCEPT: ITS HISTORICAL EVOLUTION AND INTERTWINING WITH AGEING THEORIES

The WHO 'active ageing' vision was developed using a number of diverse theoretical assumptions and is the legacy of a complex interplay of policies, theories, historical and economic events. Even though this policy achieved widespread adherence in the late 1990s, its initial traces date back to the early 1960s and were developed in the 1970s and 1980s. The following section briefly presents the interwoven linkages between developments in gerontological theory, history and policies.

According to Phillipson and Baars (2007), 3 main phases characterised European and North-American perspectives on ageing. Between World War II and the 1960s, ageing became synonymous with retirement and was approached as an individual and social problem. In the 1970s and 1980s, ageing became viewed as mainly an economic and employment problem, whereas from the 1990s onwards ageing became a global challenge.

The period after World War II witnessed the emergence of the 'social problem perspective': old age was seen as a distinct phase of individuals' life characterised by inevitable functional decline, physical decay and capability loss. Old age represented both a demographic, economic and medical burden (Victor, 2005, Phillipson and Baars,

2007). During the 1950s and 1960s, ageing became synonymous with retirement which was assumed to be a critical but necessary turning point. Retirement involved an inevitable but, in some perspectives, a beneficial loss of social roles (Cumming and Henry, 1961): disengagement theorists envisaged that in order to maintain equilibrium within society older people should withdraw from society to preserve its energy, youthfulness and ideas. 'Disengagement theory' also argued that older people withdrawing from social participation would be relieved of the social responsibilities and pressures faced by younger adults, in a functional mutual separation<sup>1</sup>. Such gradual withdrawal was also believed to be the natural response to lessened capabilities, diminished interests and expectations as well as medical conditions and illnesses. Such theory –which viewed disengagement from society as a beneficial, natural, and desirable outcome for both the individual and the social system –spread rapidly also because of relatively shorter life expectancy, earlier onset of disability, physically demanding work roles and few organized activities for older adults (Victor, 2005, Künemund and Kolland, 2007, Phillipson and Baars, 2007).

The theory of disengagement was diametrically opposed by the so-called 'activity theory', developed by Havighurst (1963). According to 'activity theory', when people age they give up many roles which usually contribute to their life satisfaction and sense of identity. Therefore, when people retire from work, become widowed or drop out of professional organisations, it is believed that they need to engage themselves in other activities that develop substitute roles for those that have been abandoned (Roy and Russell, 2005). It was argued that the maintenance in old age of the activity patterns typical of middle age and the compensation for the loss of roles by taking on new activities were the key to successful ageing (Walker, 2002, Avramov and Maskova, 2003, Victor, 2005).

Despite their diametrical perspectives, both the activity and the disengagement theories believed that "growing older represented a significant degree of discontinuity from prior life events and experiences" (Phillipson and Baars, 2007)(pg. 72).

<sup>&</sup>lt;sup>1</sup> This argument has its roots in theories of functionalism developed by Parsons (1951). The functionalist approach "views the elements of society as being functionally interdependent, with the individual and society always seeking to maintain a state of equilibrium between them" (Victor, 2005)(pg. 18); different groups function and adapt in order to ensure that the society remains both unified and stable.

During the 1970s and 1980s, continuity theory (Atchley, 1971) surfaced as an important concept and a challenge to activity and disengagement theories. This theory assumed that humans tend to maintain patterns of their former lifestyle, their sense of self-esteem and their values throughout their lives, even after retirement from paid work. Thus success is where continuity in social roles, activities of importance to the individual or personal goals is achieved, including leisure activities. The surfacing of life-course perspectives which no longer considered old age as a distinct phase of life unrelated to previous experiences slowly pushed policies to also focus on those activities of importance to the individual which were not necessarily related to the production of valued goods and services (Künemund and Kolland, 2007), such as childcare, voluntary activities, informal help, time spent for personal development, cultural expression or (self) education as well as leisure activities such as travel, sports or hobbies. It was argued that older people can stay active not only through employment but also through participating in the community as well as through hobbies and leisure activities (Clarke et al., 2005).

In the same decades, a tendency to view life after retirement as an expanded 'active leisure' phase started to become popular. Neugarten (1974) first described an emerging group of older people –called the young-old –who were relatively healthier, more educated and affluent than the previous generations and who were to challenge the ageist conceptions of what was possible in later life. Laslett (1989) further developed this concept describing it as "Third Age"<sup>2</sup>. Laslett observed a series of demographic and sociological changes in contemporary Western societies and predicted the appearance of The Third Age, a new life-course stage. The emergence of retirement and pensions as a social institutions, the increase in longevity and health combined with the increased opportunities for leisure participation resulted in the great majority of older people in industrialized countries spending many healthy, active, and potentially self-fulfilling

<sup>&</sup>lt;sup>2</sup> Laslett represented the 'First Age' as the period of childhood dependency and education, the 'Second Age' as the time of independence, employment and maturity, the 'Third Age' as the period during which people are freed from work and family constraints and have time to pursue a good quality of life. Finally, he saw the 'Fourth Age' as characterised by dependence and declines in health.

years, i.e. the Third Age<sup>3</sup>. The Third Age was conceptualized as a new conceptual (rather than age-based)<sup>4</sup> life-course stage of extended and self-fulfilling leisure and community participation following retirement (Carr and Komp, 2011, Chatzitheochari and Arber, 2011). Later life began to be associated with many opportunities for remaining productive and engaging in valued social roles. While earlier approaches conceived later life as a period of frailty, dependency and disengagement, the Third Age theoretical strand presented a more positive perception of later life and ageing which became associated with personal growth and meaningful engagement. The Third Age is "the age of personal achievement and fulfilment", where people are freed from work and family obligations and have both time and material conditions to "attain their personal objectives" (Laslett, 1987)(pg 135). This new paradigm, however, did not emphasise social divisions among older people, who are neither socially nor economically homogeneous: material disparities, disadvantages experiences in previous life-course stages, and social factors (such as class and gender) are unlikely to create similar opportunities of engagement and self-satisfaction in later life (Phillipson, 2002a, Carr and Komp, 2011). Despite these critiques, the Third Age shifted the perception of old age introducing a new phase in life where people have opportunities for continuing participation in the society and can pursue their interests through a range of activities. This was reflected in the development of theories about productive and successful ageing.

'Productive ageing' (Butler and Gleason, 1985) soon became a term to describe what some saw a more positive approach to ageing (Walker, 2002), particularly in the late 1980s when consensus on the possible negative implications of an ageing population on health-care and pension expenditure was growing (Hutchison et al., 2006). It was no longer believed that a reduction of the labour supply through early withdrawal and incentives to disengagement started in the mid-1970s' as a response to high unemployment would be an appropriate strategy to tackle an ageing and declining

<sup>&</sup>lt;sup>3</sup> Until the first half of the 20<sup>th</sup> century, adults spent virtually all their lives in the Second Age, working and caring for family. They then entered the Fourth Age, a period of dependency and decrepitude prior to death.

<sup>&</sup>lt;sup>4</sup> "The divisions between the four ages do not come at birthdays, nor do they even lie within clusters of years surrounding birthdays" (Laslett, 1987)(pg 143).

workforce (Walker, 2001). Governments began to recognise the costs of failing to use older workers and started to incentivise older people to work longer or to continue to contribute to their communities through voluntary activities (Davey, 2002). It was argued that older people had both the physical and mental abilities to perform jobs or to participate more actively in society (Avramov and Maskova, 2003). Butler and Gleason (1985) defined productive ageing as "the capacity of an individual or a population to serve in the paid workforce, to serve in volunteer activities, to assist in the family and to maintain himself or herself as independently as possible" (pg. 73-74), irrespective of impairments. Bass and Caro (2001) offered a definition of productive ageing "as activities that produce goods and services, whether paid or not, or develops the capacity to produce goods or services" (pg. 39), restricting activities to those that can be quantified with some form of economic value. Third-agers were considered the ideal candidates for productive ageing activities, thanks to their good health (Komp, 2011).

During the 1990s, the strategy of 'activeness' started to embrace a wider concept other than the mere economic one (Hao, 2008). Social activity became a crucial component of the model of 'successful aging'. Rowe and Kahn (1997) defined "successful ageing as including three components: low probability of disease and disease-related disability, high cognitive and physical capacity, and active engagement with life" (pg. 433) whose two major and most important factors were 'interpersonal relations' and 'productive activity' because they both involve contact with others as well as emotional support. Traditional definitions of successful aging to an objective judgment which did not account older people's perceptions (Strawbridge et al., 2002, Phelan et al., 2004, Bowling and Dieppe, 2005, Bowling and Iliffe, 2006).

Despite the fact that 'successful' and 'productive' ageing challenged previous assumptions and expectations for the roles of older people, both paradigms suggested a "best way, an only way" to age well. In the same decades, however, awareness around the non-existence of a 'normal ageing' paradigm arose (Ray, 2008). Also, civil rights and women's movements started to criticise the centrality of paid work, given that a large number of people (mainly women) had never been in paid employment. Older

people, it was believed, could age differently and had a variety of patterns they could choose from –compared to the prescription assumed by previous theories (Phillipson and Baars, 2007). Individuals are influenced by the society, the institutional and organisational structures and processes and their experiences: this creates both heterogeneity and inequality among ageing individuals.

The policy slogan `active ageing' -which came to the fore during the UN Year of Older People in 1999 and was subsequently promoted by the WHO -summarises this wider concept and the complexity of interacting factors affecting individual and population ageing. The WHO concept of active ageing was in fact meant to convey a more inclusive message than 'productive' and 'successful' ageing', "and not just to the ability to be physically active or to participate in the labour force". The World Health Organisation (2002) defined active ageing as "the process of optimizing opportunities for health, participation and security in order to enhance quality of life as people age" (WHO, 2002)(pg 12). The WHO movement for active ageing emphasises the connection between activity and health, focusing though on a broader range of activities than those normally associated with production and the labour market. Older people became full citizens who represent an asset and not a burden to society, thanks to their accumulated wealth of knowledge and experience. As such, older people are recognised as an integral part of families and communities and they are valued and acknowledged for their contributions and their participation (Walker and Maltby, 1997), even though they have less to give in a strictly economic sense (Victor, 2005). This implies that the concept of active ageing is also applicable to older people who have retired from paid work or who are ill or live with disabilities. The WHO definition of health, moreover, is a much more comprehensive concept which includes physical, mental as well as social wellbeing. Furthermore, under the WHO approach, ageing is not a homogeneous experience. People age together with the rest of society: they influence the wider community but their experiences are in turn shaped by social factors, culture and society. The latter might determine the resources that older people bring to later life with possible differences both within and between age-groups (Victor, 2005, Phillipson and Baars, 2007). Under the WHO definition, later life is nowadays considered as a continuation of earlier phases of life: therefore, older people are not a homogenous group but rather a complex group with its own specificities.

### 2.3 WHY ENGAGEMENT AND HEALTH ARE CONNECTED

The WHO active ageing concept assumes that active life is positively associated with enhanced health. Engagement in social and productive activities has long been considered an important factor in the study of health among older people in most gerontology theories. Such a connection between activity and health has been linked to the role gained through the activity, to the increased social relationships and their consequences on health.

In particular, Lemon et al. (1972) first proposed a theoretical framework for understanding the positive relationship between activity and health; this rests on the idea of role enhancement. Whereas the role strain theory postulates that multiple roles are associated with poor health outcomes because of the burden caused by cumulative roles (Goode, 1960), the role enhancement framework suggests that occupying several roles is expected to have direct and indirect positive effects on people's health (Sieber, 1974) thanks to complementary and buffering resources shared across different role domains. Taking part in activity promotes support in the role which in turn enhances and reinforces self-esteem, personal well-being, and personal satisfaction helping people to maintain a positive self-concept (Lemon et al., 1972). Activity, it is argued, provides individuals with a sense of usefulness and competence, enhancing control and reinforcing meaning in later life (Van Willigen, 2000). In particular, community participation offers a way of gaining approval, prestige and public recognition. Productive activities -such as work -can be attached to power, to material resources, to new developed skills or knowledge as well as to greater access to social networks, contributing towards boosting people's self-image and mental outlooks, and extending opportunities for self-expression. Social and family engagement provide people with a sense of belonging, value and attachment (Lemon et al., 1972, Berkman et al., 2000).

Activity also provides people with opportunities for companionship and sociability. The social component of activity and interpersonal relations received renewed interest in the late 1990s (Rowe and Kahn, 1997) when both contact with others and emotional support

emerged as crucial factors to ageing well. Social support and social networks are believed to have a positive impact on physical health and psychological well-being through the main effect hypothesis (i.e. social support promotes healthy behaviours) and the buffering hypothesis. Friends, family and contacts act as a cushioning mechanism of support which facilitates coping with stressful life events and 'times of transition' such as retirement and bereavement (Cobb, 1976), providing occasions for positive emotional exchanges which positively influence self-esteem and increase perceived-well-being (Askham et al., 2007). Productive activities as well as social activities within associations, local communities or family are an important channel for maintaining or developing personal relationships useful to combat loneliness and to maintain sense of meaning and purpose in life (Jerrome and Wenger, 1999, Stevens, 2001, Walker et al., 2005).

There are also plausible physiological mechanisms whereby activity may influence health (Cohen, 1988, Umberson and Montez, 2010). Firstly, there is evidence supporting the beneficial effect of engagement in activities which require moderate-to-vigorous physical activity. Numerous studies confirm that regular activity is effective in prevention of several chronic diseases and conditions, such as cardiovascular and respiratory diseases, hypertension and obesity (Bouchard, 2001, Warburton et al., 2006), with meta-analyses reporting significant protection against the occurrence of coronary heart diseases as well as cardiovascular disease mortality (Nocon et al., 2008, Sofi et al., 2008). Previous studies have also consistently found that physical activity is associated with improved well-being, and better physical health and cognitive function as well as decreased symptoms of anxiety and depression (Goodwin, 2003). More generally, for those activities which do not necessarily require physical engagement, it is believed that social activities may influence physiological processes (Umberson and Montez, 2010). The role as well as social relationships gained through activities influence the nervous system and levels of hormones, which in turn affect blood pressure and the immune system (Holmes and Joseph, 2011). For instance, Uchino et al. (1996) reviewed 81 studies to examine the evidence linking the positive aspects of social relationships to physiological processes, focusing in particular on aspects of the cardiovascular, endocrine and immune systems as potential physiological pathways by which social relationships influence physical health. Their meta-analytical study revealed that social relationships and social support were important factors in reducing cardiovascular reactivity to psychosocial stressors, and in regulating blood pressure. Higher social support was also associated with better immune system functioning. The results of the review "suggest that social support has beneficial effects on physiological processes across different age-groups. The net effect may be to biologically age the individual at a slower rate" (Uchino, 1996)(pg. 525). Therefore, engagement of older people may provide not only occasions for positive emotional exchanges and new opportunities for self-expression, but also moderate the physiological processes typically associated with chronological ageing. Exposure to social activities may produce both psychological and physiological effects that could help prevent morbidity and postpone mortality.

#### **2.3.1 EMPIRICAL EVIDENCE**

Empirical research supports the linkages between activity and health, despite the little consistency in the usage and definition of engagement, activity and participation (Bath and Deeg, 2005). Social, productive, instrumental, political, solitary, formal, low and high demand leisure activities, active and passive leisure, help and care-giving are only a few of the numerous labels used to describe various facets of social engagement (House et al., 1982, Herzog and Morgan, 1992, Glass et al., 1999, Bukov et al., 2002, Menec, 2003). Because of the absence of a homogeneous definition, studies also tend to mix up the same or similar activities under different categories of engagement (Wahrendorf et al., 2006). Just as an example, informal caring provided to people to whom the carer owes no familial obligations is labelled as volunteering by some authors (Van Willigen, 2000, Erlinghagen and Hank, 2006), whereas others categorise it separately from volunteering (Burr et al., 2007, Choi et al., 2007).

Despite differences in the way activities are labelled and participation is defined, if we delineate engagement as an umbrella concept covering activities ranging from paid work to informal volunteering and caring, empirical evidence tends to support the idea that engagement among older people is an important factor positively associated with better health (Zedlewski and Butrica, 2007). Older people who remain active tend to be in better health than those disengaged from society

Older people who work, for instance, are cross-sectionally associated with better physical and mental health, better perceived well-being and good self-rated health when compared to retired, unemployed people or homemakers (Debrand and Lengagne, 2007, Siegrist et al., 2007, Alavinia and Burdorf, 2008). Longitudinal studies seem to confirm that being in paid work is also positively associated with better health over time. Calvo (2006), using data drawn from the longitudinal US representative Health and Retirement Study, found that those who worked at baseline reported greater wellbeing, higher mood indicators and lower mortality at follow-up than those who did not work at baseline. Decreased mortality was also found among non-nationally representative populations such as the Swedish construction workers who participated in health monitoring between 1971 and 1993 (Hult et al., 2010) or all the past employees of the Shell Oil company in the United States who retired between 1973 and 2003 (Tsai et al., 2005). These studies showed that workers who retired at younger ages had a significantly higher mortality compared with the ones who retired at older age, even controlling for confounders that could affect subsequent mortality. Using longitudinal data from the Survey of Health, Ageing and Retirement in Europe (SHARE), Börsch-Supan et al. (2009) found that workers with functional limitations at baseline were more likely to retire at follow-up, showing that health plays an important role in explaining exit from employment though health limitations themselves do not explain the cross-national differences in Europe. Retirement, however, may also come as a relief to some workers (Clarke et al., 2005, Westerlund et al., 2009, Phillipson and Ogg, 2010, Westerlund et al., 2010). Westerlund and colleagues studied employees of the French national gas and Electricity Company, analysing the influence of retirement on health. Findings showed that the prevalence of depressive symptoms and physical fatigue decreased greatly one year after retirement compared with one year before retirement, particularly among workers with a chronic disease. They also found that the prevalence of workers assessing their health as less than good decreased substantially over the time period of retirement (Westerlund et al., 2009, Westerlund et al., 2010).

As regards *non-work related activities*, most studies focusing on charity work suggest that volunteering is also associated with increased life satisfaction and a slower decline in self-reported health (Van Willigen, 2000). Volunteering is also positively associated

with better cognitive health, physical functioning levels, well-being and longevity (Moen et al., 1992, Glass et al., 1999, Musick et al., 1999, Oman et al., 1999, Morrow-Howell et al., 2003, Lum and Lightfoot, 2005, Adam et al., 2007). Using SHARE and presenting merely Pearson correlations between volunteering and various measurements of health, Haski-Leventhal (2009) showed that respondents who volunteered at least monthly reported in most European countries significantly higher life satisfaction, lower rates of depression and were more optimistic about their chances to live longer. Similarly, Hank (2010) showed that the propensity to volunteer is significantly lower among SHARE respondents who reported less than good self-rated health or who presented symptoms of depression. Although volunteering seems to be associated with good health, the effect of the extent of the commitment, the number or the type of organisation older people volunteered for is unclear. Oman et al. (1999) reported that, among a community sample of 55+ older residents in California, volunteers engaged with more than two organisations had 63% lower mortality than non-volunteers over a 5-year period. Musick et al. (1999), using data from the Americans' Changing Lives (ACL) study, found a curvilinear relationship between volunteering and mortality, with older people who volunteered 40 annual hours having the greatest benefit from volunteering. Lum and Lightfoot (Lum and Lightfoot, 2005), using the American Asset and Health Dynamics Among the Oldest Old (AHEAD) Study found that having volunteered 100 annual hours significantly lowered the odds of dying. Similarly, whereas some researchers found that even a very small amount of volunteering was related to better health outcomes among older people, others found that volunteering more than 100 hours per year resulted in decreased positive self-reported health (Van Willigen, 2000, Morrow-Howell et al., 2003, Musick and Wilson, 2003). Using four waves of the USA Health and Retirement Study data collected from a sample of 7,830 individuals aged 55 to 66, Hao (2008) found that volunteers had a slower decline rate in psychological well-being in subsequent waves if they also were concurrently involved in paid work.

Numerous studies also suggest that social activities (and not necessarily only volunteering) are associated with better health. Educational, political and community-oriented activities were found to be positively associated with better perceived health

among 629 non-metropolitan older people (Young and Glasgow, 1998). Zunzunegui et al. (2004) studied 3018 people aged 65 and over living in two diverse French speaking Canadian community dwelling populations. In both settings, subjective health improved as the number of social activities increased, and community involvement was positively associated with good self-rated health. Glass et al (1999) examined associations between social, productive, and physical activity and 13 year survival (i.e. mortality from all causes during the follow-up) in a non-national representative sample of 2,761 older people aged over 65 from the City of New Haven (Connecticut, United States). Results concluded that all the activities were independently associated with survival: "more active elderly people were less likely to die than those who were less active", even though "measures of activities convey more information about the number of activities in which people participate than information about the extent of that participation" (p. 480-1). Everard et al. (2000), examining a non-representative sample of 244 65+ members of an organisation for older adults in the USA, found that both social and highdemand leisure activities were cross-sectionally associated with higher functional physical health scores whereas low-demand leisure activities were associated with better mental health scores. Using cross-sectional data from SHARE, Sirven and Debrand (2008) showed that 'social participation' (defined as engagement in volunteering, political or religious organisations, training course and social club) contributes within most European countries to the increase in the average proportion of older people reporting at least good health. The relationship between social activities and health has also been investigated in longitudinal studies. Menec (2003) examined the longitudinal relationship between 'everyday activities' (ranging from sports to solitary hobbies) and a number of health indicators such as life satisfaction, mortality, physical and cognitive function. Using the Aging in Manitoba (AIM) study -a Canadian study from which interviews carried out in 1990 and 1996 were used - the author found that greater overall activity level was associated with better function and reduced mortality, with some differences. For instance, only sport was associated with greater life satisfaction whereas attending music/art/theatre was negatively related to functional health. Mendes de Leon et al. (2003) also investigated the relationship between participation in social and productive activities (ranging from gardening to day trips, from community work to paid employment) and disability using nine waves of sequential data (1982 - 1991) from a sample of 2,812 older people (65 or older) in the New Haven, Connecticut, site of the Established Populations for Epidemiologic Studies of the Elderly. Strong and robust cross-sectional and longitudinal associations between social engagement and disability were found, although analyses "failed to provide evidence for a clear causal effect on rate of functional decline" (p 640).

People can remain active contributors to society and families even without being formally involved in any paid work or without committing themselves to any voluntary activities. There are other various ways older people can feel active, for instance being an active contributor towards family and friends. Qualitative studies suggest that good personal and social relationships represent one of the main factors affecting well-being and quality of life in older adults (Bowling, 2005, Askham et al., 2007). Social support and social networks are generally associated with better health outcomes, including higher level of satisfaction (Burholt et al., 2003), lower risk of mortality (Bowling and Grundy, 1998), lower functional decline (Uchino, 1996, Unger et al., 1999, Mendes de Leon et al., 2001) and better psychological health (Grundy and Sloggett, 2003). Litwin et al (Litwin and Shiovitz-Ezra, 2006), using data from a national sample of 1,334 Jewish-Israeli retirees, found that involvement in activities itself is significantly associated with well-being, though this association is stronger when the quality of social relationships is accounted for. Also Siegrist and Wahrendorf analysed the longitudinal relationship between quality of life among retired SHARE respondents and the reciprocity of efforts spent and rewards received in informal help: they found that those who appraised help as satisfactory and experienced reciprocity of exchange in this activity were more likely to have improved prospective quality of life (Siegrist and Wahrendorf, 2009a). Reciprocity of efforts and rewards or lack thereof may help to explain the links between social relationships and well-being of older people (Wenger, 1997, Jerrome and Wenger, 1999). Contacts with friends and relatives as well as membership in clubs or organisations are strongly and negatively associated with disability (Unger et al., 1999, Mendes de Leon et al., 2001). Unger and colleagues, for instance, using data from the longitudinal MacArthur Studies of Successful ageing (1189 men and women aged 70-79 classified as 'high functioning' were interviewed in 1988/9 with follow-up interviews scheduled between 1995 and 1997) found that those with a greater number of social relationships experienced reduced levels of functional decline.

Social participation and family relations, though, can also be negatively associated with good health. Kohli et al. (2009), using SHARE, analysed 'informal social relationships' of older European people, defined as any exchanges between family members regardless of whether the respondents were the main providers or the recipients of help and care. They found that poor self-rated health was cross-sectionally positively correlated with informal relationships, "possibly reflecting health-related needs" (pg. 333). Hank (2010) also showed that SHARE respondents reporting depressive symptoms were more likely to be 'carer', possibly as a result of the stress of providing care (Sherwood et al., 2005). Receiving 'too much' help might indicate that older people might not be in a functional healthy position to reciprocate (Henretta et al., 2002, Albertini et al., 2007) whereas caring for grandchildren in skipped generation household, for instance, could potentially add a considerable demand on a grandparent's life, being physically and emotionally challenging. Hughes et al. (Hughes et al., 2007), using a sample of 12,872 grandparents aged 50 to 80 from the Health and Retirement Study, found no evidence that caring for grandchildren had negative effects on grandparents' health and health behaviour: health disadvantages arose mainly from their own prior characteristics, not as a consequence of providing care. However, caring for grandchildren in skipped-generation households worsened older people's health, with grandmothers more likely to experience negative changes in depression and self-rated health. Similarly, Moen et al. (1992), using panel data from a sample of 313 women who were wives and mothers in 1956 and were interviewed both in 1956 and 1986, found a distinction between voluntary and obligatory roles: whereas participating in volunteer work and being a member of a club or organisation were positively related to health, paid work and caring for the infirm or aged kin were negatively associated with health (Moen et al., 1992).

To summarise, working appears generally protective of older people's health, provided that work conditions are not degrading. Being out of paid work is generally associated with (but not necessarily the cause of) negative health. Evidence also suggests that staying engaged in formal activities such as volunteering significantly benefits health and well-being of the older people as well as the organizations. Several publications document the significant positive associations between engagement and decreased mortality and depression, improved self-rated health, greater well-being and quality of life, and enhanced cognitive and functional abilities. These findings highlight the importance of activity in ageing well and also suggest that different types of activities may have different physical and psychological benefits.

It is worth mentioning, anyway, that both cross-sectional and longitudinal associations between 'engagement' and 'good health' might have been influenced by reverse causation. In fact, engagement may result from initial higher levels of health and maintain this advantage over time. Therefore, although it is plausible that being active sustains health and well-being, it is equally likely that better health provides an impetus for being more socially active in a wider variety of activities (Adelmann, 1994, Menec, 2003). As an illustrative example, if on one hand work affects the health status of older people, on the other health status can affect participation in the labour market determining the selection into and out of the workforce (Tsai et al., 2005, Barnay, 2006, Blanchet and Debrand, 2007, Alavinia and Burdorf, 2008, Hult et al., 2010). Subjects with poor health, in fact, are less likely to enter into the workplace and workers in poor health are more likely to drop out of work (healthy worker effect), biasing the higher health outcomes of workers when compared to people who no longer work. Thomas et al. (2005), for instance, analysed the impact of changes between employment and various forms of non-employment<sup>5</sup> on the psychological wellbeing of men and women aged 16-74, using the 1991 to 1998 rounds of the longitudinal British Household Panel Survey. The authors found that the prevalence of psychological distress was highest for groups not in employment; they also found that transitions from paid work into either unemployment or long term illness were associated with decreased mental health whereas transitions from non-employment into paid employment were associated with improved mental health. "This suggests that there may be health selection effects operating with healthy people being selected into employment" (Thomas et al., 2005)(pg 245). Ill people were less likely to start participating in work. This selection process for

<sup>&</sup>lt;sup>5</sup> Unemployed, retired, full time study, maternity leave, family care, and long term illness.

employment due to pre-existing health status is known as the 'healthy worker' effect, and is an instance of the broader phenomenon of reverse causality which can be found in numerous other fields (Östlin, 1989, Arrighi and Hertz-Picciotto, 1994, Siebert et al., 2001). For example researchers concerned with identifying possible benefits of volunteering have recognised that whether people choose to volunteer may be dependent on their health, making it difficult to determine the extent to which better health among volunteers compared with non-volunteers may be attributed to volunteering itself (Kent, 2011). Moreover, volunteers with high subjective well-being may gain more benefits from volunteering than other (Meier and Stutzer, 2004). Reverse causality is therefore a source of bias which makes it difficult to determine the true causal association in studies which assess the relationship of engagement in various activities with future health outcomes.

#### 2.4 LIMITATIONS OF PREVIOUS STUDIES

Previous studies generally support the idea that health is better among older people who are more active. Such an association has been found in both cross-sectional and longitudinal studies. Our review has shown that much of the research evidence on the issues of engagement and health is largely US-based and, in addition, there is relatively little comparative European work. Only recently, a number of studies have exploited the potentiality of the longitudinal Survey of Health, Ageing and Retirement in Europe (SHARE), a survey which makes comparison between European nationally representative samples possible. Most of both the US and SHARE studies, though, considered very few activities at the time and a considerable proportion focused merely on paid work, the transition into retirement, caring for a sick person and volunteering. This is clearly a legacy of past theories which were mainly concerned with the retiree's process of making post-retirement decisions in order to compensate for changes in roles and identity. Nowadays, though, studies have begun not only to account for people who are not in paid work (either unemployed, retired or homemakers) but also to account holistically for as many activities as possible. Some studies have just focused their attention on the involvement in charity work; others have concentrated upon the level of political and religious commitment only; a few have looked at specific activities such as
grandchild care. Older people, though, seem to be involved in an increasing number of simultaneous roles and activities and individuals appear to add family responsibilities to their portfolios of daily tasks without cutting back on paid work or other competing activities (Tomassini et al., 2004a, Glaser et al., 2005). People can take up a new activity in order to compensate for the loss of previous roles or give up activities due to time or physical constraints; people can also opt for multiple complementary activities performed in parallel (Van Willigen, 2000, Hank and Stuck, 2008, Kohli et al., 2009).

Moreover, the populations examined ranged from community-dwelling to residential living older adults; also specific groups of respondents either working in determined companies or having specific functional limitations were considered. Moreover, depending on the data available, each study used different methodologies as well as different definitions and measurements of both the activity and health of older people. Outcomes included –just citing some –functional health, physiological health, depression, mortality, self-rated health, psychological health, blood pressure, psychiatric morbidity, cardiovascular diseases, disability, well-being, physical health and mental health. As a consequence, the comparison of results at national levels in order to draw conclusions and explain differences or similarities between European countries is arduous: the availability of SHARE, a nationally representative study, is only recent.

Finally, both in cross-sectional and longitudinal studies reviewed, the problems of initial low-response and attrition were hardly ever mentioned, despite the fact that between 10% (Mendes de Leon et al., 2001, Mendes de Leon et al., 2003, Menec, 2003) and 30% (Unger et al., 1999, Siegrist and Wahrendorf, 2009a, Hank, 2010) of respondents were lost to follow-up. Sample attrition might bias longitudinal associations, yet this issue is generally ignored and analyses tend to be limited to complete record datasets.

# 2.5 THE CURRENT STUDY

The current research project is distinct from the previous research in several key respects. This study analyses secondary data from two surveys, the English Longitudinal Study of Ageing (ELSA) and the Survey of Health, Ageing and Retirement in Europe (SHARE). Previous joint analyses of these sources have predominantly focussed on differences between countries in health (Avendano et al., 2009, Jürges, 2009, Jürges,

2010, Crimmins et al., 2011). Demographic, socio-economic and household characteristics as well as health-related variables are comparable among these two surveys. Although in theory the nature of these surveys allows for homogeneous and standardised definitions and measurements, given that very few studies have attempted to combine ELSA and SHARE, this research project will also highlight how comparable these two studies are when indicators of different forms of engagement are considered.

Additionally, this research study adopts a multifaceted inclusive approach to activities: the concept of 'active ageing' was operationalised holistically, including as many activities as possible, within the boundaries of available data. Most studies have considered few activities at a time and focused merely on paid work, caring for a sick person and volunteering, mostly because these activities were the only ones which had the highest prevalence at older ages (Wahrendorf et al., 2006, Siegrist and Wahrendorf, 2009a, Hank, 2011). In this project, restricting analyses to respondents aged 50-69, I will include all activities mentioned in the questionnaire, given that older people can be involved in a number of simultaneous activities. Moreover, there is no-evidence base to suggest that volunteering is a better predictor of health: the odds of reporting good health indicators are equally higher among respondents who, for instance, engage in political or religious organisations (Sirven and Debrand, 2008). Ignoring respondents who participate in activities other than just volunteering or caring would exclude potentially a considerable number of respondents; moreover, this would fail to embrace the WHO policy initiative which pushes towards a gradual broadening of 'activeness' to include any continuing participation in social, cultural, economic and civic affairs.

ELSA and SHARE are moreover ongoing panel surveys. The current study therefore allows for a longitudinal analysis of the association between engagement in a mixture of different activities and health among older people. This will enable us to control for previous levels of health, which partially addresses the concern that selection effects account for the observed relationships with engagement. The problem of missing data and attrition bias will also be attended to by using multiple imputations, as discussed in Chapter 9. Whereas the common practice is to restrict analyses and findings to "complete record" dataset, this research project will also propose multiple imputations as a possible remedy to address missing data. Finally, in the belief that ageing and 'old age' have a dynamic, heterogeneous and shifting nature and are both social, cultural and economic constructs defined and shaped by social and environmental contexts (Victor, 2005, Phillipson and Baars, 2007), this study compares different countries. People age together with the rest of society: ageing people influence the wider community but their experiences are in turn shaped by social factors, culture and society. Socio-economic patterns, environmental contexts as well as factors such as age, gender and social class can influence later life and the resources that older people bring to later life with possible differences both within and between agegroups. Not necessarily all older people will be active in the same way, and the activities which keep people active may differ in different countries; similarly, the value attached to different activities could vary considerably between countries. Policies and norms could influence the wealth of knowledge, experience and energy older people might have to keep themselves active and to continue contributing to society. A comparative approach will also indirectly compare 'active ageing' policy effects, exploring what differences exist between countries in terms of 'activity levels' and its influence on older people's health.

Identifying the factors associated with engagement in different countries could help not only to determine whether and to what extent older people engage differently in different contexts but also to identify possible strategies or interventions to increase the level of participation of older people and contribute in the long-term to people's health. Investigating which activities older people prefer to engage as well as the ones which might contribute to improve or maintain a favourable health is of particular relevance to public policy, as activities could to some extent be influenced by policies.

#### 2.5.1 AIM AND OBJECTIVES OF THE CURRENT STUDY

In this study I analyse the association between a selected number of socio-economic, demographic, health, personal and household characteristics (drawn from previous research studies) and engagement. The combination and accumulation of, as well as possible conflicts between, different activities are also investigated. Moreover, making no prior assumptions about the benefit of each specific activity on health, their impact on older people's health is assessed longitudinally.

The overall aim of this research project is to assess whether activity has a positive impact on people's health. The WHO policy framework pushes boundaries towards a new balance between work, retirement and participation in a broader number of activities in which older people can contribute to maintain their health as they age. This research thesis, therefore, examines whether older people are indeed active contributors to their family, community and society, and whether there is a positive longitudinal relationship between engagement and better health. In particular, the level of 'activeness' and its longitudinal relationship with health are examined in different countries.

This project therefore aims to answer the following specific research questions:

1) Which characteristics are associated with engagement of older people?

This study investigates the baseline associations between socio-demographic and health characteristics of older people and engagement in various activities in order to gain insight into the characteristics which influence their level of engagement. Moreover, it investigates whether the level of engagement and the personal-level characteristics associated with it are common across different countries.

2) Do more 'active' people experience less health decline?

This study will analyse longitudinal associations between engagement at baseline and health at follow-up (controlling for baseline health). This analysis might help us determine whether the various forms of engagement play a similar role in different countries.

#### 3) Does attrition bias longitudinal findings?

The extent to which longitudinal associations might be biased by sample attrition will be assessed by using multiple imputations and sensitivity analyses.

The following chapter introduces the rationale behind the choice of the four countries under study in this project. Even though European countries are facing similar demographic challenges, retirement and social policies continue to differ. These different approaches might be embedded in the different European welfare state regimes.

L

# 3. WELFARE REGIMES AND THEIR INFLUENCE ON POLICIES IN FOUR EUROPEAN COUNTRIES

This project focuses on four European countries: Denmark, France, Italy and the  $UK^6$ . The choice of countries reflects the fact that, despite a general demographic convergence and although most European countries are facing similar challenges, these countries still continue to differ in a number of domains relevant to the topic of this thesis, including employment rates and retirement policies, fertility patterns (a determinant of population ageing) and patterns of household composition.

In this chapter, various approaches to distinguishing European countries and explaining persistent differences between them are briefly described, followed by a review of welfare regime classifications. Recent pension and retirement reforms in the four countries are then introduced, highlighting differences in the policies adopted which partly reflect principles and values of each distinct welfare regime. Denmark, France, Italy and the UK are purposively selected to reflect different welfare regimes, as the level of engagement of older people may indirectly relate to country-specific policies.

#### **3.1** THE 'HISTORICAL' APPROACH TO COUNTRY DIFFERENCES

Numerous are the authors who have tried to explain differences between European countries and groups of countries. Espelt et al. (2008), for instance, classify countries in three political traditions (Social democracy, Christian democracy and Late Democracies) based on their political history (such as the political party in government), as politics relate to the extent to which welfare states expand, to class inequalities, and to the redistribution of the wealth. Giuliano (2007) stresses the importance of considering norms and values. In her work, she notes that differences in behaviours such as non-marital births, the relationship between marriage and fertility, co-residence and

<sup>&</sup>lt;sup>6</sup> Although in this chapter I will refer to the UK, data used in the following chapters only refers to England, where nearly 84% of the total UK population lives (Source: Office for National Statistics).

family composition, are related to value systems rather than to economic-related factors or the provision of services. Studying the behaviour of American European immigrants' children, Giuliano notes a "clear indication that culture matters" (pg. 7) since immigrant groups, despite being exposed to the same environment, tend to replicate the behaviour of their 'national' peers. Other authors have rooted differences in historical circumstances, such as migration flows, the timing of the transition processes of proletarianisation from agricultural to industrial economy or the different political participation of the civil society and composition of unions and workers (Andreotti et al., 2001). Reher (1998) believes that the European North-South gradient and distinction between areas "where traditionally the family group has had a priority over the individual, and others where the individual values have had priority over everything else" (p. 203) should be attributed to deep historical roots. Variations in the practice of using servants or day-labourers in agriculture, for instance, influenced the departure of young men from the parental house. The Reformation -with its emphasis on the individual and the value of work -enhanced women's position in society thanks to the parallel institutionalisation of marriage as a partnership contract, whereas in the traditional Catholic settings marriage remained male-authoritarian hierarchical. These are just few of the several interwoven factors which could help explain the importance of the role that 'the family' gained over centuries in Southern European countries (Reher, 1998). For instance, in Southern countries "much of the aid given to vulnerable members of society came from the family or from individual charity, while in northern societies this was largely accomplished through public and private institutions" (p. 216). Similarly, in Southern societies, family (and not the community) was the institution taking care of older people, either through means of co-residence, transfers of goods and services through family members or proximity of households. This intergenerational familial social obligation was necessary for the survival of the society itself and still nowadays, where family is strong, women in particular tend to add elder care and other family responsibilities to their daily tasks (Glaser et al., 2004). Legacies of stronger family ties (intergenerational exchange and support, multigenerational household composition and geographical proximity) continue to be more prevalent in southern than in north-western European countries (Tomassini et al., 2004a, Tomassini et al., 2004b).

# 3.2 ESPING-ANDERSEN'S THREE WORLDS OF WELFARE CAPITALISM

Despite the appeal of the historical approach, many recent comparisons of countries pivot on the 'welfare regime' typology developed by Esping-Andersen, who did not deny the importance and complexity of different actors, trends, forces and values in explaining the differences between countries but came up with a simpler way to classify welfare regime types. Despite continuous critiques, Esping-Andersen's formulation of ideal types is still a useful tool because of its power to describe and at the same time explain variations between countries. Esping-Andersen attempted to provide a theoretical framework able to make sense of empirical findings and differences, trying this way to combine the underlying dynamics of each regime with a successful empirical attempt to encompass them.

Esping-Andersen argues that welfare regimes are clustered around three diverse regime types (conservative, liberal and social-democratic), each organised according to different logics of organisation, stratification and social integration and each encompassing different historical and political forces and nation-buildings which are linked to particular social policy and political economic configurations (Esping-Andersen, 1990). He uses two criteria to define his typology: the degree of decommodification and the social stratification system. Decommodification is the possibility for citizens to "opt out of work when they themselves consider it necessary" (pg. 23): this indicates the degree to which individuals or families can maintain an acceptable standard of living independently of labour market participation. The stratification of the society is the manifestation of the capacity of the welfare state to redistribute wealth, reduce poverty among groups and provide resources for health, education and similar services which are decisive for individual mobility and for the employment structure (Esping-Andersen, 1990).

Observing social rights and welfare-state stratification in a number of OECD countries<sup>7</sup>, Esping-Andersen finds different arrangements between state, market and the family which can be clustered in 3 different regime types. There is a clear coincidence of high

<sup>&</sup>lt;sup>7</sup> Esping-Andersen considers, in his analyses, Australia, Austria, Belgium Canada, Denmark, Finland, France, Germany, Ireland, Italy, Japan, Netherlands, New Zealand, Norway, Sweden, Switzerland, United Kingdom and United States of America.

decommodification and strong universalism in the social-democratically influenced welfare states (such as the Scandinavian countries). The Social-democratic welfare regime promotes equality through the possibility for all citizens to enjoy the same rights, with high levels of universal social benefits. This ideal type also supports individual independence providing opportunities for individuals. In order to minimise dependence on the family, the State takes direct responsibilities for caring for children, older people and marginalised groups (Rostila (2007) refers to this as the socialisation of the costs of familyhood) in order to allow all individuals and women in particular to choose work rather than household duties (Esping-Andersen, 1990). This regime, in fact, tends to maximise the labour participation of all persons (through policies aimed to develop individual capabilities) because its attainment depends on it and a failure of this achievement would be considered a social risk for all. The conservative-corporatist welfare states – such as Germany, France and Belgium – are typified by a moderate level of decommodification and are shaped by both a catholic, etatist and corporatist legacy. Social rights are not universalistic; benefits are provided only to people with a previous status of worker (with a distinction between those who served the State and those who did not) and are based on previous earnings. De facto, this welfare regime perpetuates differences between recipients and preserves differentials since rights are attached to class and status. This ideal type is also characterised -due to the church legacy -by the preservation of the traditional family where its members are the main providers of services and the interference of the State is meagre and considered necessary (mainly through employment protection of the male breadwinner) only when the family's capacity to service its members is exhausted. Finally, according to Esping-Andersen, there is also a clear correspondence between low decommodification and strong individualistic self-reliance in the liberal type of welfare regime. This is characterised by the primacy of the market and therefore policies tend to respond to possible market failure through encouraging it either passively (by guaranteeing only a minimum of benefits and by providing means-tested assistance with strict entitlement rules and modest universal transfers) or actively (by subsidising private forms of welfare system). In the Liberal Welfare states universal but low welfare benefits have never been intended to replace labour market income, and poverty is prevented by providing incentives for and access to employment removing barriers to labour market participation. People under this regime are encouraged to look for jobs (or return to the labour market as soon as possible) and to apply for private protection and this leads to a division of the population with a minority of low-income people who depend on the state and the rest who can afford private social insurance plans.

# 3.3 CRITIQUES OF ESPING-ANDERSEN'S TYPOLOGY

Esping-Andersen's typology has created much debate and led to three main criticisms, all strongly interconnected, relating to the number of clusters, the role of women and the methodology used to classify countries.

Among the critics, Esping-Andersen's typology has been strongly objected for not including a Mediterranean or Southern European welfare regime (Leibfried, 1993, Ferrera, 1996, Bonoli, 1997, Jensen, 2008) and for considering such countries as a subcategory or a variant of the continental one (Arts and Gelissen, 2002). Andreotti et al (2001), for instance, stated that "important differences make it necessary to distinguish continental European countries from Southern ones" (pg. 43). Trifiletti (1999) also argued that Southern welfare regimes lack resources to subsidise social policies because it is assumed that the family, and women specifically, have the crucial responsibility to prevent breakdown into a condition of need, being the main actors to whom care for the sick, disabled and unemployed is delegated. Moreno Minguez (2005) insisted that Southern European countries persevere in supporting family strategies associated with the male breadwinner model. As a consequence, in Southern European countries the principal referent for social and employment policies remain the family rather than the individual. Various commentators (Trifiletti, 1999, Pierson, 2000, Danna, 2008) have also argued that in these countries resources are mobilised only when the risk of the individual is perceived as a risk which can damage the ability of the family to function as it usually does. This determines low female labour participation and the economic dependence of women on their husbands' income, as argued particularly by Saraceno (1994). This also reinforces gender inequality in the family. Given the limited institutional support to families and the lack of programmes supporting 'employment and family compatibility' (Moreno Mínguez, 2005), in the Southern welfare regime women are expected to mainly work inside the family as carers who look after the other members of the family unit. In southern European welfare regime women emancipate *within* the family and not from it (Trifiletti, 1999, Andreotti et al., 2001).

Esping-Andersen's work has also been criticised for not including systematically a gender dimension in social policy and for ignoring the family's role, the provision of care as well as the contribution women have made in shaping the main contours of public policies (Pierson, 2000). Welfare regimes have an impact on issues such as female labour market participation, inequalities within households as well as the extent to which women can form an autonomous household without relying on the income of a male breadwinner (Jensen, 2008). It is not possible to divorce the study of unpaid work and caring regimes and the position of women in relation to paid work: women's labour force participation is largely influenced by the extent to which social welfare policies help both men and women to reconcile paid and unpaid work (Lewis, 1992). Women are more likely to work where social care responsibilities are transferred from the family to either the market or the state (Jensen, 2008). The provision of social care services is a necessary condition for defamilisation (defined as the degree to which adults can maintain a certain acceptable standard of living without relying on family relationships), constituting both the incentive and the opportunity for female decommodifation. Whereas in Southern European welfare regimes the public social care sector has been contained, confining women to their institutionalised role of wives and mothers within their households (Danna, 2008), social-democratic countries have expanded social care services in order to provide women with a gateway to the labour market, considering them as workers who need protection as mothers and wives. Liberal welfare state regimes were somehow driven by the idea that social services could be accessed on the market if needed, keeping the provision of social services to a minimum and treating women mainly as workers (ignoring their family roles) (Trifiletti, 1999, Jensen, 2008). Nordic countries have the lowest gap between male and female economic activity participation and provide women with social services and benefits: Scandinavian countries have the highest maternity leave compensation rates and duration (Bambra, 2007a, Thévenon, 2011).

Finally, the methods used by Esping-Andersen, the variables included in the analyses and the resulting classification of the clusters (and country membership) have also been criticised by several authors who believe that different methods and different dimensions would lead to alternative clusters and a dissimilar classification of the nations under study (Arts and Gelissen, 2002, Powell and Barrientos, 2004). For instance, Hudson and Kühner's (2009) analysis, which also includes social investments, resulted in a much more complex classification of countries capturing in a richer picture the diversity and differences of welfare states in response to the modernising pressures of the knowledge economy. Adding, substituting or attaching more importance to certain indicators can yield different classifications. Only a few countries are more homogeneous across different policy areas and tend to emerge as standard examples of the same ideal type of regime (Arts and Gelissen, 2002, Bambra, 2007a). Most countries exhibit fluctuating regime type membership, depending on which factors and indicators are used in the regime construction and which policies are under examination. Arts and Gelissen (2002) and Bambra (2007b) list the ideal welfare state regime types: classifications range from British, Liberal or Anglo-Saxon to Conservative, Corporatist or Bismarckian; variables and dimensions also range from poverty rates to social expenditure and political tradition. Clearly, very few of the mainstream typologies (Esping-Andersen, 1990, Castles and Mitchell, 1993, Leibfried, 1993, Ferrera, 1996, Bonoli, 1997, Korpi and Palme, 1998) actually agree with one another over the exact classifications of all countries. Most of the recent studies conclude that, contrary to the ideal world of welfare states, the real world is likely to exhibit hybrid forms, and most countries present a system mix (rather than being one-dimensional pure cases). Similar circumstances and challenges faced by most countries -particularly in the European Union -have produced more hybrid mixed cases rather than prototypes of distinct regimes. Despite common trends, though, differences between countries continue to persist.

# 3.4 WELFARE REGIMES: STILL A VALUABLE INSTRUMENT TO EXPLAIN COUNTRY DIFFERENCES

Despite criticisms and limitations, the welfare ideal typology is still a theoretical framework useful and much used to explain differences between countries 'belonging' to different regime types. It is undeniable that some of the major differences persisting between countries are associated with different welfare regimes types. Examples of differences attributable to different welfare regime types range from social trust to household characteristics. Variations in public social provision have big effects on issues such as income inequality, women's labour participation or household formation and characteristics (Andreotti et al., 2001, Vasconcelos Ferreira and Figueiredo, 2005). Some argue that policies reconciling family and work (and therefore the degree of economic dependence of women upon their husbands) can influence family stability (Trifiletti, 1999, Danna, 2008); others believe that the target of benefits and social assistance (individuals versus households) can have an impact on co-residence and intergenerational family transfers (Albertini et al., 2007). Cohabitation is the Southern European way to combat poverty; similarly, intergenerational transfers of resources have been linked to the generosity of welfare regimes since it is believed that older people are more likely to give more where welfare regimes provide them with more social services and allocate them more economic resources (Künemund and Rein, 1999, Attias-Donfut et al., 2005, Albertini et al., 2007). Bohnke (2008) argued that differences in support among European countries could also be explained by welfare regimes and social benefits. Family solidarity was very strong particularly among Mediterranean countries. Where social benefits were at a low level and little help was provided by the State, there was also little confidence in the social security system and people tended to seek support and help mainly through family members rather than friends and workmates which were more important in Scandinavian countries. Rostila (2007), in his work, concluded that there was empirical evidence which supported the idea that the welfare state was associated with trust. Where social security levels were higher and income inequality was lower ('socio-democratic' countries), there was a higher social cohesion and trust among citizens. Also labour market employment levels and retirement attitudes can be systematically influenced by the welfare state, directly

through (un)employment and retirement policies or indirectly through social services available, for instance, to help women reconcile work and family and facilitate their access to work. Some argue that also volunteering and more generally civic participation should not be seen in isolation from the societal context, since they are highly dependent on a country's social policies (Salamon and Sokolowski, 2001, Erlinghagen and Hank, 2006). Where provision of services by the State is lower, voluntary activities might be more 'service' oriented to compensate for their lack, compared to social-democratic countries where, for instance, voluntary work is mostly performed in the cultural or recreational domain. To a certain extent, civic participation is not merely a personal choice (Salamon and Sokolowski, 2001) but also a response of individuals to specific conditions and incentives created by policies through direct and indirect intervention: engagement in non-productive activities tends to be higher where social expenditure is high and therefore where social policies create infrastructures and opportunities for people to be socially engaged (Pichler and Wallace, 2007, Hank, 2010).

#### 3.4.1 A SENSITIVE WAY TO MAKE USE OF A WELFARE REGIMES TYPOLOGY

It appears evident that welfare states are active powerful mechanisms which influence societies on key issues such as social stratification, distribution of resources and household characteristics, working life or the evolution of employment. Regime types offer a fruitful simplified theoretical starting point for the sake of the argumentation when explaining empirical differences in trajectories among countries (Jæger, 2006).

It is also established that real welfare states are hardly ever pure types and that hybrid mixed cases are more common (Arts and Gelissen, 2002). Kasza (2002) explicitly refers to welfare regimes as an 'illusion'. The author argues that, particularly in Europe, most countries practise a disjoint set of welfare policies given the involvement of numerous actors, the historical and economic circumstances, the cumulative work of different governments, the influence of foreign models and the vastness of welfare fields ranging from income support for older people to housing and education. As a consequence, countries are not likely to exhibit the internal consistency, coherence and distinctive rationale necessary to reflect fixed principles and adhere to established values of a certain welfare regime.

Keeping in mind that single countries do not represent welfare regimes in their pure form, that evidence on the number of clusters and country membership is still inconclusive, and that –when comparing countries –differences between them (rather than welfare regimes) are being tested, there are still good reasons to continue to work on typologies, albeit with some caution. It is undeniable that differences between countries and their policies still exist. These differences are the product of complex historical cumulative interactions of interdependent multiple factors which still generate and perpetuate dissimilar dramatic outcomes for which it still makes sense to talk about regimes. In particular, this research study will consider four different welfare regimes, extending the initial Esping-Andersen's typology to also include the Southern European regime. A four-country typology, in fact, seems to better capture the gender and family dimensions, providing a more useful way to analyse and understand differences between European countries.

This research project will analyse Italy, France, Denmark and the United Kingdom 'representing' the Southern, the Continental, the Social Democratic and the Liberal regime respectively. None of these countries perfectly represents a welfare regime, though their policies on pensions and (early) retirement reflect four different 'regimes'. There are clear differences by welfare state regime in terms of how different early retirement policies were operationalised, particularly in terms of the generosity of benefits paid, the qualifying period and conditions.

Tables 3.1-3.6 present some general statistics and indicators on employment characteristics, social expenditure, social cohesion as well as household composition in the four countries under study. These indicators demonstrate that, despite similar European phenomena such as an increasing number of older people living alone or the general intention to increase female labour participation, differences still exist and somehow reflect the typology of welfare regimes they 'belong' to. These indicators also justify the use of four typologies instead of the original three proposed by Esping-Andersen: the inclusion of a Southern welfare regime typology, in fact, better accounts for the gender dimension.

In particular, Table 3.1 shows the percentage of men and women who are in paid work<sup>8</sup> by age-groups. The proportion of men working is very similar among 25-54 year old people; the younger (15-24) and older (55-64) age-groups though show country differences which are not unexpected: the employment structure, the individual mobility and the skills available to the labour market are highly influenced by the education system and in Italy and France the 'effective' average age for completion of tertiary education<sup>9</sup> is higher (Eurostat, 2009, Livi Bacci and De Santis, 2010). Similarly, the low participation rates among the older age groups are highly influenced by 'early retirement' policies: the late 1970s pension reforms in the 'Conservative' and the 'Southern' type regimes which discouraged active participation in the labour market among older people continue to have an effect on employment rates among people older than 55. Women's employment rates also reflect regime type differences: whereas Denmark (a social-democratic regime type country) seems to succeed in maintaining high employment among women and the gap between male and female participation is minimal, Italian women's labour participation is extremely low as expected in a 'male breadwinner' Southern tradition. In Italy, participation among women in their reproductive ages (25-49) is between 15% (compared to France and the UK) and 20% (compared to Denmark) lower. When the subgroup of mothers with at least one child aged less than 15 is considered, we observe that whereas in Italy only 50% are working, in Denmark more than two thirds are (see Table 3.2). Denmark is also the only country, out of the four under study, where the difference in participation between mothers of young children and other women of reproductive age is almost imperceptible. Finally, Table 3.3 shows the proportion of employed people who work part-time: in Denmark and France more than one quarter of women work part-time compared to less than 15% in Italy and more than 40% in the UK. Working women with family tasks are not numerous in Southern regimes because of the limited institutional network of support to families with children (Saraceno, 1994, Moreno Mínguez, 2005). In Socio-Democratic regimes women are treated as workers and are protected on the market if they are

<sup>&</sup>lt;sup>8</sup> Data refer to the employment rates as defined by the OECD, i.e. the proportion of the country's workingage population that is employed (or self-employed).

<sup>&</sup>lt;sup>9</sup>Completion age depends on both the duration of studies, the year of entry into higher education as well as on year repetition or other interruption to schooling

mothers: the state aims to achieve real equality of opportunities, providing services –if and when needed –to sustain such equality. In Conservative and Liberal regimes, women are primarily treated as wives and mothers but are protected on the labour market: this protection, though, also creates some gender discrimination on the labour market, either because women get some social benefits just as 'wives' through their husbands (Conservative regimes) or because they tend to mainly work part-time to have enough time to compensate for the little protection they received from the state (Liberal State). Finally, Southern regimes tend to treat women as wives and do not protect them: as a consequence, fewer women work under male conditions and mainly full time and working mothers are given very few services such as child-care provision (Trifiletti, 1999, Thévenon, 2011).

	Denmark		France		Italy		UK	
	Men	Women	Men	Women	Men	Women	Men	Women
15-24	63.9	60.5	32.5	25.9	29.9	20.8	60.4	56.7
25-34	86.3	77.3	86.3	72.0	80.1	58.2	87.9	72.5
35-44	90.2	82.8	89.2	74.9	91.2	61.3	88.9	75.0
45-54	88.2	81.5	87.3	75.1	88.1	53.5	86.3	76.7
55-64	65.6	53.5	41.6	36.0	42.7	20.8	65.7	48.1

Table 3.1 Percentage of people in paid work, by age-group and sex - 2005

This table shows the percentages of people who are employed by sex and age-groups in Italy, France, Denmark and the UK. Source: OECD Labour Force Statistics. Data accessed through the OECD website http://stats.oecd.org on 27/05/2011.

Table 3.2 Percentage of women in paid work by country and specific groups - 2005

	Denmark	France	Italy	UK
Female (24-49) employment ratio	79.4	74.6	61.1	75.2
Maternal employment ratio (mothers with child[ren] under 15)	76.5	64.9	50.0	61.4

This table shows the percentages of women in their reproductive ages (25-49) as well as the proportion of mothers (with at least one child under 15) who are employed in Italy, France, Denmark and the UK. Source: OECD Labour Force Statistics. Data accessed through the OECD website http://stats.oecd.org on 27/05/2011.

	Denmark		France		Italy		UK	
	Men	Women	Men	Women	Men	Women	Men	Women
25-49	6.4	25.4	3.9	29.2	3.6	18.3	4.2	41.1
50-64	7.3	29.6	5.8	31.2	3.8	18.4	11.3	48.1

Table 3.3 Percentage of workers working part-time, by country, sex and age-group – 2005

This table contains part-time employment as a percentage of the total employment by sex and standardised age groups (25-49, 50-64) in Italy, France, Denmark and the UK. Part-time employment is based on based on a spontaneous response by the respondent. Source: Eurostat; Employment and unemployment (Labour Force Survey). Data accessed through the Eurostat website http://epp.eurostat.ec.europa.eu/portal/page/portal/eurostat/home/ on 27/05/2011.

Table 3.4 shows the percentage of GDP spent on different cost components such as Unemployment, Old Age and Family. Whereas the proportion spent on Health is similar in all countries, other components differ quite substantially. For instance, Italy and the UK devote to unemployment protection half a point of their GDP, compared to 1.7 in France and almost 3% in Denmark (such expenditure is not a function of unemployment levels). Similarly, Italy, France and the UK spend less than 1% on programmes which aim to encourage labour market participation, compared to almost 2% in Denmark. Denmark, France and the UK also spend considerably more than Italy in Family related programmes, including family allowances and parental leaves.

			N	
	Denmark	France	Italy	UK
Old Age (including pensions)	7.2	12.8	15.0	6.8
Incapacity-Related programmes	4.5	1.9	2.0	2.4
Family	3.6	3.0	1.3	3.0
Housing	0.7	0.8	0.0	1.4
Unemployment	2.8	1.7	0.4	0.5
Active Labour market programmes	1.7	0.8	0.5	0.5
Other	0.9	0.3	0.0	0.1
Health	7.8	5.8	6.8	7.0
Total Voluntary Private Expenditures	2.3	2.6	0.5	6.3

Table 3.4 Total Social Protection expenditures as % of GDP by country, 2005

This table shows the total social-services expenditure by country. Expenditures include both cash benefits and benefits in kind; Programmes included range from pensions, family allowances, disability pension as well as parental leave. Both public and mandatory private expenditures are considered. Source: OECD Social Expenditure Dataset 2009 (ESDS International, University of Manchester) accessed through https://www.esds.ac.uk/international/access/dataset\_overview.asp#desc\_OECDSED on 24/05/2011

Table 3.5 shows the percentage of people who trust both other people and the institutions. In Denmark, both confidence in the institutions and trust in the others are much higher than the other countries under study: this has been linked to the much more 'equal' society that the socio-democratic regime countries pursue through high social security levels (Rostila, 2007).

	Denmark	France	Italy	UK
Perceived level of corruption	15	54	76	48
Confidence in institutions	75	59	49	65
Trust in others <sup>a</sup> (2008)	89	56	missing	69

Table 3.5 Social Cohesion Indicators by country, 2010

This table shows a selection of 'Social Cohesion' Indicators. Data are expressed on percentages agreeing on the various statements. a) Trust refers to 2008. Source: OECD Social Cohesion Indicators, 2010; data accessed through the OECD website on 24/05/2011

(http://www.oecd.org/document/24/0,3343,en\_2649\_34637\_2671576\_1\_1\_1\_1,00.html#data).

Table 3.6 presents the percentages of older people aged more than 65 who receive long term care in institutions and the percentage of men and women 65+ who live on their own. In Italy, very few older people live in institutions compared to the other 3 countries under study. Similarly, in France and particularly in Italy the family still takes care of older people through means of co-residence: less than 20% of older men and about 40% of older women live alone in these countries compared to more than 30% and 60% in Denmark and to a quarter and almost half women in the UK respectively.

	65+ who receive long- term care in istitutions	Men 65+ who live alone	Women 65+ who live alone
Denmark	4.8	30.5	60.6
France	6.7	18.0	40.6
Italy	2.4 <sup>a</sup>	13.6	36.9
UK	4.3 <sup>b</sup>	24.1	48.3

 Table 3.6 Percentage of people older than 65 who receive long-term care in institutions and percentage of 65+ who live alone, by country and gender

The first column of this table shows the Long-term Care Recipients in Institutions (other than Hospitals) as Percentage of Total Population, aged 65 and older, in 2008 [a) 2003; b) 2004]. Services received can be publicly or privately financed. Long-term care institutions refer to nursing and residential care facilities; Persons receiving long-term care in hospitals are excluded. Source: OECD Health Data, 2010 accessed through http://www.ecosante.fr on 24/05/2011. The second and third columns show respectively the proportion of older men and women aged more than 65 who live alone. Sources: statistics.gov.uk (UK, 2005); insee.fr (France, 1999); Rapporto Osserva Salute, Istat (Italy 2009); dst.dk (Denmark, 2005).

The following section presents a brief overview of trends in pensions, retirement policies and the more recent reforms in the four countries under study. The focus of the reforms will be up to the mid 2000 since data used in this study refer to this period of time; rather than going into the detailed technicalities of the various reforms, the general intents and trends are presented, highlighting the typical characteristics of the 4 welfare regimes.

# 3.5 PENSIONS AND RETIREMENT POLICIES

# 3.5.1 DENMARK

In Denmark, a basic public old-age pension is given to all citizens who are 65<sup>10</sup>. Such pension should secure an income by all citizens. In addition, workers also receive a supplementary earnings-based pension which depends on the individual contribution record. These occupation schemes are more flexible in terms of retirement decisions since employees can begin drawing the pension from the age of 60 with an adjustment (mainly a reduction) in entitlements (OECD, 2004a). In Denmark, workers depend less on the market since social security is universal and comprehensive. The cost of maintaining such a universalistic social system is met through tax revenue income and as a consequence, Danish pensions reforms in the last decades have focussed on containing costs through full employment (of both men and women) and prolonging working life (Aysan, 2009). The last major reforms, in fact, reformed the voluntary early retirement pensions and reduced the maximum number of years of unemployment benefits to four (1999); introduced stricter medical assessment criteria in order to be awarded a disability pension (2003); and introduced financial incentives for workers to continue working and to postpone access to their old-age pension (2004).

Denmark is tackling most of the routes and means used by workers to exit the labour market at younger ages than the official retirement age such as disability and unemployment benefits. In particular, drastic measurements have been passed to reform the voluntary early retirement pension (VERP). This was originally introduced in 1979 in order to offer fatigued older workers aged 60 and over a way to leave the labour

<sup>&</sup>lt;sup>10</sup> The normal pension age is currently 65 but will be increased gradually to age 67 in the period 2024-27. A full public old-age pension requires 40 years' residence. Shorter periods qualify for a pro-rata benefit.

market before they became entitled to the old-age pension (OECD, 2004a). Later reforms introduced the early retirement programme also for unemployed aged 50-59 (1992, 1994). De facto, in the early 1990s, the share of workers or unemployed aged 50-59 retiring through the VERP increased sharply to the extent that in 1996 the programme was terminated for unemployed, and in 1999 the entry age to access it was raised to 62 or later, provisions became less generous and the number of working years to qualify raised to 25 (from the previous 20) (OECD, 2004a).

In Denmark, reforms were mainly aimed at cost containment, met through high labour force participation and a rationalisation of the existing programmes. The reform agenda never questioned the active role that the state plays in managing social risks, compared to the market and the family. The pension regime in Denmark entails generous pension plans provided by the State and social security for its citizens as well as the emphasis on high labour force participation, both among men and women. This reflects the *Social-Democratic* welfare regime rationale which aims to provide both individual freedom (through high labour participation for both men and women) and to minimise the citizens' welfare dependence on the market (Powell and Barrientos, 2004, Aysan, 2009).

# 3.5.2 FRANCE

The French pension system is very complex, mainly because more than 180 different schemes coexist, but also because the compulsory social security pension is paralleled with complementary schemes organised on a socio-professional basis with a large number of exemptions (particularly among civil servants) (Blanchet and Pelé, 1999). Different schemes exist for wage-earners in the public and private sectors, people in the professions, artisans, shopkeepers, farmers, and numerous special schemes exist for specific blue collar government workers (OECD, 2004b). In a simplistic way, the basic general regime offers the possibility of retirement between the age of 60 and 65; benefits are calculated on a contributory basis: the pension depends on the number of years of contribution and on the average wage of a certain number of years of the career.

Three major reforms introduced in 1983, 1993 and in 2003 make the French security system even more complicated. In 1983 the nominal age of retirement was reduced to 60 from 65 in order to replace the unemployment allowances given to people over 60 who

lost their jobs with a universal pension; moreover, social benefits were introduced for workers who lost their jobs at younger ages than the decreased normal retirement age, de facto introducing new forms of preretirement incentives. France opted for an early retirement strategy to handle the increasing unemployment problem rather than unemployment or disability benefits (OECD, 2004b). Since the mid-90s efforts have been made to restrict access to early retirement, and many provisions have been abolished or tightened up but with several exceptions. In 1993, as a response to a decreasing retirement age, the period required for gaining a full pension at 60 was lengthened from 37.5 to 40 contribution years; simultaneously, the basis for calculating the reference wage was increased lowering the pension for those who retire earlier and pensions were indexed to prices instead of wages (Blanchet and Pelé, 1999). The strengthening of the conditions to be fulfilled in order to take full retirement at 60, though, did not apply to workers who belonged to special schemes; it affected mainly the private sector and scheduled a progressive application of the reform (mainly applying to the younger cohorts of workers). Moreover, early retirement measures were kept for workers in strenuous conditions, people with bad health and those who had started working at very young ages. These exceptions were adopted "to protect the unions' core clientele and avoid outright unions' opposition" even though they introduced "a strong element of intergenerational inequality into the system" (Natali and Rhodes, 2004)(pg. 10).

In 2003, another pension reform was approved. The contribution period for the public sector was equalised to the one in the private sector (and the legal age of retirement was raised to 65 in both sectors), even though the maximum pension level obtainable remained at higher levels for workers in the public sector. Nonetheless, the minimum statutory ages for retirement remain low in the public sector and the special schemes: the entitlement age in certain categories of workers remain 50 or 55 even though this is "linked to a status rather than to any real arduousness of the job itself" (OECD, 2004b)(pg. 66).

The 'conservative tradition' in the French pension system history can be seen in the emphasis on the preservation of status differentials. Rights are by institution attached to class and status rather than citizenship. The pension benefits are differentiated by

occupational groups and strongly dependent on the recipient's occupation. This also implies that men, given their relatively longer unbroken careers compared to women, have higher employment opportunities and are entitled to more generous pensions (Aysan, 2009). Unlike in Southern countries, though, women are not discouraged from participating in the labour market and opportunities such as part-time jobs are offered to women to increase their labour participation.

#### 3.5.3 ITALY

The Italian system includes a number of means-tested measures in order to protect old people against poverty and to decrease inequalities as well as additional earnings-based pension which nowadays depend on the number of years a person has worked. The latter feature of the pension is the result of several reforms.

Social provision of pension in Italy is a patchwork of independent schemes, typically characterised by different benefit payouts and eligibility requirements which penalise particularly those who work in the private sector, making the system unfair and unbalanced (OECD, 2004c). In general, in a quite simplified way, until the 1990s, the pension replacement rate was based on the final year's earnings (equal on average to 80% of the last wage) and there was no contributory system. No actuarial penalties were present for those who retired early, and access to retirement was possible from the age of 50 and after only 15 years of work in the public sector. Such a system heavily distorted the choice of workers, resulting in early retirement and low employment rates (particularly among women).

In the early 1990s, the Italian authorities began to take action to prolong working lives in order to prevent a financial crisis of the pension system. Three major reforms were passed (1992; 1995 and 1997) with the goal to shift the pension system towards a contributory one. The 1992 reform started a process of realignment redesigning coherently the pension system and smoothing differences out (Brugiavini, 1999). It was decided that at least 35 years of contribution were necessary to access the pension, irrespective of the type of occupation and sector. Also, the age at which people could retire was increased (between 57 and 65) and the pension level attainable became highly correlated with the age of retirement. With the later reforms, previous differences between civil servants and self-employed and between manual, agricultural and nonmanual workers in the private and public sectors were gradually abolished.

The official increase in the minimum retirement age and the lower accumulation of rights, however, will not have a significant impact on retired workers until after 2030 (by 2020 more than 75% will still have their pensions calculated on the old rules defined on earnings rather than contributions) (OECD, 2004c). Such long implementation period avoided the risk of electoral opposition and social mobilisation. In the negotiation process for reforming the system, the unions obtained a transition period, favouring older cohorts of workers with continued access to early retirement (if certain joint requirements of age and years of contributions were met) at the expense of their future counterparts who hold the major burden of the reform. Intra-sectoral inequity, though, was reduced: public sector workers in Italy gradually lost their privileged position and most measures, contribution rules and entitlements apply nowadays to both public and private-sector workers (Natali and Rhodes, 2004). The various reforms, despite the long implementation period, represented a "breaking point after which the social security system could no longer be as generous as in the past" (Brugiavini, 1999)(pg. 40).

The dominance of schemes characterised by defined-contributions and differentiated by occupation somehow reflects the legacy of '*conservative-corporatist*' welfare regimes, despite recent changes to defined-contribution schemes with fewer differences between private and public sectors. Italy still perseveres in maximising mainly the male labour force participation whereas the family (i.e. women) still play decisive roles in elderly and child care. The male breadwinner model and the traditional values of the family persist: the recent reforms have not yet undermined the traditional role of women, whose participation rate in the labour market remains comparably low even at younger ages (Aysan, 2009); the assumption that women should be looking after the family and their economic dependence on men due to low labour participation –typical of the *Southern regimes* –are still strong.

# 3.5.4 UK

The UK provides a flat-rate, virtually subsistence level pension to almost everyone (OECD, 2004d). A minimum pension is in fact offered according to the state pension age, regardless of the number of years of and level of contributions made (Blundell and Johnson, 1999). The UK population nearing state pension age can be split in two groups: those who only access state benefits, and those who also participate in an occupational private pension scheme<sup>11</sup>. Currently, 25% of the pensioners are in receipt of occupational and personal pensions (Department of Work and Pensions, 2010). Similar figures apply also to the workforce; particularly among the private sector workforce (self-employed included) the proportion of people who rely entirely on the basic state pension as an income when they retire peaks at almost 70% among lower/routine workers (Office for National Statistics, 2010). For these workers without any additional pension coverage, delaying retirement is not an attractive option since the replacement rate (compared to their average wages) is relatively higher. Moreover, despite efforts at reducing their generosity, unemployment and incapacity benefits and means-tested income supports<sup>12</sup> still provide an incentive to retire earlier and represent the main route of exit from the labour market before the age of state pension, particularly for lowincome earners. Also occupational schemes might provide routes to retirement at ages younger than the state pension age: in fact, as soon as full occupation pensions become due, workers tend to retire since working longer would not increase their future already relatively generous pensions (Blundell and Johnson, 1999).

Reforms in 1997, 2002 and 2007 have mainly tried to tackle the official age of entitlement to a full public old-age pension and of access to private provision schemes in order to increase the age at which people decide to retire. For instance, under the Pensions Act 2007, a gradual increase in state pension age for men and women from 65 to 68 between 2024 and 2046 was approved. It was also decided that the earliest age at which individuals can access private pension schemes will rise to 55 (from the previous age 50). Reforms have also tried to change the structure and level of generosity of

<sup>&</sup>lt;sup>11</sup> Some people also have non-occupational private pensions.

<sup>&</sup>lt;sup>12</sup> Pension credit, housing benefits and council tax benefits.

benefits so as to provide adequate pensions for lower-income earners without giving them stimulating their early retirement.

The pension regime and reforms, though, still reflect the liberal welfare regime principles: recent reforms continue to favour private pension plans and incentivise workers to opt for private schemes according to their needs and contributions (Blundell et al., 2006, Aysan, 2009) with government support through tax exemptions (Pension Policy Institute, 2004). Encouraging the market and its role, reforms also continue to perpetuate the dualism in the society and among workers, with the low-income workers and the poor, on the one hand, who have to rely on modest State provision, and, on the other hand, those who have access to occupational and private provisions.

#### **3.6 CONCLUSIONS**

While the challenges that ageing countries share are very similar, the responses can vary. Different countries have adjusted to the new challenges in ways that correspond with past institutional arrangements and are associated with different philosophies underlying welfare regimes. At the European Employment Strategies summit of 1997 in Luxembourg, it was recognised that the sustainability of the welfare state hinged on the ability to maintain high employment because this equals decreasing social expenditures, decreased number of people at risk of poverty and social exclusion as well as more production and higher GDP. At the European summit of 2000 in Lisbon, the EU Council agreed on a more precise formulation of the targets with respect to labour participation. The Council adopted the goal of achieving an overall employment rate among the working-age population of at least 70 percent in each member state by 2010; the employment rate among women was targeted at 50 percent among people between the ages of 55 and 64 (de Beer, 2007). Despite a European overall process of convergence with equal guidelines and targets for all member states, countries responded to economic and social pressure adopting quite distinct strategies and active labour market policies, which still reflect (and at the same time reinforce) different typologies of welfare regimes (Powell and Barrientos, 2004, Vasconcelos Ferreira and Figueiredo, 2005, Bambra and Eikemo, 2009). For example, pension reforms have been influenced by paths that are differentiated across welfare regimes. In the UK, a 'liberal' state, the market plays a role in the retirement and pensions system whereas the State remains a key determinant in Denmark, a 'Social-Democratic' country. In France and Italy, only recently have reforms tried to reduce the generosity of benefits and to reinforce the link between the level of benefits and the level of contributions (Reday-Mulvey, 2008). Moreover, Italy continues to reiterate a traditional role of the family and women.

Given these persistent differences in terms of provision of services, labour and in particular female participation, welfare regime typologies seem to be a useful way to understand possible different patterns of engagement between Denmark, France, Italy and the UK. This study, in fact, investigates the personal-level characteristics associated with various activities as well as the level of engagement across different countries: if country-specific differences were found, welfare regimes might provide a useful contextual framework to understand differences in the level of participation of older people in various activities such as paid work.

Denmark, France, Italy and the UK not only differ in employment rates and retirement policies; also their populations and their ageing processes present dissimilar patterns, despite a general demographic convergence, as the following Chapter Four highlights.

# 4. POPULATION AGEING IN EUROPE: AN OVERVIEW

This chapter describes briefly some of the main demographic characteristics of the four countries under study. Although all these populations are ageing, the pace of growth of the share of older individuals and the pace of change differ between them. Also future projections of the proportion and characteristics of older people vary and these patterns pose challenges specific to each country. The following paragraphs –focusing in particular on Europe –will introduce the phenomenon of ageing, its causes and the concerns arising around ageing. More detailed information on recent life expectancy and fertility trends are then presented for the four countries under study; the current population and future projections are then described.

# 4.1 AGEING AS A GLOBAL PHENOMENON

Over the past two centuries, a global transformation has occurred across most of the World: the population has been ageing, namely older individuals have become a proportionally larger share of the total population (UN, 2001). Such a phenomenon is due mainly to long term fertility declines with, more recently, increasing longevity leading to further population ageing (Livi Bacci, 1999). The rate of population aging may also be modulated by migration, although the role of international migration in changing age distributions of the host populations has been fairly modest (UN, 2000) and is less important than that of fertility and mortality (Lesthaeghe, 2000)<sup>13</sup>. While the ageing of the population is unique within every country, the global nature of ageing prefigures some common experiences among countries: "changes will impact all aspects of human life –from family composition, living arrangements and social support to

<sup>&</sup>lt;sup>13</sup> Past trends in migration seem likely to have had a more substantial effect on population totals as well as ethnic composition than variation in fertility or mortality. Their effect on age-structure, moreover, is modest (Coleman, 2003). Data on immigration, birthplace and proportion of births to immigrants are also difficult to retrieve, particularly for the pre-1990s period. For these reasons, this chapter will not consider in detail data on migration trends.

economic activity, employment rates and social security and to transfers between generations" (Peace et al., 2007)(pg. 2).

At the global level, in 1950 around 5% of the population was at least 65; by the year 2000, that proportion had increased to around 7% (with more than 28% of the world population aged 0-14). By the year 2050, more than 15% of people throughout the world are projected to be aged 65 or over (UN, 2001).

In Europe, according to the European Statistics Bureau (Eurostat, 2008), between 1960 and 2005 the percentage of children <sup>14</sup> (aged 0-14) dropped from 26.7% to 15.9% whereas older people  $(65+^{15})$  during the same period almost doubled their proportion rising from 8.8% to 15.9%. By 2050, the share of people aged 65 or more is projected to rise to almost 30% (Eurostat, 2008) and it is highly probable that by the same year those aged 80 and over will constitute at least one in ten of the European population (Grundy, 2010).

Europe is already facing major challenges in relation to the ageing population and, according to projections, will continue dealing with an ageing population, though to different extents in different countries, as will be illustrated below. For instance, whereas those aged 65+ account for about 20% of Italy's population (in 2010) with figures projected to rise to 35% by 2050, Denmark is expected to have, by the same year, one of the lower representations of older people in Europe with fewer than 25% of Danish people aged more than 65 (Eurostat, 2008).

# 4.2 'CHALLENGES' RELATED TO AN AGEING POPULATION

Increased survival is a great achievement (WHO, 2001a, 2002) although population ageing is often portrayed negatively. The main concerns arising from population ageing<sup>16</sup> centre upon pension costs (and more generally income support costs), the likely

<sup>&</sup>lt;sup>14</sup> Data presented in the 'Europe in figures – Eurostat yearbook 2008' refer to the 27 Member States of the EU (EU-27). Time-series data refer to a sum or an average for all 27 countries as if all 27 Member States had been part of the EU in periods prior to the beginning of 2007 (i.e. when EU officially included 27 Member States).

<sup>15</sup> Throughout this chapter, the conventional age of 65 will be used to describe the older proportion of the population, despite numerous critiques (Egidi, 1988, Cagiano De Azevedo and Cassani, 2005). <sup>16</sup> To reverse population ageing a rapid population growth would be necessary, which itself poses

<sup>&</sup>lt;sup>16</sup> To reverse population ageing a rapid population growth would be necessary, which itself poses challenges (Livi Bacci, 1999).

end of the recent growth seen in labour force size, the availability of family support as well as health care costs. The general assumption is that the process of population ageing will challenge younger age groups to provide for a growing share of the old. For instance, it is believed that the ageing of population will increase pressure on public expenditures: fewer people in employment (i.e. a reduced tax-base) will have to finance with larger contributions of various forms (such as taxes and social security contributions) a rapidly growing number of people outside the labour market, with negative impacts on economic growth. Given that the extra burden of taxation is likely to fall on a shrinking labour force, current demographic changes also, it has been argued, may create intergenerational inequities and tensions between younger workers and older pensioners, particularly where participation rates of older workers in the labour market remain low (OECD, 2004a, c, Grundy, 2010).

Although recent studies suggest that population ageing has a limited marginal impact on health care costs (Richardson and Robertson, 1999, Zweifel et al., 1999, OECD, 2004d), it is believed that changes in patterns of family-related behaviours and in living arrangements might reduce the availability and supply of informal care, augmenting the demand for formal long-term professional care. It has been pointed out that the decreased average number of children per women, the increased participation of women in the labour market, increased divorce rates as well as the increased proportion of older people living alone (particularly among older women) might all contribute to a reduction of the availability of family support for older people in need of assistance (Bolin et al., 2008, Eurostat, 2008).

# 4.3 AGEING IN ITALY, FRANCE, DENMARK AND ENGLAND

#### **4.3.1 DIFFERENCES WITHIN SIMILAR TRENDS**

The two major drivers of populations ageing are the shift from relatively high to low fertility rates and improvements in the overall level of mortality, particularly at older ages. The combination of these two driving factors determines the rate at which populations age.

The total fertility rate (TFR) is the average number of live children that a group of women would have if they experienced the age-specific fertility rates of a specific calendar year throughout their childbearing lifespan. The TFR in Italy has fallen from almost 2.5 in 1970 (i.e. above the replacement rate of 2.1) to 1.24 in 2000 (Figure 4.1). Denmark experienced an accelerated decline in fertility from 1967, reaching the lowest level so far in 1983 (1.4 children per woman), after which fertility began to increase again peaking almost 1.8 children per women in 2006. France has had a slightly higher birth rate than the other countries under study: from the mid 1970s until the late 1990s, the French TFR fluctuated around 1.80, reaching its lowest point in 1993/4 (with a TFR equal to 1.66); since 2001, however, the French TFR has increased to almost 2 in 2006. As for the England and Wales, after an increase by more than 30% between the early 1950s and the mid 1960s, the total fertility rate fell from 2.9 to less than 1.7 in 1977. Between the last years of the 1970s and the late 2000s, the total fertility declined gradually to reach 1.63 in 2001, its lowest level, to then recover to almost 1.9 in 2006.



Figure 4.1 Total Fertility Rates, 1950-2006 - selected countries

Total Fertility Rates (. Data refers to the period 1950-2006. Sources: Birth statistics from Insee, Division des enquêtes et études démographiques (www.insee.fr, France); Danmark Statistik Befolkning (www.statbank.dk, Denmark); Istat – Istituto nazionale di statistica (www.istat.it, Italy) and UK National Statistics (www.statistics.gov.uk; England and Wales).

The second major explanation for population ageing is improvements in the overall level of mortality, often summarised using measures of life expectancies at specific ages (for example, birth and at 65) which give a rough idea of how many years an average individual at a certain age can expect to live after this age, if there are no changes<sup>17</sup> in future death rates. As Table 4.1 shows, Italy and France have among the highest male and female life expectancies at birth and at 65. In 2006, life expectancy at birth reached

<sup>&</sup>lt;sup>17</sup> Life expectancy is in fact calculated on current mortality rates.

78.4 years for Italian men and 84.3 for French women, much higher than England and Denmark. The main differences among the four countries under study are observable when the gains in life expectancy (at birth and at 65) are compared. Whereas among Danish men life expectancy at birth has increased by around six years during the past five decades, in England and Wales men gained almost 11 years during the same period of time. In France and Italy, life expectancy for men increased by 13 and almost 14 years respectively. Gaps are even bigger if women are considered: in Italy, life expectancy at birth has increased by 16 years, almost double the gain in Denmark.

Moreover, further life expectancy of men aged 65 ranges between 16 (among Danish men) and 18 (among French men) whereas among women this ranges between 19 in Denmark and 22 in France. Increases between the mid 1950s and 2006 are quite different among the different countries: whereas Danish men and women gained only 2.3 and 4.6 years respectively, in France male life expectancy at age 65 increased by 5.9 years and for women by 7.7 years.

 Table 4.1 Life Expectancy at Birth and at 65 (in years). Comparisons between periods, selected countries

	1950-1955				2006			
	Men		Women		Men		Women	
La de la desta	At birth	At 65	At birth	At 65	At birth	At 65	At birth	At 65
Denmark	69.6	13.8	72.2	14.4	75.9	16.1	80.4	19.0
France	64.2	12.1	70.2	14.6	77.2	18.0	84.3	22.3
Italy	64.5	12.9	68.1	14.1	78.4	17.8	84.0	21.6
England & Wales	66.9	11.8	72.1	14.6	77.5	17.4	81.7	20.1

Life expectancy at birth and at 65 (in years). Data refers to the period 1950-1955 and to the year 2006. Sources: for the 1950-1955 period, data was accessed through the *Human Mortality Database* (University of California Berkeley (USA) and Max Planck Institute for Demographic Research (Germany)) available at www.mortality.org. Data for 2006 was obtained via the national statistics bureaus, www.statbank.dk (Denmark), www.insee.fr (France), www.istat.it (Italy) and www.statistics.gov.uk (England and Wales).

#### **4.3.2 CURRENT POPULATIONS**

The increasing trends of longevity together with decreasing fertility rates have contributed to shaping the current age composition and will keep transforming it over the following decades. Italy, for instance, which experienced a substantial decrease of both mortality and fertility rates, has one of the oldest age-compositions among European countries: in 2006, only 14% of the population was younger than 15, whereas people older than 65 doubled compared to 1961 representing almost 20% of the total population in 2006 (Eurostat, 2008). The relatively similar TFR trends after the 1990s in France, Denmark and England and Wales are reflected in similar proportions of people aged 0-14 at around 18% in 2006. As regards the older share of the population, in France, Denmark and England and Wales, between 15% and 16% are aged over 65. In Denmark, though, because of the relatively smaller gains in longevity, the proportion of over 65 is slightly smaller than in France and the UK (see Table 4.3).

	0-14	15-24	25-49	50-64	65-79	<b>80</b> +
Denmark	18.7	11.2	35.1	19.9	11.1	4.1
France	18.6	12.9	34.4	17.9	11.6	4.6
Italy	14.1	10.3	37.5	18.3	14.6	5.1
England and Wales	17.6	13.4	35.0	17.9	11.5	4.6

Table 4.2 Population by broad age-groups (% distribution), 2006

Population by age group, 2006 [2007 for England and Wales]. Sources: Eurostat, 2008 (Italy; France; Denmark); Office for National Statistics, 2009 (England and Wales).

Focusing in particular on people aged 65 and more, Italy shows the higher percentages of older people. In 2006, almost one in ten Italian people are aged 75 and more compared to circa 7% in Denmark and England and Wales (see Table 4.4.).

	Dentation	Percentage (and Absolute Number) of the Entire						
Country	Population	ropulation aged						
		65+	75+	85+				
Denmark	5,427,459	<b>15.16</b> (823,027)	7.01 (380,633)	1.90 (103,294)				
France	61,166,822	16.44 (10,058,762)	8.24 (5,039,778)	<b>1.96</b> (1,201,927)				
Italy	58,751,711	<b>19.73</b> (11,592,335)	<b>9.24</b> (5,431,005)	2.11 (1,244,556)				
England & Wales	54,072,643	16.07 (8,690,594)	7.81 (4,221,687)	2.17 (1,172,126)				

Table 4.3 Size and Age structure of the older population in selected countries, 2006

The first column shows the total number of residents. The following columns show the percentages and absolute values of the fraction of the population aged over 65, 75 and 85 respectively. Data Sources: www.insee.fr (France), www.statbank.dk (Denmark); http://demo.istat.it (Italy) and www.statistics.gov.uk (England and Wales).

Considering also the sex distribution of the older people in the four different countries (see Table 4.4), data show that women are more numerous than men in all age groups. This reflects higher life expectancies for women (see Table 4.1). In all countries under study, between 56% and 59% of people older than 65 are women. Restricting the analysis to people older than 85, the disproportion between men and women is even higher with only about 30% of men. This deficit of men in the oldest cohorts is augmented to some extent by war-related mortality.

		65+ (%)	75+ (%)	85+ (%)
×	Males	341,347 <b>(43.2</b> )	140,169 ( <b>38.1</b> )	29,651 ( <b>29.6</b> )
	Females	448,780 ( <b>56.8</b> )	227,651 ( <b>61.9</b> )	70,590 ( <b>70.4</b> )
×	Males	4,134,951 <b>(41.1</b> )	1,838,168 ( <b>36.5)</b>	341,669 ( <b>28.4</b> )
	Females	5,923,811 ( <b>58.9</b> )	3,201,610 ( <b>63.5</b> )	860,258 (7 <b>1.6</b> )
<u> </u>	Males	4,821,967 ( <b>41.6</b> )	1,995,778 ( <b>36.7</b> )	361,832 ( <b>29.1</b> )
Ĺ	Females	6,770,368 ( <b>58.4</b> )	3,435,227 ( <b>63.3</b> )	882,724 ( <b>70.9</b> )
Ň	Males	3,775,548 ( <b>43.4</b> )	1,648,718 ( <b>39.1</b> )	366,159 ( <b>31.2</b> )
E&	Females	4,915,046 ( <b>56.6</b> )	2,572,969 ( <b>60.9</b> )	805,567 ( <b>68.8</b> )

Table 4.4 Age-Sex Distribution of the older population, 2006

Age-Sex Distribution of people older than 65 in 2006. For each country, data show the percentage as well as absolute values of males and females in each age-group (+65; +75 and +85). Data Sources:

www.insee.fr (France), www.statbank.dk (Denmark); http://demo.istat.it (Italy) and www.statistics.gov.uk (England and Wales).

#### 4.3.3 FUTURE TRENDS

The trends in fertility and mortality discussed above affect not only the present but also the projected age distribution of populations. Figure 4.2 shows projections for the 2020 and the 2035 populations of the four countries under study. Because Italy's fertility rate is projected to remain well-below replacement level and the number of women of childbearing age has been decreasing, each birth cohort is projected to be smaller than the one above it in the population pyramid. By 2035, the base (ages 0-4) of the pyramid for Italy is projected to be not quite as wide as it for 2020, reflecting even fewer births in the later year. The 2020 pyramid has a bulge for those aged 45 to 55 years (who were born between 1965 and 1975) which reflects the rapid increase in fertility (the 'baby boom') that preceded the sharp decline of the TFR in 1973. Such a 'bulge' is projected to move up in coming years inflating the proportion and the number of older people for a period. The pyramids for Denmark and the UK<sup>18</sup> are more jagged in shape: for instance, in 2020 there are two bulges in the Danish projection, one for those aged 20-29 (which reflects the pickup of the TFR in the early 1990s) and one for those aged 55 to 59 (reflecting the high fertility rates in the early 1960s before the rapid fall after 1966). Finally, France is projected to have a more rectangular-shaped age structure. In the coming decades, though, the relatively large cohorts of people born between the 1950s and mid-1960s (aged 55-70 in 2020) will continue to shift upward swelling the number of older people by 2035. At the same time, the base (ages 0-14) of the pyramid are projected to slightly narrow by 2035. In general, Italy is projected to be the only country to have a much smaller base compared to the other four under study: the proportion of women aged 0 to 4 years is projected to be as small as 2% of the total population by 2035, compared to 2.8 in Denmark and 2.7 in France and the UK.



Figure 4.2 Projections of Age-Sex Pyramids in 2020 and 2035

<sup>&</sup>lt;sup>18</sup> In this case, data available for projections refer to the entity UK rather than to the sub-national population of England and Wales (E&W), preferred in all other tables and graphs.



Age-sex Pyramids projected at 2020 and 2035 for Denmark, Italy, France and the UK. Source: National Statistics Bureaux web-pages; projections are based on the "medium/central" assumptions. For details about hypotheses, refer to the websites www.istat.it (Italy); www.insee.fr (France); www.dst.dk (Denmark) and www.statistics.gov.uk (UK).

The changes in population age composition shown above will affect the percentage of the population that is typically defined as of working age (i.e. aged 20-64) as well as the percentage of population that is of "dependent" age  $(0-19 \text{ and } 65+)^{19}$ . People aged 20-64 are members of the population who *can* contribute to the economy, even though –of course –people older than 65 may still work and not necessarily all people aged 20-64 work. It is generally presumed that, on average, people aged 15-64 'produce' more than they 'consume' while the opposite is true for those who are younger and older. Although there is no general consensus about the 'dependency' concept and rates, the 'age-dependency' ratios are still very well known measurements used by the majority of the European and international organisations (such as OECD, Eurostat, UN as well as national statistics bureaux). The use of this indicator enables us to monitor changes in population structure (given that their evolution is a function of mortality, fertility rates

<sup>&</sup>lt;sup>19</sup> People aged less than 20 and who are older than 65 are presumed to be too young or too old to support themselves through labour market activity and therefore need to be supported by others, typically the family or the state.
and of net migration) using measures which are comparable to the main literature (Calasanti and Bonanno, 1986, Robertson, 1997, MacInnes, 2006, Verdugo, 2006).

Looking at Eurostat projections for the next decades (see Tables 4.6 and 4.7), the current lower fertility rates in Italy will have a bigger impact on the proportion of older people compared to Denmark, France and the UK, as expected looking at Figure 4.2. Under the Eurostat demographic assumptions (see Eurostat (2008) for more details), whereas in France, Denmark and the UK the relative weight of the share over 65 will equal a quarter of the total population by 2040, this threshold in Italy will be exceeded by 2030. Similarly, in Italy one in three is projected to be over 65 by 2050 compared to around 27% in France and the UK and circa 24% in Denmark. Conversely, in Italy the share of the under-15s in the population is projected to decrease from 14% to less than 12% by 2030 to then remain at the same level for the following decades. Also in France, Denmark and the UK the proportion of people under 15 is projected to fall by roughly an absolute value of 3% reaching its lowest point in 2050 at about 16% in France and Denmark and about 15% in the UK.

	2005	2010	2020	2030	2040	2050
Denmark	18.8	18.0	16.0	16.3	16.5	15.7
France	18.5	18.2	17.2	16.3	16.0	15.8
Italy	14.2	14.0	12.9	11.6	11.4	11.2
UK	18.0	17.0	16.3	15.8	14.9	14.7

 Table 4.5 Projection of the proportion of the population aged under 15 (%)

Source: Eurostat. Trend scenario, national level - base year 2004

1 able 4.0	rrojecuon	or me prop	portion of u	ie populati	on ageu ove	F 03 (70)
	2005	2010	2020	2030	2040	2050
Denmark	15.0	16.3	20.0	22.6	24.7	24.1
France	16.5	16.8	20.7	24.2	26.8	27.3
Italy	19.5	20.5	23.3	27.5	33.1	35.3
UK	16.1	16.6	19.5	22.9	25.9	26.6

Source: Eurostat. Trend scenario, national level - base year 2004

Even though the projections are based on assumptions about the future which may not prove to be correct (particularly the ones about fertility levels), certain past trends and their consequences are less uncertain. For instance, past trends of family formation (such as number of average children and rates of marriage) determine the proportion of older women who are childless and the support they might get from their family. In fact, whereas the TFR and the life expectancy indicators are synthetic 'hypothetical' summary measures which refer to what a group of people would experience if they were to experience the prevailing age-specific fertility and mortality rates of a specific calendar year throughout their lives, cohort measures such as the 'completed TFR' are actually based on the observed fertility of a real cohort of women born in a certain year who has completed childbearing and has been followed over the lifetime to count up the total number of children borne by those women.

Looking at tables 4.7 and 4.8 which show the completed TFR and the percentage of childless women by cohort respectively, we note that in both Italy and the UK the percentage of older women who are childless is currently decreasing and will do so until 2015, i.e. when those women born in 1950 and later reach the age of 65. In the coming quarter of century, by 2030, almost one in five women in both Italy and the UK will be childless. In France and Denmark, on the contrary, the proportion of childlessness among women born after 1945 has been more stable at around 10%, and therefore changes are not expected in the short and medium terms.

				N	<b>a</b>		,	<b>v</b>
	1930	1935	1940	1945	1950	1955	1960	1965
Denmark	2.38	2.36	2.24	2.06	1.90	1.84	1.90	1.89
France	2.63	2.57	2.41	2.22	2.11	2.13	2.11	2.02
Italy	2.28	2.28	2.14	2.07	1.89	1.80	1.66	1.49
UK	2.38	2.45	2.39	2.22	2.09	2.03	1.98	1.90

Table 4.7 Completed Total Fertility Rate by birth cohorts, selected years

Source: Council of Europe (Comité européen sur la population du Conseil de l'Europe, 2005)

Table 4.8 Percentage of women	childless at	age 45	5 by	birth	cohort,	various
	countries					

	1930	1935	1940	1945	1950	1955	1960	1965
Denmark	n.a.	n.a.	n.a.	8.5	10.5	11.7	11.6	12.7
France	13.0	10.5	8.3	8.1	8.3	8.3	10.2	n.a.
Italy	13.5	12.5	12.0	9.8	11.3	12.7	15.3	21.2
UK	13.0	11.4	10.7	9.8	13.9	17.0	18.9	20.5

'N.A.' stands for 'Not Available'. Data refers to women 45 years old. Source: OECD Family Database, accessed through www.oecd.org on 12/06/2011. Similarly, the distribution of family sizes has changed with differences between countries. In France, the average number of children produced by women born between 1930 and 1965 declined constantly, even though women born in 1965 still had about two children each. In Denmark and Italy, despite a similar trend, women started having less than 2 children by 1950; this means that the proportion of women aged 65 having less than 2 children will start increasing by 2015<sup>20</sup>. Finally, the UK also experienced a decreasing family size, with women born in 1960 and onwards having less than two children each. In the UK, though, women born 1935 had very large families compared to the previous and subsequent cohorts: this means that proportion of women aged 65 with the higher number of children available peaked in 2000 whereas in the following vears women are likely to have fewer and fewer children $^{21}$ .

### 4.4 **CONCLUSIONS**

Italy, France, Denmark and the UK have experienced different life-expectancy and fertility trends. For instance, France, Denmark and the UK -which were among the first countries to experience sustained below-replacement fertility in the late 1960s and early 1970s exhibit relatively high fertility in the early 2000s. Moreover, the Danish, French and British TFRs have fluctuated around levels of 1.8 in the 1990s, in sharp contrast to the pervasive TFR levels below 1.3 in Italy. Similarly, Italy and France have among the highest male and female life expectancies at birth and at 65 and, between War World II and the mid-2000s, have also experienced the highest gains in life expectancy (both at birth and at 65) compared to Denmark and the UK. The current populations of Italy, France, Denmark and the UK, as a consequence, appear to be quite dissimilar despite a general ageing phenomenon common to all four countries under study. Italy shows the higher percentage of people aged 65 and more. Also current projections suggest that both the proportion and the number of older people in Italy are increasing at a much higher rate compared to the other countries. In Italy one in three is projected to be over 65 by 2050 compared to around 27% in France and the UK and less than a quarter in Denmark. Similarly, the relative number of pensioners is expected to increase

<sup>&</sup>lt;sup>20</sup> This prediction does not account for the sex-specific mortality rates of the offspring, i.e. it does not refer to live children.<sup>21</sup> This prediction does not account for the sex-specific mortality rates of the offspring., i.e. it does not

refer to live children.

(Bongaarts, 2004), unless other substantial changes to the age of retirement will be introduced. Population ageing raises social and economic concerns, though governments are currently promoting 'active ageing' policies to achieve a much more positive vision of the ageing experience and tackle some of the issues associated with it. Before analysing whether a prolonged 'active' life may also create benefits for older people's health, chapter Five and Six introduce the datasets used: their rationale, instrument designs and implementations are described in the following Chapter whereas Chapter Six presents some descriptive statistics.

# 5. ELSA AND SHARE QUESTIONNAIRES: AN OVERVIEW

This chapter introduces the two datasets that I used in this research project. First some brief rationale for the implementation of both the English and the European longitudinal surveys on ageing are highlighted. Information on methods and sample designs follows. The development and the content of the questionnaires are then presented. Baseline and longitudinal individual response rates (by age-groups and sex) are the described. Finally, the issue of weighting and some general critiques of the questionnaires are introduced.

### 5.1 SOME BACKGROUND RATIONALE

The current phenomenon of the ageing of the population is unprecedented in Europe, as described in details in the chapter Four. The ways in which people live and experience their own old age, though, is by no means uniform: even within particular cohorts economic or gender differences may arise; moreover, expectations, policies around retirement and pension provision as well as self-perceptions of older people, their values and role in the society change over time highlighting both generational and time-period-specific effects.

Because of the above-mentioned reasons the need emerged to document the ageing process, its diversity and complexity and to investigate its implications both for individuals and the wider community and economy. It is precisely to understand the variability in the ageing phenomenon that longitudinal studies of ageing were initially established. Among the principal benefits, these studies could provide detailed pictures of the diversity of ageing trajectories, take account of changes over time and also provide some insight into directions of association. They could also present a more scientific basis for short- and long-term policies and interventions.

The English Longitudinal Study of Ageing (ELSA) was established in England with the support of various departments of the UK Government as well as the US National Institute on Aging (NIA). ELSA was shortly followed by the Study of Health and

Retirement in Europe (SHARE) in several European countries providing data to compare older people living under different arrangements of health and social services and economic policies both within Europe and with the US Health and Retirement Study (HRS) and to inform policy debate (Marmot et al., 2003b, Börsch-Supan et al., 2005, Taylor et al., 2007).

### 5.2 BRIEF HISTORY OF BOTH ELSA AND SHARE QUESTIONNAIRES

ELSA was planned to be directly comparable to its US counterpart, the Health and Retirement Study (HRS). The initial questionnaire design meetings took place in late 2000 and involved a range of diverse experts from different fields, survey specialists, academics, advisory groups as well as collaborators from the HRS. In its initial stages, experts were asked to discuss and test specific modules or review elements of the survey: this formative stage of the development of the survey instruments was characterised by lively debates (Taylor et al., 2003). Two pilots were finally conducted in August and November 2001 in order to test the survey instruments and fieldwork approach. The final version of the ELSA questionnaire was then administered to the sample in 2002.

The history of SHARE is quite similar and this questionnaire also took its cue mainly from the US HRS (Börsch-Supan and Kemperman, 2005, Kapteyn, 2006). "A very early discussion about setting up a European HRS took place during an international conference on ageing in Amsterdam in the summer of 1997" (Kapteyn, 2006)(pg. 1) when it was agreed that an instrument which could 'fill the gap' and overcome the lack of knowledge on ageing and the interaction between socio-economic and health factors in the European countries was necessary.

Within 4 years, the feasibility of an international comparative European research project was explored: in 2000 a number of European researchers came together and discussed the possibility of launching a European variant of HRS and in 2002 the first grants were obtained primarily from the European Commission (with additional funding from the US National Institute on Ageing). Once the money was secured, the creation of the SHARE questionnaire was relatively short: a pre-test was ready to be submitted in all countries by June 2003 and just after a period of about one year the first wave of data

was collected in 2004. The prompt design of SHARE questionnaire is mainly due to the previous pioneering work conducted in both the US and the UK; in fact, the SHARE questionnaire is inspired by and modelled on both the ELSA and HRS.

Collaboration with HRS and ELSA was a major advantage: their researchers were actively involved as advisors in SHARE providing support as well as technical assistance. Moreover, such a close cooperation helped SHARE to build its questionnaire so as to maximize comparability of data and therefore extend comparisons from the Anglo-Saxon countries to continental Europe. Learning from a variety of European countries with striking differences in terms of culture, history or policies and "being able to anticipate what will happen in one country by looking at another country" was also an important task for SHARE (Börsch-Supan and Kemperman, 2005).

Both ELSA and SHARE have similar features: they both are *multidisciplinary longitudinal surveys*. SHARE and ELSA (as well as HRS) consider a number of diverse domains of an individual's life. Income security, personal wealth, social networks and family as well as health are strongly linked with each other so a holistic approach –i.e. a multidisciplinary survey –was considered necessary. The time dimension was essential from the start of both surveys: observing individuals as they age over time and adapt to and are influenced by changes in family, health and wealth circumstances has the potential to provide insights into the ageing processes. Moreover, the time dimension is vital to investigate directions of associations and to better identify effects that public policies may have on the behaviour and well-being of citizens (Marmot et al., 2003a, Börsch-Supan et al., 2005). Both ELSA and SHARE follow and re-interview the same individuals every two years.

### 5.3 INSTRUMENT DESIGN AND IMPLEMENTATION

### 5.3.1 ELSA

The English Longitudinal Study of Ageing (ELSA) is a longitudinal survey of ageing and quality of life among people older than 50 living in private households in England. The design of the survey questionnaire began in 2001, when two pilots were tested. The initial draft questionnaire was tested in August 2001 in order to provide reasonable feedback about questions in each sub-model. Changes followed: in particular, the order of the modules was revised and the need of the self-completion questionnaire (with additional questions on social participation and measures of well-being and quality of life) was established. The second version of the questionnaire incorporating the resulting amendments was again tested in November 2001 (Taylor et al., 2007). The first wave of ELSA interviews then took place between March 2002 and March 2003 and the project is still ongoing with interviews being repeated biennially.

The ELSA sample is representative of people born before the first of March 1952 (i.e. older than 50 at the time of the questionnaire) and *living in private households* in England at the baseline. The sample was drawn from households that had previously responded to the Health Survey for England (HSE) in 1998, 1999 and 2001 which altogether comprised a total of 23,132 responding households and 43,200 individuals. HSE samples were drawn in two stages: postcode sectors were initially selected from the Postcode Address File and stratified by health authority and proportion of households in non-manual socio-economic groups; secondly, from each postcode sector, addresses were systematically randomly selected. Full details about the HSE are available from the Technical Reports (Erens and Primatesta, 1999, Erens et al., 2001, Prior et al., 2003).

From the households identified in the HSE, 57% of them had at least one individual older than 50, i.e. 13,203 households (see Figure 5.1). Of these, 11,578 (i.e. 88%) agreed to be re-contacted and were therefore included in the final sample frame for ELSA <sup>22</sup>. Considering only the ELSA households issued (see Figure 5.2), contact with the eligible households was possible in 95% of cases<sup>23</sup>: only 73% of the households contacted agreed to be take part into the ELSA research project. A household response rate of 70% was therefore achieved among the ELSA households issued. Within the sampled

<sup>&</sup>lt;sup>22</sup> During the fieldwork, the actual number of potential eligible households became 11,373 containing 18,576 individuals: 296 households had become ineligible in the meantime (mainly because of the death of people or because they had moved into an institution or to another country) whereas 91 new households had formed (mainly due to divorce) (Taylor et al., 2007).

<sup>&</sup>lt;sup>23</sup> The majority of the remaining 5% of the households not contacted had moved and could not be traced.



Tree diagram summarising the process of the selection of the ELSA sample. The tree reads from the top to the bottom. The shaded areas of the tree diagram represent the number of household that were dropped. (Note: Households and individuals issued individuals do not necessarily correspond to the eligible ones). This figure is taken from Taylor et al. (2007). Note: a) This percentages are calculating considering the total households containing at least one eligible individual as the base, i.e. 13,203 households.



This figure shows the number of households and individuals eligible to take part in ELSA. The response tree reads from top to bottom and shaded boxes show the subgroups of the issued sample who were not interviewed because ineligibility or non-response. This figure is taken from Taylor et al. (2007)

households, 96% of individuals responded to the questionnaire. A final individual response rate of 67% was therefore achieved (Taylor et al., 2007). This equals a total of 12,099 interviews (11,391 with respondents aged 50 years and over; 577 with partners under 50 and 131 with "new partners"). Table 5.1 shows the distribution by age-groups and sex of the achieved sample in wave 1. The full details and explanations of the sampling frame and design and on response rates have been reported elsewhere (Marmot et al., 2003a).

Age at wave 1	Male	Female	Total
<50	105	472	577
50-64	2,762	3,210	5,971
65-74	1,485	1,709	3,195
75+	983	1,373	2,356
Total	5,335	6,764	12,099

Table 5.1 Achieved	sample by a	ge and sex,	ELSA	(wave 1)	
--------------------	-------------	-------------	------	----------	--

Distribution by age and sex of the sample achieved in the first wave of ELSA; figures include the 'core sample' (i.e. all individuals older than 50) as well as 'new' and younger partners. Source: ELSA, 2002.

### 5.3.2 SHARE

The Survey of Health, Ageing, and Retirement in Europe (SHARE) is also a multidisciplinary survey which interviewed individuals aged 50 and over in several European countries (Germany, Switzerland, Belgium, Greece, Netherlands, Italy, Austria, Sweden, Denmark, France, Spain with the addition of Israel, the Czech Republic, Poland, and Ireland from 2006). Because of the initial involvement of 11 countries each speaking a different language and with different cultures and histories, the SHARE questionnaire had to be designed so as to take into account institutional and cultural variations across countries, while maintaining comparability of the collected data as far as possible. A comparable structure of the instrument was reached through meetings where the questionnaire, its sections as well as questions to be asked were discussed and agreed among all working groups and participating countries.

The resulting first draft of the questionnaire was produced in 2002 in English and in September of the same year this was piloted in the UK by the National Centre for Social Research (NatCen). After a number of experiments and pilots, the questionnaire was thoroughly revised and then translated into the languages of the different countries with the help of the Language Management Unit (Harkness, 2005). Once the translation was found to be satisfactory, in June 2003 a final pilot took place in all SHARE countries using quota samples of about 75 people in each country. This led to the final instrument, after final adjustments and amendments. In particular, country-specific questions and specific answer categories were introduced when institutions were fundamentally different (health care, pension system, education just to cite some). The final version was fielded in the participating countries by local survey agencies between April and October 2004 (Borsch-Supan, 2005, Börsch-Supan and Kemperman, 2005, Harkness, 2005).

National survey agencies, selected within each country based on a tender, were responsible for the sampling and carried out the fieldwork following the protocol and the guidelines punctiliously defined centrally by the SHARE coordinating team (MEA of the Mannheim University) who also monitored the field work in parallel to the national survey agencies. Lack of a European agency meant that the sampling design

was conducted differently in each country. Depending on the availability of and accessibility to population registers as well as the possibility of cooperating with the national statistic bureaux, sampling designs used varied from simple random selection to rather complicated multi-stage designs (full details are given elsewhere (Klevmarken et al., 2005b)). In Denmark, a simple random design was used. In France and in Italy, multi-stage designs were adopted. Only 6 regions were drawn in the French sample; the degree of urbanisation would then determine strata from which primary sampling units were drawn. In Italy, 15 municipalities (strata) were randomly drawn without replacement; then, within each stratum, a number of electoral divisions (primary sampling units) were randomly selected.

About 18,000 households and 28,000 respondents were successfully interviewed (SHARE). The target population consisted of all individuals living in residential households<sup>24</sup> who were older than 49, who resided in the sampled address and who spoke the language of the national questionnaire. The SHARE team also interviewed spouses of household members who were older than 50, regardless of their age. The overall household response among SHARE countries participating in the 2004 survey was 61.6%<sup>25</sup>. Individual response rate (i.e. the proportion of eligible individuals within the eligible households interviewed that completed the questionnaire) was in 2004 on average equal to 85.3% (see Table 5.2 for details) with "only small differences in the patterns of survey participation by gender and age group" (De Luca and Peracchi, 2005b)(pg. 355).

 <sup>&</sup>lt;sup>24</sup> Few exceptions apply: in Denmark, for instance, the target population included also individuals living in institutions for elderly people (but not those living in prisons or similar institutions for instance)
 <sup>25</sup> http://www.share-project.org/t3/share/index.php?id=97 - SHARE release 2.0.1. Last accessed on 11<sup>th</sup> of May 2010.

A DESCRIPTION OF THE OWNER OF THE					(	-/		
Country	Total	М	W	50-64	65-74	75+	Individual RR* (%)	HH RR (%)
Austria	1,893	782	1,111	949	544	356	87.5	55.6
Belgium	3,827	1,739	2,088	1,991	986	672	90.6	39.2
Denmark	1,707	771	936	916	369	330	93.0	63.2
France	3,193	1,386	1,807	1,648	759	631	92.3	81.0
Germany	3,008	1,380	1,628	1,569	886	486	86.4	63.4
Greece	2,898	1,244	1,654	1,458	712	499	92.4	63.1
Italy	2,559	1,132	1,427	1,342	785	381	79.1	54.5
Netherlands	2,979	1,368	1,611	1,693	713	459	88.0	61.6
Spain	2,396	994	1,402	1,079	701	573	73.7	53.0
Sweden	3,053	1,414	1,639	1,589	816	592	84.4	46.0
Switzerland	1,004	462	542	505	251	204	86.9	38.8
Total	28,517	12,672	15,845	16,155	8,212	5,530	85.3%	61.6%

Table 5.2 Distribution of Respondents by country, sex and age. Individual and household response rates (RR)

This table shows the sex and age distribution of respondents in SHARE countries; respondents younger than 50 are not shown. Also Individual and Household (HH) Participation Rates are given. Note: \* stands for "weighted average". Source: <u>http://www.share-project.org/</u> (last accessed on 14<sup>th</sup> of September 2009).

### 5.4 LONGITUDINAL SURVEY PARTICIPATION FOR ELSA AND SHARE

Both ELSA and SHARE are longitudinal surveys: this means that the respondents in wave 1 (the baseline) were approached and interviewed again every two years. The ELSA wave 2 took place in 2004-05 and fieldwork for the third wave took place between May 2006 and August 2007. During the same period (2006/07), the SHARE research project collected the wave 2 data providing panel data from respondents who had been interviewed during the first wave. This longitudinal dimension is of crucial dimension when studying the ageing process as this enables to put a stress on evolutions and transitions over time (Barangé et al., 2008).

Either all wave 1 respondents or a restricted version of it (which excludes baseline respondents who became ineligible over time) can be considered as the base (i.e. the denominator) to measure longitudinal response rates. Among the criteria which render respondents ineligible, death, refusal to be re-contacted and moving out of the country are the most common ones. The ELSA team also labelled as ineligible baseline members who moved into an institution in wave 2 whereas they became eligible in wave 3.

Among ELSA Cohort 1 core members, 81% (i.e. 8,781 people) were successfully interviewed also at wave 2 and almost 70% (i.e. 7,168 people) were successfully interviewed also in wave 3. These figures apply if respondents who became ineligible by the time of each subsequent wave are excluded. If such a distinction is not applied, an overall 63% of the first wave respondents participated and completed the questionnaire also at waves 2 and 3 (see Figure 5.3 for more details on participation and eligibility at various waves). Full details on fieldwork response, contact, co-operation and response rates by the time of wave 3 are given elsewhere (Scholes et al., 2008a, Scholes et al., 2009).

Figure 5.3 Responses across ELSA waves 1-3 (only Cohort 1 core members)



This figure shows the actual figures of people who responded at different waves. Source: (Scholes et al., 2009). Notation: X denotes 'Respond' at a given wave, O 'Non-Respond' whereas I stands for 'Ineligible'.

Similarly, if we just consider the proportion of SHARE baseline respondents who were successfully re-contacted and interviewed two years after, an overall 65% completed the questionnaire carried out in 2006/2007 (with percentages ranging from almost 80% in Greece to slightly more than 50% in Germany. See Table 5.3 for full details).

	Waya 1	Wave 2 respondents			
Country	(2004)	Effective #	Rate of retention (%)		
Austria	1,893	1,238	65.4		
Belgium	3,827	2,808	73.4		
Denmark	1,707	1,249	73.2		
France	3,193	1,998	62.6		
Germany	3,008	1,544	51.3		
Greece	2,898	2,280	78.7		
Italy	2,559	1,766	69.0		
Netherlands	2,979	1,777	59.7		
Spain	2,396	1,375	57.4		
Sweden	3,053	2,010	65.8		
Switzerland	1,004	696	69.3		
Total	28,517	18.741	65.7		

Table 5.3 Distribution of wave 1 SHARE respondents successfully re-interviewed at wave 2 by country

This table presents the number of respondents in SHARE surveyed in 2004 and 2006 and the number (as well as the percentage) of those who were interviewed in both waves. Source: http://www.share-project.org/ (last accessed on 14th of September 2009).

Nonetheless, not including respondents who died (757 people, i.e. 2.7%), who moved out of their original country (480 people) or who explicitly asked not to be re-contacted at all are removed, the total longitudinal response rate increases to 72%, ranging from 87% in Greece to 59% in Germany, as shown in Figure 5.4 (Schröder, 2008).



These attrition rates for Wave 1 respondents are calculated once people who moved out of the countries, people who asked not to be further re-contacted and people who have died between the first and the second wave have been accounted for. Figure from Schröder (2008).

### 5.5 STRUCTURE AND CONTENT OF THE ELSA AND SHARE QUESTIONNAIRE

SHARE and ELSA are 'sibling' studies: they both cover a number of similar issues including household composition, employment and pension details, the nature and timing of retirement and post retirement labour market activity; physical, cognitive, mental and psychological health measurements, health behaviours, social participation, social productivity and social support as well as information on housing, income, assets and expectations for the future.

### 5.5.1 ELSA QUESTIONNAIRE

In its final form, the wave 1 survey comprised a personal face-to-face interview and a self-completion questionnaire. The ELSA questionnaire is composed of 13 modules each of which addresses a specific topic ranging from health-related to family characteristics. A brief description of the questions asked within each module is highlighted in the Table 5.5.

Wave 2 and 3 included some additions to the initial content. Firstly, Wave 2 included a nurse visit for the collection of biomedical and physical performance measures; secondly, an 'End of Life' interview was conducted with relatives of deceased respondents; thirdly, a number of new set on questions were added (on quality of health care, 'effort and reward' for caring and volunteering just to cite some). The Wave 3 questionnaire included new items on dental health, consumption of fruit and vegetable and anchoring vignettes. Full details on the content of the questionnaire at waves 1-3 are reported elsewhere (Marmot et al., 2003a, Banks et al., 2006, Taylor et al., 2007, Scholes et al., 2008b, Various Authors, 2008b, Scholes et al., 2009); the full questionnaires are available from the documentation section on the ELSA website (ELSA).

Module	Content
Intr	oduction
HD	Household Demographic: basic demographics of everyone who lives in the household (sex, age and relationship to each other) including information on children living outside the household
ID	<b>Individual Demographics:</b> details about respondent's legal marital status, whether their parents were still alive or dead, number of living children and grandchildren, number of siblings and circumstances in childhood.
HE	Health: self-reported general health, long standing illnesses and disability; eyesight and hearing; specific diagnoses and symptoms; difficulty with a range of activities of daily living –(ADLs) and instrumental activities of daily living (IADL)
SP	Social Participation: extent to which respondent participates in certain types of social activity and whether they are limited or excluded from doing these activities.
WP	Work and Pensions: current work activities, current and previous pensions (with details on the amount received).
IA	<b>Income and Assets:</b> income of the respondents from a variety of sources (wages, state benefits, pensions) over the previous year. Amount of financial and non-financial assets held, any income from these assets, regular transfers from non-household members and one-off payments in the last year.
но	Housing: current housing situation (including size and quality of the accommodation), housing-related expenses, ownership of durable goods and cars and expenditure on food. Owners and Mortgagers were also asked about the value of their property and questions were asked about housing costs including mortgages and rent.
CF	<b>Cognitive Function:</b> Respondent's cognitive function including memory, mental flexibility and basic numerical ability.
EX	<b>Expectations:</b> subjective expectations in a number of dimensions (ranging from feeling about the future to optimal planning horizons).
PS	Psychosocial: Respondent's view across a variety of dimensions.
FQ	Final Questions and consents: additional information on education and ethnic origin and permission to obtain further information from various administrative sources.
MM	Measurement – Time Walked (for all individuals aged 60 or over)
SC	Self-Completion Questionnaire: social participation, social network and social relationships of the respondent. This module also contains measures of well-being and quality of life.

ELSA Questionnaire: title of each module with a short description of the questions asked.

### 5.5.2 SHARE QUESTIONNAIRE

The SHARE questionnaire has also got face-to-face, computer-aided personal interviews (CAPI) questionnaire. The main questionnaire is composed of 18 modules. A brief description of the questions asked within each module is highlighted in Table 5.6. Also a self-completion questionnaire with additional questions about issues like mental and physical health, quality of housing and social networks was given to the respondents at the end of the interview ('drop off' questionnaire as the completed questionnaire was meant to be returned via post). Extra samples ('vignettes sample' on six domains of general health) were taken in eight countries (Belgium, France, Germany, Greece, Italy, The Netherlands, Spain and Sweden) in order to collect a special self-completion questionnaire with anchoring vignette questions which are supposed to improve crossnational comparability (Kapteyn, 2006, Blanchet et al., 2007, Barangé et al., 2008).

For the second wave of the questionnaire, some adjustments to the survey instrument were made in order to account for changes between waves. In addition, an 'End of Life' interview was conducted for deceased respondents and two new modules were added (i.e. a test to measure the strength and endurance in legs and one to measure the lungs' capacity). Full details on the content and changes of the questionnaires at waves 1 and 2 are reported elsewhere (Börsch-Supan et al., 2005, Various Authors, 2008a). The generic and country-specific questionnaires for both waves are downloadable from the SHARE-website (SHARE).

Module	Content
Intr	oduction
	Cover-screen: composition of the household; identification of respondents
CV	(individuals born in 1954 or before and their partners); identification of the
	respondent for the housing and finance-related questions.
Cor	e Questionnaire
DN	Demographics: demographic characteristics, level of education, parents,
	siblings, (ex) partners, offspring (if applicable).
рн	Physical Health: general health', known diseases and treatments received,
PH	activity of daily living
RR	Behavioural risk: past and current consumption of tobacco and alcohol,
DR	physical exercise, sport practice
CF	Cognitive Function: Memory and literacy tests
мн	Mental Health: feeling of depression, appetite, irritability, existence of
	previous depression episodes
	Health Care: within the previous 12 months, visits with specialists and
HC	general practitioners, hospitalisation, stays in hospitals, home-care, money
	spent for personal health care
ЕР	Employment and Pension: current employment situation (characteristics,
	job satisfaction, income), last occupation for retired people, pension provision
GS	Grip Strength: 2 tests to assess the grip strength of each hand. Tests
	performed if this did not involve any risks for the respondent.
	Walking speed (for people older than 75): 2 timed tests assessing the length
WS	of time took by the respondent to cover 2.5 meters. Tests performed if this did
	not involve any risks for the respondent.
СН	Children: characteristics and family situation, frequency of contacts
C.D.	Social Support: during the previous 12 months, list of (non financial) help
SP	received from as well as given to family members, neighbours or mends;
<u> </u>	Financial Transforms financial halp given and reasived during the 12 months
FT	<b>Financial Transfers:</b> Infancial help given and received during the 12 months
	Housing: occupancy status (owned/rented flat) costs and purchasing
	mousing: occupancy status (owned/rented nat), costs and purchasing
по	of the housing
	Household Income: income from other members of the household not
HH	responding the questionnaire additional income from other sources
<u> </u>	Consumption: expenditure on food, telephone and other goods and services
	Assets: kinds of saving and investments and their value debts
	Activities: volunteering activities and satisfaction gained from these
	Expectations: subjective probability that certain event would happen in the
EX	future
	Interviewer Observations: information about the way the interview was
IV	conducted (filled in by the interviewer).
1. These a	uestions are asked randomly either at the beginning or at the end of the module

 Table 5.5 Overview of Content of SHARE Questionnaire (wave 1)

SHARE Questionnaire: title of each module with a short description of the questions asked.

\_\_\_\_\_

### 5.6 WEIGHTING STRATEGIES

Sample cross-sectional and longitudinal weights<sup>26</sup> were attached to both the ELSA and SHARE datasets so as to adjust for potential bias in the respondent samples and therefore enhance confidence in the representativeness of the results. Both teams provided cross-sectional adjustments to match the respondents' age-by-sex distribution to the one from the Census<sup>27.28</sup> in order to account for initial non-response in wave 1 and for drop-outs in wave  $2^{29}$ . In addition, ELSA also used the information collected from the HSE interviews to correct for differences in characteristics found between respondents and non-respondents<sup>30</sup>. This calibration procedure (Deville and Särndal, 1992) assigns weights to sample respondents in order to match known population totals obtained from census data, even though the latter may not exactly cover the same target population (Klevmarken et al., 2005a). In particular, all weights control for the size of the target population across gender and age groups with further country-specific adjustments such as by home ownership in France and geographical area in Italy and Denmark. Calibrated cross-sectional weights may help reduce the potential selectivity bias generated by this source of nonsampling error and therefore should allow less biased analyses of a particular wave. In the following chapters, distributions obtained with and without the use of cross-sectional weights will be compared to assess the effect of calibrated weights on specific key-variables such as education and labour force participation.

In addition to calibrated cross-sectional weights, SHARE and ELSA also provide weights for the longitudinal sample. The SHARE and ELSA teams have performed detailed analyses of a respondent's propensity to remain in the survey and constructed

<sup>&</sup>lt;sup>26</sup> Longitudinal weights are provided only for the subset of cases who have taken part in *all* waves whereas cross-sectional weights are defined for all cases belonging to the target population who responded in a particular wave

<sup>&</sup>lt;sup>27</sup> ELSA also used the additional information collected from the HSE interviews to correct for differences in characteristics found between respondents and non-respondents.

<sup>&</sup>lt;sup>28</sup> Non-institutionalised distribution for ELSA

<sup>29</sup> Cross-sectionals weights were computed for SHARE countries whereas they were not for the English sample since ELSA did not include a refreshment sample for the second wave, i.e. a new sample drawn to compensate for the loss of observations due to sample attrition and to include new 50+ respondents. For ELSA, wave 2 weights supplied support both cross-sectional and longitudinal analysis

<sup>&</sup>lt;sup>30</sup> The algorithm used to select Cohort 3 sample members was fallacious. The weighting adjustment made to address this shortfall is discussed elsewhere (Scholes et al., 2009).

weights accordingly in order to correct the data for the attrition as well as to adjust for initial non-response. These weights compensate for potential selectivity effects generated by sample attrition between the first and the second wave, i.e. for any differences in characteristics found between respondents and non-respondents. Therefore, such weights should reduce any bias arising from sample loss after wave 1 since they correct for non-response bias between waves 1 and 2, allowing more precise analyses of changes between waves. Longitudinal weights are based on a set of conditioning variables by using the additional information collected in the Wave 1 interview. Detailed descriptions of the weighting strategies and of the computation of calibrated cross-sectional and longitudinal weights adopted in the second wave of SHARE and ELSA are provided elsewhere (Little and Rubin, 2002, Marmot et al., 2003a, Börsch-Supan and Jürges, 2005, Taylor et al., 2007, De Luca and Rossetti, 2008, Scholes et al., 2008b, Various Authors, 2008a, Scholes et al., 2009).

### 5.7 STRENGTHS AND WEAKNESSES OF ELSA AND SHARE

ELSA and SHARE are research driven and researcher managed; this means that questionnaires aim at addressing questions scientists are interested in (Kapteyn, 2006), gathering unique, comparable and innovative information on numerous 'domains' of life. Moreover, because of the involvement of the scientific community, both projects are flexible and can quickly and relatively easily incorporate new questions of interest for research. Furthermore, SHARE and ELSA have an open and eclectic research approach: there is no one leading research hypothesis and no particular paradigm. On the contrary, various disciplines are covered.

ELSA and SHARE also aim at providing comparable cross-national multidisciplinary data. Comparability of the topics covered is guaranteed by an assiduous collaboration between the two research groups; SHARE, moreover, provides cross-national truly comparable data since the same harmonised questionnaire was administered to each European participant at the same time.

The ELSA and SHARE questionnaires not only allow researchers to compare countries at a single point in time, but also use panel data which have normally "greater capacity for modelling the complexity of human behaviour than a single cross-section or time series data" (Hsiao, 2006). Observations at different time points are ideal to capture the complexity and heterogeneity of the ageing phenomenon as well as to uncover dynamic relationships, contributing to the development of policy intervention and policy evaluations.

Despite their innovative force and potential, ELSA and SHARE in particular had to face numerous methodological problems and critiques. ELSA, for instance, has been criticised for using the HSE as a sampling frame. Among the numerous disadvantages that this frame carries, the potential loss of representativeness before the first ELSA interview is the most frequently mentioned (Taylor et al., 2007). Non-response at HSE, refusal to be re-contacted and attrition between HSE and ELSA could all potentially bias the responding sample<sup>31</sup>.

As regards SHARE, this has been criticised at various levels. Firstly, social protection, pension and retirement systems and many more domains are quite different from one country to the other. Despite the efforts made in the translation and adaptation of the questions from an English version (which is not the official language of any of the countries participating in the SHARE project) in order to allow for institutional differences, cross-cultural comparability of responses was not always addressed (Blanchet et al., 2007). "It is clear, for instance, that a term like 'excellent health' carries different meanings in different countries" (Kapteyn, 2006)(pg. 17). German people, for instance, would consider it as "an ironic exaggeration, in particular if used in the context of health" (Jürges, 2005). Similarly, it is known that the way a question is formulated can influence answers (Aureli Cutillo, 2002). Just to cite an example raised by Blanchet et al. (2007), in the module about social life there is a question asking whether the respondent has "gone to a sport, social or other kind of club". This question was translated in different ways even within countries speaking the same language. For instance, in France people were asked about their participation to "à un club sportif, à une amicable ou à un autre type de club" whereas in Belgium and Switzerland the

<sup>&</sup>lt;sup>31</sup> Nonetheless, it was believed that "the advantages of sampling from the HSE outweigh any disadvantages". In particular, "baseline data on respondents' health (details of morbidity, lifestyle, diets and blood samples) had already been collected" before they even took part in ELSA (Taylor et al., 2007) (pg. 9). Moreover, almost all the people had already been interviewed once by the NatCen (National Centre for Social Research) making them easier to approach and more likely to take part in a new study.

translation stuck closer to the English version with "club sportif, social ou d'un autre type". In Italy, the very same question was translated with "partecipazione alle attività di un circolo sportivo, ricreativo, sociale o culturale o altro tipo di associazione". The word "club" is quite specific in the French and Italian languages and for instance it is not clear whether a French player of pétanque who meets their friends on a regular basis would consider this social activity as the one of a club. The French word 'association' (similar to the one used in the Italian version) could have provoked different answers (Blanchet et al., 2007).

Secondly, despite all the efforts to homogenise data and questions in the SHARE project, comparability of money-related questions still raises problems and concerns, especially because older people might have referred to their previous national currency rather than Euros<sup>32</sup>. Information about finances is also considered to be sensitive by many respondents. Data on consumption, earnings or money exchange are less precise and need more caution in use (Kalwij and van Soest, 2005, Blanchet et al., 2007, Christelis, 2008).

Finally, the sampling design strategies for SHARE have been criticised because each country used different designs depending on the availability of registers of households and individuals but also depending on the institution which was appointed to implement the study. Lack of access to national sampling frames could potentially yield less efficient estimates or increased systematic errors. In Greece, Switzerland and Austria, for example, the national statistics offices decided *not* to collaborate and therefore sampling units were drawn from telephone directories. Furthermore, probability sampling procedures used to draw national samples are not standardized across countries and, in some cases, even within the same country sampling designs differed between the first and second wave<sup>33</sup> (De Luca and Rossetti, 2008). Finally, "only exceptionally it became possible to identify people in institutions (homes for elderly) in the sampling frame" (Klevmarken et al., 2005b) and therefore for countries where the

<sup>&</sup>lt;sup>32</sup> Euro coins and banknotes entered circulation on 1 January 2002, i.e. only two years before the SHARE wave 1 interviews.

<sup>&</sup>lt;sup>33</sup> In Belgium, the sampling design changed from three-stage sampling of telephone numbers in Wave 1 to two-stage sampling of households in Wave 2. In Denmark, the sampling design changed from simple random sampling of households in Wave 1 to simple random sampling of individuals in Wave 2.

population over 50 living in institutions was not included in the sampling frame, calibration against census population totals which do include those people is questionable and can potentially yield erroneous estimates of the variances (Klevmarken et al., 2005b).

Bearing in mind the possible limitations and strengths of the datasets, the following Chapter introduces some initial descriptive statistics and address the issues of data quality and representativeness of ELSA and SHARE.

# 6. ELSA AND SHARE: DATA QUALITY AND SOME DESCRIPTIVE ANALYSES

This chapter presents some descriptive statistics for a number of variables of interest for this research project, such as socio-demographic characteristics, work and retirement, socio-economic position, social activity and health, paying special attention to the issue of data quality and comparability. This chapter is structured in the following way: a brief description of issues such as initial response rate, non-response bias and data quality is introduced. Some descriptive analyses are then presented for Denmark, England, France and Italy. Where possible, the latter statistics will be compared with other sources (such as census data) as a way of assessing whether results from SHARE and ELSA are representative of the populations from which the samples are drawn. Throughout the Chapter, comparability of ELSA and SHARE variables are discussed.

### 6.1 INITIAL RESPONSE RATE, NONRESPONSE BIAS AND DATA QUALITY

Initial non-response is a major concern in surveys: this, in fact, may be thought of as pre-study attrition (Sivo et al., 2006) and this could be even more selective than attrition at later waves (Pyy-Martikainen and Rendtel, 2008). Initial non-response raises two main problems: survey participants may systematically differ from those who refused to participate and, secondly, low initial response rate may result in an inadequate sample size, leading to inaccurate estimations and low power to detect precisely relationships (Baroudi and Orlikowski, 1989, Shadish et al., 2002, Sivo et al., 2006). Non-response bias can have significant detrimental effects on the precision of survey estimates as well as on their generalisability and representativeness if findings are drawn from a self-selected group of respondents whose characteristics differ consistently from the non-respondents (Pinsonneault and Kraemer, 1993). The bias of estimates might be increased also by item non-response if respondents who do not answer certain questions differ from those who answer them (de Leeuw et al., 2003, Pauwels and Svensson, 2008). Item non-response occurs when a respondent fails to provide useful answers to

particular items of the questionnaire, perhaps either because of the sensitivity of the question or because of its difficulty (Shoemaker et al., 2002).

Ideally surveys should pursue a close to 100% response rate and have no item nonresponse even though, in reality, surveys enjoying such a high response are few. Despite the fact that there is no reliable vardstick for determining what is an 'acceptable' response rate (Smith and Nutbeam, 1990), it is believed that a high response rate is an indicator of the quality of all survey estimates (Fogliani, 1999, Singleton and Straits, 2005) though even a 15% non-response might suggest that inferences should be interpreted cautiously (Sivo et al., 2006). Moreover, even when response rates are high, certain specific subgroups of the population could still deliberately decide not to take part in the study or not to answer certain questions (Jones and Lang, 1980, Groves, 2006). If those who respond to survey questionnaires differ in some systematic way from non-respondents, this will induce bias (Smith and Nutbeam, 1990, Kuskowska-Wolk et al., 1992, Bostrum et al., 1993, Hill et al., 1997). As a consequence, it is important to check the reliability of the data and assess whether nonresponse bias does occur or not, regardless of the initial response rate. Despite the fact that most surveys nowadays provide weights to account for bias and adjust for it, weighting techniques require the assumption that respondents and non-respondents –within a weighting class -share the same values on key variables. This is unknowable and untestable (Groves, 2006) and some adjustments may actually make matters worse (Brick et al., 2003, Little and Vartivarian, 2003).

Among the several techniques to check whether there is non-response bias, one of the most common is to compare the respondent-based estimates with those from other more comprehensive sources, such as census data (Groves and Couper, 1998, Pyy-Martikainen and Rendtel, 2008). Comparing the distribution of age, gender, education and other socio-demographic variables with those from other sources thought to be reliable will provide some indication of the extent of possible bias due to non-response in order to understand the potential impact of non-response on the ability of surveys to describe large populations without bias and within measurable levels of uncertainty (Smith and Nutbeam, 1990, Hill et al., 1997, Groves, 2006). Such comparisons are needed particularly for Italy and Denmark where household response rates were quite

low (54% and 65% respectively). The SHARE team claims that response rates compare fairly favourably to the ones of other European surveys (De Luca and Peracchi, 2005a). They also state that "comparisons with [...] European Union Labour Force Survey, the European Community Household Panel and the European Social Survey show that the SHARE data produces very similar distributions of key concepts such as employment, income, education and health" (Börsch-Supan and Jürges, 2005)(pg. 34). Nonetheless, when comparing the response rate and data of SHARE to the ones obtained in the other above-mentioned European surveys, we note that in some cases these studies have similarly low response rates (see Table 6.1). Consistency of data and response rate between equally low-response rate surveys is a practice that does not assure validity (Roth and BeVier, 1998) nor obviate the need for further comparisons with alternative sources.

Item non-response, on the other hand, was a minor issue for noneconomic variables (e.g. those referring to demographics, physical and mental health, employment variables, expectations or activities) both for SHARE (Kalwij and van Soest, 2005, Christelis, 2008) and ELSA datasets (Taylor et al., 2003, Cheshire et al., 2006).

Table 0.1 Response Rate of SHARE and other European Surveys											
somen were		Offi	cial						1082 10		
		Euro	stat								
		surv	eys	0.236							
	SHARE	ECHP	LFS	ESS	ESS	EVS	EES	ISSP			
	2004	1994	1996	2002	2004	99-00	1999	2002	Avg.		
Denmark	63.2	62.0	75.0	68.0	65.1	57.0	59.0	66.1	64.6		
Germany	63.4	47.0	(a)	57.0	50.0	42.0	49.0	42.7	47.9		
Italy	54.5	(a)	61 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	44.0	104 - LL	68.0		TO-ST	56.0		
Netherlands	61.6	(a)	59.0	68.0	-	40.0	30.0	46.6	48.7		
Spain	53.0	67.0	(a)	53.0	54.8	24.0	-	(a)	49.7		
Sweden	46.9	the - Alth	(a)	69.0	65.8	41.0	31.0	57.2	52.8		
Austria	55.6	-	-	-	62.4	77.0	49.0	63.9	63.1		
France	81.0	79.0	(a)	-	-	42.0	44.0	20.3	46.3		
Greece	63.1	(a)	-	80.0	78.8	82.0	28.0	-	67.2		
Switzerland	38.8	-		34.0	46.9	S. F.	-	32.8	37.9		
Weighted Average	61.6	62.0	63.2	55.6	54.9	46.4	43.9	36.7	50.8		

This table shows the household response rate of SHARE and other European Surveys. Notes: (a) no pre-screening response rate reported, (-) country not in sample. ECHP: European Community Household Panel; EU-LSF: European Labour Force Survey; ESS: European Social Survey; EVS: European Values Study; EES: European Election Study; ISSP: International Social Survey Project. Source: De Luca and Peracchi, (2005a, b).

### 6.2 DESCRIPTIVE STATISTICS AND ASSESSMENT OF DATA QUALITY

The following sections present some descriptive statistics for each country separately, including distributions of age, education, marital status, labour force participation and some health indicators. Some of these descriptive tables will also be compared with data obtained from national official sources to assess quality of data. Since the ELSA and SHARE teams provide sampling weights which adjust for nonresponse, these will be used when making inferences about the populations under study. This should give us an indication of how well this method of compensation meets the needs of this particular study. Also the sampling design (primary sampling units and strata) is taken into account in order to estimate correct point estimates for the population and their correct variances<sup>34</sup>.

### 6.3 SOCIO-DEMOGRAPHIC CHARACTERISTICS

### 6.3.1 AGE AND SEX COMPOSITION

In all countries considered, women outnumbered men in most age-groups and the age distribution of the female samples was older than that of the male samples reflecting the higher survival rates of women at older ages. Between 16% and 20% of both men and women were aged 50–59 but the proportion of women aged 80 and over was close to twice as high as that for men (see Table 6.2).

Table 6.3 shows the age-sex distributions obtained from census data or national population estimates (based on updated census results)<sup>35</sup> as well as both the unweighted and the weighted distributions for the SHARE and ELSA respondents, providing 95% confidence intervals for the survey estimates. For most countries, when the calibration weights provided by the SHARE and ELSA teams were *not* used, the age-sex

<sup>&</sup>lt;sup>34</sup> Strata with only one primary sampling unit pose problem for the estimation of the precision since there is no estimate of the variability between sampling units. In order to overcome this problem, the variability of the single unit is set equal to the overall mean variability. This conservative approach (that tends to give higher estimates of the variability) is believed preferable than overstating precision.

<sup>&</sup>lt;sup>35</sup> Data refers to the 2001 census for England and to the updates of the 2000/1 census results for the SHARE countries. Age-sex distributions were obtained from the National Statistics Bureaux, i.e. Danmarks Statistik, DST, for Denmark (www.dst.dk), Istituto nazionale di statistica, ISTAT for Italy (www.istat.it), Office for National Statistics, ONS, for England (www.statistics.gov.uk) and Institut national de la statistique et des études économiques, (*INSEE*), for France (www.insee.fr).

distributions did not match census data. There appeared to be a general underestimation of the youngest (50-54) and the oldest respondents (75+) and an over-representation of the in-between age-groups (55-74). In Italy, for instance, men and women aged 50-54 were heavily under-represented in the SHARE survey as were respondents aged more than 79. On the contrary, the 60-74 year old respondents tended to be over-represented when weights were not taken into account. Also among French and Danish women the un-weighted proportion of those aged 80+ was lower in the SHARE samples than in the census. Similar patterns were observable also among English respondents. When calibrations weights were taken into account, the age-sex distributions of the surveys' respondents matched census data quite reasonably with only a few exceptions. English men, Italian and French respondents aged 50-54, for instance, were still underrepresented even in the weighted samples.

	Denmark		Fra	nce	Ita	ly	England				
	Μ%	F %	M %	F %	M %	F %	M %	F %			
	(N)	(N)	(N)	(N)	(N)	(N)	(N)	(N)			
50.50	20.41	20.24	18.29	18.79	16.45	17.13	18.48	18.81			
30-39	(317)	(328)	(575)	(617)	(356)	(474)	(1,916)	(2,250)			
60.60	13.97	14.89	12.58	13.60	14.30	16.05	13.74	14.67			
00-09	(228)	(229)	(374)	(432)	(421)	(525)	(1,611)	(1,787)			
70.70	8.40	10.51	9.70	12.95	10.04	13.78	9.98	12.49			
/0-/9	(147)	(188)	(299)	(389)	(278)	(274)	(1,174)	(1,391)			
801	3.93	7.66	4.55	9.54	4.03	8.23	4.20	7.63			
007	(65)	(112)	(155)	(235)	(70)	(109)	(485)	(777)			
Total	46.71	53.29	45.13	54.87	44.81	55.19	46.35	53.65			
(N)	(757)	(857)	(1,363)	(1,673)	(1,125)	(1,382)	(5,186)	(6,205)			

Table 6.2 Distribution of Respondents by country, sex and age

Each column shows the percentage of male and female in each age group in each country. Source: ELSA, 2002; SHARE, 2004. Weighted data.

		Male				Female				
		S	EL	SA	Ś	ELSA				
	Age group	Censu	Weighted (95% CI)	Un-Weighted (95% CI)	Censu	Weighted (95% CI)	Un-Weighted (95% CI)			
	50-54	22.3	20.8 (19.6; 22.0)	17.6 (16.5; 18.6)	19.3	18.2 (17.1; 19.3)	18.4 (17.4; 19.3)			
Q	55-59	18.4	19.1 (18.0; 20.2)	19.7 (18.6; 20.8)	15.9	16.8 (15.9; 17.7)	18.6 (17.7; 19.6)			
	60-64	15.6	15.7 (14.7; 16.7)	15.5 (14.6; 16.5)	13.7	14.1 (13.2; 14.9)	14.0 (13.2; 14.9)			
A.	65-69	13.8	14.0 (13.0; 14.9)	15.4 (14.4; 16.4)	12.6	13.3 (12.4; 14.1)	14.5 (13.6; 15.4)			
GL	70-74	11.8	12.1 (11.2; 13.0)	13.0 (12.1; 13.9)	12.0	12.4 (11.6; 13.2)	12.7 (11.8; 13.5)			
EN	75-79	9.1	9.4 (8.6; 10.2)	9.5 (8.7; 10.3)	10.8	10.9 (10.1; 11.7)	9.5 (8.7; 10.2)			
	80+	9.0	8.9 (8.2; 9.7)	9.3 (8.5; 10.1)	15.7	14.3 (13.4; 15.3)	12.3 (11.5; 13.2)			
		100	100	100	100	100	100			
			SHA	ARE		SHA	RE			
	50-54	20.9	23.7 (20.4; 26.7)	22.3 (19.3; 25.3)	18.2	17.6 (14.9; 20.3)	17.7 (15.2; 20.3)			
	55-59	22.8	20.0 (17.0; 23.0)	19.5 (16.7; 22.4)	19.8	20.4 (17.6; 23.2)	20.5 (17.8; 23.2)			
RK	60-64	17.1	17.9 (15.0; 20.7)	18.0 (15.2; 20.7)	15.3	16.1 (13.5; 18.7)	15.7 (13.3; 18.2)			
V	65-69	13.0	12.0 (9.6; 14.4)	12.1 (9.8; 14.5)	12.4	11.8 (9.5; 14.1)	11.0 (8.9; 13.1)			
Z	70-74	10.0	10.3 (8.1; 12.5)	11.2 (9.0; 13.5)	10.4	10.3 (8.3; 12.3)	11.3 (9.2; 13.4)			
DE	75-79	7.9	7.7 (5.7; 9.6)	8.2 (6.2; 10.1)	.9.3	9.4 (7.5; 11.3)	10.6 (8.5; 12.7)			
	80+	8.4	8.4 (5.7; 9.6)	8.6 (6.6; 10.1)	14.6	14.4 (11.8; 16.7)	10.6 (8.5; 12.7)			
		100	100	100	100	100	100			
	50-54	22.6	20.2 (18.0; 22.4)	21.3 (19.4; 23.3)	19.1	16.3 (14.7; 17.8)	17.5 (16.0; 19.0)			
	55-59	19.7	20.3 (17.4; 23.5)	20.8 (17.6; 24.1)	16.6	18.0 (16.2; 19.7)	19.2 (17.4; 21.0)			
E	60-64	14.0	14.2 (12.4; 16.0)	14.2 (12.5; 15.8)	12.2	13.0 (11.5; 14.5)	13.7 (12.3; 15.1)			
ž	65-69	13.2	13.7 (11.7; 15.6)	13.3 (11.7; 14.9)	12.5	11.7 (10.3; 13.2)	12.2 (10.8; 13.7)			
N S	70-74	12.0	11.8 (10.0; 13.6)	12.2 (10.4; 14.0)	12.7	12.9 (11.0; 14.9)	12.9 (11.1; 14.6)			
	75-79	9.3	9.7 (8.0; 11.4)	9.7 (8.2; 11.3)	11.3	10.6 (9.1; 12.1)	10.4 (9.0; 11.8)			
	80+	9.3	10.1 (8.4; 11.8)	8.4 (7.0; 9.9)	15.7	17.3(15.0; 19.7)	14.0 (12.4; 15.6)			
		100	100	100	100	100	100			
	50-54	19.1	15.4 (12.5; 18.2)	11.3 (9.4; 13.3)	16.7	13.3 (11.3; 15.3)	14.4 (12.6; 16.2)			
	55-59	17.5	21.3 (18.4; 24.2)	20.3 (17.8; 22.7)	14.5	17.7 (15.6; 19.9)	19.9 (17.7; 22.1)			
×	60-64	17.0	16.5 (14.3; 18.8)	19.5 (17.1; 22.0)	15.4	15.5 (13.3: 17.7)	21.1 (18.9: 23.1)			
E	65-69	14.9	15.4 (13.1; 17.6)	17.9 (15.7; 20.0)	14.1	13.6 (11.9; 15.3)	16.9 (14.9; 18.9)			
Ê	70-74	12.8	15.0 (12.8; 17.2)	16.2 (14.1; 18.2)	13.4	14.6 (12.0; 17.2)	12.1 (11.0; 14.2)			
	75-79	9.6	7.4 (5.8; 9.0)	8.5 (6.9; 10.1)	11.7	10.4 (8.7; 12.1)	7.7 (6.4; 8.9)			
	80+	9.0	9.0 (7.0; 11.0)	6.2 (4.8; 7.6)	14.1	14.8 (11.8; 18.0)	7.8 (6.4; 9.4)			
		100	100	100	100	100	100			

Table 6.3 Distribution of the ELSA and SHARE respondents by age group and sex (with and without weights); comparisons with the census data

This table compares the age-sex distribution between census (ONS for England) and ELSA and SHARE respondents respectively and its 95% confidence intervals obtained with weighted and unweighted data. Sources: ONS, 2001; ELSA, 2002; DST (Denmark), 2004; INSEE (France), 2004; ISTAT (Italy), 2004; SHARE 2004.

### 6.3.2 EDUCATION

When the level of education was considered, differences both between countries and within country (by age-groups) were quite striking as expected, reflecting few educational opportunities for older cohorts and for women. In general, men were more educated than women and the level of education decreased as age increased though differences in levels between countries were quite dramatic. Classifying older people into three categories of the ISCED-97<sup>36</sup> (low education: levels 0-2; medium education: 3; high education: levels 4-6), it was clear that both male and female younger respondents achieved higher educational levels than older ones. Differences by counties and gender were remarkable: almost 94% of Italian men older than 80 had low education versus 24% in Denmark. Almost 50% of Danish women aged 50-59 had high education compared to 23% in France and England and to 13% in Italy (see Table 6.4).

			50-59		60-69		70-79		)+	Total	
	%	M	F	М	F	М	F	M	F	Male	Female
~	Low	11.1	14.9	16.0	31.1	20.5	46.0	23.4	62.5	<b>15.3</b> (119)	32.4 (288)
V	Medium	53.5	35.0	48.6	43.3	46.0	32.9	44.0	27.3	<b>49.9</b> (390)	35.8 (311)
Σ,	High	34.5	49.4	35.4	25.6	33.5	21.1	29.0	8.3	<b>34.2</b> (240)	31.2 (254)
E	Other*	0.8	0.7	-	-	-	-	3.5	0.2	0.6 (5)	0.5 (4)
	N	315	328	228	229	147	188	64	112	754	857
	Low	31.6	40.2	50.2	55.5	64.0	77.1	64.9	85.7	<b>47.2</b> (625)	<b>60.5</b> (976)
U	Medium	39.9	35.0	27.8	26.2	22.2	14.7	18.6	9.4	<b>30.5</b> (414)	23.6 (405)
Z	High	27.6	23.5	20.4	17.6	13.1	7.3	14.8	5.3	21.1 (282)	15.1 (256)
R	Other*	0.9	1.3	1.7	0.7	0.6	0.8	1.7	-	<b>1.2</b> (15)	0.8 (14)
	N	561	611	366	424	294	385	115	231	1,336	1,651
	Low	59.8	66.2	73.0	75.8	81.8	86.3	93.3	94.2	71.9 (844)	78.2 (1,103)
	Medium	27.7	20.4	17.3	15.5	9.8	10.4	3.3	5.0	<b>18.2</b> (189)	<b>14.1</b> (184)
	High	11.7	13.4	9.7	8.2	8.4	2.4	3.4	0.8	<b>9.6</b> (90)	7.3 (91)
	Other*	0.8	-	-	0.5	-	1.0	-	-	0.3 (2)	0.4 (3)
	N	356	473	421	525	278	274	70	109	1,125	1,381
	Low	29.5	35.6	47.7	49.7	60.3	63.3	68.4	73.0	<b>45.0</b> (2,339)	<b>51.2</b> (3,109)
Z	Medium	29.7	31.4	21.5	20.8	13.5	12.2	11.2	9.6	<b>22.2</b> (1,144)	<b>20.9</b> (1,360)
ILA	High	35.8	22.4	26.2	17.5	19.5	11.7	13.5	8.4	<b>27.4</b> (1,450)	<b>16.6</b> (1,097)
Z	Other*	5.0	10.6	4.6	11.9	6.7	12.8	6.9	8.9	5.4 (280)	11.2 (712)
E	N	1.945	2,325	1,612	1,789	1,174	1.390	482	774	5,213	6.278

 Table 6.4 Level of Education (ISCED-97) by gender, age-group and country

The Level of Education classification follows the ISCED-97 standards (low education: levels 0-2; medium education: 3; high education: levels 4-6). Note: Other \* stands for "still in school", "other", "don't know", "refusal". Sources: SHARE, 2004; ELSA, 2002. Weighted data.

<sup>36</sup> International standard classification of education; full details available at http://www.uis.unesco.org/TEMPLATE/pdf/isced/ISCED\_A.pdf

However, as in the case of the age-sex distributions already discussed, there were differences between the distributions of educational attainment obtained from the SHARE and ELSA surveys and those provided by the relevant national censuses. Converting the highest degree obtained by each enumerated person into ISCED-97 codes following the official guidelines provided by UNESCO and OECD (1999), in most countries there was an under-representation of people with low education levels and an over-representation of middle and high levels. For instance, the proportion of Italian female SHARE respondents aged 55-69 who reported a high level of education was almost twice as high as that reported from the Census. Similarly, French men and women aged 50-64 with high education are over-represented in the SHARE sample (see Tables 6.A1-6A.4 in the appendix for full details).

### 6.3.3 MARITAL STATUS

Between 70% (Denmark) and 76% (Italy) of men and between 54% (Denmark) and 58% (England) of women were married<sup>37</sup>. Whereas the proportion of men who were married started to decline only at ages above 80, the proportion of women who were married declined as age increased (reflecting earlier widowhood among women). For men, the age group including the highest proportion married was the 60-69 year-old age group in Italy, England and Denmark and the 70-79 one in France; for women, it was the 50-59 age-group in all countries. In England, Denmark and France around 10% of the sample were separated or divorced whereas less than 5% of the Italian SHARE respondents were divorced or separated. Finally, the proportion of widows was for all age groups and countries much higher than the proportion of widowers: for instance over a third of men aged 80 or over were widowed compared to more than two-thirds of 80+ women. In general, between 7% (France) and 9% (Italy, Denmark and England) of men and between 25% (England and Denmark) and 33% of women (Italy) were widowed: these gender-differences reflect higher female survival rates and the common pattern of age differences between spouses with women generally marrying men older than themselves (Table 6.5).

<sup>&</sup>lt;sup>37</sup> First and following marriages were merged together for the ELSA dataset.

SHARE and ELSA data on marital status compared reasonably well with census data, with a few exceptions particularly among respondents aged 80 and over. Among older respondents, single (never-married) and widowed people were over-represented among Italian men whereas French and English older single people (80+) tended to be under-represented; divorced and separated were inflated to the detriment of single people among Danish men (data not shown).

		50-59		60-69		70-79		80+		Total	
	%	М	F	М	F	М	F	M	F	Male	Female
	Single	12.9	9.1	5.0	1.6	6.2	4.7	-	7.1	8.2 (61)	5.8 (46)
	Married	69.7	67.1	73.1	63.1	70.3	44.6	50.0	14.2	<b>69.3</b> (524)	<b>54.0</b> (473)
ž	Divorced	15.4	18.0	12.7	18.5	8.9	11.4	17.3	6.7	<b>13.6</b> (102)	15.2 (124)
	Widowed	1.7	5.8	9.2	16.8	14.5	39.3	31.9	72.0	<b>8.8</b> (66)	<b>25.0</b> (214)
	N	314	328	228	229	147	188	64	112	753	857
	Single	6.9	7.7	5.8	6.5	6.8	5.5	4.3	7.5	<b>6.3</b> (85)	<b>6.9</b> (117)
	Married	78.8	69.5	77.5	63.3	81.0	48.4	62.5	20.9	73.2 (1,038)	54.5 (924)
E	Divorced	11.8	14.7	13.9	11.2	3.6	6.3	4.3	5.2	<b>9.9</b> (130)	<b>10.2</b> (174)
	Widowed	2.4	8.1	2.6	19.0	8.5	39.8	28.9	66.4	6.5 (82)	<b>28.4</b> (434)
	Ν	561	610	365	424	294	384	115	231	1,335	1,649
	Single	12.7	7.8	10.2	6.6	7.7	8.3	9.8	16.8	10.5 (81)	8.9 (86)
	Married	77.5	78.0	81.0	61.8	76.7	40.6	57.2	13.5	76.6 (964)	<b>54.3</b> (973)
E	Divorced	6.6	4.3	4.7	4.1	1.2	1.6	-	2.2	4.2 (28)	3.3 (29)
	Widowed	3.2	9.6	4.1	27.5	14.3	48.9	33.0	67.5	8.7 (52)	<b>33.3</b> (291)
	N	356	472	421	525	278	273	70	109	1,125	1,379
	Single	8.4	4.6	7.5	3.6	4.3	5.4	2.0	7.5	6.7 (324)	<b>4.9</b> 306)
	Married	76.1	73.6	78.2	68.0	75.5	48.0	57.8	18.3	75.0 (3,916)	<b>58.2</b> (3,654)
EN	Divorced	13.3	16.9	9.0	12.4	5.5	6.0	2.0	2.5	<b>9.3</b> (463)	11.1(732)
	Widowed	2.1	4.9	5.3	15.9	14.7	40.6	38.2	71.7	9.0 (482)	25.8 (1,512)
	N	1.915	2.250	1.611	1.786	1.174	1.391	485	777	5,185	6.204

Table 6.5 Distribution of legal marital status, by gender, age-group and country

Each column shows the percentage of single, married, divorced and widowed respondents by age-group and age. 'Single' stands for 'never married'; 'Married' includes remarried and people cohabiting with a partner. 'Divorced' includes also 'Separated'. Sources: SHARE, 2004; ELSA, 2002; Weighted data.

### 6.3.4 HOUSEHOLD COMPOSITION AND OTHER CHARACTERISTICS

Information on household composition was collected separately from legal marital status in SHARE and ELSA, determining who people lived with regardless of their marital status. Table 6.6 shows that the majority of both men and women lived with their 'spouse/partner only' in all countries except Italy, where more than 40% of both men and women lived 'in other arrangements'<sup>38</sup>. The proportion of people who lived alone ranged between 15% (France) and 23% (Denmark) for men and between 27% (Italy) and 40% (Denmark) for women. The figures showed age and gender differences. The proportion of people who lived on their own increased with age. For example, in France while 11% of men and 16% of women aged 50–59 reported living alone, these proportions for people aged 80 or over were 32% and 66% respectively. Conversely, the proportion of people who lived 'in other arrangements' decreased with age.

		Tuble of Distribution of itring arrangements, by age, sex and country										
		50-	-59	60-69		70-79		80+		Total		
	%	M	F	M	F	М	F	M	F	Male	Female	
$\square$	Alone	17.6	24.3	20.7	32.5	27.1	53.5	46.6	79.8	22.7 (169)	<b>40.3</b> (335)	
Y	Partner	51.6	56.7	73.7	64.0	68.8	45.2	53.4	16.3	<b>61.4</b> (476)	<b>50.7</b> (446)	
D	Other	30.8	19.0	5.6	3.5	4.1	1.3	-	3.9	15.9 (112)	9.00 (76)	
	N	317	328	228	229	147	188	65	112	757	857	
	Alone	11.0	16.4	15.7	26.8	14.5	42.3	32.2	65.9	15.2 (201)	33.7 (536)	
2	Partner	40.9	41.0	62.1	57.4	75.6	46.6	63.5	22.2	<b>56.5</b> (768)	<b>43.1</b> (734)	
E	Other	48.1	42.6	22.2	15.8	9.9	11.1	4.3	11.9	28.3 (394)	23.2 (403)	
	N	575	610	374	424	299	384	115	231	1,363	1,649	
	Alone	13.0	9.9	13.8	20.8	19.4	41.9	35.4	55.0	16.7 (92)	<b>27.8</b> (210)	
ے	Partner	18.1	27.1	43.5	40.2	57.4	34.7	49.1	12.2	37.8 (481)	30.6 (525)	
E	Other	68.9	63.0	42.7	39.0	23.2	23.4	15.5	32.8	45.5 (552)	41.6 (647)	
	N	356	474	421	525	278	274	70	109	1,125	1,382	
	Alone	13.2	13.2	16.1	23.2	20.4	43.7	39.7	69.3	18.0 (937)	<b>31.1</b> (1,913)	
Z	Partner	41.1	46.7	63.2	61.1	69.2	44.6	52.0	16.5	54.7 (2,943)	45.8 (2,917)	
G	Other	45.6	40.1	20.6	15.7	10.4	11.7	8.3	14.2	27.3 (1,306)	<b>23.1</b> (1,375)	
	N	1,916	2,250	1,611	1,787	1,174	1,391	485	777	5,186	6,205	

Table 6.6 Distribution of living arrangements, by age, sex and country

'Partner' includes respondents who live only with their partner or spouse. 'Other' includes all other arrangements such as couples or single adults living with their children and/or parents and/or other relatives and/or other not-related people. Sources: SHARE, 2004; ELSA, 2002. Weighted Data.

<sup>&</sup>lt;sup>38</sup> The wording 'other arrangements' include couples or single adults living with their children and/or parents, couples or single adults living with other relatives as well as couples and adults living with other not-related people.

ELSA and SHARE respondents were also asked about the number of living children<sup>39</sup> and parents and whether they lived in or outside the household "since this has important implications for social relationships and social support" (Marmot et al., 2003a)(pg. 25). Differences between countries, sex and age-groups were quite striking: whereas in Denmark only 8% of women and 14% of men lived with at least one child, in Italy the percentages were 38 and 41 respectively (see Table 6.7). The results also showed a clear age pattern: younger respondents were more likely to be living with some of their children. For example, in France 45% of men and 41% of women aged 50–59 lived with at least one child on the household; the proportions for men and women aged 80 or over who lived with at least one child were respectively 3% and 10%.

	nousenola (76), by age group, gender and country											
	Age	50-59	60-69	70-79	80+	Total						
		% (N)	% (N)	% (N)	% (N)	% (N)						
DK	Men	28.6 (88)	4.9 (10)	3.6 (4)	-	14.6 (102)						
	Women	17.8 (56)	3.1 (7)	1.3 (3)	3.9 (5)	8.4 (71)						
2	Men	44.9 (259)	20.5 (76)	7.3 (22)	2.6 (3)	25.7 (360)						
ί <b>λ</b> η	Women	40.8 (259)	14.2 (60)	8.2 (33)	10.3 (24)	<b>21.2</b> (376)						
r	Men	62.4 (232)	39.0 (182)	21.4 (67)	11.4 (10)	41.2 (491)						
Ĩ	Women	59.8 (296)	35.4 (184)	19.8 (59)	27.9 (40)	38.0 (579)						
EN	Men	41.5 (760)	16.9 (236)	8.3 (98)	6.9 (32)	23.9 (1,150)						
	Women	36.2 (771)	11.1 (195)	9.6 (125)	11.5 (84)	19.6 (1,175)						

 Table 6.7 Distribution of respondents living with at least one child in the same household (%), by age group, gender and country

Each line shows the percentage of men or women who live with their children in each age-group. Sources: SHARE, 2004; ELSA, 2002. Weighted Data

<sup>&</sup>lt;sup>39</sup> Information on whether children were natural, stepchildren, adopted or fostered was also collected; both SHARE and ELSA did not collect the full fertility history of women but only the number of children alive at the time of the survey was asked to both men and women. Full details on children's characteristics can be found elsewhere (Marmot et al., 2003a, Börsch-Supan and Jürges, 2005).

### 6.4 EMPLOYMENT AND SOCIO-ECONOMIC POSITION

The SHARE and ELSA questionnaires asked respondents a number of questions about their current employment situation. Workers were also asked questions on job characteristics and pensioners were asked questions about their last occupation before retirement.

### 6.4.1 EMPLOYMENT STATUS

ELSA and SHARE respondents were initially asked to describe their current employment situation; a set of joint crosscheck questions were then asked to those who described themselves as not-employed to assess whether they did any paid activities in the month prior to the interview. In particular, the SHARE questionnaire asked respondents to self-assess their current job situation, describing it from six mutually exclusive answers: 'retired', 'employed or self-employed (including working for family business)', 'unemployed', 'permanently sick or disabled', 'homemaker' or 'other'. ELSA posed a similar question even though the categories 'employed' and 'selfemployed' were separated and instead of 'homemaker' the wording 'looking after home or family' was used. In this project, SHARE and ELSA respondents were labelled as 'workers' if they had self-described their current situation as employed or self-employed and they were not temporarily away from any work, including seasonal work. In theory, respondents who did some paid work during the four weeks prior to the interview (even for just a few hours) could also be considered as workers (regardless of their economic description). Although differences between these two measurements were almost imperceptible, the first definition was considered in the following analyses as this accounts for people who describe themselves as workers and actually work on a regular basis. As for retired people, the definition of retirement was left to the respondents: this means that this category might include people who have withdrawn from the labour market permanently as well as those who are in receipt of a pension even if they never were in employment.

In all four countries considered 'retired' people represented the largest group: between 48% (Denmark) and 67% (Italy) of male respondents were 'retired' whereas for females percentages ranged between 45% (Italy) and 56% (Denmark). The second most frequent
category was 'employed or self-employed' with the exception of Italian older women who assessed themselves as homemakers in 39% of cases (see Table 6.8). Whereas 20% of French older women and almost 17% of English female respondents called themselves 'homemakers', less than 3% of Danish women did so. Conversely, 33% of Danish women were workers compared to less than 15% of Italian women. As for men, 28% of Italian older men described themselves as employed compared to 32% in France, 39% in England and 43% in Denmark. Unemployment was higher among men and figures were higher in Denmark than in the other countries reflecting the fact that in Denmark people remain part of the labour force until older ages than in other countries and so are more likely to describe themselves as unemployed than as retired. Finally, whereas in Italy around 1% of respondents called themselves 'permanently sick or disabled' in England figures were above 6% (see Table 6.8).

	Denmark		France		Ita	ly	England		
	Μ	F	Μ	F	M	F	M	F	
Retired	48.5	56.0	60.2	50.8	67.7	45.7	49.5	48.5	
(Self) Employed	43.3	33.3	31.9	23.9	27.9	13.1	39.0	27.0	
Unemployed	5.6	3.9	3.9	2.9	3.0	1.1	2.0	0.4	
Sick/Disabled	2.2	3.9	3.2	1.9	1.3	0.8	7.2	6.1	
Homemaker	0.1	3.0	0.6	20.0	0.1	39.2	1.1	16.9	
Other*	0.2	-	0.2	0.5		0.1	1.3	1.1	
N Observations	751	856	1 313	1 626	1 1 1 8	1 377	5 186	6 205	
Missing	6	1	50	47	7	5	-	-	

Table 6.8 Distribution of self-reported current economic status, by sex and country

Notes: \*includes also 'refusal' and 'don't know' and 'Semi-retired' in the ELSA survey; moreover, the wording "homemaker" includes people who described their situation as "looking after home or family" in ELSA. Source: SHARE, 2004; ELSA 2002; weighted data

Since most of the people did not work after the age of 70 (see Figure 6.1, where the categories 'Unemployed', 'Permanently sick or disabled', 'Homemaker' have been grouped together), some more detailed descriptions are presented for respondents aged 50-69. As expected from welfare regime descriptions and differences in pension ages highlighted in chapter 3, findings reflected expected differences in labour participation rates between countries and between men and women both between and within countries. Italian and French people exited the labour force at earlier ages than in Denmark and England even if the relationship between age and proportion of workers

did not seem to be linear. Around the age of 60 there was a considerable drop of the proportion of workers which became less than 10% after the age of 65 in all countries.

Comparing the SHARE and ELSA proportions of respondents aged 50-69 in paid work with the OECD<sup>40</sup> figures, we note that in most cases the sampled data fitted reasonably well with a few exceptions. For instance, French female workers aged 60-64 were overrepresented whereas 55-59 Danish male workers were under-estimated in SHARE. When calibrated weights were not used, Danish men aged 50-59 who were in paid work were also underestimated and so were Italian men aged 60-64. As regards the working status among English ELSA respondents, regardless of the use of calibrated weights or not, there seemed to be an under-representation of the proportion of men in paid work (in all age-groups) and of women aged 60-64 (see Table 6.9).

<sup>&</sup>lt;sup>40</sup> OECD does not provide data for England only; for the latter, proportions of people in paid work were obtained through the ONS website.



These figures graphically represent the proportion of the self-reported economic activity grouped into three categories (employed; retired; all others including homemakers, sick and unemployed people) among Danish, French, Italian and English respondents at each age point. Sources: SHARE, 2004; ELSA, 2002. Weighted data.

				50-54	55-59	60-64	65-69
	L	OECD		85.0	81.4	49.3	13.9
SK	lei	SHADE	W	80.8 (74.0; 86.2)	72.7 (64.4; 79.6)	46.3 (37.7, 55.1)	9.1 (4.7; 16.7)
[Y]	<b>F</b>	SHAKE	U	80.5 (74.8; 84.8)	73.6 (65.8; 80.2)	46.3 (38.1, 54.7)	9.8 (5.2; 17.8)
Ň	u	OECD SHARE		78.6	72.4	27.0	70
E	me	Olei	W	80 8 (73 5 86 5)	67 7 (60 0: 74 4)	27.0	72(35:155)
a	No.	SHARE	U U	80 2 (73 2: 85 8)	67 0 (59 8: 73 6)	23.3(10.0, 31.7)	64(29, 135)
			0	00.2 (75.2, 05.0)	07.0 (39.8, 73.0)	22.2 (10.0, 50.0)	0.4 (2.3, 13.3)
	1	OECE		84.3	60.1	14.7	4.2
E	Iel	SHADE	W	83.6 (77.5; 88.4)	62.0 (56.1; 67.5)	17.3 (9.6; 29.4)	2.1 (0.8; 5.8)
N.		SHAKE	U	83.6 (77.6; 88.2)	60.1 (54.1; 65.9)	14.3 (7.8; 24.6)	2.2 (0.8; 6.0)
V	a	OFCI		71.2	40.5	12.4	<u> </u>
FR	nei	UECL	<b>N</b>	/1.3	49.3	12.4	2.1
	Voi	SHARE	W	(7.0, 100, 100, 100, 100, 100, 100, 100, 1	50.0 (46.5; 65.2)	19.3 (12.5; 28.5)	0.5 (0.0; 4.0)
		l	U	07.8 (62.4; 72.8)		19.5 (13.7; 27.6)	0.1 (0.0; 3.5)
	_	OECI	)	78.3	50.0	29.2	10.6
	Aen	OECI	<b>)</b> W	<b>78.3</b> 76.7 (67.0; 84.1)	<b>50.0</b> 49.4 (40.8; 58.2)	<b>29.2</b> 22.1 (15.4; 30.6)	<b>10.6</b> 9.7 (5.6; 16.4)
ΓX	Men	OECI SHARE	<b>W</b> U	78.3 76.7 (67.0; 84.1) 78.6 (70.0; 85.2)	<b>50.0</b> 49.4 (40.8; 58.2) 48.6 (41.1; 56.3)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6)	<b>10.6</b> 9.7 (5.6; 16.4) 7.9 (5.0; 12.5)
ALY.	n Men	OECI SHARE	W U	78.3 76.7 (67.0; 84.1) 78.6 (70.0; 85.2)	<b>50.0</b> 49.4 (40.8; 58.2) 48.6 (41.1; 56.3)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6)	<b>10.6</b> 9.7 (5.6; 16.4) 7.9 (5.0; 12.5)
ITALY	nen Men	OECI SHARE OECI	) W U	78.3 76.7 (67.0; 84.1) 78.6 (70.0; 85.2) 42.8	50.0 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) 25.2 26.5 (20.0 0.0 0)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b>	<b>10.6</b> 9.7 (5.6; 16.4) 7.9 (5.0; 12.5) <b>3.0</b>
ITALY	Vomen Men	OECI SHARE OECI SHARE	W U W	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)	50.0 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) 25.2 26.5 (20.9; 33.0)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6)	<b>10.6</b> 9.7 (5.6; 16.4) 7.9 (5.0; 12.5) <b>3.0</b> 5.3 (1.5; 16.7)
ITALY	Women Men	OECI SHARE OECI SHARE	W U W U	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)         46.2 (39.5; 53.1)	<b>50.0</b> 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) <b>25.2</b> 26.5 (20.9; 33.0) 23.7 (18.7; 29.6)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8)	10.6           9.7 (5.6; 16.4)           7.9 (5.0; 12.5)           3.0           5.3 (1.5; 16.7)           2.1 (0.7; 5.9)
ITALY	Women Men	OECI SHARE OECI SHARE ONS (E	W U W U	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)         46.2 (39.5; 53.1)         85.1	50.0 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) 25.2 26.5 (20.9; 33.0) 23.7 (18.7; 29.6) 75.4	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8) <b>52.2</b>	<b>10.6</b> 9.7 (5.6; 16.4) 7.9 (5.0; 12.5) <b>3.0</b> 5.3 (1.5; 16.7) 2.1 (0.7; 5.9) <b>15.2</b>
D ITALY	len Women Men	OECI SHARE OECI SHARE ONS (E	) W U W U N) W	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)         46.2 (39.5; 53.1)         85.1         80.8 (78.0; 83.3)	50.0 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) 25.2 26.5 (20.9; 33.0) 23.7 (18.7; 29.6) 75.4 69.0 (66.1; 71.9)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8) <b>52.2</b> 42.6 (39.2; 46.0)	<b>10.6</b> 9.7 (5.6; 16.4) 7.9 (5.0; 12.5) <b>3.0</b> 5.3 (1.5; 16.7) 2.1 (0.7; 5.9) <b>15.2</b> 9.2 (7.2; 11.2)
AND ITALY	Men Women Men	OECI SHARE OECI SHARE ONS (E ELSA	W U W U N) W U	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2) <b>42.8</b> 47.9 (41.2; 54.7)         46.2 (39.5; 53.1) <b>85.1</b> 80.8 (78.0; 83.3)         81.1 (78.6; 83.7)	<b>50.0</b> 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) <b>25.2</b> 26.5 (20.9; 33.0) 23.7 (18.7; 29.6) <b>75.4</b> 69.0 (66.1; 71.9) 69.2 (66.3; 72.0)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8) <b>52.2</b> 42.6 (39.2; 46.0) 42.7 (39.3; 46.1)	10.6         9.7 (5.6; 16.4)         7.9 (5.0; 12.5)         3.0         5.3 (1.5; 16.7)         2.1 (0.7; 5.9)         15.2         9.2 (7.2; 11.2)         9.2 (7.2; 11.2)
LAND ITALY	Men Women Men	OECI SHARE OECI SHARE ONS (E ELSA	W U W U N) W U	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)         46.2 (39.5; 53.1)         85.1         80.8 (78.0; 83.3)         81.1 (78.6; 83.7)	<b>50.0</b> 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) <b>25.2</b> 26.5 (20.9; 33.0) 23.7 (18.7; 29.6) <b>75.4</b> 69.0 (66.1; 71.9) 69.2 (66.3; 72.0)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8) <b>52.2</b> 42.6 (39.2; 46.0) 42.7 (39.3; 46.1)	10.6         9.7 (5.6; 16.4)         7.9 (5.0; 12.5)         3.0         5.3 (1.5; 16.7)         2.1 (0.7; 5.9)         15.2         9.2 (7.2; 11.2)         9.2 (7.2; 11.2)
VGLAND ITALY	en Men Women Men	OECI SHARE OECI SHARE ONS (E ELSA ONS (E	W U W U N) W U N)	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)         46.2 (39.5; 53.1)         85.1         80.8 (78.0; 83.3)         81.1 (78.6; 83.7)         71.7	50.0 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) 25.2 26.5 (20.9; 33.0) 23.7 (18.7; 29.6) 75.4 69.0 (66.1; 71.9) 69.2 (66.3; 72.0) 57.6	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8) <b>52.2</b> 42.6 (39.2; 46.0) 42.7 (39.3; 46.1) <b>26.0</b>	10.6         9.7 (5.6; 16.4)         7.9 (5.0; 12.5)         3.0         5.3 (1.5; 16.7)         2.1 (0.7; 5.9)         15.2         9.2 (7.2; 11.2)         9.2 (7.2; 11.2)         8.7
ENGLAND ITALY	omen Men Women Men	OECI SHARE OECI SHARE ONS (E ELSA ONS (E	W U W U N) W U W	78.3         76.7 (67.0; 84.1)         78.6 (70.0; 85.2)         42.8         47.9 (41.2; 54.7)         46.2 (39.5; 53.1)         85.1         80.8 (78.0; 83.3)         81.1 (78.6; 83.7)         70.4 (67.7; 73.1)	<b>50.0</b> 49.4 (40.8; 58.2) 48.6 (41.1; 56.3) <b>25.2</b> 26.5 (20.9; 33.0) 23.7 (18.7; 29.6) <b>75.4</b> 69.0 (66.1; 71.9) 69.2 (66.3; 72.0) <b>57.6</b> 55.7 (52.8; 58.6)	<b>29.2</b> 22.1 (15.4; 30.6) 20.6 (15.1; 27.6) <b>9.1</b> 6.0 (3.4; 10.6) 5.2 (3.0; 8.8) <b>52.2</b> 42.6 (39.2; 46.0) 42.7 (39.3; 46.1) <b>26.0</b> 23.0 (20.2; 25.9)	10.6         9.7 (5.6; 16.4)         7.9 (5.0; 12.5)         3.0         5.3 (1.5; 16.7)         2.1 (0.7; 5.9)         15.2         9.2 (7.2; 11.2)         9.2 (7.2; 11.2)         8.7         7.4 (5.7; 9.4)

Table 6.9 Percentages of the SHARE and ELSA respondents 'in paid work', by gender and age (<70). Comparisons with OECD statistics

This table shows the percentage of people aged 50-69 in paid work by age-group and gender. Survey data are presented with 95% confidence intervals; both data obtained with calibrated weights (W) and without weights (U) are presented. Sources: OECD, 2004; SHARE 2004; ONS, 2001, ELSA, 2002.

#### **6.4.2 HOUSEHOLD INCOME AND WEALTH**

ELSA and SHARE respondents were asked detailed questions on their finances and wealth (pension, income, transfers and financial and non-financial assets). Those who refused or did not give exact answers were asked whether their circumstances were higher or lower than suggested monetary amounts (Taylor et al., 2007)<sup>41</sup>. Even though this method significantly reduced the number of observations for which no information is collected, when all the components of wealth and income of the households were considered together, the proportion of respondents with at least one missing value was very high (almost 60% among SHARE countries). Just as an example, more than 12% of pension amounts were missing. Moreover, whereas ELSA wealth and income variables were asked at the 'benefit unit' level, SHARE measures were asked at the 'couple' level but referred to the household as unit of observation. Given that in SHARE a household may consist of more than one "couple", it is not always possible to distinguish to which member of the household the income and wealth measures refer. Comparisons are further complicated by the fact that the total household income is calculated differently: ELSA, for instance, excluded housing and council tax benefits and SHARE considered income before tax. Finally, in wave 1, SHARE respondents of Euro countries were allowed to choose whether to report financial amounts in Euros or in their local pre-Euro currencies.

As an indicator of wealth, house ownership was therefore considered. People were asked in both questionnaires whether they owned the house where they lived (mortgage-free or not) or whether they paid a monthly rent or mortgage. Table 6.10 shows the percentage of older people by sex and age-group who owned a mortgage-free house or live rent-free (i.e. the proportion of people who did not have to pay monthly a mortgage or a rent). Whereas 80% of the Italian sample owned a mortgage/rent-free house, only one quarter of the Danish older people did. In particular, whereas more than 70% of Italian men aged 50-59 did not pay any monthly mortgage or rent, only 56% of the French

<sup>&</sup>lt;sup>41</sup> For example, if a respondent did not know how much the last payment they received from a particular pension was, then they would have been asked an unfolding bracket question such as "Was it less than  $\pounds 600$ , more than  $\pounds 600$ , or what?" If the respondent said they received "less than  $\pounds 600$ ", then they could have been asked "Was it less than  $\pounds 300$ , more than  $\pounds 300$ , or what?"

counterpart was in the same position compared to 33% of English men and 14% of Danish men aged 50-59.

		<u>.</u>	A	ge	<i>J</i>					
	50-	-59	60-	-69	70-	-79	8	)+	Tota	l (N)
	Μ	F	Μ	F	Μ	F	M	F	Male	Female
Denmark	13.9	16.7	30.7	34.2	35.76	33.6	36.9	31.9	<b>24.8</b> (187)	<b>27.1</b> (222)
N	314	324	227	225	145	185	64	107	750	841
France	55.9	58.7	68.2	72.5	83.14	79.5	79.3	68.8	<b>67.5</b> (898)	<b>68.7</b> (1,112)
Ν	566	606	363	417	293	380	115	225	1,337	1,628
Italy	70.9	74.3	81.1	82.3	85.02	83.0	96.1	85.4	<b>79.6</b> (924)	<b>80.4</b> (1,130)
N	351	468	416	516	276	274	70	106	1,113	1,364
England	33.1	40.1	63.5	69.5	73.45	67.0	68.2	59.9	<b>54.0</b> (2,876)	<b>57.3</b> (3,564)
N	1,908	2,234	1,604	1,779	1,170	1,387	484	773	5,166	6,173

Table 6.10 Percentage of mortgage-free house ownership, by age and sex and country

This table shows the percentage of people who live in a mortgage and rent-free flat/house, by age and sex in Denmark, France, Italy and England. Source: ELSA, 2002; SHARE, 2004. Weighted data.

## 6.5 HEALTH STATUS AND ITS MEASUREMENT

Physical and mental health as well as physical and cognitive functioning were key areas of focus for this research study since they could influence and be influenced by people's employment patterns, their socio-economic position and the levels at which in particular older people participated in society. ELSA and SHARE collected measurements of different facets of health, reflecting the complexity of this concept which goes beyond the mere 'absence of disease' (Saylor, 2004, Blanchet et al., 2007, Kirsten et al., 2009) and is defined by the WHO (2006) as "a state of complete physical, mental and social well-being and not merely the absence of disease or infirmity".

In its operationalisation, health status usually draws on three distinct main models: a medical model which envisages health in relationship with certain physical or psychological standards; a functional model which defines health as the capability of performing daily activities in the social environment and a subjective model which

gleans the health status of individuals from their personal perception. SHARE and ELSA adopted a similar strategy in collecting health information: in both questionnaires questions on physical pain and self-reported diagnosed chronic conditions, on self-reported functional health (problems with activities of daily limits or disabilities), on subjective health, mental health as well as cognitive functioning and health-related behaviours were included.

These measurements are mainly self-assessed and present some limitations. For instance, it is well known that the way people evaluate their medical status and include it in the perception of their health differs depending on the country-specific socioeconomic position of the respondent (Etilé and Milcent, 2006, Bago d'Uva et al., 2008, Jürges et al., 2008, Dowd and Zajacova, 2010). Difference in self-rated health (SRH) between countries and subgroups of the population do not necessarily correspond to differences in 'true' health. SRH might also be sensitive to cultural environment as well as to different health expectations (Sherwood et al., 2005, Haski-Leventhal, 2009). As Sen (2002) stated quite clearly in an obviously extreme case, "a person brought up in a community with a great many diseases and few medical facilities may be inclined to take certain symptoms as 'normal' when they are preventable" (pg 860) and this could potentially bias the perception people have of their own health status. Similarly, differing expectations for health could potentially also lead to differences in response category cut points. For example, a 90 year old man who struggles to climb the stairs might characterise himself as having "mild difficulties" in moving around, but a 40 year old man with the same mobility might describe himself as having "moderate to high difficulties". Clearly, different response category cut-points about mobility and functional limitations could also lead to potentially incomparable answers (Salomon et al., 2004). Furthermore, when chronic conditions are considered, this thesis examines co-morbidity without distinguishing between various possible combinations of chronic diseases (such as cardiovascular diseases, chronic conditions or visual impairments). Moreover, respondents might report chronic diseases differently, depending on their level of education and memory. In addition, self-reported morbidity might not be an accurate measure of the prevalence of chronic conditions at the population level because only respondents who have a 'diagnosed' condition would report it and because of recall bias (Heliövaara et al., 1993). Finally, depression symptoms scales ask for symptoms experienced in various temporal frames (the week or the month prior to the interview, and 'recently'): although diagnosed depression is a chronic disease, it is likely that respondents who experienced symptoms "some of the time" would answer "yes", overstating the prevalence of the items and therefore the prevalence of "respondents at risk" of depression.

Keeping in mind the above-mentioned limitations, the following subsections present a selection of health-related outcomes and only consider the variables which are present in both questionnaires and are fully comparable.

## 6.5.1 SELF-RATED HEALTH (SRH)

Each respondent was asked to self-assess their health status with a generic question (Would you say your health is ... ). SHARE respondents were asked to evaluate their health using two different scales: the 'Rand' scale includes 'excellent', 'very good', 'good', 'fair' and 'poor' whereas the 'Europ' formulation comprises 'very good', 'good', 'fair', 'bad' and 'very bad'.

In both studies, respondents were randomly allocated to two subgroups; this allocation determined the order of and the scale used for the self-assessed general health question, i.e. whether they had to answer with the 'Rand' or 'Europ' scale before or after a detailed series of questions about their physical health. Both the position as well as of the formulation used to self-assess general health are important factors which can influence response, as discussed by several authors (Jürges, 2005, Bago d'Uva et al., 2008, Bowling and Windsor, 2008, Jürges, 2008, Jürges et al., 2008). Despite that, since in the second and following waves, SHARE and ELSA research teams decided to drop the 'Europ' scale, data on the 'Rand' self-classification of health only is presented. Moreover, all the 'Rand' answers were grouped together regardless of where they were placed in the questionnaire<sup>42</sup>.

<sup>&</sup>lt;sup>42</sup> Because of the random allocation procedure, there is no reason to believe that people who answered the self-assessment health question at the end of the questionnaire systematically differ by socio-economic or demographic characteristics to the ones who answer it at the beginning of the health module. Similarly, it is assumed that any effect of the position of the question on the answer will be similar among men and women in all countries under study.

Men in all countries reported better health than women. Similarly, an age gradient was clear: older people were more likely to report poor health than people in the younger age groups; for example, among French respondents 14% of men and 12% of women aged 50-59 rated their health as excellent compared with 1% and 4% respectively of those aged over 80. Conversely, only 6% of Italian women and men in the youngest age group (50-59) reported poor health whereas among 80+ this percentage increased to 19% among men and 23% among women. As regards countries, whereas more than 20% of Danish men reported their health as excellent, only 13% did so in England and 8% in France and Italy; similarly, 19% of Danish women said they had excellent health compared with 13% in England, 7% in France and only 5% in Italy (see Table 6.11). Such differences in health suggest that health status varies across welfare state regimes, with people in Scandinavian countries having better general health than Southern welfare regimes, as several other papers have shown (Bambra, 2006, Eikemo et al., 2008). These disparities in self-perceived health between countries were thought to reflect broader differences in their welfare state regimes and the implementation of policies aimed at reducing social inequalities (Navarro et al., 2006, Chung and Muntaner, 2007, Espelt et al., 2008). Suggested explanations for the better performance of the Scandinavian welfare state regimes were numerous. It was believed that health was enhanced where the welfare provision was more generous and income redistribution was higher, providing better protection against the health effects of a low socioeconomic position and reducing the welfare burden on families and women in particular (Torsheim et al., 2006, Bambra, 2007a). The implementation, over a sustained period of time, of a variety of policies (such as universal access to generous welfare services and high levels of employment amongst both men and women) might also have resulted in narrower income inequalities and higher levels of decommodification, both of which are associated with better population health (Bambra, 2006, Navarro et al., 2006, Chung and Muntaner, 2007). These differences in reporting health might also depend on cultural differences that a term like 'excellent health' carries in different countries (Jürges, 2005, Kapteyn, 2006).

Distributions of self-rated health were compared with other sources in France, Italy and England. Whereas in France and Italy surveys on health were conducted by their

National Statistics Bureaux, in England the 2001 Census asked directly whether over the previous 12 months the person's health had on the whole been 'good', 'fairly good' or 'not good'. In France, the 2007 INSEE survey "Vie quotidienne et Santé" investigated, among other measures, self-perceived health. The health survey had almost 238,000 respondents representative of the whole country and reached a response rate of more than 80%<sup>43</sup>, representing a similar reliable source for data quality. In Italy, the 2005 ISTAT "Condizione di salute e ricorso ai servizi sanitari"<sup>44</sup> survey sampled 60.000 household and had a response rate of about 84%<sup>45</sup>.

Table 6.11 Distribution of Self-Reported Health, by sex, age-group and country

		50	-59	60-	69	70-	79	80	H	Το	tal
	%	M	<u> </u>	M	F	M	F	M	F	Male	Female
	Excellent	23.7	26.9	25.0	18.9	10.5	12.6	14.0	3.9	20.9 (158)	<b>18.5</b> (159)
X	V. Good	29.2	32.2	30.5	32.2	31.8	30.0	10.9	19.6	<b>28.6</b> (213)	<b>29.9</b> (255)
Z	Good	27.0	23.1	21.6	23.1	32.0	25.9	21.5	34.4	25.8 (197)	25.3 (211)
Ž	Fair	14.2	11.3	16.8	19.3	19.7	28.0	45.6	35.2	<b>18.5</b> (140)	20.2 (179)
E	Poor	5.6	6.6	6.0	6.4	6.0	3.5	8.0	6.8	<b>6.0</b> (45)	<b>6.0</b> (53)
	N	314	328	228	229	147	188	64	112	<u>75</u> 3	857
	Excellent	13.6	11.6	7.7	6.5	3.1	2.7	0.9	3.6	8.5 (116)	<b>6.8</b> (115)
E	V. Good	22.3	20.3	15.6	14.9	7.9	8.7	7.0	6.1	15.7 (205)	<b>13.8</b> (233)
12	Good	44.0	45.9	45.9	46.8	42.7	43.5	32.5	34.2	<b>43.1</b> (571)	43.6 (726)
3	Fair	14.9	18.1	22.2	26.0	30.3	33.8	35.0	36.1	<b>22.4</b> (291)	<b>26.9</b> (435)
E	Poor	4.9	3.9	8.4	5.8	16.0	11.3	24.6	20.0	<b>10.3</b> (136)	<b>8.9</b> (133)
	N	554	607	361	423	291	382	113	230	1,319	1,642
	Excellent	12.8	8.1	7.8	7.1	7.5	0.9	4.0	-	8.4 (93)	4.8 (65)
	V. Good	18.4	15.5	15.6	11.7	13.4	6.0	6.9	3.9	15.3 (165)	10.3 (159)
5	Good	49.7	44.5	40.1	40.4	34.5	32.5	33.0	23.0	41.7 (470)	37.1 (529)
2	Fair	13.1	25.9	29.9	33.4	35.6	41.7	36.7	50.3	25.7 (296)	35.7 (467)
Ĩ.	Poor	6.1	6.1	6.4	7.4	12.8	18.9	19.4	22.8	<b>8.9</b> (95)	<b>12.1</b> (159)
	N	353	473	418	524	278	273	70	109	1,119	1,379
Γ	Excellent	16.4	16.2	14.2	13.9	9.1	8.2	6.0	9.2	13.2 (657)	12.7 (768)
9	V. Good	32.2	31.2	28.4	30.3	27.8	25.7	24.3	22.9	<b>29.4</b> (1,486)	28.7 (1,753)
3	Good	30.9	31.0	28.4	33.1	31.3	33.2	30.3	31.2	<b>30.2</b> (1,543)	32.2 (1,965)
15	Fair	13.7	15.6	21.1	17.0	22.1	23.9	27.0	27.0	<b>18.9</b> (986)	<b>19.5</b> (1,201)
Ž	Poor	6.8	5.5	7.9	5.7	9.8	9.0	12.4	9.8	<b>8.3</b> (430)	<b>6.9</b> (427)
Γ	N	1,892	2,235	1,585	1,768	1,155	1,374	470	737	5,102	6,114

Each column shows the percentages of men or women in each particular age-group who reported their health as excellent, very good, good, fair or poor. Sources: SHARE, 2004; ELSA, 2002. Weighted data.

<sup>44</sup>Health data can be consulted and downloaded at www.istat.it/sanita/sociosan/. More details about the survey can be found on http://www.istat.it/dati/dataset/20080131\_00/notametodologica.pdf <sup>45</sup>www.agenas.it/agenas\_pdf/Supplementi\_Monitor/indagine\_ISTAT\_multiscopo/Quaderno\_Monitor\_ISTAT.pdf

<sup>&</sup>lt;sup>43</sup> www.insee.fr/fr/publications-et-services/docs\_doc\_travail/Docf1001.pdf

Bearing in mind that data were not fully comparable (the English census and the Italian survey used different response scales<sup>46</sup>), some health-effect bias was present. In fact, for instance, whereas the proportion of English people reporting good health in the census data halved as age-group increased, the decline was not as sheer among ELSA respondents. Similarly, even if data from the France and Italy were also affected by potential non-response bias having being collected in surveys, both INSEE and ISTAT surveys sampled a considerable amount of households and respondents rendering these reliable sources. Compared to these surveys, French SHARE respondents tended to report similar self-perceived health with a few exceptions among respondents older than 74. Among Italian SHARE respondents, on the contrary, the proportion of people reporting 'good health' was higher at older ages compared to the ISTAT survey data (see Table 6A.5 for details).

## **6.5.2 CHRONIC DISEASES**

SHARE and ELSA also asked participants whether they were diagnosed by a practitioner to have chronic illnesses/conditions<sup>47</sup>. Table 6.12 shows the age-sex distribution of the total reported number of chronic illnesses derived summing all reported chronic illnesses (this indicator ranged from 'no reported conditions' to 'multiple chronic illnesses'). Treating these conditions as a continuous variable is advantageous because –despite the fact that each specific morbidity may have a specific and unique effect –persons with multiple conditions tend to have more rapid declines in health status and well-being and to be more likely to have greater decrements in functioning (Stewart et al., 1989, Fried et al., 1999, Wolff et al., 2002).

<sup>&</sup>lt;sup>46</sup> Whereas SHARE questionnaire classifies health as "Molto buona; Buona; Discreta, Cattiva or Molto cattiva", ISTAT survey uses the following categories "Molto bene, Bene, Discretamente, Male e Molto male". Similarly, the English census health classifications are 'good', 'fairly good' or 'not good' whereas in ELSA the scale comprises 'very good', 'good', 'fair', 'bad' or 'very bad'.

<sup>&</sup>lt;sup>47</sup> The conditions mentioned are 'heart attack or other heart problems', 'stroke or cerebral vascular disease', 'diabetes', 'chronic lung disease', 'asthma', 'arthritis or rheumatism', 'osteoporosis', 'cancer or malignant tumor'. 'stomach, duodenal, or peptic ulcer', 'Parkinson's disease, 'cataract', 'hip or femoral fracture' or 'high blood cholesterol'. In ELSA, the conditions 'stomach or duodenal ulcer, peptic ulcer' and high blood cholesterol' are not present.

Men (with a few exceptions) had 'no chronic conditions' more often than women, regardless of their age. Also, the percentage of people who were told by a doctor that they had chronic illnesses as well as the number of declared illnesses increased with age: for example, just around 45% of Italian male respondents and 42% of Italian women aged between 50 and 59 said that they had no chronic illnesses compared with 23% of men and 12% of women aged 80 or over. Similarly, whereas only around 3% of Danish respondents aged 50-59 declared more than 4 chronic illnesses, 16% of men and 21% of women aged 80 or more had more than 4 chronic illnesses. Differences between countries were mainly noticeable at older ages: 22% of Italian and English men aged 80+ declared no chronic illnesses compared to 17% among Danish men and 11% among French men; similarly, only 12% of English women aged 80+ declared more than 4 chronic illnesses compared with 18% of French and 21% of Danish and Italian women in the same age-group. For the younger age-groups (50-69) the age-sex distributions of the number of chronic condition were very similar in all four countries under study (see Table 6.12 for full details).

		50-	59	60-	69	70-	79	80	<u>+  </u>	Tot	tal	
	#	М	F	М	F	М	F	М	F	Male	Female	
	0	53.0	42.2	33.8	28.5	18.7	19.1	17.4	11.8	38.1 (286)	<b>29.4</b> (255)	
RK	1	24.3	33.8	30.6	31.7	35.2	30.9	34.5	19.1	<b>29.0</b> (228)	30.5 (259)	
I	2	14.3	15.1	17.0	20.9	14.9	19.9	19.3	30.2	15.7 (116)	<b>19.9</b> (172)	
N N	3	5.2	6.1	13.5	10.3	17.9	18.1	12.5	17.6	10.5 (75)	11.3 (97)	
E	4+	3.1	2.8	5.1	8.7	13.3	12.0	16.4	21.3	<b>6.7</b> (52)	<b>8.9</b> (75)	
	Ν	317	328	228	230	147	188	65	112	757	858	
						10.0						
	0	48.8	42.1	31.0	25.9	19.8	16.3	11.4	15.7	<b>33.9</b> (468)	27.4 (470)	
E	1	31.9	35.3	32.4	33.9	30.9	26.1	31.2	27.3	<b>31.8</b> (434)	<b>31.4</b> (535)	
Z	2	12.9	14.4	21.1	22.9	24.3	27.2	21.9	22.0	<b>18.5</b> (256)	<b>20.8</b> (346)	
×	3	2.8	4.7	11.7	9.7	14.7	17.2	22.5	16.8	<b>9.8</b> (127)	<b>11.0</b> (184)	
E	4+	3.6	3.5	3.8	7.6	10.3	13.2	13.0	18.2	<b>6.0</b> (83)	<b>9.4</b> (149)	
	Ν	579	628	374	432	300	389	115	235	1,368	1,684	
		45.0	10.0	00.0	02.0	01.1	16.6	00.7	110	22 E (2(C))		
		45.0	42.2	28.8	23.9	21.1	15.5	22.7	11.9	32.5 (368)	<b>25.7</b> (367)	
×		35.1	28.0	33.9	34.2	28.2	24.4	34.2	25.4	<b>33.1</b> (365)	28.5 (399)	
F	2	13.3	17.4	21.4	17.8	23.6	22.4	16.1	21.7	18.4 (208)	19.4 (278)	
Ē	3	4.9	7.8	7.7	13.6	14.38	14.6	19.3	19.2	9.2 (106)	12.9 (161)	
	4+	1.7	4.6	8.2	10.5	12.9	23.1	7.8	21.7	6.8 (74)	13.5 (177)	
	N	357	474	421	525	278	274	70	109	1,126	1,382	
	0	41.9	39.4	31.3	25.6	24.6	18.6	21.5	17.4	<b>32.3</b> (1,253)	26.5 (1,271)	
9	1	32.4	32.3	32.0	31.6	29.6	26.1	33.7	29.8	31.7 (1,260)	30.2 (1,465)	
A	2	14.9	17.7	19.5	23.0	24.2	26.5	23.3	25.9	19.5 (782)	22.8 (1,108)	
5	3	7.7	7.8	10.2	13.2	13.0	17.6	14.1	15.3	10.4 (410)	13.1 (626)	
Ž	4+	3.2	2.8	7.0	6.6	8.6	11.1	7.4	11.6	<b>6.1</b> (245)	7.4 (360)	
	N	1,198	1,339	1,326	1,548	997	1,191	429	692	3,950	4,830	

 
 Table 6.12 Percentage distribution of number of chronic illnesses, by sex age-group and country

This table shows the percentage of people reporting a specific number of chronic conditions by age-group and gender in Denmark, France, Italy and England. Sources: SHARE, 2004; ELSA, 2002. Weighted data.

#### **6.5.3 PHYSICAL FUNCTIONING**

Physical functioning is the ability of a person to perform normal activities of daily living without restrictions (WHO, 1980, Verbrugge and Jette, 1994, WHO, 2001b). In both SHARE and ELSA, questions about physical functioning were divided in three main domains: activities of daily living (ADLs); instrumental activities of daily living (IADLs); mobility and both upper and lower limb function. SHARE (except Denmark) and ELSA excluded respondents who received long-term care in institutions who are the most disabled: this exclusion is most relevant for France and England where the proportions of people older than 65 (women in particular) living in institutions are higher (see Chapter 3).

#### **MOBILITY AND ARM FUNCTIONING**

SHARE and ELSA asked whether, because of a health problem, respondents had any difficulties expected to last more than three months doing a number of activities used as measures of mobility (or lower-limb function) (MacInnes, 2006) and physical (upper-limb) functioning (Verdugo, 2006), here reported jointly (Calasanti and Bonanno, 1986).

The activities considered included 'walking 100 metres', 'sitting for about two hours', 'getting up from a chair after sitting for long periods', 'climbing several flights of stairs without resting', 'climbing one flight of stairs without resting', 'stooping, kneeling, or crouching', 'reaching or extending your arms above shoulder level', 'pulling or pushing large objects like a living room chair', 'lifting or carrying weights over 5 kilos<sup>48</sup> like a heavy bag of groceries' and 'picking up a small coin<sup>49</sup> from a table'.

Comparatively high rates of difficulty were reported by both sexes with 'climbing several flights of stairs' (English men up to 29%, Italian and English women up to 40%), 'stooping, kneeling or crouching' (men up to 31% in England, women up to 40% in England and Italy), 'lifting or carrying heavy weights' and 'getting up from a chair after sitting for long periods' (men up to 19% in Denmark, women up to 29% in England) (Table 6.A6).

<sup>&</sup>lt;sup>48</sup> 10 pounds in the ELSA questionnaire.

<sup>&</sup>lt;sup>49</sup> 5p coin in the ELSA questionnaire.

In all countries, women always reported more difficulty with mobility and arm functioning than men at all ages. For instance, whereas 66% of all Danish men reported no mobility difficulty, less than half of the female sample reported no difficulty. The prevalence of reported difficulty with mobility and arm function increased with age: 64% of English respondents aged 50–59 reported no difficulty compared with 19% of those aged 80 and over. As for countries, French and Danish respondents aged 50-69 tended to report no limitations more often than the English and Italian ones. For older age-groups (70+), Danish SHARE participants tended to report no limitations more often than French, Italian and English respondents. For the latter countries, no clear pattern emerged from the data (see Table 6.13).

				·	a	nu cou	<u>uu y ( /</u>	0/			
		50-	-59	60-	-69	70-	<b>.79</b>	80	)+	То	tal
	%	M	F	M	F	M	F	M	F	Male	Female
	0	76.2	64.5	67.5	52.3	52.0	39.3	36.8	19.5	<b>65.9</b> (503)	49.7 (429)
R	1	13.8	11.5	16.1	14.7	16.7	17.8	9.9	13.5	14.7 (106)	13.9 (122)
IA	2	2.8	6.5	5.9	12.4	15.0	14.6	15.5	17.9	7.0 (54)	11.3 (96)
Ž	3	2.3	5.0	5.8	6.8	5.9	10.2	14.2	8.2	5.0 (38)	7.0 (59)
E	4+	4.9	12.5	4.8	13.8	10.4	18.2	23.6	41.0	7.4 (55)	18.1 (151)
8	Ν	316	328	228	229	147	188	65	112	756	857
	0	78.9	61.0	64.7	48.5	45.8	26.7	29.7	16.4	<b>62.9</b> (864)	<b>42.0</b> (724)
E	1	8.6	17.0	16.2	21.8	15.5	17.4	16.6	11.7	13.0 (177)	17.4 (293)
Z	2	5.9	7.7	7.7	10.1	12.1	16.5	13.9	11.8	8.5 (116)	11.1 (182)
3	3	2.1	4.3	4.5	7.3	5.7	13.2	8.9	12.3	4.2 (56)	8.5 (140)
HE	4+	4.6	10.0	6.9	12.3	20.9	26.2	30.9	47.8	<b>11.4</b> (149)	<b>31.0</b> (331)
	N	575	614	373	432	299	389	115	235	1,362	1,670
	0	71.1	55.6	59.6	46.7	39.0	29.1	29.6	13.0	<b>56.5</b> (644)	40.0 (575)
	1	15.5	15.0	16.8	14.9	22.6	11.2	13.2	10.9	17.3 (193)	13.4 (193)
13	2	6.1	10.6	8.8	13.1	14.9	10.1	12.7	10.8	<b>9.6</b> (106)	11.2 (158)
LA	3	2.7	8.5	7.0	7.7	6.2	11.1	11.5	8.2	<b>5.6</b> (65)	8.9 (128)
1	4+	4.7	10.4	7.8	17.7	27.3	38.5	33.0	57.0	<b>11.0</b> (117)	<b>26.5</b> (328)
	N	356	474	421	525	278	274	70	109	1,125	1,382
	0	63.9	50.3	51.4	39.9	39.6	25.0	25.9	18.9	<b>51.6</b> (2,618)	37.1 (2,296)
	1	15.5	16.9	16.1	17.8	20.0	15.5	18.0	10.4	<b>16.8</b> (883)	15.9 (999)
	2	5.1	10.9	7.9	11.6	10.3	11.5	10.1	10.1	7.5 (402)	11.1 (704)
5	3	4.0	6.0	6.0	7.5	8.3	12.4	9.4	13.8	<b>6.0</b> (321)	<b>9.0</b> (552)
Ż	4+	11.5	15.9	18.6	23.2	21.8	35.6	36.6	46.8	18.1 (958)	<b>26.9</b> (1,644)
	N	1,915	2,247	1,608	1,780	1,174	1,391	485	777	5,182	6,195

Table 6.13 Difficulty with mobility and upper-limb functions, by age-group, sex

This table shows the percentages of older people in Denmark, France, Italy and England reporting various number of difficulty with mobility items by age and sex. Sources: SHARE, 2004; ELSA, 2004. Weighted data.

### ACTIVITIES OF DAILY LIVING (ADLs)

To assess Activities of Daily Living (ADLs), respondents of both ELSA and SHARE surveys were shown a card from which they had to select the activities they had any difficulty with for at least three months because of a physical, mental, emotional or memory problem. The activities considered for the ADLs were the following: 'dressing, including putting on shoes and socks', 'walking across a room', 'bathing or showering', 'eating, such as cutting up your food', 'getting in or out of bed', 'using the toilet, including getting up or down'.

Particularly high rates of difficulty were reported in all countries and both among men and women for dressing (up to 10% for French men and Italian women) and bathing (6% and 10% among Italian men and women respectively) (see Table 6.A7).

The prevalence of reported difficulty with ADLs increased with age and similar patterns were observed in all countries: 95% of Italian male respondents aged 50–59 reported no difficulty compared with 66% of those aged 80 and over. As regards sex, there were in general very little differences between the sexes for reported difficulty with ADLs although –in general –disability was slightly higher in men up to age 69 in England and Italy and up to age 79 in France and Denmark, and in women over the age of 70 (England and Italy) and 80 (France and Denmark). The higher prevalence of ADL limitations among women at older ages is consistent with several other studies (Robertson, 1997, Lesthaeghe, 2000, UN, 2000).

		country (70)											
		50-59 M F		60-	·69	70-	79	80	)+	То	tal		
	%	M	F	M	F	M	F	М	F	Male	Female		
¥	0	93.8	94.6	90.9	91.1	89.9	90.6	68.5	67.5	<b>90.1</b> (683)	<b>89.0</b> (764)		
<b>NR</b>	1	3.9	2.9	6.1	7.1	6.5	5.4	14.7	15.3	<b>5.9</b> (44)	<b>6.3</b> (53)		
W	2	1.8	1.3	1.9	0.7	-	1.3	7.3	7.3	<b>2.0</b> (14)	<b>2.0</b> (17)		
EN	3+	0.3	1.2	1.1	1.1	3.6	2.7	9.5	9.9	<b>2.0</b> (15)	<b>2.7</b> (23)		
Δ	Ν	316	328	228	229	147	188	65	112	756	857		
[+]	0	93.4	94.9	91.8	93.9	82.9	86.6	65.0	62.0	<b>87.6</b> (1,196)	87.6 (1,480)		
Ū	1	4.5	3.6	5.2	4.3	10.7	8.6	12.0	15.6	<b>6.9</b> (95)	<b>6.8</b> (109)		
Z	2	1.4	1.4	2.5	1.3	4.1	2.7	13.9	10.6	<b>3.6</b> (47)	<b>3.0</b> (43)		
Ř	3+	0.7	0.1	0.5	0.5	2.3	2.1	9.1	11.8	1.9 (23)	2.6 (38)		
H	Ν	575	614	372	432	299	389	115	235	1,361	1,670		
	0	94.7	95.7	92.7	93.2	86.7	82.7	65.5	60.0	<b>90.0</b> (1,026)	<b>86.2</b> (1,218)		
X	1	3.2	3.0	4.1	3.6	6.6	6.8	17.6	15.0	<b>5.5</b> (50)	<b>6.0</b> (70)		
IV	2	0.4	0.6	0.8	1.3	3.6	4.1	8.3	8.1	1.7 (19)	<b>2.9</b> (34)		
	3+	1.7	0.7	2.4	1.9	3.1	6.4	8.6	16.9	<b>2.8</b> (30)	<b>4.9</b> (60)		
	Ν	356	474	421	525	278	274	70	109	1,125	1,382		
								_					
	0	87.1	88.0	81.7	82.3	75.1	72.6	63.2	58.4	<b>80.8</b> (4,156)	7 <b>8.6</b> (4,890)		
Z	1	6.4	5.2	8.9	9.4	14.6	13.2	15.9	20.2	<b>9.8</b> (525)	10.4 (630)		
ILA	2	2.5	3.3	3.9	3.8	4.3	6.8	10.4	11.0	<b>4.0</b> (212)	<b>5.4</b> (335)		
N N	3+	4.0	3.5	5.5	4.5	6.0	7.4	10.5	10.4	5.4 (161)	5.6 (340)		
	N	1,915	2,247	1,608	1,780	1,174	1,391	485	777	5,182	6,195		

Table 6.14 Difficulty with Activities of daily living (ADLs) by sex, age-group and country (%)

This table shows the percentage of older people in Denmark, France, Italy and England reporting various number of difficulties with activities of daily living (ADLs) by age-group and sex. Sources: SHARE, 2004; ELSA, 2002. Weighted data.

#### **INSTRUMENTAL ACTIVITIES OF DAILY LIVING (IADLS)**

To assess Instrumental Activities of Daily Living (ADLs), respondents of both ELSA and SHARE surveys were asked to select the activities they had any difficulty with for at least three months because of a physical, mental, emotional or memory problem. The activities considered for the IADLs were the following: 'using a map to figure out how to get around in a strange place', 'preparing a hot meal', 'shopping for groceries', 'making telephone calls', 'taking medications', 'doing work around the house or garden', 'managing money, such as paying bills and keeping track of expenses'.

Comparatively high rates of difficulty were reported for 'doing work around the house and garden', 'shopping for groceries' as well as 'using a map'; percentages reporting these difficulties were almost double among women compared to men (see Table 6.A8). The prevalence of reported difficulty with IADLs increased with age, with less than 6% of male respondents aged 50–59 reporting difficulty compared with around 40% in Italy and France and almost 50% in Denmark of those aged 80 and over (see Table 6.15). In particular, there was a massive increase in the percentage of people reporting at least one difficulty with IADLs in the age-group 80+. Women always reported more difficulties with IADLs than men, regardless of their age and the country. Looking at country differences, whereas SHARE respondents had similar patterns and similar percentages of respondents reporting no difficulties, ELSA respondents tended to report at least one difficulty with IADL activities more often, particularly at younger ages.

The distributions of people who reported limitations with ADLs and IADLs for the French and Italian SHARE respondents was compared to the one obtained in the INSEE and ISTAT health surveys conducted in 2007 and 2005 respectively. Although the latter surveys asked slightly different questions<sup>50</sup>, given the number of respondents, these health surveys give a good indication of the prevalence of reported limitations. Table 6.16 shows the proportion of French people who reported having at least one IADL or ADL: The INSEE and SHARE distributions compared fairly well particularly for people aged more than 70; the younger SHARE respondents (aged 50-69), on the contrary, reported fewer health limitations in daily activities compared to the INSEE respondents. Table 6.17 shows the proportion of Italian people who reported having at least one serious ADL limitation. In both surveys it appeared than women reported more ADL limitations than men and the proportions increased as age increased. Whereas SHARE proportions were roughly three times higher than the ISTAT ones for people aged under 75, such ratio decreased for older respondents: this might suggest that older SHARE respondents were slightly healthier, though it could also imply that limitations were mainly serious as age increases. The two distributions were therefore hard to compare.

<sup>&</sup>lt;sup>50</sup> SHARE daily living activities are "1. Dressing, including putting on shoes and socks; 2. Walking across a room; 3. Bathing or showering; 4. Eating, such as cutting up your food; 5. Getting in or out of bed; 6. Using the toilet, including getting up or down". ISTAT questionnaire lists, on the contrary, the following ones: "1. Dressing and undressing; 2. Getting up and down on a chair; 3. Bathing or showering; 4. Eating, such as cutting up your food; 5. Getting in or out of bed; 6. Washing hand and face". ISTAT only reports the proportion of people who had '*serious*' difficulties.

	6							<u> </u>	- <u>p</u> -	Total	
		<u> </u>	-59	<u> </u>	69	<u> </u>	-79	80	+	To	tal
	%	M	F	M	F	M	F	M	F	Male	Female
¥	0	94.8	86.3	92.8	84.0	81.7	73.6	51.6	50.5	<b>88.2</b> (664)	7 <b>8.1</b> (670)
<b>NRI</b>	1	3.1	9.0	3.9	10.7	8.3	11.9	28.2	13.3	<b>6.4</b> (49)	10.6 (90)
Ň	2	1.1	2.7	1.4	2.5	4.0	8.1	2.6	12.5	<b>1.8</b> (15)	<b>5.1</b> (44)
EN	3+	1.0	2.0	1.9	2.8	6.0	6.4	17.6	23.7	3.6 (28)	<b>6.2</b> (53)
0	N	316	328	228	229	147	188	65	112	756	857
	0	95.0	92.0	93.1	89.1	77.9	74.6	61.4	43.8	<b>87.4</b> (1,197)	<b>78.7</b> (1,343)
IJ	1	3.1	5.0	4.1	7.4	9.7	16.4	7.8	17.2	<b>5.3</b> (73)	<b>10.4</b> (165)
3	2	0.7	1.3	1.0	1.6	4.5	5.2	6.9	10.3	<b>2.2</b> (30)	3.8 (60)
<b>A</b>	3+	1.2	1.7	1.8	2.0	7.9	3.9	23.8	28.7	<b>5.1</b> (61)	7.1 (102)
	N	575	614	372	432	299	389	115	235	1,361	1,670
	0	96.9	92.5	92.3	88.3	89.1	74.38	61.8	43.9	<b>90.5</b> (1,022)	<b>79.4</b> (1,134)
X	1	1.0	4.4	3.5	7.9	4.6	9.46	16.6	12.1	<b>4.0</b> (48)	7.8 (110)
A	2	0.6	1.4	1.9	1.9	1.5	5.88	3.9	13.2	1.5 (16)	4.4 (51)
E	3+	0.9	1.7	2.3	1.9	4.8	10.28	17.7	30.8	4.0 (39)	8.4 (87)
	N	356	474	421	525	278	274	70	109	1,125	1,382
	0	88.3	84.0	83.8	81.3	80.6	70.7	59.7	50.0	82.7 (4,266)	75.3 (4,689)
Z	1	5.8	9.0	7.6	9.8	11.3	14.6	17.6	18.2	8.6 (454)	11.9 (736)
TA	2	2.9	3.8	4.8	4.8	3.7	8.1	11.7	12.6	4.5 (239)	<b>6.3</b> (391)
2	3+	3.0	3.2	4.0	4.1	4.4	6.6	11.0	19.2	4.2 (223)	6.5 (379)
E	N	1,915	2,247	1,608	1,780	1,174	1,391	485	777	5,182	6,195

 Table 6.15 Distribution of number of difficulties with Instrumental Activities of daily living (IADL), by sex and age-group –Denmark

This table shows the percentage of older people in Denmark, France, Italy and England reporting various number of difficulty with instrumental activities of daily living (IADL) by age and sex. Sources: SHARE, 2004; ELSA, 2004. Weighted data.

# Table 6.16 Percentage distribution of INSEE and SHARE respondents who reported at least either one ADL or IADL limitation by age-group and sex

	50-54	55-59	60-64	65-69	70-74	75-79	80-84	85+
INSEE	12.7	15.6	15.7	20.2	27.5	35.5	48.8	70.1
SHARE	10.5	8.7	11.8	12.8	25.3	34.8	46.9	69.6

This table compares the age distribution of INSEE 'Survey of Health' and SHARE respondents who reported at least either one ADL or one IADL limitation. Sources: INSEE, 2007; SHARE 2004.

#### Table 6.17 Percentage distribution of ISTAT and SHARE respondents who reported at least one Activity of Daily Living (ADL) limitation, by age-group and sex

	45-54*		45-54* 55-64		65	-69	70-	-74	75-79		80+	
	M	W	Μ	W	Μ	W	M	W	M	W	M	W
ISTAT	0.5	0.6	1.0	1.3	2.7	3.1	4.7	5.6	8.4	12.1	25.3	36.8
SHARE	5.5	3.4	5.2	5.4	7.6	9.4	12.7	17.3	14.4	17.6	32.8	43.8

This table compares the age-sex distribution of people who reported at least one ADL between ISTAT 'Survey of Health' and SHARE respondents. Sources: ISTAT, 2005; SHARE 2004.

#### 6.5.4 MENTAL HEALTH

In addition to physical health and physical functioning, the SHARE and ELSA questionnaires also included a module focusing on mental health. The two research projects, though, used two different instruments to measure symptomatic mental illness. ELSA included an abbreviated eight-item version of the Center for Epidemiologic Studies Depression Scale (CES-D) (Radloff, 1977) whereas SHARE used the EURO- $D^{51}$  12 item scale that was developed and validated by the EURODEP Concerted Action Programme (Prince et al., 1999). Both asked respondents whether they had experienced any depressive symptoms, such as restless sleep, being unhappy and so on in the week (ELSA) or month (SHARE) prior to interview. Those who reported four or more symptoms on both the CES-D and EURO-D scales were classified as being 'depressed': "those scoring above this level would be likely to be diagnosed as suffering from a depressive disorder for which therapeutic intervention would be indicated" (Steffick, 2000, Dewey and Prince, 2005).

There were clear sex differences in the rates of people who reported suffering from depression on the CES-D and the EURO-D scale. At all ages, women were more likely than men to be suffering from depressive symptoms: 14% of men reported three or more symptoms compared with 22% of women in Denmark; 20% of English men were classified as depressed against 28% of women; around one quarter of French and Italian men were depressed compared to more than 41% of French and Italian women. As regards the relationship between age and depression, patterns were not always clear. In general, rates of depression were stable in the 50-59 and 60-69 age-groups and then increased in older ages for both sexes. 63% of Italian women aged 80+ were depressed compared with 39% aged 50-59; in Denmark, though, there was an initial drop and rates got higher only in the 80+ group: for instance, 24% of Danish women aged 50-59 were depressed compared to 16% among the 60-69 aged ones and 18% among the women aged 70-79 (see Table 6.18).

<sup>&</sup>lt;sup>51</sup> CES-D is in theory also asked in SHARE in a separate drop-off questionnaire. Unfortunately, though, the general drop-off response rate is round two third, which means that almost one older person out of three interviewed did not return by mail the additional questionnaire given to them.

	symptoms, by gender, age-group and country													
Age	50	-59	60-	-69	70-	-79	80	)+	Tota	I (N)				
	M	F	Μ	F	Μ	F	M F		Male	Female				
DK	15.6	23.8	11.5	16.0	12.9	18.0	21.4	35.5	14.3 (98)	<b>22.1</b> (185)				
N	309	322	227	229	143	184	59	109	738	844				
FR	20.5	41.2	23.4	40.2	25.4	39.7	39.2	44.7	<b>24.2</b> (303)	<b>41.2</b> (640)				
N	523	584	347	408	274	362	104	207	1,248	1,561				
IT	21.7	38.7	21.8	37.2	28.1	47.1	47.4	62.9	<b>25.5</b> (271)	44.0 (569)				
Ν	353	472	416	518	277	268	68	108	1,114	1,366				
EN	18.8	25.1	19.4	23.7	19.1	31.5	25.0	34.0	<b>19.6</b> (1,012)	27.5 (1,712)				
Ν	1,912	2,244	1,599	1,774	1,164	1,386	477	771	5,152	6,175				

 Table 6.18 Percentage of respondents reporting three or more depressive symptoms, by gender, age-group and country

This table shows the percentages of people who reported three or more symptoms on the depression scale, by age and gender. Whereas data for England is based on the 8-point depression scale CES-D, data for SHARE countries is based on the 12-item EURO-D measurement. Source: ELSA, 2002; SHARE, 2004

Finally, both ELSA and SHARE measured 'Quality of life' using the CASP-19 (Hyde et al., 2003, Wiggins et al., 2004) and the shorter version CASP-12 questionnaire (von Dem Knesebeck et al., 2007) respectively. Both contained questions on four subdomains of quality of life, i.e. control, autonomy, self-realisation and pleasure. In SHARE, though, these questions were included in the drop-off questionnaire and because of the low response rates (only 54% of the SHARE respondents completed the drop-off questionnaire) and potential selection bias, quality of life was not considered here.

## 6.6 FORMAL, INFORMAL AND FAMILY INVOLVEMENT

The growing interest in the way older people engage outside paid work and contribute to the community and society at large was reflected in both ELSA and SHARE projects which included several questions about activities and social participation.

Unfortunately, though, harmonisation between the ELSA and SHARE questions on formal and informal activities as well as family-related involvement was difficult. Most of the questions addressed to respondents were not the same and when they were either temporal references or the response scales were different. Just to mention some examples, whereas SHARE referred to 'care for a sick or disabled adult', ELSA mentioned 'care for someone'. Moreover, with regards to care and help, whereas SHARE considered both help given and received in and outside the household (personal care, practical household help and help with paperwork), ELSA concentrated only on help received.

Similarly, when inquiring about the frequency of the engagement, whereas SHARE referred to the previous 4 weeks and used 3 categories of involvement ('almost daily', 'almost every week' or 'less often'), ELSA considered the year and therefore answers ranged from 'twice a month or more' to 'less than once a year'. Finally, although the activity domains and categories were similar (training courses, religious and political organisations, volunteering and so on), whereas SHARE investigated actual engagement ELSA explored membership with the consequent impossibility of comparison between questionnaires. Both questions, moreover, were not exempt from criticism: although membership is highly correlated with activity, the former measure might lead to an overestimation of actual engagement; on the other hand, since involvement could be performed on a rather irregular basis, SHARE figures -referring only to the previous month- could underestimate the prevalence of engagement in various activities. Furthermore, SHARE asked whether people had looked after their grandchildren without the presence of their parents whereas this activity was not explicitly mentioned in ELSA. Finally, whereas ELSA explored other social/leisure activities such as 'reading a daily newspaper', 'going to the cinema' or 'going on a trip', such information was not collected in SHARE. As a final remark, activities such as getting together with friends or just visiting family members to have a chat were not included in either the SHARE or ELSA questionnaires.

For the above mentioned reasons, descriptive statistics about activities other than work are presented separately for the two questionnaires. In this descriptive section all available variables are included, although a more limited range were selected for inclusion in the multivariate analysis presented in Chapter Seven.

## 6.6.1 ACTIVITIES IN SHARE

SHARE respondents were asked whether or not they were involved during the *last month* in the following activities: voluntary or charity work, care for a sick or disabled adult, help to friends and neighbours, attendance at an educational or training course, attendance at a sport, social or other kind of club, participation in a religious or political organisation. For each activity people were engaged with, participants were asked the frequency of the engagement and whether they always received adequate appreciation from others<sup>52</sup>. Nothing was asked about what kind of sport or the organization people had taken part into, the type of activity done or the kind of charity work. Moreover, it is well known that the concept of volunteering, for instance, has different culture-specific definitions and nuances which may have lead to different reporting of participation rates (Anheier and Salamon, 1999, Wilson, 2000, Kendall, 2009); similarly, certain categories such as 'social or sport club' carry different meanings in different countries (Blanchet et al., 2007), as discussed in Chapter Five.

Questions about help and care followed. First, respondents were asked whether in the previous 12 months any family member from outside the household, any friend or neighbour had given them any kind of help (personal care, practical household help or help with paperwork). Then respondents were asked whether in same timeframe (i.e. in the previous year) they had personally given any kind of help to a family member from *outside* the household, a friend or neighbour. For up to three people, respondents were also asked the frequency with which they received or gave 'help altogether'<sup>53</sup> (on a 4-

<sup>&</sup>lt;sup>52</sup> For the activities, the responses are 'yes' or 'no' whereas the response options about the appreciation statement are 'Strongly agree', 'Agree', 'Disagree' and 'Strongly disagree'.

<sup>&</sup>lt;sup>53</sup> It is therefore not possible to distinguish between 'personal care', 'practical household help' and 'help with paperwork'.

point scale -'Almost daily', 'Almost every week', 'Almost every month' or 'Less often'). Similar questions were then asked about receiving and giving help to members *in* the household: SHARE participants were asked whether there was someone living in this household whom they had helped regularly or from whom they had received help regularly (i.e. daily or almost daily for at least 3 months) with personal care only (i.e. washing, getting out of bed, or dressing). Finally, respondents who had at least one grandchild were asked whether during the last twelve months, they had looked after their grandchild(ren) without the presence of their parents and -if they did- with what frequency of grandchild care (ranging from 'almost daily' to 'less often' than monthly).

Although SHARE provided information on the frequency of commitment for specific activities, because of the different temporal frames (previous month and previous year), Tables 6.19-21 show only the percentages of people who indicated the listed activities<sup>54</sup>. The proportions of older people who had committed to various activities varied by age, sex and activity. In general, people's commitment decreased with age. For instance, 21% of Danish men and 17% of Danish women aged 50-59 volunteered compared with 7% of those aged 80+. For specific activities, a gender difference was detectable: the percentage of women looking after grandchildren or giving help (to members either inside or outside the household) was higher than men in all countries and regardless of the age-group considered. Finally, whereas the percentage of people who took part into a religious or political organisation was similar in all countries and for both sexes (with figures ranging between 3% and 6%), differences were observed for attendance at training and educational courses: around 10% of the Danish men attended a course against 1% of French and Italian male respondents. Similarly, 31% of Danish women had gone to a sport, social or other club compared with 18% of French and 4% of Italian women. In general, Danish respondents did more activities than their French and Italian counterparts, regardless of their age and sex with a few exceptions: a higher proportion of French and Italian respondents gave help to household members.

<sup>&</sup>lt;sup>54</sup> No distinctions were made between the kind of help or care given or received; similarly, the frequency of engagement was not considered in these summary tables.

	· · · · · · · · · · · · · · · · · · ·			<u> </u>	Age	·····	
	Activities	50-59	60-69	70-79	80+	Total (N)	Eligible Total N
	Voluntary or charity work	21.2	18.4	18.7	6.6	18.7 (140)	746
	Care for a sick or disabled adult	6.1	4.3	3.0	7.1	5.1 (38)	746
	Help to family/friends/neighbours	42.5	38.8	19.6	11.4	34.7 (261)	746
	Attended training course	16.8	7.0	1.4	1.3	9.8 (69)	746
	Gone to a sport/social/other club	30.6	32.1	30.2	24.5	30.5 (231)	746
	Taken part in a religious org.on	2.8	3.6	5.6	9.2	4.1 (31)	746
	Taken part in a political org.on	6.3	7.6	9.8	-	6.8 (49)	746
	Received help from outside HH*+	27.3	26.2	26.9	26.0	<b>26.8</b> (144)	552
	Given help to outside the HHold members*	60.1	47.7	43.6	17.8	<b>49.9</b> (371)	749
	Received personal help from inside the household* ^ <	6.6	4.7	10.9	14.4	<b>7.8</b> (12)	174
	Given help to inside the HHold members* @	3.5	2.9	5.2	2.6	<b>3.6</b> (19)	581
	Looked after grandchild(ren) * •	63.0	67.3	36.5	8.0	<b>51.8</b> (237)	461
Mei	TOTAL	312	228	145	64		
	Voluntary or charity work	16.6	18.7	19.9	7.4	<b>16.5</b> (144)	848
	Care for a sick or disabled adult	8.2	6.1	3.0	2.9	<b>5.9</b> (47)	848
	Help to family/friends/neighbours	47.4	27.6	19.6	5.5	30.5 (266)	848
	Attended training course	20.8	4.7	8.2	0.8	11.0 (87)	848
	Gone to a sport/social/other club	32.0	36.5	31.4	18.9	31.3 (271)	848
	Taken part in a religious org.on	5.0	4.2	8.4	5.8	<b>5.6</b> (50)	848
	Taken part in a political org.on	5.1	1.5	4.0	0.8	3.3 (28)	848
	Received help from outside HH*+	35.5	29.2	39.1	52.1	<b>37.7</b> (233)	615
	Given help to outside the HHold members*	59.3	46.6	34.1	10.5	<b>43.9</b> (378)	853
	Received personal help from inside the household* $^{\prime}$	10.4	2.3	1.7	21.4	<b>7.0</b> (18)	210
en	Given help to inside the HHold members* @	3.0	4.5	3.1	22.8	<b>4.5</b> (25)	523
<b>M</b>	Looked after grandchild(ren) * •	79.8	81.4	33.3	2.5	<b>58.1</b> (354)	604
Ň	TOTAL	328	228	186	111		

Table 6.19 Involvement in other activities, by age-group and sex –Denmark

This table shows the percentage of Danish older people who did the listed activities by age-group and gender. Each cell then indicates the percentage of either men or women in a specific age-group who did a specific activity. Notes: \* indicates an activity done in the last twelve months; + this question is only asked to the family respondent and not to all eligible household members; ^ indicates an activity asked only to participants living in a household with more than 1 person; ✓ indicates that the question was only asked to people living with someone and who reported no difficulty with mobility; • indicates that the question was only asked to respondents who had at least one living grandchild. Source: SHARE, 2004; Bases vary; the age-group totals shown are for all SHARE sample members for the highest row total.

Age							
	Activities	50-59	60-69	70-79	80+	Total (N)	Eligible Total N
	Voluntary or charity work	16.7	25.3	10.9	9.4	17.1 (220)	1,273
	Care for a sick or disabled adult	5.0	4.9	5.1	1.8	4.7 (59)	1,273
ĺ	Help to family/friends/neighbours	31.9	27.9	18.7	10.4	<b>25.8</b> (341)	1,273
	Attended training course	5.4	4.7	1.0	-	0.4 (50)	1,273
	Gone to a sport/social/other club	16.6	24.0	22.3	10.5	19.3 (240)	1,273
ł	Taken part in a religious org.on	3.8	6.0	5.2	10.4	5.4 (66)	1,273
	Taken part in a political org.on	8.1	4.2	2.5	-	5.0 (63)	1,273
	Received help from outside HH*+	7.2	10.8	15.1	38.1	13.3 (123)	941
Mei	Given help to outside the HHold members*	34.1	31.1	21.3	7.9	<b>27.8</b> (373)	1,306
	Received personal help from inside the household* $^{\prime}$	10.9	10.3	16.4	23.6	<b>14.5</b> (55)	405
	Given help to inside the HHold members* @	3.4	5.5	6.4	6.6	<b>4.9</b> (55)	1,107
	Looked after grandchild(ren) * •	54.2	60.5	40.1	16.8	<b>47.3</b> (359)	751
	TOTAL	549	356	289	112		
<u> </u>	Voluntary or charity work	12.3	16.0	11.6	6.7	12.2 (198)	1,595
1	Care for a sick or disabled adult	8.7	9.0	8.1	4.4	7.9 (128)	1,595
	Help to family/friends/neighbours	31.6	30.2	19.3	6.6	24.1 (409)	1,595
	Attended training course	9.0	2.4	1.6	0.9	4.2 (72)	1,595
	Gone to a sport/social/other club	16.8	18.0	19.7	17.0	17.9 (279)	1,595
	Taken part in a religious org.on	4.1	5.9	6.6	6.2	5.5 (87)	1,595
	Taken part in a political org.on	4.2	1.9	0.5	-	2.2 (35)	1,595
ner	Received help from outside HH*+	13.6	15.1	28.7	60.2	27.5 (293)	1,111
Won	Given help to outside the HHold members*	38.2	33.6	22.2	9.5	<b>28.4</b> (477)	1,621
	Received personal help from inside the household* ^ <	3.5	5.8	12.5	35.1	10.7 (59)	589
	Given help to inside the HHold members* @	4.7	6.1	11.9	14.3	7.4 (81)	1,097
	Looked after grandchild(ren) * •	73.0	69.6	36.3	8.4	49.8 (546)	1,046
	TOTAL	602	420	376	223		

Table 6.20 Involvement in other activities, by age-group and sex -France

This table shows the percentage of French older people who did the listed activities by age-group and gender. Each cell then indicates the percentage of either men or women in a specific age-group who did a specific activity. Notes: \* indicates an activity done in the last twelve months; + this question is only asked to the family respondent and not to all eligible household members; ^ indicates an activity asked only to participants living in a household with more than 1 person; ✓ indicates that the question was only asked to people living with someone and who reported no difficulty with mobility; • indicates that the question was only asked to respondents who had at least one living grandchild. Source: SHARE, 2004; Bases vary; the age-group totals shown are for all SHARE sample members for the highest row total.

			/ <b>•</b>		Age		
	Activities	50-59	60-69	70-79	80+	Total (N)	Eligible Total N
	Voluntary or charity work	7.2	9.4	6.2	2.5	7.3 (79)	1,113
	Care for a sick or disabled adult	2.6	2.5	-	1.0	1.8 (24)	1,113
	Help to family/friends/neighbours	14.4	11.6	4.6	5.3	10.5 (114)	1,113
	Attended training course	0.9	0.4	0.3	-	0.5(7)	1,113
	Gone to a sport/social/other club	7.8	12.5	6.5	2.4	8.5 (94)	1,113
	Taken part in a religious org.on	4.3	3.9	1.5	1.4	<b>3.4</b> (35)	1,113
	Taken part in a political org.on	4.2	2.3	2.8	-	<b>2.9</b> (33)	1,113
	Received help from outside HH++	16.3	12.9	8.4	40.0	15.7 (100)	775
	Given help to outside the HHold members*	30.5	23.1	12.6	14.7	<b>22.7</b> (238)	1,115
	Received personal help from inside the household* ^ <	13.9	5.7	12.4	31.5	<b>12.9</b> (52)	432
	Given help to inside the HHold members* @	6.5	6.7	6.9	9.8	<b>6.9</b> (76)	1,024
_	Looked after grandchild(ren) * •	30.0	43.6	32.58	14.73	<b>33.0</b> (190)	558
Men	TOTAL	351	417	278	69		
	Voluntary or charity work	7.8	9.8	4.4	0.7	6.5 (89)	1,372
ł	Care for a sick or disabled adult	7.4	2.7	2.1	-	3.6 (54)	1,372
	Help to family/friends/neighbours	21.8	15.9	9.2	1.5	14.0 (208)	1,372
	Attended training course	3.3	0.9	0.6	-	1.5 (19)	1,372
	Gone to a sport/social/other club	6.2	4.8	2.9	-	4.1 (50)	1,372
Women	Taken part in a religious org.on	7.0	6.7	7.8	0.6	6.2 (88)	1,372
	Taken part in a political org.on	3.6	0.7	0.6	1.2	1.6 (18)	1,372
	Received help from outside HH++	13.5	14.9	25.9	41.5	<b>22.1</b> (181)	1,037
	Given help to outside the HHold members*	34.3	27.3	17.1	3.5	<b>23.4</b> (344)	1,375
	Received personal help from inside the household* $\wedge \checkmark$	7.7	7.9	19.0	44.4	15.8 (93)	665
	Given help to inside the HHold members* @	9.0	8.3	10.6	17.6	<b>9.9</b> (125)	1,167
	Looked after grandchild(ren) * •	58.3	61.8	33.4	6.9	41.9 (372)	801
	TOTAL	473	520	273	109		

Table 6.21 Involvement in other activities, by age-group and sex -Italy

This table shows the percentage of Italian older people who did the listed activities by age-group and gender. Each cell then indicates the percentage of either men or women in a specific age-group who did a specific activity. Notes: \* indicates an activity done in the last twelve months; + this question is only asked to the family respondent and not to all eligible household members; ^ indicates an activity asked only to participants living in a household with more than 1 person; ✓ indicates that the question was only asked to people living with someone and who reported no difficulty with mobility; • indicates that the question was only asked to respondents who had at least one living grandchild. Source: SHARE, 2004; Bases vary; the age-group totals shown are for all SHARE sample members for the highest row total.

## 6.6.2 ACTIVITIES IN ELSA

In the 'Work and Pension' module, respondents were initially asked whether in the month prior to the interview they did voluntary work, cared for anyone<sup>55</sup> and attended a formal educational or training course. In the self-completion section of the interview, respondents were asked whether they were members of any of the following organisations: political party, trade union or environmental groups; tenants' groups or residents' groups; neighbourhood watch; religious groups; charitable associations; education, art or music groups or evening classes; social club; sports clubs, gym or exercise classes; or any other. In the same section, they were asked whether they were involved social and civic activities<sup>56</sup>.

Overall, women younger than 80 were more likely than men to report volunteering, caring for a disable, looking after someone and attending training courses. The younger the age, the bigger was the difference between men and women: for instance, 26% of English women aged 50-59 looked after someone compared with 15% of men; among 70-79 year old people, 17% looked after someone against 16% of men. The proportion of older people who were members of an organisation varied by age, sex and type of organisation. For membership of a church or religious organisations, for instance, there was an increasing trend by age for both sexes. See Table 6.22 for full details.

Participation in cultural activities declined dramatically as age increased. For instance, whereas 42% of men and 45% of women aged 50-59 went to the cinema more than once in a year, among those aged 80+ percentages went down to 7% and 10% respectively. Clear declines in participation were observable also for having hobbies, taking holidays and outings. An increase in participation was, on the contrary, evident when rates of people who voted and read a daily newspaper were considered (see Table 6.A9).

<sup>&</sup>lt;sup>55</sup> Respondents were asked whether they have looked after anyone in the *week* prior the interview, including partners or other people in your household; if so, respondents were then asked the relationship to the recipient of care as well as the number of hours of commitment.

<sup>&</sup>lt;sup>56</sup> Respondents were asked whether they voted in the last general election, read a daily newspaper, had a hobby, had been on a holiday in the UK in the last 12 months, had been on a holiday abroad in the last 12 months, had been on a day trip in the last 12 months, used email or the Internet, and had a mobile phone. Finally, ELSA respondents were asked how often -if at all - they went to the cinema, ate outside the house, went to a museum or art gallery and went to the theatre or opera. For each activity, respondents said whether they went more than twice a month, once or twice a month, a couple of times a year, once a year, less than once a year, or never.

Δσε							
	Activities	50-59	60-69	70-79	80+	Total (N)	Eligible Total N
	Voluntary or charity work+	8.6	11.4	12.7	6.9	10.1 (543)	5,186
	Care for sick/disabled adult+	8.8	7.3	7.2	4.7	7.6 (396)	5,186
	Attended training course+	7.5	3.6	1.9	1.0	4.6 (238)	5,186
	Look after anyone (past week)	15.0	16.3	15.6	12.8	15.2 (778)	5,104
	Member of political party*	23.5	16.3	14.0	15.9	<b>18.8</b> (830)	4,530
	Member of tenants group*	14.7	20.2	22.6	19.7	18.4 (852)	4,530
	Member of religious group*	11.5	16.2	21.5	22.8	15.9 (735)	4,530
	Member of charitable ass.on*	15.3	15.6	15.8	14.0	15.4 (706)	4,530
	Member of educational classes*	9.1	10.7	9.5	4.4	<b>9.3</b> (435)	4,530
	Member of social club*	22.2	26.3	26.0	23.1	<b>24.3</b> (1,111)	4,530
	Member of sport club*	22.3	18.5	14.3	9.2	18.5 (833)	4,530
	Member of other clubs*	29.1	26.7	24.1	21.4	<b>26.7</b> (1,226)	4,530
Men	TOTAL	1,916	1,611	1,174	485		
<u> </u>	Voluntary or charity work+	11.0	16.6	13.0	5.8	12.3 (794)	6,205
	Care for sick/ disabled adult+	14.6	12.5	7.8	3.0	10.8 (682)	6,205
	Attended training course+	11.4	4.7	2.0	0.8	5.9 (385)	6,205
1	Look after anyone (past week)	26.3	25.9	16.6	8.5	21.4 (1,325)	6,118
	Member of political party*	15.2	8.8	9.5	9.1	11.4 (613)	5,340
omen	Member of tenants group*	15.6	20.8	19.4	16.8	<b>18.1</b> (985)	5,340
	Member of religious group*	17.7	27.2	29.9	35.9	<b>25.3</b> (1,334)	5,340
	Member of charitable ass.on*	19.4	20.9	17.9	19.0	<b>19.5</b> (1,055)	5,340
	Member of educational classes*	17.3	18.7	13.8	8.5	15.9 (882)	5,340
	Member of social club*	15.2	20.1	19.5	22.4	18.4 (977)	5,340
	Member of sport club*	24.6	22.4	12.2	5.3	<b>18.9</b> (1,030)	5,340
	Member of other clubs*	19.8	22.4	19.7	17.6	20.3 (1,109)	5,340
Ň	TOTAL	2,250	1,787	1,391	777		

Table 6.22 Activities and Organisational memberships, by age-group and sex – England

This table shows the percentage of English older people who did the listed activities by age-group and gender. Each cell then indicates the percentage of either men or women in a specific age-group who did a specific activity. Notes: Memberships (\*) have no temporal references; + indicates activities in the prior month whereas 'look after someone' refers to the week before. Source: ELSA, 2002.

## 6.7 CONCLUSION

Findings showed expected variations by age and sex in most of the variables considered. For instance, in all countries, men reported better health and were less likely to report at least three depressive symptoms than women. Similarly, the proportion of older people living alone was higher among women, particularly in older age groups. Moreover, the proportion of respondents reporting functional limitations and at least two chronic diseases increased with age in all subpopulations under study. Findings also pointed to large differences between countries, particularly when variables such as education, paid work and self-rated health were considered. Just as a few examples, Danish respondents –both men and women –had higher levels of education compared to French, Italian and English respondents. Italian and French people exited the labour force at earlier ages than in Denmark and England; also, the proportion of Italian women in paid work was much smaller than that of female respondents in the other countries. Similarly, the proportion of Danish men and women who reported their health as excellent was respectively three and four times higher than in France and Italy.

Initial low response rates have had some negative effect on the representativeness of the datasets; using calibration weights provided by the ELSA and SHARE teams partly compensates for nonresponse bias. Both surveys were no exception to the common pattern of non-response: younger respondents (particularly men and people in paid work) were under-represented. The few health-related indicators considered also suggested that people 'in good health' may have been over-represented. Data was not fully representative of the countries and therefore the generalization of the findings might not hold had the non-respondents also been included in the sample. Moreover, given that this research study investigates the relationships between health and engagement, non-response bias is a potential limitation of this study to be considered throughout the analysis and the interpretations of the findings. Additionally, France, Italy and England included only respondents in private households. Given that the proportions of people resident in communal establishments such as residential and nursing homes are higher in England and France than Italy, this might bias statistics particularly among older women. However, this source of bias is likely only to be

important in the oldest age-groups (80+) as proportions in younger age groups in institutions are very low (Coleman, 2003).

It is also important to bear in mind that whereas for most health-related, socio-economic and demographic variables the two questionnaires are fully comparable, when analysing activities and participation, unfortunately, ELSA and SHARE asked the very same set of questions only about paid work. Participation in other activities was, on the other hand, not fully comparable and therefore, given the different questions and timeframes they refer to, some practical compromises were necessary when setting the stage for the analyses as described in the following chapter.

# 7. HOUSEHOLD AND PERSONAL CHARACTERISTICS OF OLDER PEOPLE WHO ENGAGE IN WORK, FORMAL, INFORMAL AND FAMILY ACTIVITIES

This chapter aims to explore the personal and household characteristics which are associated with engagement of older people in each of the four different countries under study – Denmark, France, Italy and England – in each of the three main activities considered (specifically work, formal activities, informal and family-related activities). Cross-sectional analyses are presented separately for each activity and country. Finally, only people aged 50-69 with full records are considered in the following analyses because participation rates are particularly low among people older than 70, especially when paid work is considered (see 6.2.1 and 6.5 for more details on participation rates).

## 7.1 ACTIVITIES AND THEIR DETERMINANTS

Given that the holistic definition of 'active ageing' includes continuing participation in a number of different domains (Hutchison et al., 2006) which go beyond 'employment and participation in economically productive activities' to also incorporate 'community participation' and other meaningful activities common among older people which contribute to their personal well-being and quality of life, 3 major areas of involvement were considered: *labour market participation, participation in formal activities* and *participation in informal and family-related activities*.

Many factors can contribute to the decision to engage in the various forms of activities, including personal health situation, family responsibilities, the interaction of choices within couples, the satisfaction gained from such engagement or working conditions as well as the availability of social security and early retirement pathways (Gollac and Volkoff, 2000, Debrand and Lengagne, 2007, Künemund and Kolland, 2007), as described in the following sections.

26

#### 7.1.1 LABOUR MARKET PARTICIPATION

Participation in the labour market may depend on a number of different household and personal factors including respondents' health characteristics. Household and family factors are relevant as family responsibilities (such as having parents, partners or other kin members in need of assistance) could lead people to reduce paid work, especially where family-oriented norms and traditions are strong and social services are lacking or are limited. Participation in the workforce, then, could also be affected by the extent to which older people have to supply informal care to their elderly parent(s), particularly among women (Dentinger and Clarkberg, 2002a, Pavalko and Henderson, 2006, Bolin et al., 2008). Moreover, married couples may decide on retirement/work decisions based on each other's health (Gotlibovski and Weiss, 2008) and work status (Coile, 2004). For instance, Gotlibovski and Weiss (2008) found that when facing health problems, the members of the couple tend to react in different ways: "the husband reacts to his wife's illness by raising his work and the wife reacts to her husband's sickness by reducing her work" (pg. 81). Similarly, kin obligations to other family members, such as children living at home, may be associated with parents' labour participation (Cohen and Bianchi, 1999, Szinovacz et al., 2001, Raymo and Sweeney, 2006a, b), although with differences across countries (Manacorda and Moretti, 2006, Becker et al., 2010).

Personal socio-demographic and economic characteristics are also hypothesised to be associated with working status. For example, age affects pension eligibility. Similarly, education has an impact on age of entry to the labour market and type of job and is therefore expected to be associated with working status; people with higher education are more likely to have better quality jobs and are therefore more likely to remain engaged in paid work longer (Gollac and Volkoff, 2000, Blanchet and Debrand, 2007, Debrand and Lengagne, 2007, Siegrist et al., 2007). In the cohorts under study, education level itself was influenced by gender which in turn may have had an impact on work status because of the identification of women as the main family caretakers (and men as bread-winners) (Szinovacz et al., 2001, Dentinger and Clarkberg, 2002b).

Finally, health characteristics may be associated with participation in the labour market (Barnay and Debrand, 2006, Siegrist et al., 2007): for instance, people who are

depressed might be more likely to be out of work; similarly, people who experience difficulties in every-day activities or mobility are hypothesised to be less likely to be formally employed, both because these characteristics make it harder to find and retain employment and possibly because unemployment may contribute to depression or other forms of ill health (Bartley, 1994, Gallo et al., 2000, Thomas et al., 2005, Alavinia and Burdorf, 2008).

#### 7.1.2 FORMAL, INFORMAL AND FAMILY PARTICIPATION

A number of authors stressed the importance of societal context, governmental policies as well as values and expectations (Mendes de Leon et al., 2001, Salamon and Sokolowski, 2001, Erlinghagen and Hank, 2006, Hank, 2010, Hank and Erlinghagen, 2010) in shaping both informal and formal participation. It is believed that older people's level of commitment depends on the societal values people attach to such experiences and the personal benefits they gain from it (Van Willigen, 2000). Informal and formal activities are culture specific (Zunzunegui et al., 2004) but also vary within country by socio-economic position, gender and physical functioning (Unger et al., 1999, Grundy and Holt, 2001, Grundy and Sloggett, 2003). In particular, household characteristics are expected to be associated with both informal and family activities: it is believed that older people living with other family members, having at least one child still living in the household as well as having living parents (particularly if in bad health) are more likely to end up helping or caring for other kin members in and outside the household. Similarly, the area of residence of respondents could impact on their participation. For example, living in rural areas or small settlements may be associated with greater informal participation but fewer opportunities for engagement in formal activities (Smith, 1994). Also demographic characteristics such as age and sex might be associated with formal, informal and family participation. It is hypothesised that the gender of older people (and the responsibilities and obligations attached to it) could influence informal participation over the formal one, especially in those countries where care is considered a family obligation and responsibility (Herzog and Morgan, 1992, Dentinger and Clarkberg, 2002b, Raymo and Sweeney, 2006a, Burr et al., 2007, Kohli et al., 2009). Finally, health conditions such as depression and functional limitation could be associated with lower levels of participation in the above mentioned activities (Hao, 2008).

# 7.1.3 ACCUMULATION OR COMPETITION BETWEEN PARTICIPATION IN FORMAL OR INFORMAL ACTIVITIES AND EMPLOYMENT STATUS

It is hypothesised that people less involved in formal employment may be more active in other spheres of participation since they have more time to dedicate to them. However, it could also be argued that family and informal involvement might compete with more formal participation: retired people (who have therefore been formally involved in the society at some point in their life) might for example be more likely to engage in formal activities rather than informal or family ones whereas those who never formally engaged might be more likely to engage in informal and family-related activities (Hank and Stuck, 2008, Kohli et al., 2009). Older people who retire can take up a new activity in order to compensate for the loss of previous active roles or give up activities due to time or physical constraints. Family responsibilities may constrain women's involvement in nonfamily roles such as paid or charity work, though some studies suggested that particularly women appear to add family responsibilities without cutting back on work (Tomassini et al., 2004b, Glaser et al., 2005). Older people can in fact opt for multiple complementary activities performed in parallel (Van Willigen, 2000, Hank and Stuck, 2008).

# 7.2 METHODS

## 7.2.1 SAMPLE

Results in this chapter are based on cross sectional analyses of data from the 2004 Survey of Health, Ageing, and Retirement in Europe (SHARE) for Denmark, France and Italy and from the 2002 baseline wave of the English Longitudinal Study of Ageing (ELSA) for England. Analyses were restricted to respondents aged 50-69 who had complete cases, with no missing values in any of the variables considered in the multivariate analyses. Differences in the number of observations considered, where observable, were due to item non-response.

### 7.2.2 MEASURES

#### **OUTCOME MEASUREMENTS AND DESCRIPTIVE TABULATIONS**

Work – people were classified as workers if they described their current situation as "Employed or self-employed (including working for family business)" and if they were not "temporarily away from any work, including seasonal work", as it was thought most appropriate to include only those with a current commitment to labour market activity.

Formal social involvement - This was conceptualised and defined to include non-kin social activities linked to some kind of formalised association or group and formally organised, performed within an established structure with a regular and routine schedule. Formal social involvement thus included organised voluntary work; attendance at training courses, and participation in political organisations, religious organisations or sport, social or other kind of clubs. Although most studies tend to focus on volunteering, it was decided to be as inclusive and holistic as possible and therefore to consider a wider range of the activities which older can theoretically enjoy. Grouping such a variety of different activities together was possible since no underlying model or 'hierarchy of activities' is explicit and no a priori assumptions were made about the correlation between activities: It was therefore assumed that all activities have the same weight and importance. In SHARE countries, the formal social involvement variable is coded 1 if the respondent reported active participation in at least one of the abovementioned activities almost every week or more often during the month prior to the interview<sup>57</sup>. For ELSA respondents, only those who volunteered or attended a training course in the month prior to the interview are coded as 1<sup>58</sup>. In this case, the frequency of

<sup>57</sup> SHARE questionnaire specifically asks respondents "Have you done any of these activities in the last month? 1. Done voluntary or charity work; 4. Attended an educational or training course; 5. Gone to a sport, social or other kind of club; 6. Taken part in a religious organization (church, synagogue, mosque etc.); 7. Taken part in a political or community-related organization". For each of the activities, in case of an affirmative answer, the respondents are then asked "How often in the last four weeks did/have you /Do voluntary or charity work/ Attended an educational or training course/ Go to a sport, social or other kind of club/ Taken part in a religious organization (church, synagogue, mosque etc.)/ Taken part in a political or community-related organization (church, synagogue, mosque etc.)/ Taken part in a political or community-related organization? 1. Almost daily; 2. Almost every week; 3. Less often.

<sup>58</sup> ELSA respondents are asked in the Work and Pensions Module "Did you do any of these activities during the last month, that is since [date a month ago]? 3. Voluntary work; 6. Attended a formal educational or training course".
engagement was not considered<sup>59</sup> because the corresponding questions did not consider any specific temporal frame; it was also decided to not define as 'active' people who were just members of charitable, religious or social organisations because no information was available on frequency of participation in activities and membership alone does not necessarily indicate active participation. Since among the major scope of this research project is the assessment of whether more engaged people have better health to start with and whether they experience lower health declines over time, it seems reasonable to only consider people who are actively involved on a regular basis. Active engagement is meant as both an opportunity as well as a commitment which demands time and energy and helps people structure their day (Deeming, 2009).

<u>Informal and family-related participation</u> – this indicator represents engagement in activities which involve family members and are done in a more casual and less structured way. In SHARE, these activities include providing care for sick or disabled adults, help to family, friends or neighbours and others from outside the household; help to other household members, and looking after grandchildren without the presence of their parents. The informal and family-related participation indicator is coded 1 if the SHARE respondent reported that they did at least one of the above-listed activities almost every week or more often<sup>60</sup>. Introducing the *at least weekly* frequency allows comparisons with formal social activities, otherwise problematic since most of the

<sup>59</sup> In ELSA questionnaire, respondents are asked "how often -if at all- do you do any voluntary work?" but there is no specific reference to any temporal frame: declaring to volunteer more than twice per month does not necessarily entail that the specific month prior to the interview the respondent volunteered that often.

<sup>60</sup> To construct the indicator of informal and family engagement, a series of questions are considered. In particular, the following are taken into account: "Have you done any of these activities in the last month? 2. Cared for a sick or disabled adult; 3. Provided help to family, friends or neighbors". If so, was it "1. Almost daily; 2. Almost every week or 3. Less often". In the Social participation module, respondents are also asked "In the last twelve months, have you personally given any kind of help (i.e. personal care, practical household help, help with paperwork) to a family member from outside the household, a friend or neighbour?" and in case of an affirmative answer they are then asked "In the last twelve months, how often altogether have you given such help? Was it... 1. Almost daily 2. Almost every week 3. Almost every month 4. Less often". SHARE questionnaire also specifically asks "During the last twelve months, have you regularly or occasionally looked after [your grandchild/your grandchildren] without the presence of the parents?" If so, "On average, how often did you look after the child(ren) of [child name] in the last twelve months? Was it 1. Almost daily; 2. Almost every week; 3. Almost every month; 4. Less often. Finally, respondents are asked: "Is there someone living in this household whom you have helped regularly (i.e. daily or almost daily) during the last twelve months with personal care, such as washing, getting out of bed, or dressing?"

family-related activities refer to the year preceding the interview (rather than the month). For ELSA respondents, it was decided to only include those who reported that they cared for someone sick or disabled in the last month and those who cared for anyone in the past week, despite the discrepancy in time frame<sup>61</sup>.

### **Descriptive tabulations: Paid Work**

Bearing in mind the different definitions of formal and informal indicators, Tables 7.1-7.3 show the percentage of people who were actively involved in the above-described activities by age and sex. Table 7.1 shows the percentage of people who describe themselves as workers and shows large variations by country, gender and age group. Half of the female Danish sample were working compared with only 21% of Italian women. Similarly, almost 60% of the Danish men work compared to 54% of English men, 47% of French men and 40% of Italian male respondents. Differences between countries for both men and women are even more striking as age increases: whereas more than 80% of French, English and Danish men aged 50-54 work, less than 10% of French men aged 60-69 work compared to more than 31% of Danish male respondents and more than a quarter of English men (see Table 7.1 for more details). Differences between countries are evident among women also at younger ages: two third of Danish women aged 55-59 work compared to 56% in England and France and one quarter of the Italian female population.

### **Formal Activities**

Table 7.2 shows the percentage of people who were involved in formal activities. Differences between countries are noticeable (for instance, in all age groups Danish respondents consistently had participation rates between two and three times higher than those of the Italians). Differences by age were minor, although rates are slightly higher in the older age groups among SHARE respondents. In all countries considered, there was also little variation in formal participation by gender, though English women seem to report higher rates of formal involvement compared to men in all age groups. These

<sup>61</sup> ELSA family and informal engagement was assessed considering the following questions: "Did you do any of these activities during the last month? 4 Cared for a sick or disabled adult"; and "Did you look after (i.e. actively provided care for) anyone in the past week (including your partner or other people in your household)?"

differences between European countries confirm that civic participation is highly dependent on country-specific societal context, the available opportunities as well as values and expectations of what is socially appropriate (Salamon and Sokolowski, 2001, Erlinghagen and Hank, 2006).

### Informal and Family-related activities

Table 7.3 shows the percentage of people who were involved in informal and familyrelated activities by age group, gender and country. In all countries, women tend to be more involved than men in informal and family-related activities in all age-groups considered, though the gap between genders is not similar in all countries: for instance, in Italy only 29% of men were involved in family and informal activities compared with more than 47% of women whereas in Denmark 37% of men and 44% of women are informally engaged. The pattern of involvement by age varies by gender; whereas among men this seems to be u-shaped, with a decrease of informal involvement in the 55-59 age-group, among women the trend seems more linear and stable.

### Accumulation of activities

Table 7.4 shows the distribution of the respondents engaged in all the possible combinations of activities by gender and country. Interestingly, in all countries under study between 17% (Denmark) and 35% (Italy) of the respondents were not engaged in any of the above-mentioned activities; such category represents the highest proportion of respondents among Italian respondents, French and English women. On the opposite extreme, only between 7% (Denmark) and 2% (Italy and England) of the samples reported to be engaged simultaneously in paid work, formal and informal activities. Excluding the two extreme cases, the majority of the samples were engaged mainly with paid work only, with smaller proportions of respondents who combined work with any other activities. At the same time, the percentages of people engaged only in either formal or informal activities were generally higher than the proportion of respondents who combined either of them with work. For instance, 19% of the French women were engaged only in informal activities whereas 13% were engaged in both work and informal activities. A few exceptions to these trends were observed: among Danish respondents, the percentages of workers who also participated in either formal or

informal activities were higher than those only formally or informally engaged. Moreover, among French and Italian women, almost 20% and 33% of respondents were engaged only in informal activities, category which represented the second highest after 'no engagement'.

			Dy age	e-group,	sex and	country	·	
	50	)-54	55	-59	60	-69	Tota	l (N)
	M	F	M	F	M	F	Male	Female
Denmark	80.8	80.8	72.6	67.6	31.3	16.5	<b>58.4</b> (314)	49.5 (276)
N	169	152	144	176	228	229	541	557
France	83.6	68.3	62.0	56.0	9.7	10.4	<b>46.8</b> (425)	<b>40.2</b> (415)
N	274	289	276	314	360	421	910	1,024
Italy	76.6	47.9	49.4	26.5	16.0	5.7	<b>39.9</b> (270)	<b>21.2</b> (177)
N	126	199	226	274	419	522	771	995
England	80.7	70.4	69.0	55.7	26.8	15.4	54.5 (1,847)	42.4 (1,667)
N	896	1,085	1020,	1,165	1,611	1,786	3,527	4,036

Table 7.1 Percentage of people aged 50-69 in paid work, by age-group, sex and country

Sources: Denmark, France and Italy data obtained from SHARE; English data obtained from ELSA. Data refer to 2004 and 2002 respectively. Own calculations

 

 Table 7.2 Percentage of people aged 50-69 involved in formal activities, by age-group, sex and country

	50	-54	55-	-59	60	-69	Tota	al (N)
	M	F	M	F	Μ	F	Male	Female
Denmark	36.4	34.2	34.8	35.7	39.0	41.1	37.1 (200)	<b>37.6</b> (207)
N	167	151	143	175	228	228	538	554
France	23.8	22.5	26.8	25.5	34.9	24.0	29.3 (251)	<b>24.1</b> (243)
N	267	283	268	308	349	416	884	1,007
Italy	12.5	13.8	8.00	14.1	15.9	14.1	12.5 (100)	14.0 (124)
N	125	199	226	273	417	521	768	<i>993</i>
England	15.3	18.5	13.2	19.8	14.2	18.9	14.1 (503)	19.1 (786)
N	896	1,085	1,020	1,165	1,611	1,786	3,527	4,036

Source: Denmark, France and Italy data obtained from SHARE; English data obtained from ELSA. Data refer to 2004 and 2002 respectively. Own calculations

		acuv	ities, by	age-git	up, sex	anu cou	шиу	
	50	)-54	55-	-59	60	-69	Total (N)	
	Μ	F	Μ	F	M	F	Male	Female
Denmark	39.6	43.1	32.1	41.3	37.6	45.1	36.8 (200)	43.4 (241)
N	167	152	144	176	228	228	539	556
France	36.0	37.1	27.8	50.5	35.1	46.9	33.1(302)	45.3 (460)
N	270	284	269	310	351	417	890	1,011
Italy	29.1	40.9	22.3	50.7	33.3	47.8	<b>28.9</b> (236)	47.1 (483)
N	125	199	225	273	417	520	767	992
England	17.8	28.4	16.3	31.0	17.7	28.3	17.3 (613)	<b>29.1</b> (1,160)
N	896	1,085	1,020	1,165	1,611	1,786	3,527	4,036

Table 7.3 Percentage of people aged 50-69 involved in informal and family-related activities, by age-group, sex and country

Source: Denmark, France and Italy data obtained from SHARE; English data obtained from ELSA. Data refer to 2004 and 2002 respectively. Own calculations

 Table 7.4 Percentage distribution of the engagement in the combination of the three forms of activities, by gender and country

	Denn	nark	Fra	nce	Ita	ly	Eng	land
	M	F	M	F	М	F	М	F
No Activities	16.9	17.7	24.7	25.7	37.2	35.1	32.4	32.9
Only paid work	23.6	20.7	25.8	18.0	26.0	11.4	39.2	25.4
Paid Work & Formal Engagement	13.9	9.4	6.9	4.9	3.1	1.6	5.4	5.5
Paid Work & Informal Engagement	14.4	12.4	9.2	12.9	9.2	6.4	6.3	8.0
Only Formal Engagement	8.8	8.8	9.4	6.1	4.7	4.8	5.6	7.5
Only Informal Engagement	8.0	11.6	11.1	19.3	14.9	33.0	7.8	14.3
Formal & Informal Engagement	7.9	12.4	8.0	8.6	3.2	5.8	1.9	4.0
Paid Work & Formal & Informal	6.5	7.0	4.9	4.5	1.7	1.9	1.4	2.4

Source: Denmark, France and Italy data obtained from SHARE; English data obtained from ELSA. Data refer to 2004 and 2002 respectively. Own calculations

The differences apparent in these tables were investigated further using multivariate logistic regressions in which associations between the outcome measures and a range of socio-demographic and health indicators were analysed.

### **CO-VARIATES**

<u>Demographic characteristics</u>. — Age was grouped into either three (50-54; 55-59; 60-69) or four (50-54; 55-59; 60-64; 65-69) categories depending on the activity considered. In order to facilitate comparisons between countries and subpopulations it was decided not to further split the 60-69 age-group when labour market participation was considered: in this case, in fact, the proportion of people working after 60 is small

in some subpopulations and almost equal to zero after the age of 65. Although retirement policies determine official pension ages (see Chapter 3), 60 is the median effective age at which older European male and female workers withdraw from the labour force (Romans, 2007). As regards education, qualifications were re-coded to three categories representing low, mid and high education using the ISCED classification (UNESCO - UIS, 2006). The presence of children in the household was dichotomised (y/n) as 1 if at least one child or child-in-law (regardless of their age) still lived in the household and 0 otherwise. Marital status was dichotomised into living with a spouse/partner or not. In the analysis of participation in informal and family-related activities, other variables were derived indicating whether the respondent lived alone, whether their parents were still alive, whether if alive they were in good health or not. Additionally, a categorical variable indicating the health status of co-residents aged over 50 was constructed distinguishing those who lived with other older members who rated their health as good or better and those who rated it as less than  $good^{62}$ . Finally, the area of residence was dichotomised into 'urban' and 'rural'<sup>63</sup> on the basis of respondents' reports.

<u>Economic Characteristics</u>. — A dummy variable indicating 'mortgage free house ownership' was included as it was hypothesised that people who have to pay monthly a rent or a mortgage might be more likely to work. It was decided not to use equivalised income available to the household since between 77% and 88% of the household income originates from pension and work earnings, both highly correlated with employment status. Also, in order to study the interaction between the various forms of engagement, a series of dummy variables indicating whether the respondent was actively involved in

 $<sup>^{62}</sup>$  This variable includes the following categories: 'the respondent lives with another 50+ adult or a younger partner whose health was self-assessed as more than good', 'the respondent lives with another 50+ adult or a younger partner who self-assessed their health as less than good', 'the respondent lives alone', 'the respondent is the only 50+ adult in the household', and 'the respondent lives with a younger partner or other 50+ people who refused to answer the questionnaire and for whom there is therefore no information on their self-rated health status'. This categorisation of the respondents was necessary to account for all the possible combinations of household composition and also to account for the possibility of the household members to not take part into the study.

<sup>&</sup>lt;sup>63</sup> SHARE respondents are asked to describe the area they live in out of five possible options: 1. A big city; 2. The suburbs or outskirts of a big city; 3. A large town; 4. A small town or 5. A rural area or village. Large town, small town and village were all grouped together under 'rural area' for the scope of this analysis.

paid work, formal engagement or informal participation were included. People who were not in paid work were also further split into two groups: those who had retired within two years prior to the interview and those who had not worked for more than two years in order to assess whether people's attitude towards formal and informal or family engagement changed in proximity to retirement. Finally, since it wass believed that respondents living together might coordinate their work decision, the working status of the other members of the household was considered<sup>64</sup>.

<u>Health Characteristics</u> — The engagement of older people may also be associated with health. Only a selection of the numerous variables available (fully described in Chapter 6.4) was considered in this analyses. In particular, we took into account a summary variable representing the total reported number of chronic illnesses (dichotomised to 'less than 2' against '2 or more' chronic diseases); a dummy variable indicating whether the respondent self-reported at least one functional limitation (NAGI, ADL or IADL); a dummy variable indicating whether the respondents self-reported their general health as at least good and finally an indicator of depressive symptoms (constructed considering symptoms on the EURO-D or on the CES-D depression scale).

## 7.2.3 STATISTICAL MODELLING

Logistic regressions for each activity were estimated for men and women separately in each country. Models were fitted also for SHARE countries altogether (with country dummy variables). The models presented in this chapter include a selection of the hypothesised explanatory variables based on a number of criteria such as the parsimony of the model, the comparability between the female and male subpopulations in the same country and between countries as well as the sampling adequacy and the multicollinearity of the independent variables, i.e. whether independent variables might

<sup>&</sup>lt;sup>64</sup> This variable includes the following categories: 'the respondent lives with at least another 50+ adult or a younger partner who is currently not working', 'the respondent lives with another 50+ adult or a younger partner who is currently working', 'the respondent lives alone', 'the respondent is the only 50+ adult in the household', and 'the respondent lives with a younger partner or other 50+ people who refused to answer the questionnaire and for whom there is therefore no information on their self-assessed economic position'. This categorisation of the respondents was necessary to account for all the possible combinations of household composition and also to account for the possibility of the household members to not take part into the study.

contain redundant (not significant) information if they were considered altogether in the model (Agresti, 2002).

Multicollinearity between the explanatory variables was investigated with logistic regressions with variables considered in blocks (i.e. socio-demographic, the economic and the health-related variables): it was assessed whether deleting one or more of the redundant predictors would reduce standard errors of the other estimated effects.

Also other criteria were adopted to decide which variables would be kept in the model. Firstly, variables with too few observations in subcategories which could not be combined were not considered in order to avoid reporting implausibly large odds ratios and standard errors (Pedhazur, 1997, Greenland et al., 2000, Agresti, 2002)<sup>65</sup>. Secondly, it was decided to select those variables which were comparable between subgroups of the various countries. Thirdly, the criterion of readability of the model was adopted: "the selection process becomes harder as the number of explanatory variables increases, because of the rapid increase in possible effects and interactions. There are two competing goals: The model should be complex enough to fit the data well. On the other hand, it should be simple to interpret, smoothing rather than over-fitting the data" (Agresti, 2002)(pg. 211).

Finally, it was decided to restrict the analyses only to those respondents with no missing information, i.e. to those with data on all variables in the model. Between 3% (Denmark and Italy) and 10% (France) of respondents were dropped.

All analyses were performed using Stata, version 11 (Stata Corp, 2009).

<sup>&</sup>lt;sup>65</sup> No formal rules exist to decide which variables not to include: some guidelines suggests that at least between 10 and 30 outcomes of each type should occur for every predictor (Peduzzi et al., 1996, Hosmer and Lemeshow, 2000) so as to also guarantee a sufficient number of independent variables to be included in the model (Agresti, 2002).

The main final models considered were as follows:

- For participation in **paid work**, the main model (see Table 7.5) considered as explanatory variables age, education, personal ill-health indicators, wealth assessed through mortgage-free house and marital status. Also other variables such as the working status or health characteristics of the other household members or the presence of at least one child in the household were explored in separate models.
- For participation in **formal activities**, the main model (see Table 7.11) accounted for explanatory variables such as age, education, personal health-characteristics, the area of residence as well as marital status. Similarly, whether the respondent lived alone or not was investigated in a different model. Furthermore, in order to assess whether –controlling for socio-economic and demographic characteristics –formal engagement was associated with participation in paid work and informal and family-related activities an additional model was fitted (see Table 7.12).
- For participation in **informal and family-related activities**, among the explanatory variables considered in the main model (see Table 7.14), age, education, personal health-characteristics, having grandchildren and having parents still alive were included. Other household characteristics (health of the other members living in the household, whether the respondent lived alone and marital status) were also explored. Finally, dicothomous variables indicating the respondents' participation in paid work and formal activities were added to the analysis to assess 'competition' between activities (see Table 7.17).

### 7.3 RESULTS

### 7.3.1 LOGISTIC REGRESSION: ANALYSIS OF PROPORTIONS IN PAID WORK

Table 7.5 summarises the results from the fully adjusted logistic regression models of being in paid work in each of the four countries separately for both men and women. In all populations considered, age was associated with the probability of being a worker, though this linear negative relationship was not significant among Danish men: being 55-59 was not significantly associated with reduced odds of being in paid work among this subpopulation. In all countries, those with the highest level of education were more likely than those with the lowest level of education to be in paid work. Differences between those with mid and low levels of education were only significant among Italian respondents and English women. Individual self-assessed bad health was negatively associated with the probability of working although with differences between subgroups: Danish and French women had significantly lower odds of being a worker if they reported at least one functional limitation and reported depressive symptoms, whereas among Italian women reporting at least 2 chronic diseases was the only factor negatively and significantly associated with lower odds of working. Men were less likely to be in paid work if they had some kind of limitation, with the exception of Italian male respondents. Italian men's health characteristics were never significantly associated with the probability of being in paid work: although the direction of the association was negative, those with 'ill-health' did not differ much compared to respondents in 'good health'. On the contrary, among English respondents, all three health indicators (depressive symptoms, chronic diseases and functional limitations) were significantly and negatively associated with the probability of being in paid work. Living in a mortgage or rent free house was negatively associated with being in paid work in all countries except among Danish women. Finally, findings suggested that the marital status of the respondent was differentially associated with the probability of being in paid work: English and Danish men who were married or lived with a partner were significantly more likely to be in paid work whereas the relationship was negative among Italian and English women. These findings may reflect different family values and gender role-division and in particular withdrawal of women from the labour market in response to family needs and presence of children (Thompson and Walker, 1989, Szinovacz et al., 2001).

In order to examine whether and how household characteristics may be associated with the participation of the respondents in the labour market, it was decided to explore in separate models whether variables accounting for the characteristics (health or work status) of the other household members or the presence of at least one child in the household would be associated with the odds of being in paid work. Although fully adjusted logistic regression models of being in paid work were fitted for each of the above-mentioned variables separately, comments and tables refer only the results for the specific variables. The associations found between being in paid work and the other explanatory variables (i.e. age, education, personal ill-health indicators and wealth assessed through mortgage-free house) remained in fact similar in terms of significance and direction.

Table 7.6 considers the presence of at least one child in the household: this variable appears to be an important factor associated with older people's engagement in paid work. Statistically significant differential associations were found: in Denmark and England there was a positive association with work both among men and women whereas in France and Italy the direction of association was negative among women (significantly in Italy). Also the health of other members of the household was explored to assess whether having other kin members in potential need of assistance could be associated with a reduction of paid work. The direction of associations between the self-rated health of the other members living in the household and the odds of being in paid work suggested that respondents who lived with another members who rated their health as more than good were more likely to be in paid work than those who lived with members who self-assessed their health as fair or poor. This association, though, was only significant among English male respondents (data not shown).

Also the working status of the other older members living in the household was explored<sup>66</sup>. Table 7.7 shows that respondents who lived with another adult in paid work were significantly positively associated with the odds of being in paid work in all countries (except among Italian women), lending support to the idea that older people might tend to coordinate their retirement decisions which are, to a certain extent, a household rather than a personal decision (Smith and Moen, 1998, Henkens, 1999, Coile, 2004, Gotlibovski and Weiss, 2008). Findings also suggested that people who either lived alone or who were the only 'adults' in the households had higher odds of being in paid work compared to those respondents who lived with others not in paid work.

 $<sup>^{66}</sup>$  See Table 7.A1 in the Appendix for the distribution of the economic status of the members in the household.

Table 7.5 Participation in Paid Work-Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions with legal marital status, by country and sex

	8		9	(	C C			
	DENN	<b>AARK</b>	FRA	NCE	ITA	LY	ENGI	<b>AND</b>
	Male	Female	Male	Female	Male	Female	Male	Female
Age 50-54	-							
	0.71	0.55 **	0.27 ***	0.61 **	0.33 ***	0.45 ***	0.61 ***	0.59 ***
Age: 55-59	(0.39-1.29)	(0.31-0.97)	(0.15-0.47)	(0.41-0.91)	(0.18-0.61)	(0.28 - 0.70)	(0.48-0.76)	(0.48-0.71)
	0.11 ***	0.04 ***	0.02 ***	0.06 ***	0.06 ***	0.06 ***	0.11 ***	0.10 ***
Age: 00-09	(0.06-0.18)	(0.02-0.08)	(0.01-0.04)	(0.04-0.08)	(0.03-0.12)	(0.03-0.12)	(0.09-0.13)	(0.08-0.12)
Education: LOW								
	1.80	1.42	1.36	1.34	2.08 ***	4.22 ***	1.16	1.57 ***
Education: MIDDLE	(0.96-3.37)	(0.74-2.73)	(0.82-2.22)	(0.95-1.89)	(1.31-3.29)	(2.72-6.53)	(0.94 - 1.43)	(1.29-1.90)
	2.07 **	3.97***	3.43 ***	3.04 ***	6.93 ***	12.1 ***	1.30 **	1.71 ***
Education: HIGH	(1.04-4.16)	(2.04-7.75)	(1.89-6.24)	(2.08-4.45)	(4.12-11.6)	(4.89-29.9)	(1.06-1.59)	(1.39-2.10)
<2 Chronic diseases	5							
	0.69	0.68	0.79	0.85	0.72	0.48 **	0.47 ***	0.55 ***
2+ Chronic diseases	(0.43-1.13)	(0.40-1.15)	(0.48-1.31)	(0.57-1.26)	(0.43-1.22)	(0.23-1.00)	(0.39-0.58)	(0.45-0.66)
No Limitations								
1+ NAGI, IADL or	0.31 ***	0.33 ***	0.59 **	0.71 **	0.71	0.85	0.50 ***	0.59 ***
ADL Limitation	(0.19-0.52)	(0.19-0.57)	(0.39-0.91)	(0.51-0.98)	(0.45-1.22)	(0.50-1.43)	(0.42-0.60)	(0.50-0.70)
<3 Depressive Symptoms					-			
3+ Depressive	0.65	0.47 **	0.64	0.67 **	0.65	1.01	0.43 ***	0.58 ***
Symptoms	(0.34-1.26)	(0.24-0.88)	(0.36-1.12)	(0.49-0.92)	(0.38-1.09)	(0.60-1.70)	(0.34-0.54)	(0.47-0.71)
Not Married								
	2.73 ***	1.17	0.79	0.77	1.01	0.59 **	1.58 ***	0.80 ***
Married	(1.64-4.53)	(0.71-1.91)	(0.48-1.31)	(0.56-1.08)	(0.48-2.08)	(0.37-0.95)	(1.28-1.96)	(0.67-0.95)
<b>Paying Rent or Mortgage</b>								
Monthly Rent/	0.50 **	1.07	0.86	0.65 **	0.66	0.54**	0.52 ***	0.61 ***
Mortgage free	(0.28-0.89)	(0.58-1.94)	(0.60-1.22)	(0.44-0.94)	(0.39-1.08)	(0.31-0.92)	(0.44-0.63)	(0.51-0.72)
<b>Observations (N)</b>	532	542	854	971	762	977	3,338	3,556

Source: SHARE, 2004; ELSA, 2002; weighted data - own calculations. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively.

÷

Regressions, by country and sex. Selected results obtained when presence of at least one child in the household was considered Table 7.6 Participation in Paid Work-Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic

DENMARK         FRANCE         ITALY         ENGLAND           No children in Household         Male         Female         Male         Female         Male         Female         Male         Female         F					-				
No children in Household         Male         Female         Male         Female         Male         Female         Female		DENN	<b>IARK</b>	FRA	NCE	ITA	ТΥ	ENG	LAND
No children in Household         1.17         2.26 ***         0.81         1.22         0.60 ***         1.62 ***         1.25 **           At least 1 child lives         2.32 **         1.17         2.26 ***         0.81         1.22         0.60 ***         1.62 ***         1.25 **           In the Household         (1.18-4.56)         (0.52-2.52)         (1.42-3.62)         (0.57-1.12)         (0.69-2.18)         (0.41-0.89)         (1.33-1.97)         (1.03-1.52)           Number of Observations         532         542         854         971         762         978         3,339         3,556		Male	Female	Male	Female	Male	Female	Male	Female
At least 1 child lives         2.32 **         1.17         2.26 ***         0.81         1.22         0.60 ***         1.62 ***         1.25 **           in the Household         (1.18-4.56)         (0.52-2.52)         (1.42-3.62)         (0.57-1.12)         (0.69-2.18)         (0.41-0.89)         (1.33-1.97)         (1.03-1.52)           Number of Observations         532         542         854         971         762         978         3,339         3,556	No children in Household								
in the Household         (1.18-4.56)         (0.52-2.52)         (1.42-3.62)         (0.57-1.12)         (0.69-2.18)         (0.41-0.89)         (1.33-1.97)         (1.03-1.52)           Number of Observations         532         542         854         971         762         978         3,339         3,556	At least 1 child lives	2.32 **	1.17	2.26 ***	0.81	1.22	0.60 ***	1.62 ***	1.25 **
Number of Observations         532         542         854         971         762         978         3,339         3,556	in the Household	(1.18-4.56)	(0.52-2.52)	(1.42-3.62)	(0.57-1.12)	(0.69-2.18)	(0.41-0.89)	(1.33-1.97)	(1.03-1.52)
Number of Observations         532         542         854         971         762         978         3,339         3,556									
	Number of Observations	532	542	854	971	762	978	3,339	3,556

Source: SHARE, 2004; ELSA, 2002; weighted data – own calculations. Among the other covariates, this specification included age, education, personal ill-health indicators and wealth assessed through mortgage-free house. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively.

Regressions, by country and sex. Selected results obtained when employment status of the other household members was Table 7.7 Participation in Paid Work-Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic considered

	DENN	AARK	FRA	NCE	ITA	ЛУ	ENGI	<b>UND</b>
	Male	Female	Male	Female	Male	Female	Male	Female
Other member(s) in								
Household NOT in paid work			_					
Other member(s) in	3.91 ***	1.95 **	2.25 ***	2.36 ***	2.59 ***	1.55	4.03 ***	2.87 ***
Household in naid work	(2.29-6.68)	(1.01-3.76)	(1.32-3.89)	(1.45-3.84)	(1.39-4.84)	(0.80 - 3.04)	(3.30-4.92)	(2.32-3.54)
	0.57	1.09	1.58	1.64 ***	1.58	3.49 ***	1.06	1.57 ***
Living alone	(0.29-1.12)	(0.57-2.09)	(0.77-3.25)	(1.13-2.38)	(0.54-4.62)	(1.54-7.93)	(0.81-1.38)	(1.24-1.97)
Only 50+ Member	3.44	1.23	0.98	1.66	2.74 ***	1.12	2.29 ***	2.88 ***
in the Household	(0.66-17.8)	(0.40-3.78)	(0.49-1.95)	(0.96-2.87)	(1.53-4.58)	(0.62-2.03)	(1.30-4.04)	(2.07-4.00)
I iving with other members	1 93	1.42	0.89	1.68	2.09	1.28	2.21 ***	2.14 ***
who refused to answer	(0.82-4.52)	(0.48-4.31)	(0.40-1.97)	(0.60-4.71)	(0.95-4.58)	(0.60-2.73)	(1.36-3.60)	(1.49-3.08)
Number of Observations	532	541	836	945	761	977	3,338	3,556
C CITABE 2004. ELCA	2000	1 data 2000 d	-laulations A.	nong the other	correctate this	e enerification	included age	education

personal ill-health indicators and wealth assessed through mortgage-free house. The employment status of the other members in the household another member currently in paid work', 'the respondent lived alone', 'the respondent is the only adult in the household', and 'the respondent included the following categories: 'the respondent lives with other members currently not in paid work', 'the respondent lives with at least lived with other members who refused to answer the questionnaire and for whom there is therefore no information on their self-assessed economic position'. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively. UWII Calvulativits. Alivery uv Source: SHAKE, 2004; ELSA, 2002; Weignicu uala

Table 7.9 summarises results from the fully adjusted logistic regression models of being in paid work when the SHARE countries are considered all together. The probability of being a worker significantly decreased with age. Ill-health was also associated with decreased odds of being in paid work: women declaring more than 2 chronic diseases and men having at least one functional limitation and having depressive symptoms were associated with decreased odds of being in paid work. Being married was significantly negatively associated with the odds of being in paid work only among women. It was also found (data not shown) that both among men and women living with at least one other adult in paid work were significantly more likely to be in paid work compared to respondents who lived with other members not in paid work. Results also confirmed a statistically significant –though differential –association between paid work and the presence of at least one child living in the household: men were associated with higher odds of being in paid work whereas women with lower odds.

When countries were considered altogether, however, the interaction between country and education was included in order to account for the different distributions of education (see Table 6.5) and of paid work by education (see Table 7.8) in the three countries under study. If an interaction term was not included, we would end up assuming that high education was more beneficial in Italy merely because 70% of Italian men with higher education were in paid work against 63% in Denmark and France, even though the proportion of Italian male respondents with high education was less than 10% compared to more than 30% in Denmark. Without an interaction term, the multivariate model (see Table 9) would suggest, for instance, that Danish and Italian older men had similar probabilities to be in paid work, despite the much higher percentage of male workers in Denmark (58%) compared to Italy (40%).

Table 7.8 Respondents (%) in paid work by education, gender and country

	Ita	ly	Fra	nce	Denr	nark	TOTAI	SHARE
Education	M	F	М	F	M	F	Male	Female
LOW	29.7	11.3	33.8	29.7	42.0	24.8	31.3	18.1
LOW	(142/535)	(76/750)	(119/353)	(146/479)	(31/71)	(32/126)	(292/959)	(254/1,355)
MIDDIE	54.8	38.1	49.7	43.2	59.4	41.5	52.5	41.2
MIDDLE	(81/165)	(56/161)	(163/324)	(140/323)	(170/290)	(91/220)	(414/779)	(287/704)
THOM	70.3	57.5	63.0	58.2	62.9	71.5	65.2	59.4
HIGH	(46/70)	(45/84)	(138/223)	(124/213)	(111/177)	(153/209)	(295/470)	(322/506)

Each cell represents the percentage of respondents (in brackets the actual number) with a specific education level who were in paid work. Source: SHARE, 2004; weighted data.

### Table 7.9 Participation in Paid Work – Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions with and without education and considering the interaction term between education and country –all SHARE countries considered all together by sex

	Mo	del	Mo	del	Mo	del
	(without e	education)	(with ed	ucation)	(with int	eraction)
	Male	Female	Male	Female	Male	Female
Age: 50 - 54						
	0.34 ***	0.57 ***	0.33 ***	0.57 ***	0.33 ***	0.57 ***
Age: 55-59	(0.23-0.49)	(0.43-0.75)	(0.22-0.48)	(0.43-0.75)	(0.22-0.48)	(0.43-0.76)
	0.05 ***	0.07 ***	0.05 ***	0.07 ***	0.05 ***	0.07 ***
Age: 00-09	(0.04-0.08)	(0.04-0.10)	(0.03-0.08)	(0.04-0.10)	(0.03-0.08)	(0.05-0.10)
Paying				<u></u>		
Rent/Mortgage		0. (0. +++		0 (0 +++	0.50.44	0 (0 ++++
Rent/Mortgage	0.72 **	0.68 ***	0.72 **	0.62 ***	0.73 **	0.63 ***
free	(0.53-0.96)	(0.53-0.88)	(0.54-0.97)	(0.46-0.82)	(0.55-0.98)	(0.47-0.84)
No Limitations				····		
1+ Limitation	0.60 **	0.75 **	0.67 **	0.77	0.66 **	0.76
(NAGI, IADL, ADL)	(0.43-0.84)	(0.56-0.99)	(0.48-0.92)	(0.58-1.02)	(0.48-0.91)	(0.57-1.01)
<2 Chronic						
diseases						
2+ Chronic	0.77	0.58 **	0.75	0.65 **	0.74	0.67
diseases	(0.57-1.06)	(0.38-0.88)	(0.54-1.03)	(0.44-0.96)	(0.53-1.02)	(0.45-1.00)
<3 Depr.						
symptoms						
3+ Depressive	0.66 **	0.75 **	0.69 **	0.79	0.66 **	0.79
symptoms	(0.47-0.93)	(0.56-0.99)	(0.49-0.97)	(0.58-1.07)	(0.48-0.91)	(0.59-1.07)
Not Married						
Married	0.95	0.69 **	0.98	0.71 **	0.96	0.74 **
	(0.62-1.46)	(0.52-0.90)	(0.65-1.49)	(0.54-0.93)	(0.63-1.47)	(0.56-0.92)
Education:						
LOW			1 77 444	0.01.+++	1.05 +++	
Education:				2.21 ***	1.95 ***	4.23 ***
MIDDLE			(1.27-2.48)	(1.65-2.97)	(1.20-3.17)	(2.75-6.51)
Education:			4.63 ***	5.23 ***	7.41 ***	
HIGH			(3.12-0.87)	(3.35-8.17)	(4.13-13.3)	(4.03-27.0)
Italy	0.07	0.07 ***	0.71			254 +++
France	0.90	$(1 \leq 2 27)$			0.90	3.34 TT (2.22 5.20)
	1 41	(1.00-3.37)	(0.49 - 1.03)	(1.33-2.03)	(0.34-1.50)	(2.33-3.39)
Denmark		(2.96 ***	0.99	1./0 ***	1.13	(1.42.5.42)
	(0.99-2.02)	(2.05-4.55)	(0.02-1.57)	(1.14-2.53)	(0.33-2.44)	(1.43-5.43)
France " Middle Ed					(0.73)	$(0.29^{+++})$
Frence *	· · · · · · · · · · · · · · · · · · ·				0.46 **	(0.17-0.32)
France " Link Fd					(0.40 ***	(0.08.0.64)
Denmark *		┢━━━━━	<u> </u>		1 12	0.00-0.04)
Middla Fd		}			(0.48-2.62)	(0.17-0.81)
Denmark *					0.38	0.17-0.01)
High Fd	1				(0 14-1 01)	(0.13_1.27)
mgn Eu						(0.13-1.27)
Observations	2,140	2,473	2,129	2,463	2,129	2,463

Source: SHARE, 2004; weighted data – own calculations. Fully adjusted logistic results for participation in paid work for SHARE countries considered altogether, accounting for country dummy variables. Three different sub-models are considered: the first ignores education, the second includes it and the third considers the interaction between education and country. **\*\***, **\*\*\***: significant at the 0.05 and 0.01 levels, respectively.

	0	gender -	SHARE resp	ondents		
		Male			Female	······
	Italy	France	Denmark	Italy	France	Denmark
Education: LOW	1.00	0.90 (0.54-1.50)	1.13 (0.53-2.44)	1.00	3.54 *** (2.33-5.39)	2.78 *** (1.43-5.43)
Education: MIDDLE	1.95 *** (1.20-3.17)	1.49 (0.80-2.06)	2.49 *** (1.44-4.31)	4.23 *** (2.75-6.51)	4.47 *** (2.94-6.77)	4.36 *** (2.70-7.01)
Education:	7.41 ***	3.10 ***	3.23 ***	11.3 ***	9.45 ***	12.7 ***

Table 7.10 Result Summary in terms of odds ratios of being in paid work considering the interaction terms between country and level of education, by gender - SHARE respondents

Source: SHARE, 2004; weighted data. Own Calculations. Each cell represents the odds for men or women of being in paid work by country and education level obtained considering the linear combination of Education, country and their interaction term form Table 9. **\*\***, **\*\*\***: significant at the 0.05 and 0.01 levels, respectively

(4.63-27.6) (5.95-15.0)

(7.64-21.3)

(1.67-6.24)

### 7.3.2 LOGISTIC REGRESSIONS FOR FORMAL ACTIVITIES

(1.86-5.19)

(4.13-13.3)

HIGH

Table 7.11 shows the results of the regression models for the participation in formal activities fitted in each of the SHARE and ELSA subpopulations<sup>67</sup>. Age is generally not significantly associated with formal engagement, even though age-groups 60-64 and 65-69 were positively and significantly associated with participation in formal activities among specific subpopulations such as Danish, French and English women and French and Italian men. Higher education was significantly associated with higher odds of formal engagement in all subpopulations, except Danish and Italian men. In particular, Italian men with middle education had higher odds of being formally engaged than those with high education (although this may reflect the small proportions in the latter group) whereas neither categories were associated with formal engagement among Danish male respondents. The role of ill-health in older people's engagement in formal activities was unclear. Among Danish respondents, none of the variables suggesting bad health were significantly associated with formal involvement. Conversely, in the other subpopulations a statistically significant -though differential -association was found: less than very good health (in England and among French men) and reporting depressive symptoms (among French men and Italian respondents) were paralleled with a lower propensity to report formal engagement, while reporting more than 2 chronic diseases

<sup>&</sup>lt;sup>67</sup> Note that although the outcome variable definition differs for the two datasets in the description of the logistic results I will generically and indistinctively refer to 'formal participation'.

was positively associated with formal engagement among English women. Finally, the legal marital status did not seem to be associated with the odds of being engaged in formal activities; only Italian women who were married had significantly lower odds of being formally engaged. Opposite results were found when, instead of marital status, the variable indicating whether the respondent lived alone or not was considered: only Italian respondents living alone had significantly higher odds of being engaged in formal activities compared to those living alone whereas no associations were found in the other populations.

In a separate model, it was also investigated whether 'bigger' areas would present more opportunities for people to engage in formal activities. Bearing in mind that people living in different countries may have in mind a different definition of 'big town' or 'rural area' based on the relative population necessary to make a community "urban" (Hartshorn, 1992), only Italian women and Danish men who lived in a big city had higher odds of being formally engaged compared to those who reported living in a small town or rural area whereas the direction and strength of the association was unclear in the other subpopulations (data not shown).

Analyses shown on Table 7.12 also included among the covariates engagement in the other two forms of engagement, i.e. paid work and participation in informal or family-related activities. When these two factors were included in the analysis, age was no longer associated with the odds of formal engagement and the pattern of association was no longer linear. The other factors on the contrary confirmed similar patterns and directions of association. When considering people's engagement in the other forms of activity, a substitutive association between formal engagement and paid work was suggested in all countries except Italy and among French men: individuals who were in paid work were significantly less likely to engage in formal activities; in Italy and among French men the direction of the association was also negative although not significant. The association between formal and informal/family engagement was positive in all countries (but not significantly among Italian respondents and Danish men), backing the hypothesis of a cumulating commitment. It is worth noting that out of all the subpopulations under study, Italy was the only country where the substitutive

association between formal engagement and paid work as well as the cumulative association between formal and informal participation was not significant.

In a separate model, it was tested whether not being in paid work for less than two years before the survey was associated with formal engagement (see Table 7.A2 in appendix). Findings showed that respondents who had 'recently retired' had lower odds of being formally engaged similarly to those who were in paid work (significantly among French women and English men) with the exception of Italian women who had higher odds of being involved in formal activities compared to respondents who had not been in paid work at all or for more than two years. Cross-sectional analyses failed to endorse the general idea that people who retire are more likely to get involved in formal activities due to their free time available and also in order to (over)-compensate for lack of work and replace the benefits previously derived from paid work (Fischer et al., 1991, Wilson, 2000). Retirement does not necessarily draw people into formal involvement (Herzog et al., 1989, Caro and Bass, 1997), this may increase only among those who already were formally engaged (Mutchler et al., 2003).

Education, ill-health and employment status seemed to be the major determinants of older people's commitment in formal participation even though there was no clear common pattern and differences between subpopulations were quite striking and difficult to interpret: commitment to formal activities among Danish men seemed to associated mainly with employment status for instance. Other factors not considered in this analysis such as societal context, culture, values and expectations must be playing a more important role in shaping older people's level of formal commitment (Salamon and Sokolowski, 2001, Hank and Stuck, 2008, Hank, 2010, Hank and Erlinghagen, 2010), given the different levels of commitment in the four countries under study.

Table 7.11 Participation in Formal Activities –Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Repressions, by country and sex

		9						
	DENN	<b>1AKK</b>	<b>FKA</b>	NCE	IIA	ТY	ENG	<u>ANU</u>
	Male	Female	Male	Female	Male	Female	Male	Female
Age 50-54								
	0.98	1.07	1.15	1.15	0.78	1.05	0.88	1.20
Age: cc-cc	(0.59–1.62)	(0.66-1.76)	(0.80 - 1.67)	(0.79–1.69)	(0.33 - 1.85)	(0.65–1.68)	(0.67 - 1.15)	(0.95–1.52)
Amo: 60 64	1.16	1.34	1.46	1.54 **	2.23 **	1.16	1.07	1.26 *
Age: 00-04	(0.68-1.94)	(0.79 - 2.28)	(0.89 - 2.37)	(1.02-2.31)	(0.98 - 5.10)	(0.66 - 2.04)	(0.80 - 1.43)	(0.97–1.62)
A no 65 60	1.30	1.99 **	2.32 ***	0.92	1.82	0.68	1.28 *	1.39 **
Age us-us	(0.74 - 2.32)	(1.10 - 3.58)	(1.37 - 3.91)	(0.60 - 1.39)	(0.85–3.89)	(0.39 - 1.16)	(0.96 - 1.73)	(1.08–1.81)
<b>Education: LOW</b>								
Education:	0.93	1.48	1.08	2.01 ***	2.67 ***	2.53 ***	2.38 ***	3.64 ***
MIDDLE	(0.52-1.86)	(0.89-2.46)	(0.74 - 1.59)	(1.45–2.78)	(1.47-4.84)	(1.45-4.42)	(1.77-3.19)	(2.88-4.60)
Education:	1.49	2.36 ***	2.08 ***	3.28 ***	1.29	3.04 ***	3.84 ***	7.52 ***
HIGH	(0.80-2.80)	(1.37-4.05)	(1.46-2.97)	(2.14–5.04)	(0.51–3.27)	(1.77–5.20)	(2.92–5.06)	(5.95–9.50)
<2 Chronic Diseases								
	0.87	0.97	1.45 *	1.07	0.82	1.11	1.16	1.29 **
2+ Unronic diseases	(0.54 - 1.37)	(0.62–1.52)	(0.95–2.21)	(0.69–1.65)	(0.45–1.47)	(0.56 - 2.19)	(0.90 - 1.50)	(1.05-1.59)
No Functional lim.s								
1+ Functional	0.67	96.0	1.17	1.18	1.44	1.50	1.02	1.02
Limitation	(0.38-1.16)	(0.63-1.48)	(0.80-1.71)	(0.89-1.56)	(0.79 - 2.62)	(0.81 - 2.78)	(0.80 - 1.28)	(0.84 - 1.24)
3 Depressive sympt.s								
<b>3+ Depressive</b>	06'0	0.75	0.65 **	0.85	0.45 **	0.61 *	0.95	0.89
Symptoms	(0.48–1.69)	(0.45–1.26)	(0.42 - 1.00)	(0.61–1.19)	(0.21-0.96)	(0.34 - 1.08)	(0.70 - 1.29)	(0.71 - 1.12)
SRH ≥ good								
Self-rated Health	0.64	1.06	0.45 ***	0.76	0.51	0.88	0.65 ***	0.64 ***
< Good	(0.35-1.14)	(0.60-1.86)	(0.26-0.77)	(0.47 - 1.24)	(0.23–1.15)	(0.52–1.51)	(0.47 - 0.89)	(0.48 - 0.84)
Not Married								
	1.47	1.06	1.10	1.12	0.68	0.49 ***	1.21	1.11
Married	(0.95-2.25)	(0.71 - 1.58)	(0.77-1.58)	(0.83 - 1.51)	(0.33 - 1.40)	(0.31-0.77)	(0.93–1.59)	(0.91 - 1.35)
N Observations	532	548	852	981	763	987	3,308	3,547

Source: SHARE, 2004; ELSA, 2002; weighted data - own calculations. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively.

....

٩,

Table 7.12 Participation in Formal Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions

•

with the 'living alo	ne' indicator	and dummy	indicators of e	engagement in	the other two	forms of activ	vity, by count	try and sex
	DENM	ARK	FRA	NCE	ITA	LY	ENG	LAND
	Male	Female	Male	Female	Male	Female	Male	Female
Age 50-54								-
	66.0	1.01	1.11	1.01	0.85	0.92	0.84	1.15
Age: 33-39	(0.59 - 1.67)	(0.60 - 1.68)	(0.77 - 1.61)	(0.68–1.49)	(0.33 - 2.19)	(0.57 - 1.51)	(0.64–1.11)	(0.91 - 1.45)
Age: 60-64	1.03	16.0	91.1	1.18	2.17	0.93	0.92	1.09
	(0.59–1.75)	(69.1-16.0)	(0.69-1.94)	(0.83-1.69)	(68.0-08.0)	(7/.1-10.0)	0.00	(0.0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0
Age 65-69	0.99 (0.52-1.90)	1.21 (0.61–2.37)	(1.02-3.21)	0.64 (0.41-1.01)	1.73 (0.68–4.36)	0.34-1.08)	0.70-1.36)	(0.86–1.54)
Education: LOW	(							
Education:	0.97	1.45	1.04	2.03 ***	2.65 ***	2.65 ***	2.40 ***	3.69 ***
MIDDLE	(0.53-1.77)	(0.87–2.40)	(0.68–1.57)	(1.47–2.79)	(1.48-4.74)	(1.48-4.77)	(1.78-3.22)	(2.91–4.67)
Education:	1.63	2.65 ***	2.12 ***	3.57 ***	1.24	3.28 ***	3.88 ***	7.77 ***
HIGH	(0.85-3.12)	(1.53-4.60)	(1.43–3.14)	(2.34–5.44)	(0.47–3.25)	(1.89–5.72)	(2.93–5.13)	(6.14–9.85)
<2 Chronic Diseases								
	0.89	0.93	1.45 *	1.04	0.78	1.07	1.14	1.28 **
2+ Chronic diseases	(0.56–1.44)	(0.58–1.46)	(0.96–2.19)	(0.68–1.61)	(0.43 - 1.42)	(0.54-2.09)	(0.88 - 1.47)	(1.04 - 1.57)
No Limitation								
1+ Limitation	0.63	0.00	1.12	1.16	1.37	1.44	0.97	1.01
(NAGI, IADL, ADL)	(0.36-1.09)	(0.58 - 1.40)	(0.76-1.64)	(0.88-1.54)	(0.75 - 2.48)	(0.79–2.62)	(0.76 - 1.23)	(0.83–1.22)
3 Dep. Symptoms								
3+ Depressive	0.88	0.70	0.63 **	0.83	0.42 **	0.61 *	0.91	0.85
symptoms	(0.46 - 1.67)	(0.42–1.16)	(0.40 - 0.98)	(0.58–1.19)	(0.19 - 0.93)	(0.34 - 1.08)	(0.67 - 1.24)	(0.67 - 1.07)
SRH > Good								
Self-rated Health	0.58	0.91	0.44 ***	0.73	0.48	0.86	0.60 ***	0.61 ***
< Good	(0.31 - 1.07)	(0.51 - 1.62)	(0.26 - 0.73)	(0.45–1.17)	(0.20 - 1.15)	(0.51 - 1.46)	(0.43 - 0.84)	(0.46 - 0.80)
Not Living Alone								
	0.63	1.03	1.23	1.11	3.00 ***	2.05 **	0.77	1.03
LIVING Alone	(0.37 - 1.08)	(0.67–1.58)	(0.77 - 1.96)	(0.79–1.57)	(1.37 - 6.61)	(1.02-4.12)	(0.56-1.07)	(0.81 - 1.28)
Not in paid work								
	0.62 **	0.47 ***	0.72	0.66 **	0.86	0.69	0.65 ***	0.74 ***
In paid work	(0.38 - 1.01)	(0.28-0.77)	(0.47 - 1.10)	(0.44-0.98)	(0.42–1.76)	(0.40 - 1.20)	(0.51 - 0.84)	(0.60-0.92)
No Informal Eng.nt								
Engaged in Fam&Inf	1.15	1.64 ***	*** <i>L</i> 9.1	1.71 ***	1.65	1.47	1.46 ***	1.37 ***
Activities	(0.78 - 1.70)	(1.12–2.39)	(1.25–2.23)	(1.31 - 2.24)	(0.96–2.85)	(0.91–2.36)	(1.14 - 1.87)	(1.14 - 1.67)
N Observations	532	548	851	981	763	988	3,309	3,547
Countries Ch	TARE 2004- FI	SA 2002 weigh	ted data – own ca	Iculations. ** **:	: significant at the	e 0.05 and 0.01 lev	vels, respectively	

5.5 .: Significalit at the 0.00 and Source: SHARE, 2004; ELSA, 2002; weighted data – own calculations. <sup>++</sup>, <sup>+</sup>

4

A 'country' factor was also confirmed when all SHARE countries were analysed together and a dummy country indicator was included in the analyses (see Table 7.13). Multivariate analyses reflected differences between the SHARE countries: both Danish and French older men and women were found to exhibit significantly higher probabilities to be engaged in informal activities compared to Italian respondents, even after controlling for socio-demographic, economic and health characteristics. Findings also suggested a case for a divide between paid work and the other forms of activities: whereas the association between formal and informal engagement was positive, the association between formal involvement and paid work was negative. Respondents who were engaged in formal activities were also associated with engagement in informal and family-related activities whereas those in paid work were more likely to have lower odds of participating in formal activities.

# 7.3.3 LOGISTIC REGRESSIONS FOR BEING INVOLVED IN INFORMAL AND FAMILY ACTIVITIES

Table 7.14 shows the results of the logistic regression model for the participation in informal and family activities in both SHARE countries and England<sup>68</sup>. Age was not significantly associated with informal engagement; the only exception were French respondents aged 55-59 who were significantly less likely to be engaged in family and informal activities compared to those aged 50-54. Similarly, no significant association was found between education and informal engagement. As for health characteristics, ill-health indicators are generally not significantly associated with informal and family engagement, with a few exceptions: Danish women who declared at least 2 chronic diseases, English women who had depressive symptoms and English men who declared at least one IADL, ADL or NAGI limitation were significantly positively correlated with informal engagement. The existence of kin members – almost a prerequisite for informal and family engagement in almost all countries. In particular, having at least one parent alive (regardless of their health status) and/or at least one grandchild increased the odds

<sup>&</sup>lt;sup>68</sup> Although the outcome variable definition differs for the two datasets, in the following description of the logistic regression results I will generically and indistinctively refer to 'informal participation'.

	M	F	M	F	M	F
	0.94	1.09	0.95	1.11	0.92	0.99
Age: 55-59 *	(0.66-1.34)	(0.84–1.43)	(0.66–1.35)	(0.85–1.45)	(0.64–1.34)	(0.75–1.30)
	1.65 **	1.37 **	1.67 **	1.39 **	1.43	1.09
Age: 60-64 "	(1.12–2.44)	(1.02–1.82)	(1.12–2.48)	(1.04–1.85)	(0.90–2.28)	(0.81–1.47)
	1.91 **	0.90	1.93 ***	0.92 **	1.60 **	0.71 **
Age 05-09	(1.30-2.80)	(0.67–1.21)	(1.31–2.84)	(0.68–1.24)	(1.01–2.53)	(0.51-0.97)
Education:	1.51 **	2.17 ***	1.52 **	2.19 ***	1.49 **	2.25 ***
MIDDLE <sup>b</sup>	(1.06-2.15)	(1.63–2.89)	(1.06-2.17)	(1.65–2.93)	(1.03-2.16)	(1.69–3.00)
Education:	2.02 ***	3.16 ***	2.03 ***	3.21 ***	2.07 ***	3.57 ***
HIGH <sup>b</sup>	(1.45-2.81)	(2.31–4.31)	(1.45–2.83)	(2.35-4.38)	(1.46–2.93)	(2.61-4.86)
1+ Limitation <sup>c</sup>	1.20	1.28	1.20	1.27	1.14	1.24
(NAGI, IADL, ADL)	(0.87-1.64)	(0.97–1.69)	(0.88-1.65)	(0.96–1.69)	(0.83-1.57)	(0.95–1.64)
2+ Chronic <sup>d</sup>	1.10	1.06	1.10	1.07	1.09	1.02
diseases	(0.81–1.48)	(0.74–1.53)	(0.81–1.49)	(0.75–1.53)	(0.80–1.48)	(0.72-1.46)
Dermanute	0.57 ***	0.74	0.58 ***	0.74	0.56 ***	0.74 **
Depressed	(0.40-0.81)	(0.55–1.00)	(0.41–0.83)	(0.55-1.00)	(0.39-0.81)	(0.55–1.00)
SRH	0.48 ***	0.81	0.49 ***	0.82	0.48 ***	0.79
< Good <sup>f</sup>	(0.32-0.75)	(0.57–1.15)	(0.32-0.75)	(0.57–1.16)	(0.31-0.75)	(0.56–1.12)
Their a Alana	1.49	1.32			1.59 **	1.37
Living Alone	(0.98-2.28)	(0.93-1.86)			(1.02-2.51)	(0.97–1.94)
Manniad			0.96	0.80		
warrieu			(0.68–1.36)	(0.61-1.05)		
In naid work					0.78	0.67 ***
					(0.54–1.14)	(0.49-0.90)
Informally					1.59 ***	1.58 ***
Engaged					(1.23–2.04)	(1.23-2.02)
France 8	2.62 ***	1.56 ***	2.64 ***	1.57 ***	2.62 ***	1.63 ***
r rance -	(1.93-3.62)	(1.18-2.07)	(1.92-3.62)	(1.19-2.08)	(1.90-3.60)	(1.23-2.16)
Donmanly 8	3.06 ***	2.17 ***	3.10 ***	2.21 ***	3.06 ***	2.33 ***
Deпшагк -	(2.12-4.45)	(1.55-3.04)	(2.14-4.52)	(1.59–3.09)	(2.08-4.50)	(1.66-3.29)
	ļ	L	<u> </u>	L	ļ	
Observations	2,147	2,517	2,146	2,517	2,147	2,516

 Table 7.13 Participation in Formal Activities –Odds Ratios and 95% Confidence

 Intervals obtained from fully adjusted Logistic Regressions when all SHARE

 countries are grouped altogether, by sex and various model specifications

Source: SHARE, 2004; weighted data – own calculations. This table shows the fully adjusted logistic results for participation in informal activity only for SHARE countries when the latter are considered altogether and country dummy variables are taken into account. Analyses are repeated for various model specifications. Reference categories: a) Age 50-54; b) Education: LOW c) No NAGI, IADL or ADL limitations d) < 2 Chronic diseases, e) Less than 3 depressive symptoms, f) Self-Rated Health ≥ good g) Italy. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively.

of being committed to informal and family activities. Also respondents who are married are associated with increased odds of being informally engaged. Married people tend to have larger networks which increases their chances of getting in touch with other family members or friends who need care or help (De Koker, 2009). Among Italian men, French men and English respondents having a parent alive, having grandchildren and being married are all simultaneously positively and significantly associated with family and informal engagement.

Other characteristics of the household were also investigated to assess whether, for instance, the presence of a member in the household who self-rated their health as less than good would increase the odds of being informally engaged. Table 15 shows that respondents who lived with other members who self-rated their health as better than good had lower odds to be involved in informal activities compared to those who lived with members with poor or bad self-rated health; this association was statistically significant among French women and English respondents. Table 16 focuses specifically on the association between informal engagement and household composition, regardless of the relationship of the other members living in the household or their health status: findings show that respondents who live dalone were generally less likely to be engaged in informal activities than those who live with other members (significantly among Danish and French men as well as English respondents).

Table 7.17 shows results obtained when engagement in paid work and formal engagement were included among the covariates. It was found that people who were in paid work had lower odds of being engaged in informal and family activities compared to those not in paid work, though this negative association was statistically significant only among English people. On the other hand, respondents formally engaged were significantly positively associated with informal engagement in all countries except among Italian women and Danish men, suggesting a complementarity of these two activities.

Table 7.18 shows the results for the logistic regression model for the participation in informal and family activities when SHARE countries are considered all together. Most of the above-discussed results were confirmed: presence of kin in or outside the

household was positively associated with higher odds of engagement in informal activities. Considering the inter-relationship between the various forms of engagement, formal engagement and paid work were differentially associated with informal engagement, though the negative association with paid work was not statistically significant among SHARE male respondents (as also observed for formal engagement – see Table 7.13). Finally, being Danish or French was not significantly associated with informal and family engagement. The slighter higher level of informal engagement in Denmark and France (37% and 33% respectively) compared to Italian men (29%) and a reverse mirrored picture for women (i.e. higher participation in Italy compared to France and Denmark) is reflected in the direction of the odds. Nonetheless, once health, age, education and the presence of a kin network were controlled for, there were no statistically significant differences between countries.

Table 7.14 Participation in Informal and Family-related Activities -Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic Regressions. by country and sex

	DENN	IARK	FRA	NCE		ALV	ENGI	AND
	Mala	Remale	Male	Famala	Mala	Female	Mala	Female
	V DO	r cultary			INTAIC 0 CC	1 OF	1VIAIC	r chiaic
A 02. 55.50 "	0.82	0./8	0.04 **	1.0/	0.00	C7.1	0.84	1.11
Age. 33-33	(0.48-1.40)	(0.46 - 1.31)	(0.44-0.94)	(1.21–2.31)	(0.36–1.22)	(0.86 - 1.82)	(0.65–1.09)	(0.90 - 1.37)
	1.29	1.12	0.89	1.36	1.18	1.26	0.86	1.05
Age: 00-04	(0.72–2.30)	(0.62–2.02)	(0.56 - 1.41)	(0.87–2.13)	(0.68–2.06)	(0.78-2.01)	(0.64–1.15)	(0.82 - 1.34)
	1.36	1.12	1.04	1.43	1.18	0.72	0.94	1.03
Age 02-09	(0.70-2.64)	(0.58-2.18)	(0.69-1.56)	(0.91–2.24)	(0.64–2.17)	(0.44 - 1.18)	(0.68–1.28)	(0.79 - 1.34)
	0.83	1.56	1.38	1.08	1.68 **	1.05	0.93	1.11
Education: MIDULE	(0.47–1.47)	(0.96-2.52)	(0.95–1.99)	(0.80 - 1.44)	(1.03-2.76)	(0.62 - 1.79)	(0.73–1.19)	(0.93 - 1.34)
	1.00	1.82 **	1.43 *	0.83	1.77	0.93	1.05	0.97
Education: HIGH	(0.54 - 1.84)	(1.08-3.04)	(0.98-2.09)	(0.58–1.20)	(0.75-4.18)	(0.46 - 1.85)	(0.83 - 1.33)	(0.79 - 1.20)
	0.67	1.64 **	1.16	1.10	1.02	1.20	1.06	0.96
2+ Curonic Diseases	(0.42 - 1.07)	(1.07-2.51)	(0.79 - 1.70)	(0.76–1.59)	(0.69–1.51)	(0.80 - 1.79)	(0.83–1.34)	(0.79 - 1.14)
1+ Limitation	1.62	1.01	1.19	1.14	1.53	1.37	1.39 ***	0.92
(NAGI, IADL, ADL)	(0.94-2.77)	(0.65-1.56)	(0.80 - 1.78)	(0.86 - 1.49)	(0.89–2.63)	(0.95 - 1.96)	(1.12–1.73)	(0.77–1.09)
3+ Depressive	0.59	1.21	1.01	0.94	1.26	0.99	1.16	1.41 ***
Symptoms	(0.30-1.16)	(0.75–1.96)	(0.72–1.39)	(0.74 - 1.20)	(0.73–2.17)	(0.70 - 1.39)	(0.89–1.52)	(1.17 - 1.70)
Self-rated Health	0.84	0.89	0.87	0.78	0.68	0.76	0.83	0.87
< Good <sup>c</sup>	(0.47–1.50)	(0.52–1.53)	(0.60 - 1.28)	(0.57-1.07)	(0.38-1.20)	(0.50 - 1.14)	(0.63–1.08)	(0.70–1.09)
No natural parents alive <sup>d</sup>								
At least 1 parent	1.85 **	1.51	1.74 ***	1.38 **	2.11 ***	1.42		
in < good health <sup>d</sup>	(1.12–3.03)	(0.95–2.42)	(1.21–2.51)	(1.00-1.91)	(1.27–3.49)	(0.95–2.13)	1.51 ***	1.97 ***
Parent(s) in	2.37 ***	1.90 **	1.50 *	1.74 ***	1.74	1.64 **	(1.22–1.88)	(1.66–2.35)
≥ good health <sup>d</sup>	(1.34-4.18)	(1.06–3.40)	(0.95–2.35)	(1.20–2.52)	(0.89–3.39)	(0.98–2.71)		
No Grandchildren								
	66.0	2.02 ***	1.97 ***	1.78 ***	2.77 ***	2.84 ***	1.61 ***	1.78 ***
At least 1 grandchid	(0.63–1.55)	(1.26–3.23)	(1.48–2.62)	(1.37–2.32)	(1.76-4.35)	(1.85-4.35)	(1.30 - 1.99)	(1.49–2.13)
	66.0	1.05	1.99 ***	1.04	2.74 ***	1.42	1.44 ***	1.24 ***
Marned	(0.64–1.52)	(0.70–1.55)	(1.32-3.01)	(0.80-1.36)	(1.45–5.20)	(0.95-2.13)	(1.11–1.87)	(1.04–1.48)
Number of Observations	514	541	826	950	734	954	3.197	3.414
				~~~	-			

Source: SHARE, 2004; ELSA, 2002; weighted data – own calculations. Reference categories: a) Age 50-54; b) Education: LOW; c) Self-Rated Health  $\geq$  good a) for ELSA, the categories used refer to 'parents dead' vs 'at least one parent alive': \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively. \*\*\*: significant 'parents categories

Male	Female				,
		Male	Female	Male	Female
0.97	0.63 **	0.66	0.82	0.38 ***	0.37 ***
(0.71–1.59)	(0.40 - 0.97)	(0.38–1.16)	(0.56–1.21)	(0.30 - 0.48)	(0.30 - 0.46)
0.46 ***	0.77	0.27 ***	0.64	0.20 ***	0.28 ***
(0.25-0.84)	(0.52-1.14)	(0.11 - 0.70)	(0.32 - 1.27)	(0.14 - 0.30)	(0.22 - 0.36)
0.65	0.83	0.51 **	0.79	0.49 **	0.51 ***
(0.31-1.35)	(0.49-1.41)	(0.11-0.97)	(0.32 - 1.27)	(0.27 - 0.90)	(0.37-0.71)
1.67	0.94	0.81	1.00	0.22 ***	0.40 ***
(0.63-4.44)	(0.51-1.71)	(0.42-1.58)	(0.55-1.83)	(0.11-0.43)	(0.28-0.57)
011	020	733	953	3.166	3.352
	0.77 0.46 *** 0.46 *** 0.25-0.84) 0.65 0.65 0.65 1.67 1.67	$\begin{array}{ccccccc} 0.00 & 0.00 \\ (0.71-1.59) & (0.40-0.97) \\ 0.46 *** & 0.77 \\ 0.46 *** & 0.77 \\ 0.25-0.84) & (0.52-1.14) \\ 0.65 & 0.83 \\ 0.65 & 0.83 \\ 0.31-1.35) & (0.49-1.41) \\ 1.67 & 0.94 \\ 1.67 & 0.94 \\ 0.63-4.44) & (0.51-1.71) \end{array}$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

fully adjusted Logistic Regressions, by country and sex. Selected results obtained when the variable indicating whether the respondent lived Table 7.16 Participation in Informal and Family-related Activities -Odds Ratios and 95% Confidence Intervals obtained from alone was considered

	DENN	AARK	FRA	NCE	IT/	ALY	ENGI	<b>DND</b>
	Male	Female	Male	Female	Male	Female	Male	Female
Living with others								
D	** 05.0	0.74	0.44 ***	0.97	1.19	0.99	0.42 ***	0.54 ***
Living Alone	(0.18-0.84)	(0.38–1.41)	(0.27-0.71)	(0.70–1.44)	(0.74 - 1.93)	(0.66–1.51)	(0.30-0.59)	(0.43 - 0.66)
Number of Observations	514	540	811	929	733	953	3,160	3,552
					•	•	11:11	Lands and and and and

Source: SHARE, 2004; ELSA, 2002; weighted data – own calculations. Among the other covariates, this specification included age, education and personal ill-health indicators.

Table 7.17 Participation in Informal and Family-related Activities –Odds Ratios and 95% Confidence Intervals obtained from fully adjusted

	DENN	IARK	FRA	NCE	ITA	VTY	ENGI	AND
	Male	Female	Male	Female	Male	Female	Male	Female
	0.79	0.75	0.57 ***	1.62 ***	0.64	1.21	0.79	1.06
Age: 55-59 <sup>a</sup>	(0.47–1.35)	(0.44–1.27)	(0.40 - 0.83)	(1.17–2.25)	(0.35 - 1.18)	(0.84 - 1.73)	(0.60 - 1.03)	(0.85-1.31)
	1.25	0.85	0.68	1.22	1.13	1.11	0.70 **	0.86
Age: 60-64 "	(0.69–2.27)	(0.45–1.64)	(0.41–1.12)	(0.74-2.01)	(0.65–1.99)	(0.66 - 1.87)	(0.52-0.95)	(0.67–1.12)
	1.27	0.76	0.72	1.31	1.13	0.64	0.64 **	0.82
Age 65-69 -	(0.62-2.62)	(0.36–1.61)	(0.40–1.29)	(0.80 - 2.14)	(0.59–2.17)	(0.39–1.05)	(0.45-0.92)	(0.62 - 1.09)
Education:	0.86	1.51	1.38 **	1.02	1.59	1.07	0.88	1.08
MIDDLE <sup>b</sup>	(0.48–1.52)	(0.93–2.46)	(0.95–2.00)	(0.75–1.38)	(0.94–2.69)	(0.60 - 1.92)	(0.68–1.13)	(0.89–1.30)
Education:	1.00	1.79 **	1.38 **	0.74	1.76	1.03	0.96	0.90
HIGH	(0.55–1.85)	(1.05–3.05)	(0.93 - 2.03)	(0.50 - 1.08)	(0.67-4.61)	(0.56 - 1.91)	(0.76–1.23)	(0.72–1.13)
	0.68	1.67 **	1.09	1.08	1.04	1.16	1.01	0.91
2+ Chronic Diseases	(0.43–1.07)	(1.09–2.57)	(0.75–1.58)	(0.75–1.55)	(0.69–1.55)	(0.79 - 1.69)	(0.79–1.29)	(0.76 - 1.10)
1+ Limitation	1.63	0.97	1.18	1.12	1.49	1.37	1.34 ***	0.89
(NAGL LADL, ADL)	(0.94-2.81)	(0.61–1.52)	(0.80-1.75)	(0.85–1.47)	(0.86-2.56)	(0.96–1.95)	(1.07–1.67)	(0.75 - 1.06)
3+ Depressive	0.58	1.23	1.05	96.0	1.32	1.02	1.15	1.43 ***
Symptoms	(0.30-1.15)	(0.75-1.99)	(0.75–1.46)	(0.74-1.23)	(0.74–2.35)	(0.72 - 1.45)	(0.87 - 1.51)	(1.18-1.72)
Salf-rated Health	0.83	0.78	0.86	0.77	0.72	0.73	0.74 **	0.83
< Cond <sup>c</sup>	(0.46–1.50)	(0.44–1.36)	(0.59–1.26)	(0.56-1.08)	(0.41 - 1.28)	(0.48 - 1.11)	(0.56-0.99)	(0.66-1.04)
14 natural narant	1 98 **	1 86 **	1.56 ***	1.58 ***	2.09 **	1.39 **	1.60 ***	2.03 ***
I · HALUTAL PAIVIT	(1.23–3.38)	(1.03–2.78)	(1.09–2.35)	(1.13 - 2.14)	(1.03-3.52)	(1.04 - 2.37)	(1.28 - 1.99)	(1.70-2.42)
	66.0	1.95 ***	1.96 ***	1.77 ***	2.82 ***	2.97 ***	1.59 ***	1.79 ***
1+ grandchild	(0.63-1.55)	(1.21–3.15)	(1.47–2.62)	(1.37–2.29)	(1.79-4.45)	(1.93 - 4.60)	(1.28 - 1.98)	(1.49-2.14)
	1.05	1.03	1.97 ***	1.03	2.62 ***	1.42	1.41 ***	1.21 ***
Married	(0.67–1.87)	(0.66–1.53)	(1.29-2.98)	(0.77–1.40)	(1.54–5.01)	(0.92 - 2.11)	(1.07 - 1.84)	(1.03 - 1.46)
	0.94	0.60	0.66	0.86	0.97	0.69	0.54 ***	0.60 ***
In paid work	(0.57-1.56)	(0.36–1.02)	(0.43-1.03)	(0.63-1.18)	(0.66–1.41)	(0.43 - 1.11)	(0.42 - 0.70)	(0.49 - 0.72)
Encaced in Formal	1.16	1.59 **	1.73 ***	1.74 ***	1.82 **	1.42	1.54 ***	1.42 ***
Activities <sup>d</sup>	(0.77–1.73)	(1.09–2.33)	(1.29–2.33)	(1.32–2.30)	(1.00–3.33)	(0.86–2.38)	(1.19-2.00)	(1.16-1.74)
			100	010	1.12	OEE	2 100	3 414
N Observations	514	540	824	949	40/	664	0,170	- 17.0
		-		Acceleration A co 51	VEA. b) Cduratio	I OW - Self	f_R ated Health >	· ocod· d) Note ti

hat the Source: SHARE, 2004; ELSA, 2002; weighted data – own calculations. Reference categories: a) Age 50-54; b) Education: LOW; c) Self-Kated Health  $\geq$  good; d) Note definition of Engagement in family and informal activities differs for SHARE and ELSA datasets. **\*\***, **\*\*\***: significant at the 0.05 and 0.01 levels, respectively.

Table 7.18 Participation in Informal and Family-related Activities –Odds Ratiosand 95% Confidence Intervals obtained from fully adjusted logistic Regressionswhen all SHARE countries are grouped altogether, by sex and considering variousmodel specifications

	With 'livi	ng alone'	With mar	ital status	Marital + activiti	
	Male	Female	Male	Female	Male	Female
A. 55 50 Å	0.62 ***	1.42 ***	0.65 ***	1.40 ***	0.62 ***	1.36 ***
Age: 55-59	(0.47-0.89)	(1.12–1.79)	(0.47–0.90)	(1.11–1.76)	(0.45–0.85)	(1.08–1.70)
A === 60 64 8	1.06	1.31	1.03	1.28	0.93	1.14
Age: 00-04	(0.76 - 1.47)	(0.97–1.76)	(0.74–1.43)	(0.95–1.74)	(0.66–1.29)	(0.81–1.60)
	1.13	0.99	1.10	0.98	0.96	0.87
Age 05-09	(0.79–1.63)	(0.71–1.39)	(0.77–1.58)	(0.71–1.36)	(0.63-1.45)	(0.63-1.23)
Education:	1.44 **	1.05	1.45 **	1.05	1.42 **	1.04
MIDDLE <sup>b</sup>	(1.04–1.98)	(0.81–1.37)	(1.05–2.00)	(0.81–1.36)	(1.02–1.98)	(0.78–1.35)
Education:	1.54 **	0.91	1.54 **	0.91	1.51	0.89
HIGH <sup>b</sup>	(1.01–2.34)	(0.66–1.26)	(1.03–2.31)	(0.65–1.26)	(0.97–2.32)	(0.65 - 1.21)
	1.37	1.25 **	1.36	1.24 **	1.34	1.22
I+ Limitation	(0.98–1.91)	(1.01–1.54)	(0.98–1.89)	(1.01–1.54)	(0.96–1.87)	(0.99–1.51)
2+ Chronic	1.03	1.17	1.03	1.17	1.01	1.16
Diseases	(0.79–1.34)	(0.91-1.50)	(0.79–1.35)	(0.91-1.51)	(0.77–1.31)	(0.90–1.47)
3+ Depressive	1.10	0.97	1.10	0.97	1.14	0.99
Symptoms	(0.80-1.51)	(0.79–1.19)	(0.80-1.51)	(0.80–1.19)	(0.82–1.58)	(0.81 - 1.21)
SRH	0.77	0.77 **	0.78	0.78 **	0.80	0.76 **
< Good '	(0.56–1.06)	(0.61-0.98)	(0.56-1.07)	(0.61-0.99)	(0.58–1.10)	(0.58-0.97)
	0.38 ***	0.79				
Living Alone	(0.25–0.57)	(0.57–1.10)				
Manufad			2.16 ***	1.22	2.12 ***	1.14
Married			(1.52-3.04)	(0.98–1.51)	(1.50–3.01)	(0.94–1.45)
1+ parent with			1.88 ***	1.36 ***	1.89 ***	1.40 ***
SRH <good d<="" th=""><th></th><th>}</th><th>(1.41-2.52)</th><th>(1.08-1.72)</th><th>(1.40-2.54)</th><th>(1.12-1.76)</th></good>		}	(1.41-2.52)	(1.08-1.72)	(1.40-2.54)	(1.12-1.76)
Parent with			1.64 **	1.67 ***	1.66 **	1.71 ***
SRH $\geq$ good <sup>d</sup>			(1.13-2.39)	(1.27–2.21)	(1.12-2.45)	(1.29-2.26)
			2.21 ***	2.25 ***	2.25 ***	2.28 ***
1+ grandeniid			(1.72–2.84)	(1.74-2.92)	(1.74–2.90)	(1.75–2.96)
In Daid work					0.84	0.75 **
In Paid work					(0.65-1.07)	(0.58-0.97)
Formally					1.69 ***	1.58 ***
Engaged					(1.30–2.21)	(1.21–2.05)
Enonce e	1.04	0.86	1.05	0.85	0.95	0.86
France	(0.76-1.43)	(0.67-1.09)	(0.76–1.45)	(0.67–1.08)	(0.69–1.31)	(0.67–1.09)
Donmark *	1.09	0.77	1.11	0.76	0.99	0.74
	(0.74–1.61)	(0.57-1.04)	(0.76-1.64)	(0.56-1.02)	(0.67–1.46)	(0.53-1.04)
<b>N</b> Observations	2,074	2,446	2,074	2,445	2,072	2,444

Source: SHARE, 2004; weighted data – own calculations. This table shows the fully adjusted logistic results for participation in informal activity only for SHARE countries when the latter are considered altogether and country dummy variables are taken into account. Analyses are repeated for each of the various model specifications.

Reference categories: a) Age 50-54; b) Education: LOW; c) SRH = good, very good, excellent; d) No natural parents alive; e) Italy. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively.

### 7.4 CONCLUSIONS

This chapter examined the baseline associations between socio-demographic and health characteristics of older people and engagement in paid work, formal and informal activities in Denmark, France, Italy and England in order to assess whether the characteristics associated with engagement were common across different countries.

It is clear that older people aged 50-69 contribute significantly to the well-being of other individuals as well as to the larger society. Using data from SHARE and ELSA, overall almost 40% of respondents aged 50-69 were in paid work, one quarter were engaged in formal activities, and one third provided regular help and care for family members, friends and neighbours. Nonetheless, almost one quarter of respondents in SHARE and one third in ELSA were not engaged in any of the above-mentioned activities.

As regards the *personal-level characteristics* associated with engagement in the three activities, some common patterns were found across different countries. Among the factors associated with the three forms of engagement, the age of the respondents was, in both SHARE and ELSA countries, consistently and significantly correlated only with paid work whereas there was no significant association between age and formal and informal engagement. Higher education was generally positively associated with both paid work and commitment in formal activities, confirming that the level of education is the most consistent predictor of civic participation (Wilson, 2000, van Ingen and Dekker, 2010).

The association between health and older people's engagement was less clear. Ill-health indicators were significantly negatively associated with work in all countries except among Italian men. The association with the other two forms of activities were weaker and differential depending on the country and the gender. Some of the positive associations between ill-health and engagement may be explained by the fact that activities such as providing help and caring for sick people could be potentially risk factors for health; it could also be possible, though, that people who were formally and informally engaged were those not in paid work for health-related reasons. Clearly, given the cross-sectional nature of these findings, it is not possible to determine whether

poor health was the cause or the consequence of being engaged in certain activities. Both mechanisms are equally plausible. Just as an example, being in paid work may have a favourable and positive impact on people's health but at the same time it might be less likely for ill people to start participating in work (Thomas et al., 2005).

Older people's *level of engagement* in paid work, formal, informal and family activities differed in each of the countries considered<sup>69</sup>, suggesting the importance of other country-specific factors in determining older people's level of engagement. Generally speaking higher proportions of 'active' older people were found in Denmark whereas Italy was characterized by lower rates of engagement. The proportion of Danish SHARE female respondents in paid work was more than double the Italian proportion; similarly, almost 60% of the Danish men were in paid work compared to 40% of the Italian male respondents. Danish respondents consistently also had participation rates in informal activities between two and three times higher than those of the Italians. Moreover, whereas in France, Italy and England the proportion of respondents who were engaged only in formal or informal activities was higher than that of respondents aged 50-69 combined work with at least one of the formal or informal activities.

Independent of the general level of participation in a country, we also found evidence for a complementary relationship between the formal and informal activities and therefore a possible accumulation of these two forms of engagement at the individuallevel. While, in our sample, paid work was negatively associated with the probability to engage in formal and informal activities, a strong positive correlation was found between active involvement in informal domain and the propensity to be engaged also in formal activities. Among the mechanisms proposed to explain these relationships, it is likely that older people engaged with informal and family-related activities might be more likely to get involved with larger social networks, including charitable, political or religious organizations, which may provide opportunities for engaging in formal activities. These findings confirm similar results reported in a number of studies (Burr et al., 2005, Choi et al., 2007, Hank and Stuck, 2008). Paid work, on the other hand, did

<sup>&</sup>lt;sup>69</sup> After controlling for socio-economic, demographic and health-related variables, differences between countries remained only when paid work and formal engagement were considered

not seem to compete with family and informal commitment (with the exception of ELSA respondents <sup>70</sup>). One possible interpretation is that, because of normative obligation or expectation, older people are available to give support to their family and friends regardless of their employment status (Caro et al., 2005).

Differences in the level of engagement suggest a connection between patterns of activity and policies such as -for paid work -the statutory retirement age, pat-time work opportunities, the social security benefits and the pension wealth (Pichler and Wallace, 2007, Hank and Stuck, 2008, Börsch-Supan et al., 2009). Similarly, the level of participation in formal activities seemed to be country-specific, suggesting that differences in the level of formal commitment may be mainly explained by the interaction between welfare regimes and the societal context, norms and values as proposed in several studies (Salamon and Sokolowski, 2001, Smith and Shen, 2002, Erlinghagen and Hank, 2006, Pichler and Wallace, 2007, Hank and Stuck, 2008, Hank, 2010, Hank and Erlinghagen, 2010). Conversely, family-related and informal participation was in all countries mainly associated with the presence of kin network and, once controlling for that, differences between countries become almost undetectable.

Longitudinal analyses may provide a better insight on the mutual relationship between participation in the various activities and health and whether maintaining an active engaged life can be beneficial for their health.

Chapter 9 investigates the longitudinal association between engagement at baseline and health at follow-up, trying to understand whether the various forms of activity contribute equally to health and its lower decline over time. Before introducing these longitudinal analyses, however, a few considerations about attritions are introduced in Chapter 8. This deals with attrition and introduces a model to better describe the characteristics of people who were not successfully interviewed at the second wave of the study.

<sup>&</sup>lt;sup>70</sup> The measure of outcome for ELSA respondents, however, is different and includes fewer activities.

# 8. ATTRITION IN ELSA AND SHARE: WHO IS MISSING AT WAVE 2

The main focus of this chapter is on attrition, i.e. respondents who dropped out of the survey between waves. After an initial introduction about missing data, some analyses on the occurrence of missing data are shown, focusing in particular on those factors hypothesised to be correlated with the outcomes of the study. Cross-tabulations (and relative chi-squared tests) comparing the groups of subjects with and without missing data at the second wave are presented (i.e. data collected at the first wave will be used to understand the nature of subsequent non-response). The influence of a selection of predictor variables on the occurrence of missing data at subsequent waves is also presented by means of a simple logistic regression analysis. In the latter case, the focus will not only merely be the significance of the variables but also the direction of the association with missingness since both are important issues in determining whether or not the pattern of missing data informative. The latter suggests that the pattern of missing data is not ignorable and has to be acknowledged when the influence of activity on health is analysed.

# 8.1 ATTRITION IN LONGITUDINAL STUDY. AN INTRODUCTION

## **8.1.1 POTENTIAL REASONS**

Sample attrition is a type of missingness which occurs when a responding unit in one wave of the panel drops out in a subsequent wave. This is almost an unavoidable problem (Ruspini, 2004, Twisk, 2007) which is often considered one of the most serious disadvantages of panel data (Fitzgerald et al., 1998, Trivellato, 1999) because of decreased precision of the estimates as the sample size decreases too and, most importantly, because non-response may not be random. As a result the sample may cease to be representative and estimates of the outcomes of interest may be biased (Uhrig, 2008).

Although preventive measures are adopted in order to minimise drop-outs, attrition might happen because of the *survey design* (such as the length of the interview, the topic of the survey, incentives and follow-up procedures), *individual and household characteristics* (such as gender, employment status, health as well as household location) and the *interaction between the interviewer and the respondents* (Lepkowski and Couper, 2002, Lynn et al., 2005, Nicoletti and Peracchi, 2005). For instance, Schröder (2008) showed how the interview length was associated (with mixed country-specific findings) with survey participation in SHARE Wave 2. In the Michigan Panel Study on Income Dynamics attrition was concentrated among lower socio-economic status respondents whereas marriage and living in an urban area were protective factors (Fitzgerald et al., 1998). Zamarro et al. (2008) found that people in poor health and with poor cognitive abilities were more likely to drop out. Several studies have shown that the interviewer characteristics as well as how they got along with the respondents were also important factors in determining people's attrition level (Groves and Couper, 1998, Schröder, 2008, Uhrig, 2008).

### 8.1.2 MISSINGNESS MECHANISMS

In the statistical literature, it is considered useful to distinguish the missing data mechanism (i.e. the relationship between missingness and the values of the variables) because most statistical methods depend on such mechanisms (Little and Rubin, 2002). Data are said to be missing completely at random (MCAR) if missingness does not depend on the (unobserved and observed) values of the data. Data are said to be missing at random (MAR) if missingness depends only on the observed data (or in other words, given the observed values, the missing data are random). Finally, if missingness depends on information that has not been recorded and this information also predicts the missing values the missing values, then the missing data mechanism is said to be not missing at random (NMAR). Although the distinction between MAR and NMAR is quite important since this affects methods, analyses and results, the observed data in a given dataset cannot be used to distinguish between MAR and NMAR mechanisms without additional untestable assumptions given that the missing data mechanism is in most

situations unknown because of the unavailability of the missing data (Kenward and Carpenter, 2007).

### 8.2 ATTRITION IN ELSA AND SHARE

Despite the impossibility of testing the missing mechanisms, it is nevertheless important to investigate whether missing data depend on earlier observations, i.e. to determine whether certain predictor variables (mainly the demographic and health status of the respondents) were related to the occurrence of missing data (Twisk, 2007). If not formally, at least these analyses of the attrition in each of the countries separately will inform us on whether the pattern of missing data was ignorable or informative and on whether attrition differed in different countries.

Because of the longitudinal nature of the study and the potential self-selectiveness of the sample, it is important to have an idea of the potential limitation of the findings. Attrition characteristics among both ELSA and SHARE respondents should help us determine whether and how the factors of interest are related to the occurrence of missing data. In particular, given that health and activity participation are mutually causally correlated, the longitudinal effect of each on the other might be overstated should characteristics of drop-outs not be accounted for. For instance, in fact, if unhealthier and less engaged people are more likely to have dropped out of the study, the longitudinal as well as cross-sectional wave-2 association between activity and health could be exaggerated.

### 8.3 ATTRITION RATES IN ELSA AND SHARE

The attrition rate was based on respondents with a completed interview in 2004-2005 (SHARE) and in 2002 (ELSA). Around a third of French and 30% of the Italian samples dropped-out of the survey between the first and the second wave, regardless of their gender. In contrast, 23% of Danish and less than 20% of English older people dropped out of the survey in the second wave (see Table 8.1). Interestingly, France experienced more attrition than the other countries under study; this could partly reflect the initially higher response rate achieved by SHARE in this country.

	Denmark % (N)	France % (N)	Italy % (N)	England % (N)
Participated	75.5	64.4	70.3	80.0
in wave 2	(832)	(1,286)	(1,249)	(6,049)
Drop out	23.0	34.5	28.7	19.0
(excluding dead)	(254)	(690)	(509)	(1,436)
Dead between	1.5	1.1	1.0	1.0(70)
waves	(16)	(22)	(18)	1.0(79)
Total	100	100	100	100
1 0181	(1,102)	(1,998)	(1,776)	(7,564)

Table 8.1 Attrition Rates for Wave 1 Respondents by country

This table shows the percentages (and number) of respondents from each country who did not take part in the second wave of SHARE and ELSA in 2006 and 2004 respectively. Source: SHARE 2006, ELSA 2004; unweighted data.

Given the very few number of people who died between the first and second wave, deaths were excluded in the analyses: not only they were not counted as attritions but also they were not considered as a specific category (Rendtel, 2002).

### **8.4 DETERMINANTS OF ATTRITION**

# 8.4.1 RESPONDENT CHARACTERISTICS AND DIFFERENTIAL NON-RESPONSE IN WAVE 2; BIVARIATE ANALYSES

Almost all the personal and household characteristics could potentially be considered among the factors associated with dropping out of the study. In this section, only a few important ones such as gender, age, employment status and participation in other activities, health status and type of house were considered. These variables are among the ones of interest in this study and it is therefore important to determine whether and how they are related to the occurrence of missing data.

Men were more likely to drop out of the study than women in all countries except Denmark, even though this relationship was significant<sup>71</sup> only in France (see Figure 8.1). This confirmed findings from previous studies where men dropped out more frequently than women (Lepkowski and Couper, 2002, Behr et al., 2005, Hawkes and Plewis, 2006), even though the effect of gender usually disappears when other factors are controlled for (Watson, 2003).

<sup>&</sup>lt;sup>71</sup> Bivariate associations were assessed with  $\chi^2$  tests.





Table 8.2 shows attrition rates by gender and **age-group** at baseline. The attrition rates did not seem to follow any particular trends except among Danish and English men where a reverse U-shaped pattern was detectable, with the highest attrition among those aged 55-59. The latter were also the only two subpopulations where the association between attrition rates and age-group was significant.

				8				
	Denr	nark	Fra	ance	Ita	ıly	Eng	land
	M **	F	M	F	Μ	F	M**	F
50-54	24.4	25.3	40.5	36.3	28.7	30.9	17.0	19.1
	(41/168)	(38/151)	(116/287)	(110/294)	(35/127)	(60/199)	(154/893)	(208/1,082)
55-59	28.1	26.9	34.3	35.8	38.1	24.3	21.4	19.6
	(43/147)	(44/171)	(97/278)	(112/322)	(81/226)	(65/275)	(213/1,006)	(222/1,153)
60-64	19.6	17.4	39.1	32.1	28.9	33.7	21.1	16.8
	(28/134)	(25/135)	(73/188)	(73/227)	(59/217)	(88/290)	(171/801)	(143/872)
65-69	11.4	25.8	34.0	27.0	30.6	33.0	19.9	19.6
	(10/88)	(25/92)	(56/175)	(53/205)	(53/193)	(68/231)	(153/783)	(172/894)
Total	22.2 (122/537)	23.9 (132/549)	37.1 (342/928)	33.4 (348/1,048)	32.1 (228/763)	30.1 (281/995)	19.7 (691/3,483)	18.8 (745/4,001)

Table 8.2 Attrition Rates for Wave 1 respondents by age-group, country and gender

This table shows the percentages and absolute number of respondents who were interviewed at wave 1 but not at wave 2 stratified by gender, country and age-group at the point of the first wave interview. Sources: SHARE 2006/4, ELSA 2004/2; weighted data. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively
Figure 8.2 shows the relationship between attrition and educational level of the respondents in all subpopulations. Among French women and English respondents there is a significant (at a 5% level) association between education and dropping out of the survey whereas in the other subpopulations, at least at a bivariate level, there is no significant relationship between attrition behaviour and education. Nonetheless, it is quite interesting to note that whereas in France, Denmark and England highly educated people are less likely to drop out of the study, in Italy the direction of the association is the opposite, although not significant. High education associated with less attrition in northern Europe and with more in southern Europe was also reported by Watson (2003).





Figure 8.3 shows the relationship between attrition and whether living in either a urban or a rural area. Zabel (1998), analysing attrition behaviour in two longitudinal surveys (the Panel Study of Income Dynamics and the Survey of Income and Program Participation) reported higher attrition among urban dwellers. However, in SHARE there was no significant urban/rural difference in attrition (but rates of attrition are generally higher among urban dwellers).



### Figure 8.3 Attrition Rates for Wave 1 Respondents by area of residence, gender and country

This figure shows the percentages of respondents who were not interviewed at wave 2 stratified by area of residence (rural or urban), gender and country at the baseline. Source: SHARE 2006/4; weighted data.

It was also found that respondents who were married at baseline were slightly less likely to drop out of the study as shown in previous research (Fitzgerald et al., 1998, 1998, Watson, 2003, Behr et al., 2005). This factor was protective among all subpopulations (significantly only among Italian male respondents), except English women (Table 8.3).

Table 8.3 Attrition Rates for Wave 1 respondents by a 'living alone' indicator, marital status and home ownership respectively, country and gender.

	Denr	nark	Fra	nce	Ita	ıly	Eng	land
	Μ	F	Μ	F	Μ	F	Μ	F
Not	25.7	24.0	40.5	32.8	45.2 ***	33.0	19.8	16.8 **
Married	(42/153)	(43/181)	(75/191)	(110/345)	(45/101)	(58/195)	(151/757)	(196/1,204)
14	20.4	23.4	34.8	32.3	28.8	28.7	19.6	19.6
Married	(78/381)	(89/368)	(249/714)	(225/688)	(183/662)	(222/798)	(540/2,725)	(549/2,797)

This table shows the percentages and absolute number of respondents who were interviewed at wave 1 but not at wave 2 tabulated by gender, country and whether the respondent was married at the point of the first wave interview. Sources: SHARE 2006/4, ELSA 2004/2; weighted data. \*\*, \*\*\*: significant at the 0.05 and 0.01 levels, respectively

Table 8.4 shows the association between attrition and health. Findings were mixed for French and Danish men. Among the other subpopulations, patterns were opposite: respondents with at least one function limitation, depressive symptoms, and who rated their health as poor or fair were less likely to be interviewed in the second wave among Danish women and English men whereas they were more likely to participate in wave 2 among Italians and French female respondents.

Similarly, SHARE respondents who reported having two or more chronic diseases at baseline were more likely to participate in the second wave of the survey (significantly among Italian women, French and Danish men) whereas among ELSA respondents there was no significant association.

				iti y und g	chuci			
	Den	mark	Fra	nce	Ita	aly	Eng	land
	М	F	Μ	F	Μ	F	M	F
Chronic	24.3 *	24.4	38.6 *	34.5	34.5	32.6 *	19.3	18.4
diseases	(99/394)	(92/382)	(263/684)	(253/731)	(173/550)	(188/614)	(492/2,533)	(495/2,729)
2+ Chronic	16.7	22.7	32.9	30.8	26.1	25.7	20.8	19.8
diseases	(23/143)	(40/167)	(79/244)	(95/317)	(55/213)	(93/381)	(199/950)	(250/1,272)
No	21.0	21.5	36.0	37.6 ***	37.0 ***	34.5 **	18.8	18.8
Limitations	(84/394)	(67/313)	(240/671)	(213/570)	(165/497)	(146/469)	(372/1,968)	(327/1,751)
1+	25.3	27.1	39.8	28.3	23.1	25.8	20.9	18.8
Limitation	(38/143)	(65/236)	(102/257)	(135/478)	(63/266)	(135/526)	(319/1,515)	(418/2,250)
Not	22.2	22.8	34.9	33.9	32.5	32.2 *	19.1 *	18.4
Depressed	(105/465)	(100/433)	(230/666)	(196/591)	(179/599)	(182/613)	(547/2,839)	(550/3,015)
Depressed	16.1	27.6	35.8	28.6	27.6	25.0	22.1	20.1
Depresseu	(11/64)	(30/111)	(66/184)	(116/400)	(43/156)	(92/373)	(144/644)	(195/986)
(SRH) =>	20.6	22.8	33.5 **	31.1 *	32.9	31.7	18.3 ***	17.4 ***
Good	(91/427)	(98/433)	(228/682)	(235/760)	(164/553)	(184/617)	(479/2,606)	(534/3,094)
SRH =	27.3	28.1	43.3	37.0	27.2	26.3	22.8	22.7
fair/poor	(29/107)	(34/116)	(90/212)	(98/269)	(58/146)	(95/376)	(189/829)	(195/877)

Table 8.4 Attrition Rates for Wave 1 respondents by selected health indicators, country and gender

Percentages and absolute number of respondents who were interviewed at wave 1 but not at wave 2 tabulated by gender, country and selected health characteristics (chronic diseases, functional limitation, depression and self-rated health) at the point of the first wave interview. Source: SHARE 2006/4, ELSA 2004/2; weighted data. \*, \*\*, \*\*\*: significant at the 0.10, 0.05 and 0.01 levels, respectively.

Finally, when considering the relationship between the employment status at baseline and attrition we found few statistically significant differences between those who were in paid work and those who were not. Even though other studies suggested that people in paid work were more likely to drop out (Watson, 2003, Behr et al., 2005), this association was significant only among French men. In England and among Danish women, on the contrary, the direction of the association is opposite (though not significant) (see Table 8.5 for full details).

ELSA and SHARE respondents (except Italian women) who were engaged in formal activities at baseline were more likely to be interviewed at the second wave and this association was significant among Danish and Italian men as well as French women and

the English subpopulations. This confirms the hypothesis that people engaged in community affairs also tend to have higher civic mindedness and are therefore more willing to keep participating in surveys (Groves et al., 2000, Lepkowski and Couper, 2002, Uhrig, 2008).

Similarly, people who were actively involved in informal and family-related activities at time 1 seemed to have lower attrition rates than those who were not engaged in such activities in almost all subpopulations (except Danish and English women), although this difference was significant only among French women, Danish men and ELSA participants.

	Den	mark	Fra	nce	Ita	aly	Eng	land
	M	F	Μ	F	M	F	M	F
No Paid	22.0	22.2	32.1 **	31.6	29.7	29.1	21.0	19.6
<u>Work</u>	(51/220)	(03/2/3)	(14//409)	(192/609)	(136/490)	(223/814)	(341/1,041)	(450/2,342)
In Paid	21.5	25.6	39.0	33.7	33.9	31.7	18.7	17.7
Work	(68/313)	<u>(67/274)</u>	(165/420)	(137/414)	(86/267)	(51/177)	(350/1,842)	(295/1,659)
No	25.0 **	24.6	36.1	34.2 ***	32.6 **	28.9	20.9 ***	20.1 ***
Formal	(84/332)	(85/341)	(223/616)	(260/763)	(202/655)	(240/865)	(628/2,984)	(640/3,218)
Formal	15.3	21.8	33.3	25.5	21.8	32.6	12.7	13.6
Acts	(32/198)	(44/205)	(79/247)	(60/243)	(18/100)	(35/124)	(63/499)	(105/783)
No Fam	23.8 *	22.9	36.4	35.5 ***	32.3	31.0	20.2	18.1 *
Inf Acts	(85/337)	(69/310)	(205/571)	(194/552)	(156/521)	(153/509)	(586/2,876)	(509/2,851)
Family/	17.4	24.9	33.3	28.5	28.6	27.7	17.3	20.6
Inf Acts	(32/194)	(62/238)	(100/298)	(130/459)	(64/234)	(122/480)	(105/607)	(236/1,150)

 Table 8.5 Attrition Rates for Wave 1 respondents by 'participation' in the three forms of activities, country and gender

This table shows the percentages and absolute number of respondents who were interviewed at wave 1 but not at wave 2 tabulated by gender, country and participation in the three activities under study (work, formal, informal and family-related) at the point of the first wave interview. Sources: SHARE 2006/4, ELSA 2004/2; weighted data. \*, \*\*, \*\*\*: significant at the 0.10, 0.05 and 0.01 levels, respectively

# **8.4.2** RESPONDENT CHARACTERISTICS AND DIFFERENTIAL NON-RESPONSE IN WAVE 2; MULTIVARIATE MODEL OF RESPONSE TO MAIN INTERVIEW

Table 8.6 shows participation in the second wave of the SHARE and ELSA survey modelled on a full range of information collected at baseline, including not only personal demographic and health characteristics but also the three variables indicating whether the respondent was actively engaged at baseline in the three forms of activities under study (i.e. work, formal participation, informal and family-related activities).

Age did not seem to be significantly associated with the likelihood of participating in the second wave. Among Danish men and French women, people aged 65-69 at baseline were more likely to stay in the study compared to the younger counterparts whereas in the other subpopulations a drop in the odds of participation was detectable as age increased(though such associations were not significant). Responders in wave 2 were more likely than non-responders to be highly educated in all countries (and significantly among English respondents, Danish and French women) except Italy.

As regards the health characteristics, fully adjusted logistic results confirmed most of the bivariate associations. Having reported at least 2 chronic diseases at baseline increased the odds of participation in all subpopulations (except English women), although significantly only among Danish men. Similarly, the direction of the association between 'depression' and participation in the second wave of the survey seemed to suggest that respondents who were depressed at baseline were less likely to drop out. As regards SRH, in most subpopulations those who rated their health as less than good were less likely to participate in the second wave (statistically significant among French and English respondents). Having at least one IADL, ADL or NAGI limitation was differentially associated with the participation in the second wave: French women and Italian men with baseline functional limitations were significantly more likely to be interviewed at follow-up whereas the opposite association held among French men.

Living in a big city was positively associated with the likelihood of dropping out in all SHARE countries, significantly among Danish women and French men. 'Living alone' was associated with participation in the second wave in a differential way depending on the country. Among Danish and English women, living alone at baseline was positively and significantly associated with participation in the second wave whereas among Italian respondents living alone were significantly more likely to drop out. No association was detectable in the other subgroups.

Finally, when engagement in the three forms of activities was considered, findings showed that paid work at baseline was significantly associated with attrition only among French men. In the other subpopulations, the direction of the association between paid work and attrition was positive among Italian respondents and Danish women whereas it was negative among Danish men, French women and ELSA respondents. People who were actively involved in formal activities at the time of the first interview were significantly more likely to participate also in the second, significantly among French women and Danish, English and Italian men. No association between formal engagement and attrition was present among Danish female and French male respondents. Being involved in informal and family-related activities at baseline also increased the odds of participating in the second wave interview in most subpopulations except Danish and English women.

(0.65-1.04) (0.77 - 1.16)(0.68-1.00) (0.72 - 1.16)(0.84-1.48) (0.68-1.24) 1.48-2.32) 1.34-2.19) (0.86 - 1.27)(0.85-1.33) (1.12-1.82) (0.83 - 1.28)(0.92 - 1.51)0.83 \*\* 1.85 \*\*\* 1.72 \*\*\* 1.42 \*\*\* Female 1.17 3.501 0.95 1.04 0.92 1.04 1.07 0.82 0.91 ENGLAND Health = good, very good, excellent d) Living in rural area, small town or big town e) Note that the definition of Engagement in family and informal activities differs for SHARE and ELSA datasets. **\*\***, **\*\*\***: significant at the 0.05 and 0.01 levels, respectively. Source: SHARE, 2004; ELSA, 2002; weighted data - own calculations. Reference categories: a) Age 50-54; b) Education: LOW; c) Self-Rated Regressions analysis including 'living alone' indicator and 'engagement' at baseline, by country and sex, (1.29-2.04) (0.73 - 1.18)(0.82-1.38) (0.87 - 1.38)(1.15-2.11) (0.92-1.51) (1.08-1.70) (0.87-1.36) (0.77-1.16) (0.75 - 1.23)(0.62 - 1.03)(0.59-1.04)(0.64-1.22) 1.36 \*\*\* 1.62 \*\*\* 1.56 \*\*\* 0.80 \* \* 61.0 3.258 Male 0.88 1.09 0.95 0.96 0.93 1.06 1.10 1.18 (0.79-2.19) (0.70-1.55) (0.41 - 1.39)0.55-1.34) 0.60-1.47) (0.82-1.48) (0.87 - 1.89)(0.83-1.84) (0.38-1.07) (0.44-1.59) (0.52-1.61) (0.61-1.61) (0.58-2.40)(0.78 - 1.60)Female 0.64 \* 0.76 0.86 0.94 0.84 1.12 1.04 1.10 1.32 0.92 0.99 1.28 1.24 973 1.17 ITALY (0.22-0.73) (0.39-1.21) (1.11-3.59) (0.28-1.17) (0.56 - 1.54)0.58-1.51) (0.66-1.63) (0.38 - 1.26)(0.50 - 1.35)0.37-1.78) (1.21-3.21) (0.75-1.87) (0.29-1.00)1.81-2.67) 0.40 \*\*\* \*\*\* 70.1 2.00 \*\* 0.54 \*\* 0.93 Male 0.69 0.69 1.04 1.46 1.18 0.93 747 0.57 0.82 0.82 (0.94-2.13) (0.45-0.86) (1.01-1.65) (0.70 - 1.64)(76.1-06.0) 1.34-3.48) 1.16-2.02) 1.20-2.45) (0.76 - 1.46)(1.11-2.03) (0.70 - 1.42)(0.85-1.59) (1.06-1.97) (0.51 - 1.27).72 \*\*\* 0.62 \*\*\* 2.16 \*\*\* 1.53 \*\*\* 1.50 \*\*\* 1.29 \*\* Female 1.42 \* 1.44 \*\* 1.33 1.00 1.16 1.08 1.05 0.81 969 FRANCE (0.88-1.54)(0.86-1.83) (0.58-1.06) (0.86-1.74) (0.38-0.85) (0.50-2.21)(0.60-1.01)(0.43-0.95) (0.68-1.49) (0.84-1.84) (0.46-2.48)(0.82-1.72) (0.64-1.61) (0.91 - 1.86)0.57 \*\*\* 0.78 \* 0.64 \*\* 0.78 \* 1.16 Male 1.23 1.05 825 1.30 1.19 1.00 1.24 1.28 1.01 (0.55-1.35) (0.63-1.59) (0.43 - 1.31)(0.36 - 1.13)(0.38-0.96) (0.46-1.41) (0.62-2.37) (0.27 - 1.30)(0.67 - 2.03)(1.05-3.73) (0.84-2.54) (0.44 - 1.26)(0.44-1.59) (0.98-2.85) 0.60 \*\* Female \*\* 86 1.67 \* 0.87 0.74 0.75 0.84 0.64 534 0.59 1.17 0.80 1.46 1.00 5 DENMARK (1.00-2.74) (1.07-2.89) (0.50-1.60)(0.78-2.87)1.25-8.44) (0.91 - 3.26)(0.83-3.51) (1.17-4.03) (0.34 - 1.21)(1.11-5.13)(0.36-1.39) (0.59-2.05) 0.59-1.66) (0.82 - 2.55)2.17 \*\* 1.65 \*\* 2.39 \*\* 3.25 \*\* 1.76 \*\* Male 0.64 1.50 1.10 0.99 1.45 524 0.89 0.71 & Informal Activities<sup>e</sup> **Education: MIDDLE** Education: HIGH<sup>b</sup> (NAGI, IADL, ADL) Formal Activities <sup>e</sup> 2+ Chronic diseases Living in big city<sup>d</sup> **Engaged in Family Observations (N)** Self-rated Health Age: 55-59 <sup>a</sup> Age: 60-64 \* Age: 65-69 \* 1+ Limitation **Living Alone Engaged** in Depressed Working < Good °

Table 8.6 Participation in wave 2 of the survey –Odds Ratios and 95% Confidence Intervals obtained from fully adjusted Logistic

# 8.5 CONCLUSIONS

Both bivariate and multivariate logistic regression analyses indicated that the pattern of attrition was not completely at random, although demographic, activity and health characteristics associated with attrition differed across countries and gender. Keeping in mind that the self-selection process varied in each subpopulation, in general highly educated respondents were more likely to stay in the study as were respondents engaged in formal and informal activities. Workers and respondents who rated their health as poor or fair were on the contrary more likely to drop out of the study. In summary, analyses suggested a differential non-response between waves 1 and 2.

When drawing conclusions on the mutual interplay between health and engagement as people age, caution is therefore warranted. The selectivity effect of the sample might over or underestimate certain findings: when commenting on results and discussing them, the self-selectivity of the sample should not be ignored. Attrition may weaken or jeopardize the generalisability of the findings and confound some of the reported associations. The following Chapter presents some longitudinal analyses in order to better understand whether engagement is beneficial for older people's health and also explores multiple imputations techniques to account for attrition in the surveys.

# 9. LONGITUDINAL ASSOCIATION BETWEEN BASELINE ENGAGEMENT, AND SELF-PERCEIVED HEALTH AND DEPRESSION AT FOLLOW-UP

Previous analyses (chapter 7) reported associations between the participation of older people in paid work, formal and informal activities and various socio-economic, demographic and health-related factors. These associations only considered one point in time.

The analyses to follow investigate longitudinal associations between the three forms of engagement at baseline and self-rated health (SRH) and depressive symptoms (depression) of respondents two years later. Using the first two waves of both ELSA and SHARE, I will investigate whether engagement in paid work, formal and informal activities are associated with health after two years (controlling for baseline measures of health) and whether such associations differ between countries and by gender.

Additionally, the association between changes or continuity over time in engagement level and health outcomes will be investigated. This might provide some insight into the mutual relationship between health indicators and different forms of engagement, even though causality between level of participation over time and health two years later cannot be demonstrated. Given the contemporaneous nature of these measurements, participants' health could experience a lower decline because they started a new activity but it is also plausible to assume that older people whose health perception improved decide to start engaging in a new activity.

The following sections will present the longitudinal outcome measurements. Also some descriptive longitudinal tables will be introduced to show changes over time in both health and in different dimensions of engagement. Results obtained from multivariate logistic and ordinal regressions, will be then presented separately for men and women.

Finally, multiple imputations will be explored in order to account for attrition in the surveys.

# 9.1 HEALTH AND ITS DETERMINANTS

Two health measurements were considered as outcomes of the longitudinal analyses: self-rated health and depressive symptomatology<sup>72</sup>.

#### 9.1.1 SELF-RATED HEALTH

**Self-rated health (SRH)** is a single-item measure in which individuals are asked to rate their health on a 5-point ordinal scale (excellent, very good, good, fair, or poor). Such an apparently simple question has the potential to portray the complex concept of 'ageing well' without merely relying on chronic conditions or functional limitations (Strawbridge and Wallhagen, 1999, Strawbridge et al., 2002). The SRH is a validated global measure of general health: even when physical health conditions are controlled for, SRH predicts outcomes such as quality of life, health care utilization and mortality (Kaplan and Camacho, 1983, Connelly et al., 1989, Wolinsky and Johnson, 1992, McCallum et al., 1994, Idler and Benyamini, 1997, Maier and Smith, 1999).

Longitudinal studies which have explored trajectories of changes in perceived health over time showed that multiple patterns of changes in subjective health exist, including substantial improvement as well as decline (Kim et al., 1997, Han, 2002). Previous studies suggest that physical and mental health conditions (Goldstein et al., 1984, Rodin and McAvay, 1992, Han, 2002, Choi, 2003, Jang et al., 2009), and health-related behaviours (Guthrie et al., 1997, Leinonen et al., 2001, Wilson et al., 2007) are strongly associated with changes in self-rated health, even though persons with equivalent levels of physical conditions do not necessarily rate their perceived health equally (Kahana et al., 1995, Jang et al., 2009). Similarly, socio-demographic and economic characteristics such as age and education are known influential mediating factors in a change in self-reported health (Choi, 2003, Buckley et al., 2004).

<sup>&</sup>lt;sup>72</sup> Longitudinal analyses were restricted to depression and SRH only because very few changes occurred between baseline and follow-up with regards to chronic diseases and functional limitations.

#### 9.1.2 DEPRESSION

The WHO's World Health Report (WHO, 2001c) estimated that by 2020 **depression** is expected to be the second highest ranking cause of disease in the developed world. Depression, even after adjusting for socio-demographic characteristics, is associated with increased coronary artery diseases and cardiovascular death and with worsened quality of life (Angermeyer et al., 2002, Black et al., 2003, Rumsfeld et al., 2005, Rapp et al., 2008, Nabi et al., 2010, Celano and Huffman, 2011), even though some researchers suggest that such associations are no longer significant after controlling for chronic health conditions (Everson-Rose et al., 2004, St John and Montgomery, 2009).

Factors including loneliness, stressful life events, disruptions in family functioning, poor nutrition, low socioeconomic status and adverse health behaviours (such as alcohol use disorders) are strongly correlated with depression (Averina et al., 2005, Rosenquist et al., 2011), even though such associations do not necessarily indicate the direction of any causal relationship.

# 9.2 METHODS

#### 9.2.1 SAMPLE

Results in this chapter are based on longitudinal analyses of data from the first and second wave of the SHARE and ELSA surveys. Initial longitudinal analyses were undertaken on records with complete data on the variables included in the analyses and were restricted to respondents aged 50-69 at wave 1 who were successfully re-interviewed at wave 2, with participation rates ranging from 80% (England) to 64% (France).

# 9.2.2 MEASURES

#### **Outcome Measurements**

This study investigates whether engagement in paid work, formal and informal activities at baseline is associated with SRH and depression at follow-up (controlling for baseline values) and with changes in status over a 48-month period.

When self-rated health at follow-up is considered, there are various ways to assess whether there has been a change in the perception of health. The SHARE questionnaire asks respondents whether they think that, compared to the previous wave, their health is better, about the same or worse. Most respondents answered that their health was about the same as in the previous wave, with women declaring a worsening of health more frequently than men in all countries. Very few people, on the contrary, declared that their health was better (see Table 9.A.1 in appendix). Such a question, though, is not present in the ELSA questionnaire. Moreover, for the SHARE respondents there is no association between the self-assessed change of health (SRHC) and the direct comparison of SRH scores at the two time points (data not shown). The absence of relationship between these two measurements of change in health might be due to different criteria used in the assessment. People adapt to changes in their health and therefore might assess their health as better or the same despite, for instance, increasing functional health limitations or diseases. SRHC, on the contrary, is assessed based on intra-individual comparison: less positive scores indicate that respondents are realistic about health changes over time (Leinonen et al., 2001).

Considering actual changes in the SRH from baseline to follow-up (calculated by subtracting SRH at baseline from SRH at follow-up) would capture the range of possible changes and their magnitude, with positive scores indicating worsening of perceived health, negative indicating improvement in health and zero implying no change. Unfortunately, this approach creates (particularly for high and small values) several cells which are extremely small or empty when cross-tabulated with predictors. Such a situation creates difficulties in fitting the models (Scott Long, 1997, Agresti, 2002). Similarly, although some authors would consider this outcome variable as continuous (Jang et al., 2009), it is not recommended to use such approach as the outcome is truly multinomial and ordered. Switching to a linear regression would make the model theoretically incorrect.

Table 9.1 considers improvement/decline in health between the two surveys as an increase or a decrease of *one point or more* on the five point scale (Excellent, very good, good, fair, poor). Changes in health depend on the starting point in most countries: those who rated their health as very good at time 1 tend to declare a worsened health at time 2 particularly among French and Italian respondents, whereas the majority of Danish and English participants reported the same health. As health decreases, the proportion of

people reporting either the same health as before or an improvement increases. Considering point changes, though, has the disadvantage of categorising into the same group people who dropped their assessment by any point (with decline in health referring to both the decrease from 'excellent' to 'very good' as well as from 'excellent' to 'poor'). This fails to capture the magnitude of change and suffers from 'floor and ceiling effects' which render the detection of changes at follow-up assessment more difficult for those who assessed their health as 'excellent' or 'poor' at baseline (Uttl, 2005, Wang et al., 2009).

In this study, the variables capturing changes over time were constructed as 4 ordinal category indicators. Changes in SRH were categorised as follows: 'improvement' if respondents whose SRH at baseline was equal to poor/fair at follow-up was greater than or equal to good; 'decline' for those with initial good or better health who rated their health as poor/fair at follow-up; two no-change groups denoting respectively those whose SRH was "greater or equal to good" at both time points and those whose SRH was "less than good" at both time points (i.e. their SRH changed from poor to fair or vice-versa or stayed equal to fair or poor)<sup>73</sup>. Such an outcome has the advantage of containing information on the relative rank order of changes in health. Of course, this scale is not exempt from criticism. In particular, this indicator assumes that certain changes (such as from poor to fair or vice versa) should be considered as negligible (mirroring the general dichotomisation between those in more than good and those in less than good health). Moreover, because of this distinction, the four categories of such indicator do not necessarily reflect same-distance changes: both a respondent who selfassessed their health as 'excellent' at baseline and as 'poor' at follow-up and a respondent who answered 'good' and 'fair' respectively would be equally considered as having worsened health. Despite the fact that this indicator does not reflect interval changes, this variable still captures an ordinal scale of possible changes in health which are ordered in line with no ambiguities and are meaningful for comparison. Moreover, this approach partly avoids the ceiling and flooring effects since, for instance, a change from 'excellent' to 'very good' health would not be defined as a decrease in health.

<sup>&</sup>lt;sup>73</sup> This indicator was coded with increasing numbers from 'decline' to 'improvement'. This means that in ordinal regressions, positive log-odds represent a change for the better, whereas negative values represent a deterioration of respondents' SRH.

Table 9.1 Percentage distribution of respondents by self reported health at baseline and change in self reported health at

follow-up, by sex and country

,008 l,041 859 838 168 467 514 163 421 Z 501 Better 17.0 (146) 254) 30.3 15.6 (160) (330) 36.7 (171)43.4 32.4 35.8 (189) 53.7 (72) (88) ł ł England Same 46.7 (196) 46.6 (403)47.5 (402) (226)56.6 46.6 (478) 48.2 (239) (498) 49.2 (251) 47.7 46.3 (96) 47.1 (75) Worse 53.3 (225) (310) 36.4 182) 36.7 (370) 22.3 15.2 20.5 213) 53.4 262) 15.0 (02) (74) ł ł 238 114 102 287 225 93 32 58 44 56 Z Better 19.8 6.69 (45) 54.7 17.0 (50) 23.4 (39) (15) (38) 33.1 9.9 (57) 8.3 8 6 ł ł Italy Same 31.4 25.0 61.5 (143) 30.5 50.8 (215) (130) 45.3 62.0 (26)(11) (55) (32) (18) 49.1 (11) 29.1 13) 30.1 Worse 68.6 (41) 66.7 17.8 70.9 59.6 23.2 (89) 14.6 (59) 18.7 (50) (20)(61) (38) ł ł Wave 2 113 273 333 144 22 Z 73 22 35 29 Е Better 19.8 36.9 63.5 24.8 (20)17.7 (46) (34) 61.7 (22) 6.4 12.7 (42) (36) ł ł France Same 43.9 52.4 61.5 (172) 36.5 28.6 65.2 (215)(32) 25.7 39.2 38.3 (31) (48) (13)(21)(49) 61.1 (88) Worse 54.5 20.8 (55) 56.1 (41) (62) 10.7 71.4 54.4 (65) 22.2 (76) (01) (20) **I4.1** (30)ł ł 106 139 30 8 0 8 Z 8 94 55 27 Better 16.6 36.0 (36) 35.6 17.3 (24) 39.4 31.0 55.5 (22) (21) (37) (17)(15) 28.1 ł 3 ł Denmark Same 37.6 49.9 45.5 71.9 52.3 (58) (38) 48.0 43.9 41.4 61.8 44.5 (55) (30) (13) ([9]) (39) (34) (4) (12) Worse 37.9 26.4 (27) 47.7 14.5 38.8 (54) (15) (50) 52.0 (53) 19.2 (18) **(4)** 6 ł ł Excellent Excellent Wave 1 Very Good Good Poor Very Good Good Fair Fair Poor **WEN** MOWEN

Sources: SHARE 2004/06 and ELSA 2002/04. Weighted data; own calculations. (18) (III)

Table 9.2 shows the distribution of changes in SRH. Most people self-assessed their health in the same category as in the baseline in most countries, although between 6% and 9% of respondents improved the assessment of their health over the years whereas between 10% and 16% of those successfully interviewed at the second wave reported a decline in health. These trends are in line with other studies which show that, on average, health deteriorates slowly over time and improvements in health are possible in some people (Kim et al., 1997, Han, 2002, Sacker et al., 2011). This table also confirms the initial cross-sectional differences in health between countries: partly given their higher baseline levels of SRH, Danish and English respondents tend to have also better SRH over time, with almost 70% of respondents assessing their health as at least good in both waves. Within-country gender differences in the distribution of health changes are tiny in England and Denmark. Differences between male and female Italian participants are, on the contrary, striking: 59% of men reported good health in both waves compared to less than 45% among women.

	Den	mark	Fra	ance	It	aly	Eng	land
	Male	Female	Male	Female	Male	Female	Male	Female
	% (N)	% (N)	% (N)	% (N)	% (N)	% (N)	% (N)	% (N)
Dottor	7.0	5.8	7.2	5.9	7.5	8.9	7.2	6.3
Detter	(29)	(24)	(42)	(41)	(40)	(64)	(195)	(209)
	70.6	68.5	66.9	62.0	58.9	44.7	67.0	69.3
Still >= good	(293)	(286)	(392)	(434)	(315)	(319)	(1,861)	(2,242)
Still <- good	12.8	14.9	14.5	18.9	19.8	30.4	16.3	14.5
Still <- good	(53)	(62)	(85)	(132)	(106)	(217)	(440)	(468)
Woreo	9.6	10.8	11.5	13.3	13.8	16.0	9.5	9.6
worse	(40)	(45)	(67)	(93)	(74)	(114)	(257)	(308)
Tot (N)	415	417	586	700	535	714	2,753	3,227

Table 9.2 Distribution of changes in SRH between wave 1 and 2, by sex & country

Sources: SHARE 2004/06 and ELSA 2002/04. Weighted data

Similarly, for depression 4 categories were considered (with categories equal to 'no longer depressed', 'not depressed at either wave', 'became depressed'; 'depressed at both times')<sup>74</sup>. Differences in changes of depression status between gender and countries were striking (see Table 9.3). Whereas among men, roughly 10% of those who were not

<sup>&</sup>lt;sup>74</sup> Also in this case, the variable was coded so that negative log-odds ratios in the ordinal regressions mean that the respondents has decreased the level of their mental health and has become depressed, whereas positive log odd ratios imply a positive change, i.e. the respondent no longer depressed.

depressed at baseline declared depressive symptoms at the second interview, among women the percentages were higher particularly in France and Italy (20% and 25% respectively). Between 50% and 65% of women continued to declare depression symptoms after two years whereas among men percentages ranged between 31% (Denmark) and 50% (Italy and England).

					~		- ••• u.						
		De	enmarl	κ.	F	France			Italy		]	England	
	Wave 2 Wave 1	Not Depressed	Depressed	N	Not Depressed	Depressed	N	Not Depressed	Depressed	N	Not Depressed	Depressed	N
EN	Not Depressed	89.3 (318)	10.7 (38)	356	91.3 (382)	<b>8.7</b> (37)	419	88.9 (366)	11.1 (51)	417	89.4 (2,057)	10.4 (235)	2,292
W	Depressed	68.7 (36)	31.3 (16)	52	63.5 (71)	36.5 (43)	114	46.4 (50)	53.6 (61)	111	48.2 (246)	51.8 (254)	500
EN	Not Depressed	87.9 (290)	12.1 (40)	330	7 <b>8.8</b> (307)	21.2 (82)	389	74.2 (314)	25.8 (114)	428	85.5 (2,107)	14.5 (358)	2,465
MOM	Depressed	51.3 (41)	48.7 (39)	80	34.4 (94)	65.7 (179)	279	39.2 (110)	60.8 (171)	281	46.5 (371)	53.5 (420)	791

Table 9.3 Percentage distribution of changes in 'depression' between wave 1 and 2,by sex and country

Sources: SHARE 2004/06 and ELSA 2002/04. Weighted data

# Co-variates

Socio-demographic and economic characteristics are factors which might be associated with changes in both self-reported health and depression. Among the variables included in the multivariate analyses, time-fixed variables were age, whether the respondent lived in a rent/mortgage free home, education and whether the respondent lived alone, all measured at baseline. It was decided not to consider changes in household composition because the number of people who got divorced or became widowed over the two-year time was very small (less than 1.5% in all countries either got divorced or lost their spouses; similarly less than 2% on average were living alone at follow-up compared to wave 1). Similarly, the only health-related variable included in the analyses was the one accounting for any limitations (IADLs, ADLs or NAGI) at time 1 to avoid the potential risk of over-fitting the model. Besides, controlling for changes in limitation over time

could cause over-adjustment bias (Schisterman et al., 2009), since it is unclear how changes in health status and changes in activity level mutually affect each other. Changes in limitations might be on the causal pathway between exposure (engagement level over time) and the outcome variables and as a consequence the effect of the exposure on the outcome could be estimated with bias.

Among the explanatory variables, the main interest of this study focused on the association between engagement at baseline (paid work, participation in formal activities, participation in formal and family-related activities) and health and changes in health at follow-up after two years. For each activity, a new variable capturing changes in level of participation was created: those who were active at baseline but no longer at wave 2 were assumed to have given up the respective activity whereas people who were inactive at time 1 and then active at time 2 were assumed to have started an activity.

Tables 9.4–9.6 show some descriptive statistics about changes in level of participation in the various activities over time. Between 20% and 30% of both male and female respondents who were in **paid work** at baseline were no longer working at the time of the second interview (see Table 9.4). The proportions were slightly higher in Italy compared to the other three countries, particularly among women. The transition from no work into work was rare in most countries: Danish men were the only exceptions with more than 10% of non-workers in 2004 working in 2006 (though that only means 17 people switching category). For this reason, it was decided to just consider three categories to capture changes in paid work: 'out of paid work at both waves', 'out of paid work for less than 2 years' and 'in paid work at time 2' (regardless of status at time 1).

					anu 2,	UY SUA	anu (	Junti					
		D	enmark	K	F	rance			Italy		]	England	
	Wave 2 Wave 1	Not in Paid Work	In paid work	N	Not in Paid Work	In paid work	N	Not in Paid Work	In paid work	N	Not in Paid Work	In paid work	N
EN I	Not in Paid work	88.9 (152)	11.1 (17)	169	97.0 (312)	3.0 (10)	322	95.2 (343)	<b>4.8</b> (11)	354	92.9 (1,123)	7.1 (87)	1,300
IW	In Paid work	24.1 (63)	75.9 (182)	245	26.4 (61)	73.6 (194)	255	29.8 (60)	70.2 (121)	181	17.3 (270)	82.7 (1,222)	1,492
AEN	Not in Paid work	95.7 (201)	4.3 (9)	210	96.5 (403)	3.5 (14)	417	98.0 (581)	2.0 (10)	591	93.9 (1.794)	6.1 (108)	1,892
NOW	In Paid work	22.2 (46)	77.8 (161)	207	21.2 (61)	78.8 (215)	276	32.8 (40)	67.2 (83)	123	23.5 (328)	76.5 (1,036)	1,364

Table 9.4 Percentage distribution of changes in 'paid work status' between wave 1and 2, by sex and country

Sources: SHARE 2004/06 and ELSA 2002/04. Weighted data

In all four countries under study, continuity in formal engagement was noticeable, particularly among SHARE respondents. This may be because the measure of engagement I used in the SHARE analyses was based on weekly participation whereas the ELSA measure refers to monthly participation. In France and Italy, more than 60% of both men and women who engaged in formal activities at baseline were still engaged after two years. Proportions of people continuing engagement in formal activities were slightly higher among Danish respondents (70% and 77% among men and women respectively) whereas figures were slightly lower among English respondents (though the ELSA indicator considered fewer activities than the SHARE one and limited the frequency of engagement to the month prior the interview). Between 10% (among Italian and English respondents) and more than 30% (in Denmark) of those who were not formally engaged at baseline became 'active' participants by the time of the second questionnaire (see Table 9.5 for more details). Changes in participation in formal activities were not significantly associated with changes in work status (p value = 0.44): the proportion of women who started to be formally engaged by time 2 was slightly higher among those who no longer were in paid work compared to those who stayed in the same work category across waves, though not significantly (results not shown).

		D	enmark		F	rance			Italy		I	England	
	Wave 2 Wave 1	NO Formal	Formal engagement	N	NO Formal	Formal engagement	N	NO Formal	Formal engagement	N	NO Formal	Formal engagement	N
Z	No Formal	70.8 (171)	29.2 (72)	243	85.9 (321)	14.1 (55)	376	90.6 (403)	9.4 (46)	449	88.8 (2,086)	11.2 (270)	2,356
W	Formal	29.4 (47)	70.6 (115)	162	37.0 (64)	63.0 (102)	166	37.4 (34)	62.6 (48)	82	50.6 (213)	49.4 (223)	436
AEN	No Formal	61.0 (152)	39.0 (97)	249	81.7 (400)	18.3 (92)	492	89.3 (559)	10.7 (65)	624	87.1 (2,236)	12.9 (342)	2,578
NOW	Formal	22.4 (36)	77.6 (125)	161	39.8 (71)	60.2 (107)	178	35.9 (32)	64.1 (57)	89	43.8 (291)	56.2 (387)	678

 Table 9.5 Percentage distribution of changes in 'formal engagement' between wave

 1 and 2, by sex and country, by sex and country

Sources: SHARE 2004/06 and ELSA 2002/04. Weighted data

As regards **informal and family-related participation**, continuity in engagement was generally higher among women. Among SHARE male respondents, only half of the sample continued engagement whereas women's continuity peaked at 75% among Italian respondents. In all SHARE countries, both among men and women, a quarter of people who were not engaged at all with these activities at baseline became 'active' at time 2. As for ELSA respondents, only a third of male respondents and about 40% of women remained engaged in informal and family-related activities: such a different pattern might be due to the fact that the indicator used to account for informal activities was slightly different compared to the one used among SHARE respondents: there was no account of frequency of commitment, and activities such as 'looking after grandchildren' were not asked about in the ELSA questionnaire.

		Ι	Denmar	<u>k</u>		France			Italy		I	England	
	Wave 2 Wave 1	Not informal engaged	Informally engaged	N	Not informal engaged	Informally engaged	N	Not informal engaged	Informally engaged	N	Not informal engaged	Informally engaged	N
Z	No Informal	73.9 (184)	26.1 (65)	249	76.5 (265)	23.5 (83)	348	75.7 (270)	24.2 (93)	363	93.8 (2,143)	6.2 (147)	2,290
Ī	Informal	49.5 (80)	50.5 (81)	161	45.8 (91)	54.2 (105)	196	44.9 (71)	55.1 (98)	169	68.8 (345)	31.2 (157)	502
_		69.0	22.0		72.2	267	·	70.7	20.2	r	0( 0	12.0	
IEN	NO Informal	(162)	(76)	238	(258)	(93)	351	(248)	(107)	355	80.2 (2,020)	(322)	2,342
NON	Informal	38.5 (67)	61.5 (107)	174	36.6 (120)	63.4 (205)	325	25.2 (88)	75.8 (270)	358	59.4 (543)	40.6 (371)	914

 Table 9.6 Distribution of changes in 'informal and family-related engagement'

 between wave 1 and 2, by sex and country, by sex and country

Sources: SHARE 2004/06 and ELSA 2002/04. Weighted data

#### 9.2.3 STATISTICAL MODELLING

For each of the four outcomes, either binary logistic regression or ordinal multinomial logistic regression methods were used. The former was used when the response variable was dichotomous (as in the case of depression at follow up controlling for baseline) whereas the latter for outcome variables presented in ordered categories (such as SRH at follow-up or health changes over time). In both cases, the influence (i.e. the probabilistic relationship) of the various explanatory factors on the outcome was examined. Both models are widely used in epidemiological studies which present the response variable in the form of categories, either dichotomous or ordinal (Anderson and Philips, 1981, Ananth and Kleinbaum, 1997).

As regards the independent variables, baseline socio-economic and demographic characteristics of the respondents were considered for each outcome variable. Additionally, functional limitations at baseline were included. In this study, the factors of major interest were the participation of older people in the various forms of engagement at baseline; nonetheless, changes in engagement levels over time were also considered.

The above-mentioned analyses were initially restricted to participants with complete data on the variables examined. Given that "complete record" analyses could be both biased and inefficient, multiple imputations were proposed as a remedy to deal with missing data (Klebanoff and Cole, 2008). In particular, multiple imputations (MI) were explored under the Missing At Random (MAR) assumption (i.e. missingness depends on variables fully observed in the data set) (Little and Rubin, 2002). In this study, though, there may be reason to believe that missingness was related to the missing data themselves. Some sensitivity analyses to the MAR hypothesis were therefore performed.

#### 9.2.3.1 A BRIEF OVERVIEW OF MULTIPLE IMPUTATION

The general idea of MI is to create a collection of datasets (usually between 5 and 20) where missing values in the original data are replaced by randomly imputed ones. The imputations are carried out so that the overall joint distribution of all variables involved is maintained. Multiple sets are created, instead of just one, in order to reproduce the variability in the original –but missing –data. Each dataset is then analysed as if there were no missing values, with the results summarised into a single summary set of findings, with standard errors that incorporate the variability in results between the imputed data sets. This method has the advantage of being robust to violations of non-normality of the variables used in the analysis (Roth, 1994, Little and Rubin, 2002, Wayman, 2003).

The initial step of MI is the definition of an imputation model. This must include all variables that are thought to influence the missing mechanism plus all variables to be considered in the analyses. Among the various methods for multiple imputation, this research study used chained equations (MICE), which use a series of conditional distributions rather than a joint distribution to impute missing data (Royston, 2005). The idea of such an algorithm is to impute, in each chained equation cycle, each missing value in each variable on the basis of a predictive distribution derived from a regression on all other variables in the imputation model. With each iteration the simulations are updated for a certain number of 'cycles' until the imputed values converge and a final imputed dataset is created. A number of authors suggest that such a method works well

in epidemiological settings (van Buuren, 2007, Lee and Carlin, 2010). Moreover, its availability in Stata renders it computationally simple to implement.

In this research study, the two main variables imputed were SRH and depression at follow-up. In particular, the imputation models used in this section included all variables found to be related to the probability of missing data on the outcomes of interest. For each of the outcome variables, a separate imputation model was used. Also, the imputation was carried out separately by country and gender, given that the analyses were carried out separately for men and women in each country. Among the variables included for each of the imputations models, I considered the following baseline variables: age-group (as a categorical variable), ADL and IADL limitations, SRH, depression, chronic disease, marital status, whether the respondent lived alone or not, engagement in formal and informal activities, employment status and whether the respondent lived in a rent or mortgage-free house. Given the large amount of missing data, the chained equation process was continued for 20 cycles and 200 imputed datasets were created in order to reduce the impact of the random sampling in multiple imputation procedures. The distributions of observed and imputed data were compared; no implausible differences were found. The results of analyses on each individual data set were combined using Rubin's rules (Little and Rubin, 2002). All analyses were performed using Stata, version 11 (Stata Corp, 2009).

#### 9.2.3.2 SENSITIVITY ANALYSES

The above-described method assumed that loss of cases over time (incomplete data) was missing at random. This means that, controlling for baseline fully-observed characteristics, the observed distribution of depression and SRH at follow-up were treated as reference distributions which could be used to impute all the missing values due to attrition. Potentially, though, the rate of depression and less than good SRH among subjects who were lost to follow-up may be different from the rates of respondents at follow-up. Missingness of those subjects who dropped out of the study might depend on the unobserved SRH and depression status themselves; i.e. people who dropped out are potentially missing not at random (MNAR). In such instances, it was decided to carry out some sensitivity analyses so that the posterior distribution of the

rate of depression and less than good SRH for drop-outs was artificially increased as compared with the reference distributions. Pattern mixture models (Daniels M. J. and Hogan, 2008) were used to assess whether and how much various plausible 'arbitrary' assumptions about the missing data mechanism affect the results. Such an approach is quite simple but flexible and easy to interpret

In this research study, given that no prior information was available, it was decided to test the robustness of the results by doing successive analyses assuming that among drop-outs the proportion of depressed and people who self-rated their health as poor or fair was increased by 20% and 33%. This way, even controlling for baseline characteristics, those who were lost to follow-up had lower SRH and were more depressed than respondents.

# **9.3** RESULTS FROM THE MULTIVARIATE ANALYSES FOR THE "COMPLETE RECORD"

# 9.3.1 SELF-RATED HEALTH AT FOLLOW UP CONTROLLING FOR BASELINE SELF-RATED HEALTH

Table 9.7 shows the results from an ordinal logistic regression model which investigates how SRH at follow-up was associated with baseline engagement controlling for SRH at baseline. Age (treated as a continuous variable) was not associated with SRH at followup: the only exceptions were Italian men and French women. In the latter cases, there was a differential association significant at a 10% level: for a one unit increase in age, the log odds of reporting a lower level of SRH were expected to decrease among French women and to increase among Italian male respondents, given all of the other variables in the model are held constant. Such a differential association might be influenced by characteristics of respondents who dropped out of the study: for instance, attrition is significantly higher among younger female French respondents (see chapter 8).

When socio-economic characteristics and household composition were considered, the direction of association between baseline level of education and SRH at follow-up suggested that respondents were less likely to report bad health the higher their level of education: only among Italian female and French male respondents was this association

not significant. Living in a rent/mortgage free home, a proxy of socio-economic status, did not have any significant effect on health reported at follow-up among SHARE respondents whereas for ELSA respondents living in rent/mortgage-free home significantly reduced the odds of reporting lower levels of SRH. Finally, living alone at baseline was not significantly associated with SRH at follow-up except among Italian and English women. Among the latter subpopulations, the association was differential: women living alone at baseline were more likely to report a lower level of SRH in Italy compared to a higher level in England. Such a differential association, though, could be partly attributable to the different attrition patterns in these two subpopulations: English women who lived alone were significantly more likely to participate in the second wave of the survey whereas Italian women living alone were significantly more likely to drop out (see chapter 8).

Focusing on SRH and functional limitations at baseline, the lower SRH was at baseline, the higher the increase in the odds of reporting low SRH at follow-up. This applied to all subpopulations. Similarly, for respondents who reported some kind of functional limitation at baseline, an increase in the log-odds of being in a lower level of SRH at time 2 was observed, when all of the other variables in the model are held constant. The latter applies to all countries with the exception of Italian and French men where the direction of the association is similar but not significant.

Finally, when the three indicators of engagement in paid work, formal and informal or family-related activities were considered, patterns arising in most subpopulations were unclear. People who were in paid work at baseline had significantly higher odds of reporting higher levels of SRH at time 2 in most countries, except among Italian and female Danish respondents for whom the association was not significant. Both formal and informal/family-related engagements at baseline were not significantly associated with SRH at follow-up.

Table 9.8 highlights the associations<sup>75</sup> between SRH at follow-up and changes in level of commitment in the various forms of activities; the other socio-economic and

<sup>&</sup>lt;sup>75</sup> These analyses do not test for any causality between SRH and changes in the levels of engagement in the various activities. Given that both changes (in health and engagement) happened simultaneously, it is

demographic factors included in the analyses are not shown in this table since results are similar to the ones described above. Working at follow-up was associated with lower odds of reporting low levels of SRH at time 2 in all countries, significantly among English respondents, French women and Danish men. Respondents who were involved in formal activities at both waves and those who became formally engaged between waves had reduced probabilities of reporting lower health in all subpopulations, significantly among English, French and female Italian respondents.

Findings did not show any clear patterns when changes in the level of informal engagement were considered. Among ELSA respondents, those who became active in informal activities had significantly lower odds of reporting lower levels of SRH; on the other hand, English men who stopped informal engagement were associated with lower health. No association was found among respondents who were 'active' at both waves. Among SHARE respondents, only French women who were 'active' in informal activities at both waves were significantly associated with lower odds of reporting lower levels of SRH; all other associations are not significant. Given the differences in the definition of 'informal and family related' participation, analyses among SHARE respondents were repeated excluding 'looking after grandchildren' from the indicator of informal commitment. Results showed no significant difference.

impossible to detect whether changes in SRH were the determinants of changes in the engagement levels or vice versa.

and 95% and 95%	Confidence Int	ervals obtaine	d from fully a	djusted ordina	up, court onur il logistic regr	ession, by gen	der and count	us nautus ry
	ITA	TY	FRA	NCE	DENN	IARK	ENG	AND
	Male	Female	Male	Female	Male	Female	Male	Female
1 1	-0.05 *	-0.02	-0.01	0.03 *	0.00	-0.01	-0.01	0.00
Age (at wave 1)	(-0.10; 0.00)	(-0.05; 0.02)	(-0.07; 0.05)	(-0.00; 0.07)	(-0.04; 0.04)	(-0.05; 0.03)	(-0.03; 0.01)	(-0.01; 0.01)
Education:	0.04	0.28	-0.12	0.26 *	0.35	0.14	0.32 ***	0.30 ***
MIDDLE <sup>*</sup>	(-0.34; 0.42)	(-0.21; 0.78)	(-0.49; 0.26)	(-0.05; 0.58)	(-0.22; 0.92)	(-0.36; 0.65)	(0.13; 0.51)	(0.14; 0.46)
Education:	0.51 *	0.10	0.28	0.44 **	0.66 **	0.57 **	0.49 ***	0.34 ***
HIGH *	(-0.02; 1.04)	(-0.46; 0.67)	(-0.20; 0.76)	(0.05; 0.82)	(0.02; 1.29)	(0.05; 1.10)	(0.30; 0.68)	(0.15; 0.53)
Monthly Rent/	-0.20	-0.08	80.0	0.21	0.15	90.0	0.22 ***	0.32 ***
Mortgage free <sup>b</sup>	(-0.70; 0.29)	(-0.43; 0.27)	(-0.25; 0.42)	(-0.08; 0.49)	(-0.29; 0.59)	(-038; 0.50)	(0.05; 0.38)	(0.17; 0.47)
l trine Alana C	-0.44	-0.48 **	0.21	0.04	-0.15	-0.06	-0.13	0.17*
LIVING AMMC	(-1.13; 0.25)	(-0.97; -0.01)	(-0.34; 0.75)	(-0.31; 0.39)	(-0.67; 0.36)	(-0.49; 0.37)	(-0.36; 0.09)	(-0.02; 0.35)
In paid work	-0.15	0.24	0.49 *	0.56 **	0.36 *	-0.34	0.35 ***	0.27 ***
at time 1 <sup>d</sup>	(-0.70; 0.41)	(-0.31; 0.79)	(-0.11; 1.09)	(0.08; 1.04)	(-0.13; 0.86)	(-0.84; 0.16)	(0.15; 0.55)	(0.10; 0.45)
Formally Engaged	0.08	0.07	0.37 *	0.19	0.12	-0.44 **	0.23 **	0.07
at time 1	(-0.39; 0.56)	(-0.37; 0.52)	(-0.07; 0.80)	(-0.11; 0.50)	(-0.28; 0.52)	(-0.83; -0.05)	(0.03; 0.43)	(-0.10; 0.24)
Informally Engaged	-0.31	0.25 *	-0.03	0.16	0.05	-0.13	-0.20 **	0.04
at time 1 <sup>f</sup>	(-0.77; 0.15)	(-0.07; 0.57)	(-0.49; 0.43)	(-0.14; 0.46)	(-0.34; 0.44)	(-0.49; 0.24)	(-0.40; -0.00)	(-0.12; 0.19)
SRH at wave 1	-0.93 **	-0.88 **	-1.15 ***	-0.95 ***	-1.49 ***	-1.32 ***	-1.22 ***	-1.40 ***
= Very good <sup>8</sup>	(-1.75; -0.10)	(-1.74; -0.01)	(-1.77; -0.52)	(-1.54; -0.37)	(-2.03; -0.94)	(-1.87; -0.76)	(-1.68; -0.76)	(-1.63; -1.16)
SRH at wave 1	-1.29 ***	-1.60 ***	-2.38 ***	-2.24 ***	-2.47 ***	-2.01 ***	-2.42 ***	-2.41 ***
= Good <sup>\$</sup>	(-2.05; -0.53)	(-2.38; -0.82)	(-2.93; -1.82)	(-2.81; -1.68)	(-3.08; -1.86)	(-2.60; -1.42)	(-2.70; -2.15)	(-2.67; -2.16)
SRH at wave 1	-3.09 ***	-3.19 ***	-4.00 ***	-4.31 ***	-3.69 ***	-3.48 ***	-3.75 ***	-4.04 ***
= Fair <sup>c</sup>	(-3.96; -2.22)	(-3.94; -2.45)	(-4.72; -3.28)	(-5.00; -3.62)	(-4.50; -2.89)	(-4.31; -2.65)	(-4.08; -3.42)	(-4.36; -3.71)
SRH at wave 1	-4.79 ***	-3.83 ***	-5.00 ***	-5.22 ***	-5.69 ***	-5.35 ***	-5.19 ***	-5.44 ***
= Poor <sup>±</sup>	(-5.78; -3.81)	(-4.80; -2.86)	(-5.88; -4.10)	(-6.44; -4.00)	(-7.56; -3.82)	(-6.54; -4.16)	(-5.69; -4.69)	(-5.90; -4.98)
Functional Limit.s	-0.26	-0.75 ***	-0.35	-0.56 ***	-0.85 ***	-0.43 **	-0.73 ***	-0.57 ***
at time 1 h	(-0.62; 0.11)	(-1.11; -0.41)	(-0.91; 0.22)	(-0.90; -0.22)	(-1.33; -0.37)	(-0.86; -0.00)	(-0.90; -0.57)	(-0.72; -0.41)
	-2.04	0.18	0.07	3.76	0.57	-0.64	0.40	0.53
Intercept I	(-5.27; 1.18)	(-1.87; 2.23)	(-3.62; 3.76)	(1.56; 5.96)	(-1.93; 3.08)	(-3.49; 2.19)	(-0.71; 1.51)	(-0.43; 1.49)
Intercont 7	-3.31	-1.23	-1.23	2.06	-1.46	-2.55	-1.73	-1.65
	(-6.46; -0.16)	(-3.25; 0.78)	(-4.83; 2.36)	(0.17; 4.30)	(-3.99; 1.08)	(-5.41; 0.30)	(-2.85; -0.61)	(-2.61; -0.68)
Intervent 3	-5.85	-3.45	4.19	-1.08	-3.14	4.10	-3.83	-3.61
	(-9.00; -2.70)	(-5.47; -1.44)	(-7.69; -0.69)	(-3.29; 1.12)	(-5.71; -0.57)	(-6.97; -1.24)	(-4.96; -2.69)	(-4.59; -2.63)
Intercent 4	-8.35	-6.41	-6.63	-3.97	-5.39	-6.67	-5.96	-5.85
	(-11.5; -5.20)	(-8.49; -4.33)	(-10.3; -2.95)	(-6.23; -1.71)	(-8.06; -2.71)	(-9.58; -3.76)	(-7.12; -4.81)	(-6.84; -4.85)
N Observations	410	412	544	676	532	713	2,721	3,154
Sources: SHARE 2004	/06, ELSA 2002/0	4; Weighted Data.	Ref categories: a	a) Low Education;	b) Paying rent/m	iortgage; c) Not A	Alone at time 1 (t	l); d) Not in

6 ( ł 1 5 ; . H - J V ; • . . Table 0.7 paid work at 11; c) Not formally engaged at 11; f) Not informally engaged at 11; g) Excellent SRH; h) No functional limit.s at 11. \* p<10; \*\* p<0.05; \*\*\* p<0.01 

Table 9.8 Association between changes in levels of engagement over time and SRH at follow-up, controlling for baseline SRH. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted ordinal logistic regression, by gender and country

	ITA	ΙΛ	FRA	NCF	DENN	IARK	FNGI	AND
	Male	Female	Male	Female	Male	Female	Male	Female
Not in paid work at both waves								
Not in paid work for less	-0.28	0.54	0.62	0.36	0.66 **	-0.60	-0.07	0.12
than 2 years	(-1.09; 0.52)	(-0.11; 1.19)	(-0.46; 1.70)	(-0.23; 0.94)	(0.01; 1.31)	(-1.30; 0.10)	(-0.38; 0.24)	(-0.14; 0.38)
	0.12	0.08	0.41	0.70 ***	1.10 ***	0.18	0.54 ***	0.46 ***
In paid work at time 2	(-0.52; 0.76)	(-0.52; 0.68)	(-0.34; 1.16)	(0.23; 1.16)	(0.48; 1.74)	(-0.38; 0.74)	(0.38; 0.71)	(0.27; 0.64)
Not active in formal activities in either waves								
	0.06	0.61 *	0.25	*** 69.0	0.45	0.21	0.04	0.18 *
No Formai → Formai	(-0.54; 0.65)	(-0.03; 1.25)	(-0.41; 0.91)	(0.23; 1.15)	(-0.08; 0.98)	(-0.25; 0.68)	(-0.21; 0.29)	(-0.02; 0.40)
	-0.37	-0.63	0.18	0.30	-0.18	-0.62	0.09	-0.07
Formal $\rightarrow$ No formal	(-1.18; 0.44)	(-1.44; 0.19)	(-0.33; 0.70)	(-0.12; 0.72)	(-0.84; 0.48)	(-1.58; 0.34)	(-0.18; 0.37)	(-0.29; 0.15)
Active in formal activities at	0.16	0.55 **	0.54 **	0.40 **	0.16	0.27	0.35 ***	0.22 **
both waves	(-0.41; 0.72)	(0.05; 1.05)	(-0.03; 1.11)	(-0.00; 0.80)	(-0.33; 0.65)	(-0.18; 0.72)	(0.09; 0.62)	(-0.01; 0.45)
Not active in informal								
activities in either waves								
No informal → informal &	0.09	0.28	-0.05	0.01	0.28	-0.05	0.43 ***	0.34 ***
family engagement	(-0.40; 0.57)	(-0.28; 0.85)	(-0.63; 0.53)	(-0.40; 0.42)	(-0.29; 0.85)	(-0.58; 0.48)	(0.11; 0.75)	(0.09; 0.59)
Informal and Family →	-0.12	0.46	-0.12	-0.11	-0.02	-0.48	-0.23 **	0.03
No more informal	(-0.71; 0.47)	(-0.05; 0.98)	(-0.78; 0.53)	(-0.57; 0.35)	(-0.54; 0.50)	(-1.07; 0.10)	(-0.47; 0.01)	(-0.16; 0.21)
Active in informal/family	-0.43	0.27	-0.26	0.35 **	0.23	-0.05	-0.03	0.20
activities at both waves	(-1.03; 0.17)	(-0.14; 0.67)	(-0.80; 0.29)	(0.01; 0.69)	(-0.29; 0.75)	(-0.49; 0.39)	(-0.36; 0.30)	(-0.04; 0.43)
Number of Observations	410	412	544	676	532	713	2,721	3,154
	Sources: S	SHARE 2004/06	J.SA 2002/04: W	/eichted Data * n<	<10: ** n<0.05: **	* p<0.01		

Controlling for age, education, baseline functional health, household composition (alone or not); household living arrangements (mortgage/rent free home or not); household living arrangements (mortgage/rent free home or not).

# **9.3.2 DEPRESSION AT FOLLOW UP CONTROLLING FOR BASELINE DEPRESSION**

Table 9.9 shows the results from a logistic regression model which investigated how depression at follow-up was associated with baseline engagement once socio-economic, demographic characteristics and depression at baseline were controlled for.

Age was significantly associated with incidence of depression symptoms at follow-up only among French women and English: among these subpopulations the odds of being depressed at follow-up increased with age; such an association though might be severely influenced by the characteristics of those who dropped out of the study (see chapter 8). The direction of association between level of education at time 1 and depression at time two suggested that highly educated respondents were less likely to be depressed at follow-up, significantly among French and Danish women and ELSA respondents, even controlling for the baseline depression. Finally, the direction of association between depression at follow-up and baseline household composition showed that survey participants who lived alone at baseline had higher odds of being depressed 24 months later. This association was significant among English respondents and Italian women. Among French men living alone the direction of association was reversed, although not significantly so.

When baseline health characteristics are analysed, findings showed a general positive association between baseline functional limitation and depression at follow-up: people who reported at least one functional limitation at baseline were more likely to be depressed 2 years later, significantly in all subpopulations except Danish respondents and French men. Moreover, baseline depression was positively and significantly associated with higher odds of depression at follow-up in all countries and subpopulations.

Finally, when we consider the three indicators of engagement in paid work, formal and informal or family-related activities at baseline, findings suggested that being in paid work at baseline was associated with lower odds of depression at follow-up, significantly in all groups except Italian and French men and Danish women. Baseline formal and informal/family engagements, however, were not associated with the likelihood of being depressed 24 months later. Only among Italian female and Danish male respondents was there any indication of a possible negative association between formal engagement at baseline and depression at follow-up, but this relationship was significant only at a 10% level.

Table 9.10 presents results obtained when changes in level of commitment in the various forms of activities were taken into account. Continuity in participation in both formal and informal engagement was not correlated with depression at follow-up, except among Italian and French women, and English men active in formal activities at both waves who had significantly lower depression at follow-up. 'Quitting' formal or informal engagement was not associated with depression, except among English men no longer involved in formal activities at follow-up who had significant higher odds of being depressed compared to those who were not engaged in either waves.

Table 9.9 Association between engagement at baseline and depression at follow-up, controlling for baseline depression. Log

Odds Ratios and	d 95% Confid	ence Intervals	s obtained fro	m fully adjust	ed logistic regi	ression, by gen	ider and count	l <b>ry</b>
	ITA	ILY	FRA	NCE	DENN	<b>1ARK</b>	ENG	LAND
	Male	Female	Male	Female	Male	Female	Male	Female
	0.02	-0.02	-0.05	-0.05 **	-0.06	-0.02	-0.03 **	-0.03 ***
Age (at wave 1)	(-0.04; 0.06)	(-0.07; 0.03)	(-0.10; 0.01)	(-0.09; -0.01)	(-0.13; 0.01)	(-0.08; 0.04)	(-0.05; -0.01)	(-0.05; -0.01)
Education: MIDN E	80.0	-0.10	0.01	-0.34	-0.21	-0.57	-0.07	-0.14
EQUCATION: MIDDLE	(-0.75; 0.92)	(-0.61; 0.41)	(-0.59; 0.61)	(-0.72; 0.03)	(-1.08; 0.67)	(-1.24; 0.10)	(-0.35; 0.21)	(-0.38; 0.08)
	-0.51	-0.24	-0.66	-0.41 **	-0.05	<b>** 68.0-</b>	-0.38 ***	-0.28 **
Equeation: HIGH	(-2.32; 1.30)	(-0.95; 0.47)	(-1.51; 0.19)	(-0.94; 0.13)	(-0.99; 0.88)	(-1.65; -0.14)	(-0.67; -0.09)	(-0.55; -0.01)
Monthly Rent/	-0.46	-0.28	-0.04	0.40 **	-0.19	-0.05	-0.31 **	-0.23 **
Mortgage free <sup>b</sup>	(-1.13; 0.21)	(-0.77; 0.21)	(-0.52; 0.44)	(0.04; 0.77)	(-0.98; 0.59)	(-0.70; 0.60)	(-0.55; -0.07)	(-0.44; -0.02)
	0.45	0.71 **	-0.37	0.26	0.10	0.16	0.45 ***	0.36 ***
Living Alone	(-0.34; 1.24)	(0.07; 1.34)	(-1.32; 0.58)	(-0.28; 0.80)	(-0.63; 0.83)	(-0.81; 0.49)	(0.15; 0.75)	(0.12; 0.58)
In paid work	-0.30	-0.51 **	-0.38	-0.48 **	++ 66.0-	-0.01	-0.39 ***	-0.36 ***
at time 1 <sup>d</sup>	(-0.88; 0.27)	(-0.99; -0.03)	(-1.11; 0.34)	(-0.79; -0.17)	(-1.69; -0.26)	(-0.70; 0.68)	(-0.67; -0.11)	(-0.58; -0.12)
Active in formal activities	0.08	-0.51 *	0.16	-0.35	-0.61 *	0.51	0.12	-0.05
at time 1 °	(-0.78; 0.94)	(-1.12; 0.11)	(-0.43; 0.76)	(-0.79; 0.10)	(-1.32; 0.10)	(-0.09; 1.11)	(-0.21; 0.45)	(-0.32; 0.21)
Active in informal/family	0.03	-0.11	0.24	-0.33	0.27	0.04	0.20	0.16
activities at time 1 <sup>f</sup>	(-0.62; 0.68)	(-0.49; 0.27)	(-0.33; 0.81)	(-0.71; 0.04)	(-0.36; 0.89)	(-0.53; 0.61)	(-0.08; 0.48)	(-0.05; 0.38)
	2.02 ***	1.30 ***	*** 89.1	1.88 ***	1.02 ***	1.94 ***	1.86 ***	1.69 ***
Depressed at time 1	(1.49; 2.55)	(0.88; 1.72)	(1.06; 2.29)	(1.55; 2.12)	(0.28; 1.77)	(1.34; 2.53)	(1.62; 2.11)	(1.49; 1.89)
Functional Limitations	0.69 ***	0.51 **	0.59	0.53 ***	0.33	0.30	0.63 ***	0.52 ***
at time 1 h	(0.27; 1.11)	(0.09; 0.93)	(-0.16; 1.35)	(0.23; 0.83)	(-0.37; 1.03)	(-0.30; 0.90)	(0.39; 0.87)	(0.31; 0.72)
	-2.83	0.31	0.40	1.65	2.50	-0.84	-0.25	-0.06
Constant	(-6.32; 0.66)	(2.45; 3.07)	(-3.24; 4.04)	(-0.70; 4.00)	(-1.25; 6.25)	(-4.70; 3.02)	(1.80; 1.30)	(-1.33; 1.20)
Number of Observations	410	412	544	676	532	713	2,721	3.154
Sources: SHARE 2004/06,	ELSA 2002/04;	Weighted Data. R	eference categor	ies: a) Low Educa	tion; b) Paying re	nt/mortgage mon	thly at time 1; c)	Not living

alone at ture 1; d) Not in paid work at time 1; e) Not formally engaged at time 1; f) Not informally engaged at time 1; g) Not depressed at time 1; h) No functional limitations at time 1. \* p<10; \*\* p<0.05; \*\*\* p<0.01

Table 9.10 Association between changes in levels of engagement over time and depression at follow-up, controlling for baseline depression. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted logistic regression, -• -

1949-197 A

			by gender an	id country				
	ITA	LY	FRA	NCE	DENM	IARK	Engl	and
	Male	Female	Male	Female	Male	Female	Male	Female
Not in paid work at both								
Not in naid work for less	-0.22	-1.03 **	-0.15	0.24	-1.80 **	0.33	-0.43 **	-0.26
than 2 vears	(-1.06; 0.61)	(-2.01; -0.05)	(-0.93; 0.63)	(-0.47; 0.95)	(-3.35; -0.25)	(-0.63; 1.31)	(-0.85; -0.01)	(-0.61; 0.08)
	-0.34	-0.06	-0.69	-0.38 **	-0.84 **	-0.27	-0.52 ***	-0.39 ***
In paid work at time 2	(-1.32; 0.44)	(-0.68; 0.56)	(-1.86; 0.48)	(-0.68; -0.08)	(-1.63; -0.05)	(-1.54; 0.34)	(-0.82; -0.23)	(-0.65; -0.13)
Not active in formal activities in either waves								
	-0.52	-0.40	0.00	-0.12	-0.18	-0.11	-0.61 ***	-0.30 *
No Formal → Formal	(-1.45; 0.41)	(-1.06; 0.26)	(-0.78; 0.78)	(-0.66; -0.43)	(-1.03; 0.67)	(-0.86; 0.65)	(-1.08' -0.14)	(-0.66; 0.06)
	-0.15	0.15	0.45	0.18	-0.07	* 06.0	0.50 **	0.16
Formal $\rightarrow$ No formal	(-1.46; 1.16)	(-0.90; 1.20)	(-0.50; 1.40)	(-0.41; 0.76)	(-1.41; 1.27)	(-0.06; 1.86)	(0.08; 0.91)	(-0.18; 0.51)
Active in formal activities at	-0.09	-1.05 ***	-0.08	-0.80 ***	-0.43	-0.22	-0.54 **	-0.32 *
both waves	(-1.07; 0.87)	(-1.73; -0.37)	(-0.80; 0.64)	(-1.38; -0.21)	(-1.20; 0.36)	(-0.95; 0.50)	(-1.05; -0.03)	(-0.67; 0.04)
Not active in informal/family								
activities in either waves								
No informal → informal &	0.62	0.27	-0.31	0.25	-0.19	-0.31	0.12	-0.35
family engagement	(-0.11; 1.35)	(-0.21; 0.76)	(-1.14; 0.52)	(-0.36; 0.86)	(-1.16; 0.78)	(-1.10; 0.48)	(-0.40; 0.64)	(-0.76; 0.06)
Informal and Family →	0.24	-0.06	0.36	-0.32	0.14	-0.53	0.14	0.13
No more informat	(-0.73; 1.22)	(-0.60; 0.48)	(-0.20; 0.93)	(-0.90; 0.27)	(-0.72; 1.00)	(-1.31; 0.24)	(-0.20; 0.47)	(-0.13; 0.38)
Active in informal/family	0.12	-0.01	-0.01	-0.20	0.25	0.24	0.37	0.07
activities at both waves	(-0.59; 0.83)	(-0.46; 0.44)	(-0.74; 0.72)	(-0.68; -0.28)	(-0.58; 1.08)	(-0.52; 1.00)	(-0.08; 0.83)	(-0.26; 0.39)
Number of Observations	410	412	544	676	532	713	2,721	3.154
T STATE INSTALLASTIC TO INTITUT	TT SUCCES IS	aichtad Data * n	<10. ** n<0 05. *	** n<0.01	-		,	

Sources: SHAKE 2004/06, ELSA 2002/04; weighted Data.  $\tau p < 10$ ;  $\tau^{\tau} p < 0.03$ ;  $\tau^{\tau} p < 0.01$ Controlling for age, education, baseline functional health, household composition (alone or not); household living arrangements (mortgage/rent free home or not), baseline SRH (all at baseline). 197

#### 9.3.3 SRH AND DEPRESSION CHANGES OVER TIME

Tables 9.11 and 9.12 show the results of ordinal logistic models of changes between waves in SRH and in depression respectively by engagement at baseline, controlling for socio-economic and demographic characteristics at baseline.

SRH changes over time display similar associations with socio-economic and demographic characteristics as SRH at follow-up. Only a few exceptions apply. English women living alone at baseline no longer had increased odds of reporting better levels of the SRH scale. Moreover, limitations at baseline were associated with increased odds of either keeping reporting a low health or with a deterioration of SRH at follow-up significantly in all subpopulations under study.

Focusing on engagement at baseline in the three activities, respondents who were in paid work at baseline were more likely to either report improvements in their SRH or to report SRH as good as at baseline, compared to those who were not working at baseline. This association was significant at a 5% level in all subpopulations except Italian male and Danish female respondents. No significant associations were found in the other subpopulations. Similarly, informal and family-related engagement was in none of the subpopulations significantly associated with longitudinal changes in SRH.

Compared to the model which considered SRH at follow-up and controlled for baseline SRH, this model with changes in SRH over time showed a stronger and more significant association with engagement in paid work at baseline whereas the longitudinal associations with the other two indicators of formal and informal activities disappeared in most subpopulations.

Table 9.12 shows the results of the ordinal logistic analysis modelling changes in depression between waves on engagement at baseline. No major differences were observed in the direction of association between changes in depression and baseline socio-economic and demographic characteristics. Limitations at baseline, in this case, were significantly associated with increased odds of becoming depressed at follow-up in all subpopulations under study.

When the main variables of interests were analysed, paid work at baseline was associated with higher odds of either being no longer depressed or remaining notdepressed also at follow-up, when all other variables in the model are held constant. This association was significant in all subpopulations, except among French male respondents. As regards formal participation, the direction of association suggested that respondents who were actively engaged at baseline had lower odds of becoming depressed, though this association is not significant. Finally, among SHARE respondents, there was neither statistically significant nor clear direction of association between informal and family-related engagement with changes in depression. This finding held also when the indicator of informal commitment did not include the activity 'looking after grandchildren' (data not shown). On the contrary, ELSA respondents who were informally engaged at baseline had significantly higher odds of becoming depressed at follow-up.

Table Error! No text of specified style in document..1 Association between engagement at baseline and changes (for the better) in SRH over time. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted ordinal logistic regression, by gender and country

	ITA	LY	FRA	NCE	DENN	<b>1ARK</b>	Eng	land
	Male	Female	Male	Female	Male	Female	Male	Female
	-0.06 **	-0.01	0.01	0.03	0.01	-0.01	0.02 ***	0.01
Age (at wave 1)	(-0.11; -0.01)	(-0.04; 0.02)	(-0.04; 0.06)	(-0.02; 0.07)	(-0.03; 0.05)	(-0.05; 0.03)	(0.00; 0.04	(-0.01; 0.03)
	0.37	0.37	-0.30	0.16	0.32	0.18	0.43 ***	0.39 ***
Equeation: MILDLE	(-0.18; 0.93)	(-0.15; 0.88)	(-0.78; 0.18)	(-0.26; 0.58)	(-0.28; 0.91)	(-0.40; 0.76)	(0.22; 0.64)	(0.20; 0.58)
	0.88 ***	0.08	0.25	0.32 **	0.38	0.59 **	0.67 ***	0.33 ***
Education: HIGH	(0.46; 1.30)	(-0.54; 0.69)	(-0.21; 0.71)	(0.03; 0.60)	(-0.28; 1.05)	(0.05; 1.14)	(0.47; 0.87)	(0.13; 0.53)
Monthly Rent/	-0.28	0.05	0.04	0.09	0.16	0.39	0.20 **	0.38 ***
Mortgage free <sup>b</sup>	(-0.85; 0.28)	(-0.23; 0.34)	(-0.27; 0.36)	(-0.18; 0.36)	(-0.35; 0.67)	(-0.14; 0.91)	(0.03; 0.38)	(0.22; 0.54)
	-0.57	-0.68 **	-0.16	-0.13	-0.05	0.08	-0.20	0.03
LIVING AIONC	(-1.19;0.05)	(-1.21; -0.03)	(-0.56; 0.23)	(-0.52; 0.26)	(-0.57; 0.48)	(-0.39; 0.56)	(-0.45; 0.05)	(-0.17; 0.25)
In paid work	-0.07	0.35 **	0.82 **	** 69.0	0.86 ***	-0.04	0.59 ***	0.42 ***
at time 1 <sup>d</sup>	(-0.65; 0.51)	(0.00; 0.71)	(0.16; 1.48)	(0.04; 1.34)	(0.28; 1.43)	(-0.60; 0.52)	(0.39; -0.79)	(0.23; 0.62)
Formally Engaged	0.25	0.01	0.31 *	0.13	0.26	-0.26	0.19 *	0.15
at time 1 °	(-0.27; 0.77)	(-0.40; 0.41)	(-0.07; 0.69)	(-0.16; 0.43)	(-0.21; 0.74)	(-0.66; 0.14)	(-0.02; 0.41)	(-0.03; 0.33)
Informally Engaged	-0.29	0.25	-0.05	0.01	0.07	-0.16	-0.13	-0.01
at time 1	(-0.75; 0.17)	(-0.15; 0.64)	(-0.50; 0.40)	(-0.30; 0.32)	(-0.36; 0.50)	(-0.57; 0.25)	(-0.34; 0.08)	(-0.17; 0.14)
Functional Limitations	-0.45 **	-0.76 ***	-0.42 **	-0.62 ***	-0.96 ***	-1.02 ***	-0.88 ***	-0.79 ***
at wave 1 <sup>g</sup>	(-0.86; -0.04)	(-1.11; -0.40)	(-0.78; -0.08)	(-0.92; -0.32)	(-1.60; -0.34)	(-1.53; -0.51)	(-1.08; -0.67)	(-0.97; -0.61)
	-1.30	1.52	3.56	4.71	4.18	2.39	4.48	3.79
Intercept I	(-4.37;1.77)	(-0.41; 3.45)	(0.72; 6.40)	(1.93; 7.49)	(1.70; 6.67)	(-0.52; 5.30)	(3.41; 5.56)	(2.70; 4.87)
C +	-4.78	-1.09	-0.32	0.93	-0.18	-1.97	0.54	-0.25
Turercept 2	(-7.82; -1.75)	(-2.96; 0.78)	(-3.20; 2.55)	(-1.93; 3.79)	(-2.56; 2.23)	(-4.76; 0.82)	(-0.49; 1.59)	(-1.29; 0.79)
1-44-2	-6.06	-2.76	-1.36	-0.25	-1.31	-3.11	-0.80	-1.45
	(-9.09; -3.03)	(-4.68; 0.85)	(-4.24; 1.52)	(-2.96; 2.46)	(-3.70; 1.07)	(-5.88; 0.34)	(-1.84; 0.24)	(-2.49; 0.41)
Number of Observations	410	412	544	676	532	713	2,721	3,154
Sources: SHARE 2004/06,	, ELSA 2002/04; W	/eighted Data. Re	ference categories	s: a) Low Educati	on; b) Paying rent	Umortgage month	ly at time 1; c) N	ot living

Alone at time 1; d) Not in paid work at time 1; e) Not formally engaged at time 1; f) Not informally engaged at time 1; g) No functional limitations at time 1. \* p<10; \*\* p<0.05; \*\*\* p<0.01

over time. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted ordinal logistic regression, by gender Table Error! No text of specified style in document..2 Association between engagement at baseline and changes in depression

			DUR	country				
	L	ALY	FRA	NCE	DEN	MARK	ENG	LAND
	Male	Female	Male	Female	Male	Female	Male	Female
Age (st wave 1)	0.02	0.02	0.02	0.06 ***	*** 60.0	0.07 ***	0.05 ***	0.04 ***
	(-0.03; 0.07)	(-0.01; 0.05)	(-0.03; 0.07)	(0.03; 0.09)	(0.04; 0.14)	(0.02; 0.12)	(0.03; 0.07)	(0.02; 0.06)
Education: MIDDLE -	0.24	0.05	-0.17	0.21	0.10	0.38	0.32 ***	0.33 ***
	(-0.31: 0.80)	(-0.42; 0.52)	(-0.56; 0.22)	(-0.14; 0.57)	(-0.70; 0.90)	(-0.19; 0.95)	(0.10; 0.56)	(0.14; 0.52)
Education: HCu <sup>2</sup>	0.97 **	0.50 *	0.40 *	0.34 *	-0.01	0.65 **	0.56 ***	0.46 ***
	(0.00: 1.95)	(-0.08; 1.09)	(-0.10; 0.90)	(-0.06; 0.74)	(-0.89; 0.87)	(0.29; 1.01)	(0.33; 0.79)	(0.24; 0.67)
Monthly Rent/	0.39	0.36 *	0.16	-0.14	0.42	-0.06	0.52 ***	0.43 ***
Mortgage free b	(-0.15: 0.93)	(-0.02; 0.73)	(-0.19; 0.52)	(-0.43; 0.16)	(-0.30; 1.15)	(-0.56; 0.44)	(0.32; 0.72)	(0.26; 0.59)
l iving Alone <sup>c</sup>	0.31	-0.85 ***	-0.32	-0.38 +	-0.29	-0.08	-0.78 ***	-0.55 ***
	(-1.08; 0.45)	(-1.41; -0.29)	(-0.77; 0.13)	(-0.79; 0.03)	(-0.94; 0.35)	(-0.56; 0.41)	(-1.01; -0.55)	(-0.74; -0.36)
In paid work	0.34 *	0.48 **	0.26	0.35 **	0.96 ***	0.44 **	0.73 ***	0.49 ***
at time 1 <sup>d</sup>	(-0.10; 0.79)	(0.12; 0.83)	(-0.27; 0.79)	(0.02; 0.68)	(0.38; 1.54)	(0.02; 0.86)	(0.50; 0.95)	(0.30: 0.67)
Formally Engaged	0.47	0.49 **	0.19	0.30	0.28	0.14	0.01	0.21 **
at time 1 °	(-0.28; 1.21)	(0.07; 0.91)	(-0.21; 0.59)	(-0.06; 0.66)	(-0.23; 0.79)	(-0.30; 0.58)	(-0.26; 0.28)	(0.00; 0.42)
Informally Engaged	0.19	0.05	-0.05	0.17	0.05	-0.23	-0.26 **	-0.34 ***
at time 1 <sup>1</sup>	(-0.73; 0.34)	(-0.24; 0.34)	(-0.55; 0.44)	(-0.09; 0.43)	(-0.45; 0.56)	(-0.67; 0.21)	(-0.50; -0.03)	(-0.51; -0.16)
<b>Functional Limitations</b>	-1.18 ***	-1.08 **	-1.06 ***	-0.97 ***	-1.05 ***	-0.75 ***	*** 66.0-	-0.89 ***
at time 1 <sup>g</sup>	(-1.58;-0.79)	(-1.47; -0.68)	(-1.56; -0.56)	(-1.24; -0.69)	(-1.59; -0.51)	(-1.19; -0.30)	(-1.18; -0.80)	(-1.06; -0.72)
Tatanat 1	0.42	1.21	0.41	3.34	4.48	3.28	1.89	1.79
	(-2.89; 3.74)	(-0.99; 3.41)	(-2.87; 3.69)	(1.43; 5.24)	(1.37; 7.59)	(0.02; 6.55)	(0.63; 3.15)	(0.72; 2.86)
Intervent 7	-0.17	0.49	-0.46	2.71	3.80	2.69	1.31	1.17
	(-3.47; 3.13)	(-1.69; 2.68)	(-3.72; 2.80)	(-0.82; 4.60)	(-0.69; 6.91)	(-0.58; 5.97)	(0.06; 2.57)	(0.10; 2.25)
Intercent 3	-0.87	-0.27	-1.20	2.12	2.45	1.84	0.50	0.40
	(-4.23; 2.49)	(-2.46; 1.91)	(-4.38; 1.98)	(0.21; 4.02)	(-0.70; 5.61)	(-1.43; 5.12)	(-0.75; 1.76)	(-0.67; 1.48)
Number of Observations	410	412	544	676	532	713	2.721	3.154
Sources: SHARE 2004/06,	ELSA 2002/04;	Weighted Data.	Reference catego	ries: a) Low Educ	ation: h) Paving	rent/mortoage mo	onthly at time 1.	Not living
Alone at time 1: d) Not in r	oaid work at time	1. e) Not formally	v ennaned at time	1. f) Not informa	Illy angeographics the second of the		and the task of a	
I man in the transman amount		1, V) 1101 IUI IUI	y viikageu ai iiiie	I, IJ IVUI IIIUIIIII	uly ciigageu ai uui	ie 1; gj ino iuncus	onal limitations at	time I.

Note: Negative log-odds ratios mean that the respondents has decreased the level of their mental health and has become depressed, whereas positive log odd ratios imply a positive change, i.e. the respondent is still or no longer depressed. \* p<10; \*\* p<0.05; \*\*\* p<0.01

# **9.4** RESULTS FROM THE MULTIVARIATE ANALYSES OF THE MULTIPLE 'IMPUTED DATASETS' UNDER MAR AND NMAR.

The above results referred to the complete-record analyses. Under the MAR assumption, missing values at follow-up were imputed and regression analyses for each outcome repeated. Only a selection of all the analyses ran are presented: Table 9.13 shows the results of the model which regressed SRH and changes between waves in SRH on baseline characteristics whereas Table 9.14 shows the findings from the regression of depression at follow-up controlling for depression at baseline.

# 9.4.1 SELF-RATED HEALTH

Under the missing at random assumption, MI should correct biases that may otherwise arise in complete record analyses. When SRH at follow-up and changes in SRH over time were examined, the inclusion of imputed values changed substantially the significance level and the direction of certain associations. In particular, age was no longer associated with SRH; English male respondents had no longer significantly increased odds of reporting better health for each unit increase in their age. Italian women who lived alone at baseline were no longer the only subpopulation experiencing a significant increase in their odds of reporting poor or worsened health. Moreover, analyses on the imputed datasets showed a significant association between functional limitations at baseline and SRH at follow-up among Italian and French men (whereas in the complete record analyses these were not significant).

When we focus on the three forms of engagement at baseline, we observe that Italian men and Danish women who were in paid work at baseline now show a similar (though not significant) direction of association with SRH and its changes over time as the other countries under study. As regards formal engagement, among SHARE respondents, MI analyses showed no significant association with SRH. The implausible association which applied only to Danish women is no longer significant when MI datasets under MAR assumptions were analysed: female Danish respondents who were engaged in formal activities had no longer significantly worsened self-reported health. Among ELSA respondents, both men and women who were actively engaged in formal activities at baseline had significantly higher odds of reporting better SRH at follow-up.
Changes in estimates for informal engagement at baseline suggested that participation in such activity was no longer associated with SRH changes in any of the countries under study.

#### 9.4.2 DEPRESSION

When depression was considered, analyses performed on the imputed datasets showed that age was no longer significantly associated with depression at follow-up. Among SHARE respondents, moreover, living in a mortgage or rent-free house at baseline was no longer significantly associated with increased odds of being depressed at follow-up among French women. Similarly, Italian women living alone were no longer significantly more likely to be depressed at follow-up. Focusing on the three dimensions of participation, whereas among SHARE respondents both formal and informal/family-related engagement were no longer associated with depression, ELSA respondents who were engaged in formal activities were significantly less likely to be depressed at follow-up. Finally, paid work is no longer associated with reduced odds of being depressed at follow-up.

#### 9.4.3 SENSITIVITY ANALYSIS

Tables 9.15 and 9.16 show a sample of the sensitivity analyses performed. These two tables refer to the logistic regression analyses of depression at follow-up on baseline characteristics under the assumption that, among non-respondents, the proportion of people who were depressed (regardless of their baseline characteristics) was higher by 20% and 33% respectively. In both scenarios, the associations between baseline characteristics and depression at follow-up were weaker and less significant. For instance, paid work at baseline was significantly associated with reduced odds of being depressed at follow-up only among Danish men; formal engagement reduced the likelihood of being depressed only among female ELSA respondents. Also, if the depression rate was arbitrarily and randomly increased among non-respondents, baseline characteristics such as 'reporting at least one functional limitation' are no longer associated with the likelihood of being depressed at follow-up.

Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted ordinal logistic regression with imputed datasets Table 9.13 Association between engagement at baseline and changes (for the better) in SRH over time.

		(nuq	r MAR), by g	ender and cou	intry	•	ı	
	ITA	LY	FRA	NCE	DENN	AARK	Eng	gland
	Male	Female	Male	Female	Male	Female	Male	Female
Are (at wave 1)	-0.03	-0.01	00.0	0.01	-0.01	-0.01	0.00	-0.00
Age (at wave 1)	(-0.08; 0.01)	(-0.04; 0.02)	(0.04; -0.04)	(-0.02; 0.05)	(-0.03; 0.05)	(-0.05; 0.03)	(-0.02; 0.02)	(-0.01; 0.01)
Education: MIDDLE <sup>2</sup>	0.17	0.42	-0.14	0.26	0.26	0.28	0.31 ***	0.31 ***
	(-0.32; 0.66)	(-0.09; 0.93)	(-0.51; 0.23)	(-0.26; 0.58)	(-0.33; 0.85; )	(-0.27; 0.83)	(0.15; 0.48)	(0.16; 0.45)
Education. UICU <sup>2</sup>	0.57 **	0.10	0.16	0.41 **	0.36	0.63 **	0.48 ***	0.31 ***
	(0.04; 1.10)	(-0.49; 0.68)	(-0.20; 0.53)	(0.03; 0.79)	(-0.29; 1.02)	(0.02; 1.23)	(0.32; 0.64)	(0.15; 0.47)
Monthly Rent/	-0.22	0.05	80.0	0.06	0.20	0.28	0.22 ***	0.34 ***
Mortgage free	(-0.72; 0.28)	(-0.28; 0.38)	(-0.23; 0.39)	(-0.22; 0.35)	(-0.33; 0.73)	(-0.24; 0.80)	(0.09; 0.36)	(0.21; 0.47)
l iving Alona <sup>c</sup>	-0.48	-0.41	-0.15	-0.13	-0.12	0.11	-0.07	0.02
	(-1.19; 0.05)	(-0.95; 0.13)	(-0.58; 0.28)	(-0.47; 0.21)	(-0.63; 0.38)	(-0.35; 0.58)	(-0.24: 0.11)	(-0.13; 0.17)
In paid work	0.15	0.34 **	0.56 **	0.52 **	0.77 ***	0.11	0.47 ***	0.33 ***
at time 1 <sup>d</sup>	(-0.40; 0.71)	(0.04; 0.641)	(0.11; 1.00)	(0.02; 1.03)	(0.21; 1.32)	(-0.42; 0.64)	(0.30; 0.64)	(0.18; 0.49)
Formally Engaged	0.23	-0.05	0.16	0.12	0.20	-0.16	0.21 ***	0.32 ***
at time 1 °	(-0.33; 0.79)	(-0.51; 0.40)	(-0.20; 0.52)	(-0.20; 0.43)	(-0.25; 0.65)	(-0.55; 0.23)	(0.07; 0.35)	(0.19; 0.45)
Informally Engaged	-0.17	0.25	0.16	0.07	10.0-	-0.06	-0.05	-0.08
at time 1 <sup>1</sup>	(-0.24; 0.59)	(-0.06; 0.57)	(-0.20; 0.52)	(-0.23; 0.36)	(-0.43; 0.42)	(-0.46; 0.34)	(-0.22; 0.11)	(-0.06; 0.23)
Functional Limitations	-0.47 **	-0.63 ***	-0.46 **	-0.54 ***	-0.94 ***	-0.94 ***	-0.77 ***	-0.67 ***
at time 1 <sup>c</sup>	(-0.88; -0.05)	(-1.01; -0.26)	(-0.91; -0.01)	(-0.83; -0.24)	(-1.58; -0.31)	(-1.43; -0.44)	(-0.93; 0.61)	(-0.80; -0.53)
T tomote	0.29	1.86	2.33	3.59	3.75	2.45	3.11	2.64
	(-2.60; 3.18)	(-0.06; 3.78)	(-0.11; 4.78)	(1.38; 5.79)	(1.15; 6.35)	(-0.36; 5.27)	(2.48; 3.73)	(2.05; 2.23)
Intervent 7	-3.03	-0.83	-1.04	0.19	-0.41	-1.76	-0.45	-0.89
	(-5.95; -0.12)	(-2.70; 1.04)	(-3.51; 1.41)	(-2.07; 2.45)	(-2.97; 2.15)	(-4.48; 0.96)	(-1.04; 0.14)	(-1.47; 0.32)
Intercent 3	-4.26	-2.30	-2.22	-1.00	-1.66	-2.98	-1.80	-2.20
	(-7.20; -1.32)	(-4.18; -0.41)	(-4.67; 0.23)	(-3.16; 1.16)	(-4.23; 0.90)	(-5.69; 0.27)	(-2.39; -1.21)	(-2.78; -1.62)
<b>Number of Observations</b>	532	541	839	951	761	779	3,306	3,488
(% imputed)	(22.2)	(23.9)	(37.1)	(33.4)	(32.1)	(30.1)	(19.7)	(18.8)
Sources: SHARE 2004/06,	, ELSA 2002/04; W	eighted Data. Imp	uted Datasets. Re	ference categories	: a) Low Educatio	n; b) Paying rent	mortgage month	v at time
1; c) Not living Alone at 1	time 1; d) Not in	paid work at time	1; e) Not formal	Ily engaged at tin	ne 1; f) Not infor	mally engaged at	time 1: g) No f	unctional
limitations at time 1. $* p < 1$	0; ** p<0.05; *** p	×0.01		•	•	•	5	

Table 9.14 Association between engagement at baseline and depression at follow-up, controlling for baseline depression. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted logistic regression with imputed datasets dar MADN ha .

		ש	Incl MUM/N	y genuel anu	country			
	ITA	LY	FRA	NCE	DENN	IARK	ENGI	AND
	Male	Female	Male	Female	Male	Female	Male	Female
	0.00	-0.03	-0.04	-0.04	-0.05	-0.00	-0.00	0.03
Age (at wave 1)	(90.06; 0.06)	(-0.07; 0.02)	(-0.09; 0.01)	(-0.08; 0.01)	(-0.11; 0.01)	(-0.06; 0.06)	(-0.01; 0.01)	(-0.01; 0.01)
	+0.0-	-0.17	-0.13	-0.30	-0.21	-0.46	-0.13	-0.17
Education: MILULLE	(-0.77; 0.68)	(-0.70; 0.36)	(-0.66; 0.40)	(-0.67; 0.07)	(-1.08; 0.66)	(-1.12; 0.20)	(-0.37; 0.10)	(-0.36; 0.01)
	-0.32	-0.18	-0.49	-0.49 *	0.01	-0.72 *	-0.32 ***	-0.26 **
Foucation: HIGH	(-1.78; 1.14)	(-0.87; 0.51)	(-1.51; 0.19)	(-1.01; 0.03)	(-0.94; 0.95)	(-1.49; 0.05)	(-0.55; -0.09)	(-0.48; -0.04)
Monthly Rent/	-0.29	-0.17	-0.15	0.16	-0.10	-0.18	-0.26 **	-0.21 **
Mortgage free <sup>b</sup>	(-0.96; 0.38)	(-0.68; 0.33)	(-0.60; 0.31)	(-0.20; 0.52)	(-0.92; 0.71)	(-0.82; 0.46)	(-0.46; -0.06)	(-0.37; -0.52)
	0.36	0.51	-0.27	0.20	0.32	0.22	0.38 ***	0.28 ***
Living Aone	(-0.59; 1.30)	(-0.15; 1.16)	(-1.00; 0.47)	(-0.26; 0.66)	(-0.40; 1.03)	(-0.83; 0.39)	(0.15; 0.60)	(0.11; 0.46)
In paid work	-0.33	-0.31	-0.31	-0.08	-0.76 **	-0.10	-0.19	-0.23 **
at time 1 <sup>d</sup>	(-1.03; 0.37)	(-0.87; 0.24)	(-0.94; 0.32)	(-0.58; 0.42)	(-1.44; -0.09)	(-0.80; 0.59)	(-0.43; 0.06)	(-0.45; -0.02)
Formally Engaged	-0.01	-0.22	0.02	-0.19	-0.55	0.27	-0.21 **	-0.30 ***
at time 1°	(-0.86; 0.85)	(-0.82; 0.38)	(-0.54; 0.59)	(-0.61; 0.22)	(-1.23; 0.14)	(-0.30; 0.85)	(-0.40; -0.02)	(-0.46; -0.15)
Informally Engaged	0.06	-0.10	0.07	-0.25	0.22	-0.03	0.13	0.15
at time 1 <sup>f</sup>	(-0.58; 0.70)	(-0.51; 0.30)	(-0.44; 0.57)	(-0.60; 0.09)	(-0.40; 0.84)	(-0.59; 0.53)	(-0.10; 0.36)	(-0.04; 0.34)
Depressed	1.87 ***	1.37 ***	1.67 ***	1.76 ***	1.15 ***	1.96 ***	1.77 ***	1.67 ***
at time 1 <sup>°</sup>	(1.30; 2.44)	(0.98; 1.76)	(1.14; 2.20)	(1.42; 2.09)	(0.42; 1.89)	(1.37; 2.55)	(1.57; 1.97)	(1.51; 1.83)
<b>Functional Limitations</b>	0.61 **	0.54 **	0.56	0.48 ***	0.27	0.28	0.66 ***	0.56 ***
at time 1 h	(0.12; 1.10)	(0.15; 0.93)	(-0.14; 1.26)	(0.17; 0.79)	(-0.42; 0.96)	(-0.29; 0.85)	(0.46; 0.86)	(0.38; 0.74)
(	-1.89	0.51	0.20	1.19	1.28	-1.30	-1.89	-1.65
Constant	(-5.52; 1.74)	(-2.06; 3.10)	(-3.07; 3.47)	(-1.06; 3.44)	(-2.33; 4.89)	(-5.11; 2.51)	(-2.76; -1.04)	(-2.33; -0.98)
N of Observations	532	541	839	951	761	977	3,306	3,488
(% imputed)	(22.2)	(23.9)	(37.1)	(33.4)	(32.1)	(30.1)	(19.7)	(18.8)
Sources: SHARE 2004/06	), ELSA 2002/04;	Weighted Data.	Imputed Datasets	. Reference catego	ories: a) Low Edu	cation; b) Paying	rent/mortgage mo	onthly at time

1; c) Not living Alone at time 1; d) Not in paid work at time 1; e) Not formally engaged at time 1; t) Not informally engaged at time 1; g) Not depressed at time 1; h) functional limitations at time 1. \* p<10; \*\* p<0.05; \*\*\* p<0.01

Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted logistic regression with imputed datasets under Table 9.15 Association between engagement at baseline and depression at follow-up, controlling for baseline depression.

INTENT	Summer In	uchi cosion a	mong non-res	ponuenus incr	eased by 20%	), by gender al	nd country	
	ITA	LY	FRA	NCE	DENN	AARK	ENG	LAND
	Male	Female	Male	Female	Male	Female	Male	Female
l ivina Alona <sup>1</sup>	0.47	0.55	-0.25	0.20	0.27	0.26	0.34 **	0.22
	(-0.53; 1.46)	(-0.15; 1.16)	(-0.96; 0.45)	(-0.26; 0.66)	(-0.42; 0.97)	(-0.85; 0.33)	(0.06; 0.61)	(-0.09: 0.53)
In paid work	-0.27	-0.28	-0.23	-0.11	-0.73 **	-0.03	-0.17	-0.20 *
at time 1 <sup>b</sup>	(-0.97; 0.42)	(-0.83; 0.28)	(-0.94; 0.48)	(-0.60 0.38)	(-1.38; -0.08)	(-0.76; 0.70)	(-0.40; 0.07)	(-0.44; 0.04)
Formally Engaged	-0.12	-0.21	-0.00	-0.22	-0.60	0.24	-0.26	-0.37 **
at time 1 <sup>c</sup>	(-1.04; 0.80)	(-0.79; 0.37)	(-0.55; 0.54)	(-0.63; 0.20)	(-1.25; 0.04)	(-0.33; 0.81)	(-0.54; 0.01)	(-0.65: -0.08)
Informally Engaged	0.05	-0.10	0.05	-0.28	0.16	0.00	0.09	0.14
at time 1 <sup>d</sup>	(-0.56; 0.66)	(-0.49; 0.28)	(-0.45; 0.54)	(-0.64; 0.07)	(-0.46; 0.78)	(-0.55; 0.55)	(-0.18; 0.37)	(-0.03: 0.31)
Functional Limitations	0.46	0.47 *	0.54 **	0.39	0.29	0.27	0.60 ***	0.49 ***
at time 1 <sup>c</sup>	(-0.34; 1.26)	(-0.04; 0.97)	(0.01; 1.07)	(-0.09; 0.88)	(-0.37; 0.95)	(-0.27; 0.81)	(0.27; 0.92)	(0.15; 0.82)
N of Observations	532	541	839	951	761	779	3.306	3.488
(% imputed)	(22.2)	(23.9)	(37.1)	(33.4)	(32.1)	(30.1)	(19.7)	(18.8)
Sources: SHARE 2004/0	6, ELSA 2002/04;	; Weighted Data	. Imputed Dataset	s. Reference cate	gories: a) Not livir	ng Alone at time 1	; b) Not in paid w	ork at time 1;

Controlling for age, education, household composition (alone or not); household arrangements (mortgage/rent free home or not), depression (all at baseline). c) Not formally engaged at time 1; d) Not informally engaged at time 1; e) functional limitations at time 1. \* p<10; \*\* p<0.05; \*\*\* p<0.01

Table 9.16 Association between engagement at baseline and depression at follow-up, controlling for baseline depression. Log Odds Ratios and 95% Confidence Intervals obtained from fully adjusted logistic regression with imputed datasets under

	Sumer				aseu ny 22 /0),	by genuer an	u country	
	ITA	LY	FRA	NCE	DENN	IARK	ENGI	AND.
	Male	Female	Male	Female	Male	Female	Male	Female
l iting Alone a	0.53	0.59	-0.24	0.19	0.25	-0.28	0.31 **	0.19
	(-0.46; 1.52)	(-0.08; 1.27)	(-0.92; 0.43)	(-0.26; 0.65)	(-0.42; 0.93)	(-0.85; 0.30)	(0.01; 0.60)	(0.16; 0.53)
In paid work	-0.23	-0.25	-0.12	-0.12	-0.72 **	-0.03	-0.16	-0.19
at time 1 <sup>b</sup>	(-0.89; 0.43)	(-0.80; 0.29)	(-0.88; 0.63)	(-0.61 -0.37)	(-1.35; -0.09)	(-0.71; 0.76)	(-0.40; 0.08)	(-0.44; 0.07)
Formally Engaged	-0.18	-0.19	-0.01	-0.23	-0.56	0.20	-0.20	-0.32 **
at time 1 <sup>c</sup>	(-1.12; 0.75)	(-0.75; 0.36)	(-0.54; 0.52)	(-0.63; 0.17)	(-1.24; 0.04)	(-0.30; 0.85)	(-0.50; 0.10)	(-0.62; -0.04)
Informally Engaged	0.05	-0.11	0.00	-0.30	0.13	0.02	0.07	0.13
at time 1 <sup>d</sup>	(-0.55; 0.65)	(-0.48; 0.27)	(-0.47; 0.47)	(-0.66; 0.06)	(-0.48; 0.73)	(-0.51; 0.55)	(-0.22; 0.36)	(-0.04; 0.30)
Functional Limitations	0.37	0.41	0.52 **	0.32	0.30	0.26	0.55 ***	0.45 **
at time 1 °	(0.51; 1.24)	(0.14; 0.96)	(0.03; 1.01)	(-0.22; 0.86)	(-0.33; 0.94)	(-0.26; 0.78)	(0.18; 0.92)	(0.07; 0.82)
N of Observations	532	541	839	951	761	977	3,306	3,488
(% imputed)	(22.2)	(23.9)	(37.1)	(33.4)	(32.1)	(30.1)	(19.7)	(18.8)
Sources: SHARE 2004/06	5, ELSA 2002/04;	Weighted Data.	Imputed Datasets	. Reference catego	ories: a) Not living	Alone at time 1:	b) Not in paid w	ork at time 1:
c) Not formally encoded a	it time 1. d) Not in	formally anges	d at time 1: a) fur	ational limitation				

Controlling for age, education, household composition (alone or not); household arrangements (mortgage/rent free home or not), depression (all at baseline). c) Not formally engaged at time 1; d) Not informally engaged at time 1; e) functional limitations at time 1. \* p<10; \*\* p<0.05; \*\*\* p<0.01

#### 9.5 CONCLUSIONS

Among the four countries under study, descriptive statistics suggested that Danish and English respondents had comparatively better self-reported health and lower prevalence of depression than Italian and French, at both baseline and follow-up. Denmark and England also showed a slower decline in these indicators over time.

Despite these differences, it appeared that -out of the three engagement indicators being active in the labour market was the only factor significantly associated with better health; this association applied to all countries. Formal assessment of these associations supported these observations and showed that respondents who were in paid work at baseline, rated their health better than those who were 'inactive' and had higher odds of increasing their SRH or maintaining it 'good'. Similarly, they had decreased odds of being or becoming depressed. Findings were in line with those from similar longitudinal studies which have shown that work has an important influence on health outcomes such as depression and SRH (Badley et al., 2000, Innstrand et al., 2011) and employment is generally associated with slower decline in mental health (Butterworth et al., 2011). The directions of such associations as well as their significance levels, though, were not consistent among all subpopulations under study.

Formal engagement appears to be longitudinally associated with good health and lower incidence of depression though such associations are rarely significant. Findings in this research study seemed to suggest that when the holistic indicator of formal engagement is considered, its direction of association with SRH and depression at follow-up was beneficial but weak. Only among ELSA respondents, was formal engagement (which mainly refers to volunteering) associated with enhanced self-rated health and positive mental health outcomes, lending support to studies which focused just on volunteering (Van Willigen, 2000, Thoits and N., 2001, Morrow-Howell et al., 2003, Musick and Wilson, 2003).

Furthermore, informal and family-related commitment at baseline was not significantly associated with follow-up SRH and depression among SHARE respondents. Only English respondents engaged in informal activities were more likely to become depressed, though this association did not hold for the other three outcomes considered. These findings were in line with the literature review which suggests that factors such as

satisfaction, intention to care, time and intensity of care as well as the conditions of the recipient and the relationship with the caregiver tend to be more important in 'predicting' health measurements than the activity itself (Hubley et al., 2003, Gruenewald et al., 2007, Yajima et al., 2007). Given that informal engagement covers a wide range of activities (from caring for sick people to looking after and helping friends and family members), and given the lack of information on the intensity of care, on the kind of help provided or on the characteristics of the recipient of care, it is not surprising that the effects of informal activities on health is unclear.

Finally, it is worth mentioning that, when the dataset restricted to participants with complete information on all the factors considered in this chapter was analysed, a few exceptions to the general trends were found. For instance, formal engagement and paid work at baseline were associated with decreased SRH and higher prevalence of depression among Danish female respondents (but not among Danish male respondents, despite similar patterns in changes of health and baseline commitment in the various activities). Similarly, it seems implausible that French and Italian men are the only exceptions among which functional limitations at baseline are not associated with SRH at follow-up; or that age is a protective factor against depression and poor SRH at follow up among English respondents and French women. In these cases, it seemed plausible to assume that attrition introduced some bias in the analyses. Indeed, analyses carried out under the MAR assumption showed how certain associations might have been mainly caused by informative drop-outs. Of course, caution is needed -just as with any statistical methodology --when imputed datasets and sensitivity analyses are considered. Indeed, analyses based on MI might also be flawed should the imputation models fail to capture the missing-data mechanism, as 'departures' from MAR analyses showed. In all cases it is undeniable that missing data can have an impact on the results and associations found, an issue which is generally either not recognised as such or "is considered a nuisance that is best hidden" (p.6) (Burton and Altman, 2004).

## **10. CONCLUSIONS AND DISCUSSION**

Various international bodies (EU, OECD, WHO) are promoting the concept of 'active ageing': in particular, the WHO has argued that ageing well can be achieved by keeping active and involved in paid work and by being engaged with the wider community through activities such as volunteering or helping family members and friends. Extending participation beyond the labour market has the potential to help ensure that more people continue to play a full and active role in the society.

The aim of this study was to assess whether being 'active' was positively associated with people's health. In particular, this research had three objectives. Firstly, this study investigated the personal and household characteristics associated with various forms of engagement among older people in four European countries (Denmark, Italy, France and England) to evaluate the extent to which different levels of engagement could be explained by demographic, socio-economic, and health characteristics of older people. Moreover, this research study assessed whether engagement in certain activities conflicted with the participation in others. In particular, it was decided to distinguish between three main activities: paid work, formal activities (such as volunteering and attending educational courses) and informal activities (such as caring for sick people and looking after grandchildren). Secondly, this study investigated whether baseline engagement was longitudinally associated with health and depression at follow-up, and the extent to which such longitudinal associations were associated with other factors. Thirdly, this research study investigated whether or not longitudinal associations found were biased by sample attrition.

Data drawn from the first two waves of two parallel European longitudinal surveys, the Surveys of Health and Retirement in Europe (SHARE) and the English Longitudinal Study of Ageing (ELSA) were used to analyse characteristics associated with older people's engagement and to better understand the reciprocal influence of activity and health in ageing populations in four European countries selected to 'represent' different welfare regimes. Descriptive results for all survey members were presented but the main analyses focused on participants aged 50-69 at baseline.

The present chapter seeks to bring together and discuss –trying to avoid repetitions –the findings from the cross-sectional and longitudinal analyses, which have already been described and situated within the literature in the chapters where findings were presented. The following sections also discuss the strengths and limitations of this research project, the implications of the study findings and possible future research.

#### **10.1** MAIN FINDINGS

### 10.1.1 CROSS-SECTIONAL ASSOCIATIONS OF OLDER PEOPLE'S CHARACTERISTICS WITH ENGAGEMENT, AND RELATIONSHIPS BETWEEN DIFFERENT FORMS OF ACTIVITIES

Cross-sectional analyses showed that older people's level of engagement in paid work, formal, informal and family activities differed in each of the countries considered. Descriptive analyses showed large variations by country, gender and age-groups particularly when paid work and formal engagement were considered. The proportion of Danish SHARE female respondents in paid work was more than double the Italian proportion; similarly, almost 60% of the Danish men were in paid work compared to 40% of the Italian male respondents. Danish respondents consistently also had participation rates in informal activities between two and three times higher than those of the Italians. Differences between countries were hardly noticeable when participation in informal and family-related activities was analysed. In all countries, women were more involved than men in informal and family-related activities in all age-groups considered. Among survey respondents, the gap between genders was higher among the Italians and French than the English and Danish.

Interestingly, in all countries under study between almost one fifth (Denmark) and a third (Italy) of the respondents were not engaged in any of the above-mentioned activities whereas only a very small proportion of the samples (between 7% in Denmark and 2% in Italy) was simultaneously engaged in paid work, formal and informal activities. Among French and Italian women, almost 20% and 33% of respondents were engaged only in informal activities – the category which represented the second highest after 'no engagement'. In the other subpopulations, among those engaged, the majority of the samples were in paid work and –with the exception of Denmark –the proportion

of respondents who were engaged only in formal or informal activities was higher than that of respondents who combined either of these activities with work. Denmark, on the contrary, seems the only country where there were more respondents who combined work with at least one of the formal or informal activities than those who just engaged in the latter alone.

# Association of socio-economic, demographic, health-related, personal and household characteristics with the various forms of engagement

Cross-sectional multivariate analyses showed that, in both SHARE and ELSA countries, age of the respondents was consistently and significantly correlated only with paid work whereas there was no significant association between age and formal and informal engagement. These findings suggested that paid work was the only domain where factors such as statutory retirement age and pension benefit entitlements were probably more important than age itself in shaping older people's participation. In fact, whereas involvement in paid work dropped significantly after the age of 60 in all countries, such patterns were not present among the other two spheres of engagement.

Level of education was the most consistent predictor of participation in work and, in certain subgroups and countries, higher education levels were also associated with higher involvement in formal activities whereas its association with the participation of older people in informal activities was undetectable. These findings were in line with previous studies (Hank and Stuck, 2008, De Koker, 2009, Kohli et al., 2009), suggesting that education may have the ability to stimulate civic skills and values, increase awareness of problems and enhance social skills and networks (Wilson, 2000, van Ingen and Dekker, 2010) boosting participation in formal activities and civic organizations.

The role of health in older people's engagement was less clear. Ill-health indicators were negatively associated with work in all countries but not significantly so among Italian male respondents. The associations with the other forms of activities were much weaker and differential depending on country and gender: bad health tended to be negatively associated with participation in formal activities whereas it was positively associated with family and informal participation. Such cross-sectional positive association may be explained by the fact that providing help, caring for sick people and looking after grandchildren could be potential risk factors for health (Schulz and Beach, 1999b, Hirst, 2003, Hughes et al., 2007).

#### Relationship between the various forms of engagement

Analyses also showed that the association between informal and formal engagement was positive, suggesting complementarity between the two activities and therefore a possible accumulation of these two forms of engagement. This confirmed results reported in a number of both European and American studies (Burr et al., 2005, Caro et al., 2005, Burr et al., 2007, Choi et al., 2007, Hank and Stuck, 2007, Kohli et al., 2009), providing some evidence for the existence of a general motivation to be active in different activities even when general individual characteristics are controlled for. Paid work, on the contrary, was negatively associated with the probability to engage in formal activities, for both men and women. Time constraints may be a reason. However, paid work was not a deterrent to family and informal commitment (with the exception of England<sup>76</sup>). In other studies, respondents seemed to engage, either by choice or because of sense of normative obligation, in helping and caring for kith and kin even when they were also engaged in paid work (Caro et al., 2005, Choi et al., 2007). This relationship applied to all countries, i.e. both countries where family relationships and ties were believed stronger and countries where family support was high and ties weaker (Reher, 1998, Trifiletti, 1999, Pierson, 2000, Giuliano, 2007).

#### The Role of Welfare Regimes and Policies

In the fully adjusted regression models, socio-economic, demographic and health-related variables showed similar associations with the three variables of interest, both among men and women and in all four countries under study. As a consequence, differences between countries in the proportion of older people who engaged in paid work or formal activities suggest that country-specific factors, such as elements of welfare state regimes, play an important role in determining older people's level of engagement, (Pichler and Wallace, 2007, Hank and Stuck, 2008). The Italian statutory retirement age, social security benefits and pension wealth (see Chapter 3) may be an incentive for

<sup>&</sup>lt;sup>76</sup> The measure of outcome for ELSA respondents, however, is different and includes fewer activities.

Italian respondents to retire at relatively younger ages compared to the other countries under study. Similarly, policies rather than socio-demographic and health gender- and country-specific differences may also help to explain the general lower participation of Italian women in the labour market. Moreover, the fact that Italian male respondents with ill-health (whose health indicators did not differ much from French and Danish respondents –see sections 6.4.1, 6.4.2 and 6.4.3) did not have significantly lower odds of being in paid work reinforces that idea of an 'unused capacity' for active work among Italian respondents. Where retirement is offered at a relatively lower age, people tend to take it regardless of health limitations, whereas in countries with a higher retirement age early retirement is more dependent on ill-health. (Blanchet and Debrand, 2007, Debrand and Lengagne, 2007, Siegrist et al., 2007, Börsch-Supan et al., 2009, Siegrist and Wahrendorf, 2009b, Westerlund et al., 2009).

Similarly, differences in the level of engagement in formal activities reinforce the idea that older people's engagement in volunteering, political and religious organisations is shaped mainly by the societal context, norms and values as proposed in several studies focusing primarily on volunteering (Salamon and Sokolowski, 2001, Smith and Shen, 2002, Erlinghagen and Hank, 2006, Pichler and Wallace, 2007, Hank and Stuck, 2008, Hank, 2010, Hank and Erlinghagen, 2010). As an illustrative example, even accounting for differences in the educational level and in the level of participation in paid work, Danish respondents reported much higher levels of engagement in formal activities compared to French and Italian participants. Paraphrasing Salaman and Sokolowski (2001), findings suggest that the level of engagement in formal activities did not depend on mere individual spontaneous decisions but was affected also by larger social and institutional forces, values and expectations of what is socially appropriate.

Finally, factors associated with family-related and informal participation were similar in all countries under study as were the levels of engagement in such activities. This challenged the idea of a divide between the Nordic and Southern experiences of informal engagement. The patterns and levels of informal engagement were similar among all countries under study, despite the strong family traditions in southern European countries (Reher, 1998) and the higher levels and expenditure for formal care provisions in the Northern countries (Rostgaard, 2002, Bettio and Plantenga, 2004, Viitanen, 2007).

#### 10.1.2 LONGITUDINAL ASSOCIATION BETWEEN BASELINE ENGAGEMENT AND HEALTH AT FOLLOW-UP

The overall aim of this research project was to assess whether activity had a positive impact on people's health. The cross-sectional associations between the three forms of activities seemed to suggest that work and formal engagement were associated with good health, whereas informal and family activities were associated with bad health among certain subpopulations. Given the nature of these analyses, however, it was not possible to disentangle the direction of these associations. Although it is plausible that being in paid work or formal activities, for instance, sustain health and well-being, it is equally plausible that better health provides an impetus for being more socially active in a wider variety of activities. As illustrated by several authors, for example, subjects with poor health are less likely to enter into the workplace whereas workers in poor health are more likely to drop out of work (healthy worker effect), and the higher health outcomes of workers when compared to people who are not in paid work may reflect these selection effects, rather than indicating a causal effect of work on health (Tsai et al., 2005, Barnay, 2006, Barnay and Debrand, 2006, Blanchet and Debrand, 2007, Alavinia and Burdorf, 2008, Hult et al., 2010). Both mechanisms are equally plausible: being in paid work could have a favourable and positive impact on people's health but at the same time it might be less likely for ill people to enter into work (Thomas et al., 2005). Similarly, caring for sick people could be a risk factor for older people's health though it is equally possible that care providers are those who are not in paid work for health reasons.

Longitudinal data provide a better way to understand whether engagement was beneficial for older people's health. The first two waves of both ELSA and SHARE were used to investigate whether engagement in paid work, formal and informal activities at baseline were associated with health after two years (controlling for baseline measures of health) and whether such associations differed between countries and gender. In particular, self-rated health and depression were analysed. Among the three domains of engagement considered, it appeared that, at follow-up, only respondents who were in paid work at baseline rated their health better than those who were 'inactive' and had higher odds of improving their SRH or maintaining it at a 'good' level. Similarly, they had decreased odds of being or becoming depressed, in line with results from similar longitudinal studies (Badley et al., 2000, Butterworth et al., 2011, Innstrand et al., 2011). Formal engagement at baseline was rarely significantly associated with health at follow-up. This weak relationship, though, may be partly explained by the fact that formal engagement was low at baseline; moreover, because formal engagement 'competes' with paid work, and paid work explains most of the health outcomes at follow-up, the effect of formal engagement might be weakened as a result. Informal and family-related commitments at baseline were not significantly associated with follow-up SRH and depression. Caregiver status alone does not appear to result in lower levels of quality of life, poorer mental and physical health (Hubley et al., 2003).

#### **10.1.3 SAMPLE ATTRITION AND ITS IMPACT**

The findings described above were obtained limiting the analyses to the dataset restricted to participants with complete information on all the factors considered. It is however important to recall that the ELSA and SHARE surveys both suffered from attrition over time. In particular, between 20% (England) and 35% (France) of the samples dropped out of the survey between the first and the second waves. Multivariate logistic regression analyses indicated that –although the attrition mechanism varied in each subpopulation –highly educated respondents were more likely to stay in the study as were respondents engaged in formal and informal activities whereas workers and respondents who rated their health as poor or fair were more likely to drop out of the study. This selectivity effect of the sample might therefore weaken the generalisability of the findings and confound some of the reported associations. Findings might overestimate the level of active people at follow-up as well as the effect of activities on health over time. In order to account for attrition in the survey, multiple imputation techniques and sensitivity analyses were explored. Analyses carried out under the MAR assumption showed how certain associations might have been mainly caused by

informative drop-outs: for instance, formal engagement and paid work at baseline were no longer associated with decreased SRH and higher prevalence of depression among Danish female respondents when analyses based on imputed datasets were considered. Changing assumptions on the pattern of missingness, we also showed how sensitive findings are to 'departures' from MAR assumptions: assuming a higher proportion of people with depressive symptoms among drop-outs, for instance, weakened significantly the associations between baseline characteristics and depression at follow-up. Attrition, in this research, has had an undeniable impact on the results and associations found which should not be ignored. Even though MI is not a panacea (especially for cases such as France where almost 40% of male respondents dropped out of the study), it is important to acknowledge the impact that attrition had on longitudinal associations and to recall that different hypotheses on the characteristics of people who dropped out of the study could yield substantially different findings.

#### **10.2 STRENGTH OF STUDY**

The current research project investigated associations between engagement and health comparing data collected from SHARE and ELSA, two surveys built to maximize comparability of data and therefore to extend comparisons from the Anglo-Saxon countries to continental Europe. The standardisation of these two datasets included substantial data management as well as file manipulations, necessary to harmonise measurements such as education level and create comparable and homogeneous indicators of health and engagement. The UK has mainly been compared with the US "because of similarities in the demographic, economic and social contexts alongside important differences in institutional systems, for example in relation to health and social care, retirement provision and retirement incentives" (Scholes et al., 2009)(pg. 7). This project, instead, compared the UK with three other Continental European countries. Few studies have compared ELSA and SHARE; furthermore they mainly examined differences in health by gender, wealth or education (Avendano et al., 2009, Jürges, 2009, Jürges, 2010, Crimmins et al., 2011). This study included, among the principal variables of interest, not only demographic, socio-economic, household characteristics and health-related variables but also, and primarily, various indicators of engagement.

Moreover, this study compared the distribution of age, gender, education and other socio-demographic and health variables with those obtained from other sources such as censuses and other surveys with better response rates and higher number of respondents. These comparisons were believed necessary in order to provide some indication of the extent of possible bias due to initial non-response and in order to assure validity on the ability of the surveys to be representative of and describe large populations without bias. Assessment of data quality for SHARE has never been formally carried out and distributions of variables such as employment, education and health were compared only with those obtained in other surveys with similarly low-response rates and low number of respondents.

Similarly, most studies using SHARE and ELSA longitudinal data mainly relied on the use of longitudinal weights to correct for bias due to attrition and account for panel attrition, without assessing the characteristics of respondents who dropped out. This research, in contrast, not only assessed formally whether respondents who were successfully re-interviewed at follow-up differed from those who dropped-out in each of the countries under study, but also explored multiple imputations techniques to account for attrition in the surveys. Whereas the common practice is to restrict analyses and findings to "complete record" dataset which could however lead to biased results, this research study proposed multiple imputations as a possible remedy to address missing data. Findings confirmed that the self-selectivity of the sample cannot just be ignored without weakening or jeopardising the generalisability of the findings and confounding some of the reported associations.

Additionally, this research study adopted a multifaceted inclusive approach to activities: the concept of 'active ageing' was operationalised holistically, including as many activities as possible, within the boundaries of available data. Most studies have considered very few activities at a time and focused merely on paid work, the transition into retirement, caring for a sick person and volunteering while this research project tried to bring together the various dimensions older people engage in, given that older people can be involved in an increasing number of simultaneous activities. Moreover, this study compared four different countries, whose policies are believed to reflect principles and values of distinct welfare regimes. This comparative approach explored *indirectly* the existing influence between welfare state policies and participation of older people, without studying in detail specific policies or programmes. Our finding that the levels of participation varied between countries –even after controlling for potential confounders –suggests that social and economic policies are important in shaping the way older people engage with the wider society. Governments may encourage participation in the labour market or in other activities by adopting specific policies to promote employment of the older workers or programmes which promote activity in non-work time. This represents one possible explanation of the observed country differences at the macro level. This work showed the potential of cross-national, longitudinal and comparative individual- level data to provide a first assessment of a larger research question linking micro-data findings with macro data and specific policies which were not analysed in this study.

Furthermore, this study identified characteristics associated with engagement in different countries. This could help identify possible strategies or interventions to increase the level of participation of older people and contribute to improving or maintaining favourable health. This might be of particular relevance to public policy. A better understanding of the determinants of participation in paid work and formal activities, for instance, may help policy interventions target those groups which are likely to be most vulnerable and less likely to participate in the various forms of activities, alleviating the effects socioeconomic status or household characteristics exert on health and engagement.

#### **10.3 LIMITATIONS**

A number of limitations apply to both the cross-sectional and the longitudinal findings of this study. In particular, when longitudinal findings are analysed, it is important to recall that, although this study employs longitudinal data with controls for confounding variables, it is not possible to make causal conclusions. In fact, this study was not a true experiment in which people were randomly assigned to groups of workers and nonworkers, for instance. Findings seem to suggest that engagement itself relates to late life health. Nonetheless, given that the indicators used to summarise the various forms of engagement are grossly summative measures which did not consider the significance attached to them and given that panel data -not experimental data -were used, alternative causal interpretations are possible. For instance, when we try to understand the underlying mechanisms linking 'activities' to 'health', the value attached by society to specific activities or the role perceived by the individual may be responsible for such links. Activities can also indirectly enhance health by providing individuals with power, prestige, privilege, status, and emotional gratification or by providing them with social support or continued income. Conversely, initial health could determine individual's capacity for engagement: engagement may lead to and result from higher levels of health. As cross-sectional findings showed, individuals engaged in paid work and formal engagement were already 'healthier' than those who were not engaged in such activities: these respondents might have merely maintained this advantage over time. As a consequence, even accounting for baseline health, the effect of the baseline participation on health at follow-up might be overestimated simply because only those in good health were engaged at baseline. These direct and indirect causal connections between engagement and health merit more careful exploration.

Both ELSA and SHARE surveys suffered from initial low response. As a consequence, data quality and its representativeness are affected and therefore the generalization of the findings might not hold had the non-respondents also been included in the sample. In particular, younger respondents (particularly men and people who were in paid work) were under-represented whereas there was an indication of an over-representation of people 'in good health'. Given the scope of this research study, the reciprocal influence of health and engagement might potentially be jeopardised and biased.

Moreover, in addition to the methodological weaknesses of ELSA and SHARE and the problems due to institutional and cultural differences arising when comparing different countries (as highlighted in Chapter 5.7), other limitations are mainly due to the availability of the data. Neither questionnaire asked detailed questions about activities; nothing is known about hours of commitment in educational and training classes; no information was collected on the number of voluntary organisations people are involved with, rendering hard to distinguish between 'super' and 'regular' volunteers, for

instance. Similarly, SHARE and ELSA do not distinguish different kinds of voluntary activities nor do they distinguish the kind of help provided to family or friends. Moreover, the discretionary or 'obligatory' nature of the activities carried out is not investigated: the same activity, though, could be beneficial or detrimental depending on the value and the meaning attached to the specific experience as well as people perceive it. Furthermore, questions on the efforts and rewards or on satisfaction gained are only asked for specific activities such as paid work and the workplace. Similarly, the quality of social relationships obtained through civic participation or informal activities as well as the feeling of reciprocity of efforts and rewards or lack thereof are not accounted for. Additionally, the SHARE respondents were not asked to specify the recipient of their care and both questionnaires failed to ask whether caring for sick relatives represented a burden for the carer. Finally, activities such as getting together with friends or just visiting family members to have a chat could not be considered in this research study since neither questionnaire contemplated leisure or solitary activities. It could be argued that those who have a more active social life are less prone to engaging in other formal social activities. Only ELSA enquired about activities such as visits to museum or overnight trips though these questions investigated the average frequency on a yearly basis rather than the actual participation in a specific temporal frame. The datasets used, therefore, did not provide enough information able to capture the complexity of the experiences and activities older people are engaged in, the appraisal of the engagement experience, the meaning attached to it and the personal benefits or satisfaction gained from it.

It is also worth mentioning that the measurements considered in this study were sensitive to the time frame they referred to. The activity indicators mainly focused on the month prior to the interview and this, for certain activities, resulted in 'seasonal' changes. Particularly for formal engagement (see 6.5.1), the percentage of SHARE respondents who declared they were involved at least weekly in the month prior to the interview was significantly lower if the interview was held between July and September (data not shown). Furthermore, when interpreting the longitudinal results one must keep in mind the indicators of both engagement and depression refer to the month prior to the interview (i.e. nothing is known about changes in the rest of the 2-year between-

interview interval); also, changes in health refer to changes in SRH occurring between the two dates of the interviews.

A further limitation concerns the fact that all data rely on self-reports: measurements such as SRH or participation in certain informal and formal activities (such as volunteering) may be sensitive to cultural differences of definitions (Cnaan et al., 1996, Jylhä et al., 1998). Health expectations may vary according to culture: direct gender and cultural comparisons of self-rated health outcomes, in general, should be made with caution.

#### **10.4 IMPLICATIONS OF FINDINGS**

Analyses in this study advance our understanding of the relationship between engagement and health in later life, given that 'active ageing' is emerging as a hallmark of 'new ageing'. Older people aged 50-69 seem to be actively engaged with the wider society, their community and family: overall, almost 40% of SHARE and ELSA respondents aged 50-69 were in paid work, one quarter were engaged in formal activities, and one third provided regular help and care for family members, friends and neighbours. In our sample, older people seem to have opportunities for continuing participation in the society, engaging through a range of different activities (and not just paid work). Older people are therefore recognised as an integral part of families, communities and the wider society thanks to their contributions and their participation in a broader range of activities than those normally associated with the labour market.

Evidence seems to support the WHO concept of 'active ageing' and the connection between activity and health. Indeed, the current research study supports the general assumption that engaging with the wider society contributes to the maintenance of good self-rated health and non-depressive symptoms among European older people aged 50-69 years, despite the above-mentioned restrictions and limitations. In particular, the findings illuminate that engagement in work is an important (but not necessarily causal) pathway to health in late life (Hinterlong et al., 2007, Wahrendorf and Siegrist, 2010). This positive relationship was consistent across Europe, both among men and women.

However, if 'active ageing' is a significant policy initiative which pushes boundaries towards a gradual broadening of the idea of engagement providing new opportunities to older people to combine various forms of participation, new balances between work and other forms of engagement are still to be found. In fact, in all countries of interest in this study, independent of the general level of participation, paid work was negatively associated with the probability to engage in formal or informal activities; on the other hand, a strong positive correlation was found between active involvement in informal activities and the propensity to be engaged in formal activities. Finally, if being 'active' is the new paradigm for older people, overall participation rates also indicate that between 17% (Denmark) and 35% (Italy) of respondents were not regularly engaged in any of the above-mentioned activities.

Overall results also indicate that participation rates and the general health of respondents at baseline and follow-up varied by country, suggesting that policies may exert effects on both occupational levels (particularly among women) through labour policies and social protection measures (Dragano et al., 2010) and health, through the magnitude and generosity of their welfare state regimes, such as policies to tackle unemployment and relative poverty (Bambra and Eikemo, 2009).

The longitudinal association between paid work and health at follow-up, nonetheless, does not necessarily suggest that encouraging a fuller and longer participation of older citizens in the labour market would necessarily contribute towards weakening the detrimental advance of depression and towards increasing the self-rated health of the general population. Improvements in population level participation may not automatically lead to improvements in population health, given that the causal pathway between engagement and health is not straightforward. Similarly, assuming that encouraging older people to stay in paid work -regardless of the health conditions of the worker, the occupational class and the quality of the workplace -is beneficial for their health would be reductive. Indeed, a number of factors such as the well-being in the workplace and work conditions which are equally important in determining both the likelihood of workers to stay in their work and affect their health (Gollac and Volkoff, 2000, Phillipson, 2002b, Debrand and Lengagne, 2007, Siegrist et al., 2007, Alavinia and Burdorf, 2008) were not considered in this study. Even if modern pension reforms have mainly been promulgated in order to limit the increasing cost of an ageing society, it would seem important to pay attention to people's working lives, including the quality

223

of the work environment itself, in order to increase the level of participation in paid work and maintain the health of people as they age.

#### **10.5** FUTURE RESEARCH

Given the limitations mentioned above, more refined analyses are desirable to explore the relationships between engagement and health more precisely. For instance, given that several authors have shown that the beneficial effect of activities also relies on the feeling of reciprocity of efforts and rewards or lack thereof (Wenger, 1997, Jerrome and Wenger, 1999), on the experience of reward, autonomy and personal control, on the level of satisfaction, on the work conditions and quality of work (Gollac and Volkoff, 2000, Debrand and Lengagne, 2007, Siegrist et al., 2007, Alavinia and Burdorf, 2008, Siegrist and Wahrendorf, 2009a) as well as on the quality of social relationships obtained through civic participation or informal activities, it would be helpful to have more detailed information about these aspects of the participation in the various activities. Moreover, it would also be desirable to have data on both high and lowdemand leisure activities (such as theatre or reading or just gathering with friends), given that these are also associated with higher mental health scores (Everard et al., 2000).

Furthermore, more detailed information about the kind and intensity of care or help, and the characteristics of the recipient of care might be helpful, since numerous studies suggest that factors such as satisfaction, intention to care, time and intensity of care as well as the conditions of the recipient and the relationship with the caregiver tend to be more important in 'predicting' health measurements than the activity itself (Hubley et al., 2003, Gruenewald et al., 2007, Yajima et al., 2007). The simplistic assumption that helping a friend carries the same meaning as caring for a family member may partly explain the lack of significant results in our analyses. Evidence also suggests that older people helping their partners or providing support to their friends could derive mental health benefits (Beach et al., 2000, Brown et al., 2003, Pinquart and Sörensen, 2003 ) or generate high levels of distress and depression when the burden becomes too heavy (Schulz and Beach, 1999a; Fredman et al., 2008, Schulz and Sherwood, 2008 , Kuzuya et al., 2011). More detailed information would help us to factor into the model the level

of burden or stress older people experience in the various informal activities they engage in and the way this relates to health.

It would be beneficial to also have larger samples for each of the country under study. For instance, the weak association between formal engagement at baseline and health at follow-up might have been weakened as a result of the low participation at baseline combined with small sample sizes. Effects and associations are harder to detect in smaller samples; ideally, then, it would be important to perform the same analyses using equally comparable surveys with bigger sample sizes in order to boost the statistical power of the relationships and associations found.

Given that a third round of the SHARE questionnaire has recently become available, it would also be interesting to assess whether the same longitudinal relationships would hold if a third wave were considered. Using observations in three time points would allow us to examine whether engagement at baseline and at the first follow-up are longitudinally and equally associated with health at the last point in time. The use of a third wave would also allow us to determine how changes in activities and other socioeconomic characteristics between baseline and the wave two were associated with health at time three, controlling for baseline characteristics. As SHARE continues with further waves, the potential for more detailed analyses increase substantially, although such potential is paralleled with the increasing problems due to attrition.

Finally, although these analyses suggested that some country-specific differences may have been affected by policy/welfare regimes, limited attention to national-level policies and programmes restricts the conclusions that may be drawn in this regard. For example, there was no attention to the degree to which any relevant policies are implemented in each country. Although beyond the scope of this study, additional country-specific research in this area could provide useful insights into policy implications in this field. Moreover, this research project did not link micro-data estimations to macro-data on welfare policies potentially relevant to participation in various activities or health. Future work may investigate variations across Europe in the diversity of the characteristics of older people engaged in various activities relating patterns of engagement to policies, encompassing analyses of both micro- and macro-level data using a range of existing sources, including EUROSTAT and OECD statistics, to assemble contextual information.

## **BIBLIOGRAPHY**

- ADAM, S., BONSANG, É., GERMAIN, S. & PERELMAN, S. 2007. Retraite, activités non professionnelles et vieillissement cognitif. Une exploration à partir des données de Share. *Economie et Statistique*, 403 - 404, 83 - 96.
- ADELMANN, P. K. 1994. Multiple Roles and Psychological Well-being in a National Sample of Older Adults. *J Gerontol*, 49, S277-285.
- AGRESTI, A. 2002. Categorical Data Analysis, Hoboken, New Jersey, John Wiley & Sons.
- ALAVINIA, S. M. & BURDORF, A. 2008. Unemployment and retirement and illhealth: a cross-sectional analysis across European countries International Archives of Occupational and Environmental Health, 82, 39-45.
- ALBERTINI, M., KOHLI, M. & VOGEL, C. 2007. Intergenerational transfers of time and money in European families: common patterns different regimes? *Journal of European Social Policy*, 17, 319-334.
- ANANTH, C. V. & KLEINBAUM, D. G. 1997. Regression models for ordinal responses: a review of methods and applications. *International Journal of Epidemiology*, 26, 1323-1333.
- ANDERSON, J. A. & PHILIPS, P. R. 1981. Regression, Discrimination and Measurement Models for Ordered Categorical Variables. Journal of the Royal Statistical Society. Series C (Applied Statistics), 30, 22-31.
- ANDREOTTI, A., GARCIA, S. M., GOMEZ, A., HESPANHA, P., KAZEPOV, Y. & MINGIONE, E. 2001. Does a Southern European Model Exist? Journal of European Area Studies, 9.
- ANGERMEYER, M. C., HOLZINGER, A., MATSCHINGER, H. & STENGLER-WENZKE, K. 2002. Depression and Quality of Life: Results of a Follow-Up Study. International Journal of Social Psychiatry, 48, 189-199.
- ANHEIER, H. K. & SALAMON, L. M. 1999. Volunteering in Cross-National Perspective: Initial Comparisons. Law and Contemporary Problems, 62, 43-65.
- ARRIGHI, H. M. & HERTZ-PICCIOTTO, I. 1994. The Evolving Concept of the Healthy Worker Survivor Effect. *Epidemiology*, 5, 189-196.
- ARTS, W. & GELISSEN, J. 2002. Three worlds of capitalism or more? A state-of-theart report. Journal of European Social Policy, 12, 137-158.
- ASKHAM, J., FERRING, D. & LAMURA, G. 2007. Personal Relationships in Later Life In: BOND, J., PEACE, S., DITTMANN-KOHLI, F. & WESTERHOF, G. J. (eds.) Ageing in society. European perspectives on gerontology (third edition). London: SAGE publications.
- ATCHLEY, R. C. 1971. Retirement and Leisure Participation: Continuity or Crisis? *The Gerontologist*, 11, 13-17.
- ATTIAS-DONFUT, C., OGG, J. & WOLFF, F.-C. 2005. European patterns of intergenerational financial and time transfers. *European Journal of Ageing*, 2, 161-173.
- AURELI CUTILLO, E. 2002. Lezioni di Statistica Sociale. Fonti, strumenti e metodi., Roma, CISU, Centro di Informazione e Stampa Universitaria.

- AVENDANO, M., GLYMOUR, M., BANKS, J. & MACKENBACH, J. 2009. Health Disadvantage in US Adults Aged 50 to 74 Years: A Comparison of the Health of Rich and Poor Americans With That of Europeans. Am J Public Health, 99, 540-548.
- AVERINA, M., NILSSEN, O., BRENN, T., BROX, J., ARKHIPOVSKY, V. L. & KALININ, A. G. 2005. Social and lifestyle determinants of depression, anxiety, sleeping disorders and self-evaluated quality of life in Russia. Social Psychiatry and Psychiatric Epidemiology, 40, 511-518.
- AVRAMOV, D. & MASKOVA, M. 2003. Active Ageing in Europe –Volume 1 (Population Studies Series No. 41), Strasbourg, Council of Europe Publishing.
- AYSAN, M. F. Welfare regimes for ageing populations: divergence or convergence? XXVI IUSSP International Population Conference, 2009 Marrakech Morocco.
- BADLEY, E. M., WANG, P. P., COTT, C. A. & GIGNAC, M. A. M. 2000. Determinants of Changes in Self-Reported Health and Outcomes Associated with those Changes: a report of the longitudinal components of the 1994–1996 National Population Health Survey. University Health Network.
- BAGO D'UVA, T., O'DONNELL, O. & VAN DOORSLAER, E. 2008. Differential health reporting by education level and its impact on the measurement of health inequalities among older Europeans. *Int. J. Epidemiol.*, 37, 1375-1383.
- BAMBRA, C. 2006. Health Status and the Worlds of Welfare. Social Policy and Society, 5, 53-62.
- BAMBRA, C. 2007a. Defamilisation and welfare state regimes: a cluster analysis. International Journal of Social Welfare, 16, 326-338.
- BAMBRA, C. 2007b. Going beyond The three worlds of welfare capitalism: regime theory and public health research. Journal of Epidemiology and Community Health, 61, 1098-1102.
- BAMBRA, C. & EIKEMO, T. A. 2009. Welfare state regimes, unemployment and health: a comparative study of the relationship between unemployment and selfreported health in 23 European countries. Journal of Epidemiology and Community Health, 63, 92-98.
- BANKS, J., BREEZE, E., LESSOF, C. & NAZROO, J. 2006. Retirement, health and relationships of the older population in England: The 2004 English Longitudinal Study of Ageing (Wave 2), London, Institute for Fiscal Studies.
- BARANGÉ, C., EUDIER, V. & SIRVEN, N. 2008. SHARE, the Survey of Health, Ageing and Retirement in Europe goes longitudinal. *Questions d'économie de la* santé 137, 1-4.
- BARNAY, T. 2006. Health impact on the labour force. For IRDES (Institute of Research and Documentation in Health Economics).
- BARNAY, T. & DEBRAND, T. 2006. Effects of health on the labour force participation of older persons in Europe. *Health Economics Letter*, Issues in health economics IRDES
- BAROUDI, J. J. & ORLIKOWSKI, W. J. 1989. The Problem of Statistical Power in MIS Research. *MIS Quarterly*, 13, 87-106.
- BARTLEY, M. 1994. Unemployment and ill health: understanding the relationship. Journal of Epidemiology and Community Health, 48, 333-7.

- BASS, S. A. & CARO, F. G. 2001. Productive Aging: A Conceptual Framework. In: MORROW-HOWELL, N., HINTERLONG, J. & SHERRADEN, M. (eds.) Productive Aging: Concepts and Challenges. Baltimore: Johns Hopkins University Press.
- BATH, P. A. & DEEG, D. 2005. Social engagement and health outcomes among older people: introduction to a special section. *European Journal of Ageing*, Volume 2, 24-30.
- BEACH, S. R., SCHULZ, R., YEE, J. L. & JACKSON, S. 2000. Negative and Positive Health Effects of Caring for a Disabled Spouse: Longitudinal Findings From the Caregiver Health Effects Study. *Psychology and Aging*, 15, 259-271.
- BECKER, S., BENTOLILA, S., FERNANDES, A. & ICHINO, A. 2010. Youth emancipation and perceived job insecurity of parents and children. Journal of Population Economics, 23, 1047-1071.
- BEHR, A., BELLGARDT, E. & RENDTEL, U. 2005. Extent and Determinants of Panel Attrition in the European Community Household Panel. *European Sociological Review*, 21, 489-512.
- BERKMAN, L. F., GLASS, T., BRISSETTE, I. & SEEMAN, T. E. 2000. From social integration to health: Durkheim in the new millennium. *Social Science & Medicine*, 51, 843-857.
- BETTIO, F. & PLANTENGA, J. 2004. Comparing Care Regimes in Europe. Feminist Economics, 10, 85-113.
- BLACK, S. A., MARKIDES, K. & RAY, L. A. 2003. Depression Predicts Increased Incidence of Adverse Health Outcomes in Older Mexican Americans With Type 2 Diabetes. *Diabetes Care*, 26, 2822-2828.
- BLANCHET, D. & DEBRAND, T. 2007. Souhaiter prendre sa retraite le plus tôt possible: santé, satisfaction au travail et facteurs monétaires. *Economie et statistique* 403 404, 39-62.
- BLANCHET, D., DEBRAND, T., DOURGNON, P. & LAFERRÈRE, A. 2007. Santé, vieillissement et retraite en Europe. *Economie et Statistique*, 403-404, 3-18.
- BLANCHET, D. & PELÉ, L.-P. 1999. Social Security and Retirement In France. In: GRUBER, J. & WISE, D. A. (eds.) Social Security and Retirement around the World. Chicago: NBER (National Bureau of Economic Research).
- BLUNDELL, R., EMMERSON, C. & WAKEFIELD, M. 2006. The importance of incentives in influencing private retirement saving: known knowns and known unknowns. London: The Institute for Fiscal Studies.
- BLUNDELL, R. & JOHNSON, P. 1999. Pensions and Retirement in the United Kingdom. In: GRUBER, J. & WISE, D. A. (eds.) Social Security and Retirement around the World. Chicago: NBER (National Bureau of Economic Research).
- BOHNKE, P. 2008. Are the poor socially integrated? The link between poverty and social support in different welfare regimes. *Journal of European Social Policy*, 18, 133-150.
- BOLIN, K., LINDGREN, B. & LUNDBORG, P. 2008. Your next of kin or your own career?: Caring and working among the 50+ of Europe. Journal of Health Economics, 27, 718-738.
- BONGAARTS, J. 2004. Population Aging and the Rising Cost of Public Pensions. Population and Development Review, 30, 1-23.

- BONOLI, G. 1997. Classifying Welfare States: a Two-dimension Approach. Journal of Social Policy, 26, 351-372.
- BORSCH-SUPAN, A. 2005. Introduction. In: AUTHORS, V. (ed.) Health, Ageing and Retirement in Europe. First Results from the Survey of Health, Ageing and Retirement in Europe. Mannheim: Research Institute for the Economics of Aging (MEA).
- BÖRSCH-SUPAN, A., BRUGIAVINI, A. & CRODA, E. 2009. The role of institutions and health in European patterns of work and retirement. *Journal of European Social Policy*, 19, 341-358.
- BÖRSCH-SUPAN, A., BRUGIAVINI, A., JÜRGES, H., MACKENBACH, J., SIEGRIST, J. & WEBER, G. 2005. Health, Ageing and Retirement in Europe. First Results from the Survey of Health, Ageing and Retirement in Europe, Mannheim, Mannheim Research Institute for the Economics of Aging (MEA).
- BÖRSCH-SUPAN, A. & JÜRGES, H. 2005. The Survey of Health, Aging, and Retirement in Europe Methodology, Mannheim, Research Institute for the Economics of Aging.
- BÖRSCH-SUPAN, A. & KEMPERMAN, M.-L. 2005. The SHARE Development Process. In: BÖRSCH-SUPAN, A. & JÜRGES, H. (eds.) The Survey of Health, Aging, and Retirement in Europe – Methodology. Mannheim: Mannheim Research Institute for the Economics of Aging (MEA).
- BOSTRUM, G., HALLQUIST, J., HAGLUND, B. J., ROMELSJÖ, A., SVANSTRÖM, L. & DIDERICHSEN, F. 1993. Socioeconomic differences in smoking in an urban Swedish population. The bias introduced by non-participation in a mailed questionnaire. Scand J Social Med, 21, 77-82.
- BOUCHARD, C. 2001. Physical activity and health: introduction to the dose-response symposium. *Medicine & Science in Sports & Exercise*, 33, S347-S350.
- BOWLING, A. 2005. Ageing Well Quality of Life in Old Age, Maidenhead, Open University Press.
- BOWLING, A. & DIEPPE, P. 2005. What is successful ageing and who should define it? BMJ, 331, 1548-1551.
- BOWLING, A. & GRUNDY, E. 1998. The association between social networks and mortality in later life. *Reviews in Clinical Gerontology*, 8, 353-361.
- BOWLING, A. & ILIFFE, S. 2006. Which model of successful ageing should be used? Baseline findings from a British longitudinal survey of ageing. *Age Ageing*, 35, 607-614.
- BOWLING, A. & WINDSOR, J. 2008. The effects of question order and responsechoice on self-rated health status in the English Longitudinal Study of Ageing (ELSA). Journal of Epidemiology and Community Health, 62, 81-85.
- BRICK, J. M., LÊ, T. & WEST, J. Dealing with movers in a longitudinal study of children. In: CANADA, S., ed. Statistics Canada Symposium - Challenges in Survey Taking for the Next Decade, 2003.
- BROWN, S. L., NESSE, R. M., VINOKUR, A. D. & SMITH, D. M. 2003. Providing Social Support May Be More Beneficial Than Receiving It. *Psychological Science*, 14, 320-327.

- BRUGIAVINI, A. 1999. Social Security and Retirement in Italy. In: GRUBER, J. & WISE, D. A. (eds.) Social Security and Retirement around the World. Chicago: NBER (National Bureau of Economic Research).
- BUCKLEY, N. J., DENTON, F. T., LESLIE ROBB, A. & SPENCER, B. G. 2004. The transition from good to poor health: an econometric study of the older population. *Journal of Health Economics*, 23, 1013-1034.
- BUKOV, A., MAAS, I. & LAMPERT, T. 2002. Social Participation in Very Old Age: Cross-Sectional and Longitudinal Findings From BASE. J Gerontol B Psychol Sci Soc Sci, 57, P510-517.
- BURHOLT, V., WENGER, G. C., LAMURA, G., PAULSSON, C., VAN DER MEER, M., FERRING, D. & GLUECK, J. 2003. European Study of Adult Well-Being: Comparative Report on Social Support Resources (for the European Commission). Centre for Social Policy Research and Development, University of Wales, Bangor. .
- BURR, J. A., CHOI, N. G., MUTCHLER, J. E. & CARO, F. G. 2005. Caregiving and Volunteering: Are Private and Public Helping Behaviors Linked? *The Journals* of Gerontology Series B: Psychological Sciences and Social Sciences, 60, S247-S256.
- BURR, J. A., MUTCHLER, J. E. & CARO, F. G. 2007. Productive Activity Clusters Among Middle-Aged and Older Adults: Intersecting Forms and Time Commitments. J Gerontol B Psychol Sci Soc Sci, 62, S267-275.
- BURTON, A. & ALTMAN, D. G. 2004. Missing covariate data within cancer prognostic studies: a review of current reporting and proposed guidelines. *British Journal of Cancer*, 91, 4-8.
- BUTLER, R. N. & GLEASON, H. P. 1985. Productive aging: Enhancing vitality in later life, New York, Springer.
- BUTTERWORTH, P., LEACH, L. S., STRAZDINS, L., OLESEN, S. C., RODGERS,
  B. & BROOM, D. H. 2011. The psychosocial quality of work determines whether employment has benefits for mental health: results from a longitudinal national household panel survey. Occupational and Environmental Medicine.
- CAGIANO DE AZEVEDO, R. & CASSANI, B. 2005. Invecchiamento e svecchiamento demografico: ricadute sociali. Stastica & Società, IV, 21-27.
- CALASANTI, T. M. & BONANNO, A. 1986. The social creation of dependence, dependency ratios, and the elderly in the United States: A critical analysis. Social Science & Comp.; Medicine, 23, 1229-1236.
- CALVO, E. 2006. Does Working Longer Make People Healthier and Happier? [Online]. Available: <u>http://mpra.ub.uni-muenchen.de/5606/</u> [Accessed 13 October 2009.
- CARO, F. G. & BASS, S. A. 1997. Receptivity to Volunteering in the Immediate Postretirement Period. Journal of Applied Gerontology, 16, 427-441.
- CARO, F. G., BRUNER-CANHOTO, L., BURR, J. A. & MUTCHLER, J. E. 2005. Motivation for active ageing: results for a pilot study. *In:* WORKING PAPER, G. I. (ed.). Boston: University of Massachussetts.
- CARR, D. C. & KOMP, K. S. 2011. Gerontology in the Era of the Third Age. Implications and Next Steps, New York, Springer.

- CASTLES, F. G. & MITCHELL, D. A. D., ), . 1993. Worlds of Welfare and Families of Nations. In: CASTLES, F. G. (ed.) Families of Nations. Patterns of Public Policy in Western Democracies. Aldershot: Dartmouth Publishing.
- CELANO, C. M. & HUFFMAN, J. 2011. Depression and Cardiac Disease: A Review. Cardiology in Review, 19, 130-142.
- CHATZITHEOCHARI, S. & ARBER, S. 2011. Identifying the Third Agers: An Analysis of British Retirees' Leisure Pursuits. Sociological Research Online, 16, 3.
- CHESHIRE, H., COX, K., LESSOF, C. & TAYLOR, R. 2006. Methodology In: BANKS, J., BREEZE, E., LESSOF, C. & NAZROO, J. (eds.) Retirement, health and relationships of the older population in England: THE 2004 ENGLISH LONGITUDINAL STUDY OF AGEING (Wave 2). London: The Institute for Fiscal Studies.
- CHOI, N. G. 2003. Determinants of self-perceived changes in health status among preand early-retirement populations. *International Journal of Ageing and Human development*, 56, 197-222.
- CHOI, N. G., BURR, J. A., MUTCHLER, J. E. & CARO, F. G. 2007. Formal and Informal Volunteer Activity and Spousal Caregiving Among Older Adults. *Research on Aging*, 29, 99-124.
- CHRISTELIS, D. 2008. Item Non-response. In: BÖRSCH-SUPAN, A., BRUGIAVINI, A., JÜRGES, H., KAPTEYN, A., MACKENBACH, J., SIEGRIST, J. & WEBER, G. (eds.) First Results from the Survey of Health, Ageing and Retirement in Europe (2004-2007). Starting the Longitudinal Dimension. Mannheim: Research Institute for the Economics of Aging (MEA).
- CHUNG, H. & MUNTANER, C. 2007. Welfare state matters: a typological multilevel analysis of wealthy countries. *Health Policy*, 80, 328-339.
- CLARKE, L., EVANDROU, M. & WARR, P. 2005. Family and Economic Roles In: WALKER, A. (ed.) Understanding Quality of Life in Old Age. Maidenhead: Open University Press.
- CNAAN, R. A., HANDY, F. & WADSWORTH, M. 1996. Defining Who is a Volunteer: Conceptual and Empirical Considerations. *Nonprofit and Voluntary Sector Quarterly*, 25, 364-383.
- COBB, S. 1976. Presidential Address-1976. Social support as a moderator of life stress. *Psychosom Med*, 38, 300-314.
- COHEN, P. N. & BIANCHI, S. M. 1999. Marriage, children, and women's employment: what do we know? *Monthly Labour Review*, 22-31.
- COHEN, S. 1988. Psychosocial models of social support in the etiology of physical disease. *Health Psychology*, 7, 269-297.
- COILE, C. 2004. Retirement Incentives and Couples' Retirement Decisions. Topics in Economic Analysis & Policy., 4.
- COLEMAN, D. A. 2003. Mass migration and population change. Zeitschrift für Bevölkerungswissenschaft, 28, 183-215.
- COMITÉ EUROPÉEN SUR LA POPULATION DU CONSEIL DE L'EUROPE 2005. Evolution démographique récente en Europe 2004, Strasbourg, Cedex.

- COMMISSION OF THE EC. 1999a. New Paradigm in Ageing Policy Conference communication, [Online]. Available: <u>http://ec.europa.eu/employment\_social/soc-prot/ageing/news/paradigm\_en.htm</u> [Accessed 18/10/2009.
- COMMISSION OF THE EC 1999b. Towards A Europe for All Ages Promoting Prosperity and Intergenerational Solidarity. Brussels: COM (1999) 221 final.
- CONNELLY, J. E., PHILBRICK, J. T., SMITH, G. R. J., KAISER, D. L. & WYMER, A. 1989. Health perceptions of primary care patients and the influence on health care utilization. *Medical Care*, 27, S99-109.
- CRIMMINS, E. M., KIM, J. K. & SOLÉ-AURÓ, A. 2011. Gender differences in health: results from SHARE, ELSA and HRS. *The European Journal of Public Health*, 21, 81-91.
- CUMMING, E. & HENRY, W. E. 1961. Growing old. The process of disengagement, New York, Basic Books.
- DANIELS M. J. & HOGAN, J. W. 2008. Missing data in longitudinal studies: Strategies for Bayesian modeling and sensitivity analysis, Boca Raton, Chapman & Hall.
- DANNA, D. 2008. La demolizione dello stato sociale e le disparità di genere La finanziarizzazione dell'economia mondiale e la sua crisi : come restituire potere ai cittadini? Pomezia.
- DAVEY, J. A. 2002. Active Ageing and education in mid and later life. Ageing & Society, 22, 95-113.
- DE BEER, P. 2007. Why work is not a panacea: a decomposition analysis of EU-15 countries. *Journal of European Social Policy*, 17, 375-388.
- DE KOKER, B. 2009. Socio-demographic determinants of informal caregiving: coresident versus extra-resident care. European Journal of Ageing, 6, 3-15.
- DE LEEUW, E. D., HOX, J. & HUISMAN, M. 2003. Prevention and Treatment of Item Nonresponse. Journal of Official Statistics, 19, 153-176.
- DE LUCA, G. & PERACCHI, F. 2005a. Survey Participation in the First Wave of SHARE. In: BÖRSCH-SUPAN, A. & JÜRGES, H. (eds.) The Survey of Health, Aging, and Retirement in Europe – Methodology. Mannheim: Research Institute for the Economics of Aging (MEA).
- DE LUCA, G. & PERACCHI, F. 2005b. Survey Response. In: BORSCH-SUPAN, A., JURGES, H., BRUGIAVINI, A., MACKENBACH, J., SIEGRIST, J. & WEBER, G. (eds.) Health, Ageing and Retirement in Europe. First Results from the Survey of Health, Ageing and Retirement in Europe. Mannheim: Research Institute for the Economics of Aging (MEA).
- DE LUCA, G. & ROSSETTI, C. 2008. Sampling Design and Weighting Strategies in the Second wave of SHARE. In: BÖRSCH-SUPAN, A., BRUGIAVINI, A., JÜRGES, H., KAPTEYN, A., MACKENBACH, J., SIEGRIST, J. & WEBER, G. (eds.) First Results from the Survey of Health, Ageing and Retirement in Europe (2004-2007). Starting the Longitudinal Dimension. Mannheim: Research Institute for the Economics of Aging (MEA).
- DEBRAND, T. & LENGAGNE, P. 2007. Pénibilité au travail et santé des seniors en Europe? Economie et Statistique, 403-404, 19-38.
- DEEMING, C. 2009. 'Active ageing' in practice: a case study in East London, UK. Policy & Politics, 37, 93-111.

- DENTINGER, E. & CLARKBERG, M. 2002a. Informal Caregiving and Retirement Timing among Men and Women. Journal of Family Issues, 23, 857-879.
- DENTINGER, E. & CLARKBERG, M. 2002b. Informal Caregiving and Retirement Timing among Men and Women: Gender and Caregiving Relationships in Late Midlife. *Journal of Family Issues*, 23, 857-879.
- DEPARTMENT OF WORK AND PENSIONS 2010. The Pensioners' Incomes Series 2008-09. In: EVANS, J. & HEATHER, R. (eds.).
- DEVILLE, J. C. & SÄRNDAL, C. E. 1992. Calibration Estimators in Survey Sampling. Journal of the American Statistical Association, 87, 376-82.
- DEWEY, M. & PRINCE, M. J. 2005. Mental Health. In: AUTHORS, V. (ed.) Health, Ageing and Retirement in Europe First Results from the Survey of Health, Ageing and Retirement in Europe. Mannheim: Research Institute for the Economics of Ageing.
- DOWD, J. B. & ZAJACOVA, A. 2010. Does Self-Rated Health Mean the Same Thing Across Socioeconomic Groups? Evidence From Biomarker Data. Annals of Epidemiology, 20, 743-749.
- DRAGANO, N., SIEGRIST, J. & WAHRENDORF, M. 2010. Welfare regimes, labour policies and unhealthy psychosocial working conditions: a comparative study with 9917 older employees from 12 European countries. Journal of Epidemiology and Community Health.
- EGIDI, V. 1988. Stato di salute e morbosità della popolazione, Secondo rapporto sulla situazione demografica italiana, Roma, IRP-CNR.
- EIKEMO, T. A., BAMBRA, C., JUDGE, K. & RINGDAL, K. 2008. Welfare state regimes and differences in self-perceived health in Europe: A multilevel analysis. Social Science & Medicine, 66, 2281-2295.
- ELSA. English Longitudinal Study of Ageing [Online]. http://www.ifs.org.uk/elsa/index.php.
- ERENS, B. & PRIMATESTA, P. 1999. Health Survey for England 1998. Volume 2: Methodology & Documentation. In: ERENS, B. & PRIMATESTA, P. (eds.). London: The Stationery Office.
- ERENS, B., PRIMATESTA, P. & PRIOR, G. 2001. Health Survey for England. The Health of Minority Ethnic Groups 1999. Volume 2: Methodology & Documentation,. *In:* ERENS, B., PRIMATESTA, P. & PRIOR, G. (eds.). London: The Stationery Office.
- ERLINGHAGEN, M. & HANK, K. 2006. The participation of older Europeans in volunteer work. Ageing & Society, 26, 567-584.
- ERVIK, R. 2006. Pitfalls and Possibilities of Active Ageing. Conceptualisations and Policies. Towards a Multiage Society: The Cultural Dimension of Age Policies. Brussels.
- ESPELT, A., BORRELL, C., RODRIGUEZ-SANZ, M., MUNTANER, C., PASARIN, M. I., BENACH, J., SCHAAP, M., KUNST, A. E. & NAVARRO, V. 2008. Inequalities in health by social class dimensions in European countries of different political traditions. *Int. J. Epidemiol.*, 37, 1095-1105.
- ESPING-ANDERSEN, G. 1990. The three worlds of welfare capitalism, Oxford, Polity Press.

- ETILÉ, F. & MILCENT, C. 2006. Income-related reporting heterogeneity in selfassessed health: evidence from France. *Health Economics*, 15, 965-981.
- EUROSTAT 2008. Europe in figures Eurostat yearbook 2008. Luxembourg: Office for Official Publications of the European Communities: European Commission.
- EUROSTAT, E. C.-. 2009. Key Data on Education in Europe, Brussels, Education, Audiovisual and Culture Executive Agency (EACEA P9 Eurydice).
- EVERARD, K. M., LACH, H. W., FISHER, E. B. & BAUM, M. C. 2000. Relationship of Activity and Social Support to the Functional Health of Older Adults. J Gerontol B Psychol Sci Soc Sci, 55, S208-212.
- EVERSON-ROSE, S. A., HOUSE, J. S. & MERO, R. P. 2004. Depressive Symptoms and Mortality Risk in a National Sample: Confounding Effects of Health Status. *Psychosomatic Medicine*, 66, 823-830.
- FERRERA, M. 1996. The 'Southern Model' of Welfare in Social Europe. Journal of European Social Policy, 6, 17-37.
- FISCHER, K., RAPKIN, B. D. & RAPPAPORT, J. 1991. GENDER AND WORK HISTORY IN THE PLACEMENT AND PERCEPTIONS OF ELDER COMMUNITY VOLUNTEERS. *Psychology of Women Quarterly*, 15, 261-279.
- FITZGERALD, J., GOTTSCHALK, P. & MOFFIT, R. 1998. An Analysis of Sample Attrition in Panel Data. The Journal of Human Resources. Special Issue: Attrition in Longitudinal Surveys, 33 251-299.
- FOGLIANI, M. 1999. Low Response Rates and Their Effects on Survey Results. Methodology Advisory Committee paper.
- FREDMAN, L., CAULEY, J. A., SATTERFIELD, S., SIMONSICK, E., SPENCER, S. M., AYONAYON, H. N., HARRIS, T. B. & FOR THE HEALTH ABC STUDY GROUP 2008. Caregiving, Mortality, and Mobility Decline: The Health, Aging, and Body Composition (Health ABC) Study. Arch Intern Med, 168, 2154-2162.
- FRIED, L. P., BANDEEN-ROCHE, K., KASPER, J. D. & GURALNIK, J. M. 1999. Association of Comorbidity with Disability in Older Women: The Women's Health and Aging Study. *Journal of Clinical Epidemiology*, 52, 27-37.
- GALLO, W. T., BRADLEY, E. H., SIEGEL, M. & KASL, S. V. 2000. Health Effects of Involuntary Job Loss Among Older Workers: Findings From the Health and Retirement Survey. J Gerontol B Psychol Sci Soc Sci, 55, S131-140.
- GIULIANO, P. 2007. "Ties that Matter: Cultural Norms and Economic Behavior in Western Europe". Prepared for the "Labor Market Institutions and Public Policy Responses: A Symposium to Honor Lloyd Ulman" conference held on 27th Oct 2007.
- GLASER, K., EVANDROU, M. & TOMASSINI, C. 2005. The health consequences of multiple roles at older ages in the UK. *Health & Social Care in the Community*, 13, 470-477.
- GLASER, K., TOMASSINI, C. & GRUNDY, E. 2004. Revisiting convergence and divergence: support for older people in Europe. *European Journal of Ageing*, 1, 64-72.
- GLASS, T. A., MENDES DE LEON, C., MAROTTOLI, R. A. & BERKMAN, L. F. 1999. Population based study of social and productive activities as predictors of survival among elderly Americans. *BMJ*, 319, 478-483.

- GOLDSTEIN, M. S., SIEGEL, J. M. & BOYER, R. 1984. Predicting changes in perceived health status. Am J Public Health, 74, 611-614.
- GOLLAC, M. & VOLKOFF, S. 2000. Les conditions de travail, Paris, Éditions La Découverte.
- GOODE, W. J. 1960. A Theory of Role Strain. American Sociological Review, 25, 483-496.
- GOODWIN, R. D. 2003. Association between physical activity and mental disorders among adults in the United States. *Preventive Medicine*, 36, 698-703.
- GOTLIBOVSKI, C. & WEISS, Y. 2008. Retirement Decisions of Married Couples. Social security - Journal of welfare and social security studies, 76, 73-96.
- GREENLAND, S., SCHWARTZBAUM, J. A. & FINKLE, W. D. 2000. Problems due to Small Samples and Sparse Data in Conditional Logistic Regression Analysis. *Am. J. Epidemiol.*, 151, 531-539.
- GROVES, R. & COUPER, M. 1998. Nonresponse in Household Interview Surveys, New York, John Wiley & Sons.
- GROVES, R. M. 2006. Nonresponse Rates and Nonresponse Bias in Household Surveys. *Public Opinion Quarterly*, 70, 646-675.
- GROVES, R. M., SINGER, E. & CORNING, A. 2000. Leverage-Saliency Theory of Survey Participation: Description and an Illustration. *Public Opinion Quarterly*, 64, 299-308.
- GRUENEWALD, T. L., KARLAMANGLA, A. S., GREENDALE, G. A., SINGER, B.
  H. & SEEMAN, T. E. 2007. Feelings of Usefulness to Others, Disability, and Mortality in Older Adults: the MacArthur Study of Successful Aging

J Gerontol B Psychol Sci Soc Sci, 62, 28-37.

- GRUNDY, E. 2010. The care of older people in the United Kingdom: problems, provision and policy. In: CEPAL (ed.) Experiencias internacionales y propuestas para consolidar la red nacional de cuido de las personas adultas mayores en Costa Rica. San José.
- GRUNDY, E. & HOLT, G. 2001. The socioeconomic status of older adults: How should we measure it in studies of health inequalities? J Epidemiol Community Health, 55, 895-904.
- GRUNDY, E. & SLOGGETT, A. 2003. Health inequalities in the older population: the role of personal capital, social resources and socio-economic circumstances. *Social Science & Medicine*, 56, 935-947.
- GUTHRIE, J. R., DUDLEY, E. C., DENNERSTEIN, L. & HOPPER, J. L. 1997. Changes in physical activity and health outcomes in a population-based cohort of mid-life Australian-born women. *Australian and New Zealand Journal of Public Health*, 21, 682-687.
- HAN, B. 2002. Depressive Symptoms and Self-Rated Health in Community-Dwelling Older Adults: A Longitudinal Study. *Journal of the American Geriatrics Society*, 50, 1549-1556.
- HANK, K. 2010. Societal Determinants of Productive Aging: A Multilevel Analysis Across 11 European Countries. *Eur Sociol Rev*, jcq023.
- HANK, K. 2011. How "Successful" Do Older Europeans Age? Findings From SHARE. The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 66B, 230-236.

- HANK, K. & ERLINGHAGEN, M. 2010. Dynamics of Volunteering in Older Europeans. The Gerontologist, 50, 170-178.
- HANK, K. & STUCK, S. 2007. Volunteer Work, Informal Help, and Care among the 50+ in Europe: Further Evidence for 'Linked' Productive Activities at Older Ages. Discussion Papers of DIW Berlin.
- HANK, K. & STUCK, S. 2008. Volunteer work, informal help, and care among the 50+ in Europe: Further evidence for `linked' productive activities at older ages. Social Science Research, 37, 1280-1291.
- HAO, Y. 2008. Productive Activities and Psychological Well-Being Among Older Adults. J Gerontol B Psychol Sci Soc Sci, 63, 864-72.
- HARKNESS, J. 2005. SHARE translation procedures and translation assessment. In: BORSCH-SUPAN, A. & JURGES, H. (eds.) The Survey of Health, Ageing and Retirement in Europe - Methodology. Mannheim: Research Institute for the Economics of Ageing (MEA).
- HARTSHORN, T. A. 1992. Interpreting the City: An Urban Geography, Hoboken, New Jersey, John Wiley & Sons.
- HASKI-LEVENTHAL, D. 2009. Elderly Volunteering and Well-Being: A Cross-European Comparison Based on SHARE Data. Voluntas: International Journal of Voluntary and Nonprofit Organizations, 20, 388-404.
- HAVIGHURST, R. 1963. Successful Aging. In: WILLIAMS, R. H., TIBBITS, C. & DONAHUE, W. (eds.) Processes of Aging, vol 1. Chicago: University of Chicago Press.
- HAWKES, D. & PLEWIS, I. 2006. Modelling non-response in the National Child Development Study. Journal of the Royal Statistical Society: Series A (Statistics in Society), 169, 479-491.
- HELIÖVAARA, M., AROMAA, A., KLAUKKA, T., KNEKT, P., JOUKAMAA, M. & IMPIVAARA, O. 1993. Reliability and validity of interview data on chronic diseases The mini-Finland health survey. Journal of Clinical Epidemiology, 46, 181-191.
- HENKENS, K. 1999. Retirement intentions and spousal support: a multi-actor approach. The Journal of Gerontology. Series B. Psychological Science and Social Science, 54, S63-73.
- HENRETTA, J. C., GRUNDY, E. & HARRIS, S. 2002. The influence of socioeconomic and health differences on parents' provision of help to adult children: a British–United States comparison. Ageing & Society, 22, 441-458.
- HERZOG, A. R., KAHN, R. L., MORGAN, J. N., JACKSON, J. S. & ANTONUCCI, T. C. 1989. Age Differences in Productive Activities. *Journal of Gerontology*, 44, S129-S138.
- HERZOG, A. R. & MORGAN, J. N. 1992. Age and Gender Differences in the Value of Productive Activities: Four Different Approaches. *Research on Aging*, 14, 169-198.
- HILL, A., ROBERTS, J., EWINGS, P. & GUNNELL, D. 1997. Non-response bias in a lifestyle survey. *Journal of Public Health*, 19, 203-207.
- HINTERLONG, J. E., MORROW-HOWELL, N. & ROZARIO, P. A. 2007. Productive Engagement and Late Life Physical and Mental Health. *Research on Aging*, 29, 348-370.

- HIRST, M. 2003. Caring-related inequalities in psychological distress in Britain during the 1990s. Journal of Public Health, 25, 336-343.
- HOLMES, W. & JOSEPH, J. 2011. Social participation and healthy ageing: a neglected, significant protective factor for chronic non communicable conditions. *Globalization and Health*, 7, 43.
- HOSMER, D. W. & LEMESHOW, S. 2000. Applied logistic regression New York, John Wiley & Sons.
- HOUSE, J. S., ROBBINS, C. & METZNER, H. L. 1982. The associations of social relationships and activities with mortality: prospective evidence from the Tecumseh community health study. *American Journal of Epidemiology*, 116, 123-140.
- HSIAO, C. 2006. Panel Data Analysis Advantages and Challenges [Online]. Institute of Economic Policy Research (IEPR). Available: http://ideas.repec.org/p/scp/wpaper/06-49.html.
- HUBLEY, A. M., HEMINGWAY, D. & MICHALOS, A. C. 2003. A Comparison of Older Informal Caregivers and Non-Caregivers Living in Non-Metropolitan Areas. Social Indicators Research, 61, 241-258.
- HUDSON, J. & KÜHNER, S. 2009. Towards productive welfare? A comparative analysis of 23 OECD countries. Journal of European Social Policy, 19, 34-46.
- HUGHES, M. E., WAITE, L. J., LAPIERRE, T. A. & LUO, Y. 2007. All in the Family: The Impact of Caring for Grandchildren on Grandparents' Health. J Gerontol B Psychol Sci Soc Sci, 62, S108-119.
- HULT, C., STATTIN, M., JANLERT, U. & JÄRVHOLM, B. 2010. Timing of retirement and mortality A cohort study of Swedish construction workers. Social Science & Medicine, 70, 1480-1486.
- HUTCHISON, T., MORRISON, P. & MIKHAILOVICH, K. 2006. A review of the literature on Active Ageing, Canberra.
- HYDE, M., WIGGINS, R. D., HIGGS, P. & BLANE, D. B. 2003. A measure of quality of life in early old age: the theory, development and properties of a needs satisfaction model (CASP-19). Aging & Mental Health, 7, 186 - 194.
- IDLER, E. L. & BENYAMINI, Y. 1997. Self-Rated Health and Mortality: A Review of Twenty-Seven Community Studies. Journal of Health and Social Behavior, 38, 21-37.
- INNSTRAND, S. T., LANGBALLE, E. M. & FALKUM, E. 2011. A Longitudinal Study of the Relationship between Work Engagement and Symptoms of Anxiety and Depression. *Stress and Health*, n/a-n/a.
- JÆGER, M. M. 2006. Welfare Regimes and Attitudes Towards Redistribution: The Regime Hypothesis Revisited. *European Sociological Review*, 22, 157-170.
- JANG, Y., CHIRIBOGA, D., KIM, G. & CHO, S. 2009. Changes in Perceived Health and Depressive Symptoms: A Longitudinal Analysis with Older Korean Americans. Journal of Immigrant and Minority Health, 11, 7-12.
- JENSEN, C. 2008. Worlds of welfare services and transfers. Journal of European Social Policy, 18, 151-162.
- JERROME, D. & WENGER, G. C. 1999. Stability and change in late-life friendships. Ageing & Society, 19, 661-676.
- JONES, W. H. & LANG, J. R. 1980. Sample Composition Bias and Response Bias in a Mail Survey: A Comparison of Inducement Methods. *Journal of Marketing Research*, 17, 69-76.
- JÜRGES, H. 2005. Cross-Country differences in general health. In: AUTHORS, V. (ed.) Health. ageing and retirement in Europe. First results from the Survey of Health. Ageing and Retirement in Europe. Mannheim: Research Institute for the Economics of Ageing.
- JÜRGES, H. 2008. Self-assessed health, reference levels, and mortality. Applied *Economics*, 40, 569.
- JÜRGES, H. 2009. Healthy Minds In Healthy Bodies: An International Comparison Of Education-Related Inequality In Physical Health Among Older Adults. *Scottish Journal of Political Economy*, 56, 296-320.
- JÜRGES, H. 2010. Health inequalities by education, income and wealth: a comparison of 11 European countries and the US. Applied Economics Letters, 17, 87 91.
- JÜRGES, H., AVENDANO, M. & MACKENBACH, J. 2008. Are different measures of self-rated health comparable? An assessment in five European countries. European Journal of Epidemiology, 23, 773-781.
- JYLHÄ, M., GURALNIK, J. M., FERRUCCI, L., JOKELA, J. & HEIKKINEN, E. 1998 Is self-rated health comparable across cultures and genders?

. J Gerontol B Psychol Sci Soc Sci, 53, S144-52.

- KAHANA, E., REDMOND, C., HILL, G. J., KERCHER, K., KAHANA, B., JOHNSON, J. R. & YOUNG, R. F. 1995. The Effects of Stress, Vulnerability, and Appraisals on the Psychological Well-Being of the Elderly. *Research on Aging*, 17, 459-489.
- KALWIJ, A. & VAN SOEST, A. 2005. Item Response. In: BÖRSCH-SUPAN, A., BRUGIAVINI, A., JÜRGES, H., MACKENBACH, J., SIEGRIST, J. & WEBER, G. (eds.) Health, Ageing and Retirement in Europe. First Results from the Survey of Health, Ageing and Retirement in Europe. Mannheim: Research Institute for the Economics of Aging (MEA).
- KAPLAN, G. A. & CAMACHO, T. 1983. Perceived health and mortality: A nine-year follow-up of the human population laboratory cohort. *American Journal of Epidemiology*, 117, 292-304.
- KAPTEYN, A. D. SHARE: A Survey of Health, Ageing and Retirement in Europe. Conference proceedings on Longitudinal Social and Health Surveys in an International Perspective, 25-27 January 2006 2006 Montreal.
- KASNECI, D. 2007. Active ageing: The EU policy response to the challenge of population ageing. European Papers on the New Welfare. 8, 32-61.
- KASZA, G. J. 2002. The Illusion of Welfare. Journal of Social Policy, 31, 271-287.
- KENDALL, J. 2009. The value of volunteering in Europe in the noughties: what would Beveridge have thought? [Online]. Kent: School of Social policy, Sociology and Social Research - University of Kent. [Accessed Accessed on the 17th December 2009.
- KENT, M. 2011. Volunteering and Health for Aging Populations. *Today's Research on Aging*. National Institute on Aging.
- KENWARD, M. G. & CARPENTER, J. 2007. Multiple imputation: current perspectives. Statistical Methods in Medical Research, 16, 199-218.

- KIM, S. H., WOLDE-TSADIK, G. & REUBEN, D. B. 1997. Predictors of perceived health in hospitalized older persons: a cross-sectional and longitudinal study. *Journal of the American Geriatrics Society*, 45, 420-6.
- KIRSTEN, T. G. J. C., VAN DER WALT, H. J. L. & VILJOEN, C. T. 2009. Health, wellbeing and wellness: An anthropological eco-systemic approach. *Health SA Gesondheid*, 14, 1-7.
- KLEBANOFF, M. A. & COLE, S. R. 2008. Use of Multiple Imputation in the Epidemiologic Literature. Am. J. Epidemiol., kwn071.
- KLEVMARKEN, A. N., SWENSSON, B. & HESSELIUS, P. 2005a. The SHARE Sampling Procedures and Calibrated Design Weights. In: BORSCH-SUPAN, A. & JURGES, H. (eds.) The Survey of Health, Ageing and Retirement in Europe Methodology. Mannheim: Research Institute for the Economics of Ageing (MEA).
- KLEVMARKEN, N. A., SWENSSON, B. & HESSELIUS, P. 2005b. The SHARE Sampling Procedures and Calibrated Design Weights. In: BÖRSCH-SUPAN, A. & JÜRGES, H. (eds.) The Survey of Health, Aging, and Retirement in Europe Methodology. Mannheim: Research Institute for the Economics of Aging (MEA).
- KOHLI, M., HANK, K. & KUNEMUND, H. 2009. The social connectedness of older Europeans: patterns, dynamics and contexts. *Journal of European Social Policy*, 19, 327-340.
- KOMP, K. S. 2011. The political economy of the Third Age. In: CARR, D. C. & KOMP, K. S. (eds.) Gerontology in the era of the third age. New York: Springer.
- KORPI, W. & PALME, J. 1998. The Paradox of Redistribution and Strategies of Equality: Welfare State Institutions, Inequality, and Poverty in the Western Countries. American Sociological Review, 63, 661-687.
- KÜNEMUND, H. & KOLLAND, F. 2007. Work and retirement. In: BOND, J., PEACE, S., DITTMANN-KOHLI, F. & WESTERHOF, G. J. (eds.) Ageing in society. European perspectives on gerontology. 3rd Edition ed. London: SAGE publications,.
- KÜNEMUND, H. & REIN, M. 1999. There is more to receiving than needing: theoretical arguments and empirical explorations of crowding in and crowding out. Ageing & Society, 19, 93-121.
- KUSKOWSKA-WOLK, A., HOLTE, S., OHLANDER, E.-M., Å., B., HOLMBERG, L., ADAMI, H.-O. & BERGSTROM, A. 1992. Effects of different designs and extension of a food frequency questionnaire on response rate. Completeness of data and food frequency responses. *Int J Epidemiol*, 21, 1144-1150.
- KUZUYA, M., ENOKI, H., HASEGAWA, J., IZAWA, S., HIRAKAWA, Y., SHIMOKATA, H. & AKIHISA, I. 2011. Impact of Caregiver Burden on Adverse Health Outcomes in Community-Dwelling Dependent Older Care Recipients. American Journal of Geriatric Psych, 19, 382-391 10.1097/JGP.0b013e3181e9b98d.

LASLETT, P. 1987. The Emergence of the Third Age. Ageing & Society, 7, 133-160.

LASLETT, P. 1989. A Fresh Map of Life: The Emergence of the Third Age, London, Weidenfeld and Nicolson.

- LEE, K. J. & CARLIN, J. B. 2010. Multiple Imputation for Missing Data: Fully Conditional Specification Versus Multivariate Normal Imputation. *American Journal of Epidemiology*, 171, 624-632.
- LEIBFRIED, S. 1993. Towards a European Welfare State? On integrating poverty regimes into the European Community. In: JONES, C. (ed.) New Perspectives on the welfare state in Europa. London and New York,: Routledge.
- LEINONEN, R., HEIKKINEN, E. & JYLHÄ, M. 2001. Predictors of decline in selfassessments of health among older people -- a 5-year longitudinal study. Social Science & Medicine, 52, 1329-1341.
- LEMON, B. W., BENGTSON, V. L. & PETERSON, J. A. 1972. An Exploration of the Activity Theory of Aging: Activity Types and Life Satisfaction Among Inmovers to a Retirement Community. J Gerontol, 27, 511-523.
- LEPKOWSKI, J. & COUPER, M. 2002. Nonresponse in the Second wave of Longitudinal Household Surveys. In: GROVES, R., DILLMAN, D., ELTINGE, J. & LITTLE, R. (eds.) Survey Nonresponse. New York: John Wiley & Sons.
- LESTHAEGHE, R. 2000. Europe's demographic issues: fertility, household formation and replacement migration. *Working paper*, United Nations Expert Group Meeting on Policy Responses to Population Ageing and Population.
- LEWIS, J. 1992. Gender and the Development of Welfare Regimes. Journal of European Social Policy, 2, 159-173.
- LILLARD, L. A. & PANIS, C. W. A. 1998. Panel Attrition from the Panel Study of Income Dynamics: Household Income, Marital Status, and Mortality. *The Journal of Human Resources*, 33, 437-457.
- LITTLE, R. J. & VARTIVARIAN, S. 2003. On weighting the rates in non-response weights. *Statistics in Medicine*, 22, 1589-1599.
- LITTLE, R. J. A. & RUBIN, D. B. 2002. Statistical Analysis With Missing Data, New York, Wiley.
- LITWIN, H. & SHIOVITZ-EZRA, S. 2006. The association between activity and wellbeing in later life: what really matters? Ageing & Society, 26, 225-242.
- LIVI BACCI, M. 1999. Introduzione alla Demografia, Torino, Loescher Editore.
- LIVI BACCI, M. & DE SANTIS, G. 2010. Studenti, accelerate il passo! La rivista 'IL Mulino'. Bologna: Il Mulino.
- LUM, T. Y. & LIGHTFOOT, E. 2005. The Effects of Volunteering on the Physical and Mental Health of Older People. *Research on Aging*, 27, 31-55.
- LYNN, P., BUCK, N., BURTON, J., JÄCKIE, A. & LAURIE, H. 2005. A Review of Methodological Research Pertinent to Longitudinal Survey Design and Data Collection, Colchester, University of Essex.
- MACINNES, J. 2006. WORKLIFE BALANCE IN EUROPE: A RESPONSE TO THE BABY BUST OR REWARD FOR THE BABY BOOMERS? European Societies, 8, 223-249.
- MAIER, H. & SMITH, J. 1999. Psychological Predictors of Mortality in Old Age. The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 54B, P44-P54.
- MANACORDA, M. & MORETTI, E. 2006. Why do most of Italian youths live with their parents? Intergenerational transfers and household structure. *Journal of the European Economic Association*, 4, 800-829.

- MARMOT, M., BANKS, J., BLUNDELL, R., LESSOF, C. & NAZROO, J. 2003a. Health. wealth and lifestyles of the older population in England. The 2002 English longitudinal study of ageing, London, Institute for Fiscal Studies.
- MARMOT, M., BANKS, J., BLUNDELL, R., LESSOF, C. & NAZROO, J. 2003b. Introduction. In: MARMOT, M., BANKS, J., BLUNDELL, R., LESSOF, C. & NAZROO, J. (eds.) Health, wealth and lifestyles of the older population in England: The 2002 English Longitudinal Study of Ageing. London: Institute for Fiscal Studies.
- MCCALLUM, J., SHADBOLT, B. & WANG, D. 1994. Self-rated health and survival: a 7-year follow-up study of Australian elderly. *American Journal of Public Health.* 84, 1100-1105.
- MEIER, S. & STUTZER, A. 2004. Is Volunteering Rewarding in Itself? In: IEW -WORKING PAPERS (ed.). Institute for Empirical Research in Economics: University of Zurich.
- MENDES DE LEON, C. F., GLASS, T. A. & BERKMAN, L. F. 2003. Social Engagement and Disability in a Community Population of Older Adults: The New Haven EPESE. Am. J. Epidemiol., 157, 633-642.
- MENDES DE LEON, C. F., GOLD, D. T., GLASS, T. A., KAPLAN, L. & GEORGE, L. K. 2001. Disability as a Function of Social Networks and Support in Elderly African Americans and Whites: The Duke EPESE 1986-1992. J Gerontol B Psychol Sci Soc Sci, 56, S179-190.
- MENEC, V. H. 2003. The Relation Between Everyday Activities and Successful Aging: A 6-Year Longitudinal Study. J Gerontol B Psychol Sci Soc Sci, 58, S74-82.
- MOEN, P., DEMPSTER-MCCLAIN, D. & WILLIAMS, R. M., JR. 1992. Successful Aging: A Life-Course Perspective on Women's Multiple Roles and Health. *The American Journal of Sociology*, 97, 1612-1638.
- MORENO MÍNGUEZ, A. 2005. The persistence of male breadwinner model in southern European countries in a compared perspective: Familism, employment and family policies [Online]. Available: http://www.mariecurie.org/annals/volume4/soc1.pdf [Accessed 16 April 2010.
- MORROW-HOWELL, N., HINTERLONG, J., ROZARIO, P. A. & TANG, F. 2003. Effects of Volunteering on the Well-Being of Older Adults. The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 58B, S137– S145.
- MUSICK, M. A., HERZOG, A. R. & HOUSE, J. S. 1999. Volunteering and mortality among older adults: findings from a national sample. J Gerontol B Psychol Sci Soc Sci, 54, S173-180.
- MUSICK, M. A. & WILSON, J. 2003. Volunteering and depression: the role of psychological and social resources in different age groups. *Social Science & Medicine*, 56, 259-269.
- MUTCHLER, J. E., BURR, J. A. & CARO, F. G. 2003. From Paid Worker to Volunteer: Leaving the Paid Workforce and Volunteering in Later Life. Social Forces, 81, 1267-1293.
- NABI, H., KIVIMÄKI, M., SUOMINEN, S., KOSKENVUO, M., SINGH-MANOUX, A. & VAHTERA, J. 2010. Does depression predict coronary heart disease and cerebrovascular disease equally well? The Health and Social Support

Prospective Cohort Study. International Journal of Epidemiology, 39, 1016-1024.

- NATALI, D. & RHODES, M. 2004. Trade-offs and Veto Players: Reforming Pensions in France and Italy. *French Politics*, 2, 1-23.
- NAVARRO, V., MUNTANER, C., BORRELL, C., BENACH, J., QUIROGA, A., RODRÍGUEZ-SANZ, M., VERGÉS, N. & PASARÍN, M. I. 2006. Politics and health outcomes. *Lancet* 1033-1037.
- NEUGARTEN, B. L. 1974. Age Groups in American Society and the Rise of the Young-Old. The ANNALS of the American Academy of Political and Social Science, 415, 187-198.
- NEY, S. 2005. Active aging policy in Europe: Between path dependency and path departure. *Ageing International*, 30, 325-342.
- NICOLETTI, C. & PERACCHI, F. 2005. Survey Response and Survey Characteristics: Microlevel Evidence from the European Community Household Panel. Journal of the Royal Statistical Society. Series A (Statistics in Society), 168, 763-781.
- NOCON, M., HIEMANN, T., MÜLLER-RIEMENSCHNEIDER, F., THALAU, F., ROLL, S. & WILLICH, S. N. 2008. Association of physical activity with allcause and cardiovascular mortality: a systematic review and meta-analysis. European Journal of Cardiovascular Prevention & Rehabilitation, 15, 239-246.
- OECD 1999. Classifying Educational Programmes. Manual for ISCED-97 Implementation in OECD Countries, Paris, OECD.
- OECD 2000. Reforms for an Ageing Society, Paris, OECD [Organisation for Economic Co-Operation and Development].
- OECD 2004a. Ageing and Employment Policies Denmark, Paris, OECD [Organisation for Economic Co-Operation and Development].
- OECD 2004b. Ageing and Employment Policies France, Paris, OECD [Organisation for Economic Co-Operation and Development].
- OECD 2004c. Ageing and Employment Policies Italy, Paris, OECD [Organisation for Economic Co-Operation and Development].
- OECD 2004d. Ageing and Employment Policies United Kingdom Paris, OECD [Organisation for Economic Co-Operation and Development].
- OFFICE FOR NATIONAL STATISTICS 2010. Pension Scheme Membership. In: DAFFIN, C. (ed.) Pension Trends.
- OMAN, D., THORESEN, C. E. & MCMAHON, K. 1999. Volunteerism and Mortality among the Community-dwelling Elderly. J Health Psychol, 4, 301-316.
- ÖSTLIN, P. 1989. The 'health-related selection effect' on occupational morbidity rates. Scandinavian Journal of Public Health, 17, 265-270.
- PARSONS, T. 1951. The Social System, London, Routledge & Kegan Paul.
- PAUWELS, L. & SVENSSON, R. 2008. How Serious Is the Problem of Item Nonresponse in Delinquency Scales and Aetiological Variables? *European Journal of Criminology*, 5, 289-308.
- PAVALKO, E. K. & HENDERSON, K. A. 2006. Combining Care Work and Paid Work. Research on Aging, 28, 359-374.
- PEACE, S., DITTMANN-KOHLI, F., WESTERHOF, G. J. & BOND, J. 2007. The ageing World. In: BOND, J., PEACE, S., DITTMANN-KOHLI, F AND

WESTERHOF, G (ed.) Ageing in Society: European Perspectives on Gerontology. 3rd Edition ed. London: SAGE publications.

- PEDHAZUR, E. J. 1997. Multiple regression in behavioral research, Orlando, FL., Harcourt Brace.
- PEDUZZI, P., CONCATO, J., KEMPER, E., HOLFORD, T. R. & FEINSTEIN, A. R. 1996. A simulation study of the number of events per variable in logistic regression analysis. *Journal of Clinical Epidemiology*, 49, 1373-1379.
- PENSION POLICY INSTITUTE. 2004. Tax Relief and Incentives for Pension Saving [Online]. London. Available: http://www.pensionspolicyinstitute.org.uk/uploadeddocuments/PPI\_ACE\_report \_\_\_\_\_\_final.pdf [Accessed 10 February 2011.
- PHELAN, E. A., ANDERSON, L. A., LACROIX, A. Z. & LARSON, E. B. 2004. Older Adults' Views of "Successful Aging"—How Do They Compare with Researchers' Definitions? Journal of the American Geriatrics Society, 52, 211-216.
- PHILLIPSON, C. 2002a. Transitions from work to retirement: Developing a new social contract. *Policy Press.* Bristol.
- PHILLIPSON, C. 2002b. Transitions from Work to Retirement: Developing a New Social Contract. Bristol: The Policy Press.
- PHILLIPSON, C. & BAARS, J. 2007. Social Theory and Social Ageing. In: BOND, J., PEACE, S., DITTMAN-KOHLI, F. & WESTERHOF, G. (eds.) Ageing in Society: European Perspectives on Gerontology. 3rd Edition ed. London: SAGE publications.
- PHILLIPSON, C. & OGG, J. 2010. Active Ageing and Universities: engaging older learners. Universities UK.
- PICHLER, F. & WALLACE, C. 2007. Patterns of Formal and Informal Social Capital in Europe. Eur Social Rev, jcm013.
- PIERSON, P. 2000. Three Worlds of Welfare State Research. Comparative Political Studies, 33, 791-821.
- PINQUART, M. & SÖRENSEN, S. 2003 Associations of stressors and uplifts of caregiving with caregiver burden and depressive mood: a meta-analysis. J Gerontol B Psychol Sci Soc Sci, 58, 112-28.
- PINSONNEAULT, A. & KRAEMER, K. L. 1993. Survey research methodology in management information systems: an assessmen. J. Manage. Inf. Syst., 10, 75-105.
- POWELL, M. & BARRIENTOS, A. 2004. Welfare regimes and the welfare mix. European Journal of Political Research, 43, 83-105.
- PRINCE, M., REISCHIES, F., BEEKMAN, A., FUHRER, R., JONKER, C., KIVELA, S., LAWLOR, B., LOBO, A., MAGNUSSON, H., FICHTER, M., VAN OYEN, H., ROELANDS, M., SKOOG, I., TURRINA, C. & COPELAND, J. 1999. Development of the EURO-D scale--a European, Union initiative to compare symptoms of depression in 14 European centres. *The British Journal of Psychiatry*, 174, 330-338.
- PRIOR, G., DEVERILL, C., MALBUT, K. & PRIMATESTA, P. 2003. Health Survey for England 2001. Methodology & Documentation. In: PRIOR, G., DEVERILL, C., MALBUT, K. & PRIMATESTA, P. (eds.). London: The Stationery Office.

- PYY-MARTIKAINEN, M. & RENDTEL, U. 2008. Assessing the impact of initial nonresponse and attrition in the analysis of unemployment duration with panel surveys. AStA Advances in Statistical Analysis, 92, 297-318.
- RADLOFF, L. S. 1977. The CES-D Scale: A Self-Report Depression Scale for Research in the General Population. *Applied Psychological Measurement*, 1, 385-401.
- RAPP, M. A., GERSTORF, D., HELMCHEN, H. & SMITH, J. 2008. Depression predicts mortality in the young old, but not in the oldest old: results from the Berlin Aging Study. *The American Journal of Geriatric Psychiatry*, 16, 844-52.
- RAY, R. E. 2008. Coming of age in critical gerontology. Journal of Aging Studies, 22, 97-100.
- RAYMO, J. M. & SWEENEY, M. M. 2006a. Work-Family Conflict and Retirement Preferences. J Gerontol B Psychol Sci Soc Sci, 61, S161-169.
- RAYMO, J. M. & SWEENEY, M. M. 2006b. Work-Family Conflict and Retirement Preferences. The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 61, S161-S169.
- REDAY-MULVEY, G. 2008. The Strategy of the Four Pillars in a Long-life Society. European Papers on the new Welfare, 9, 94-103.
- REHER, D. S. 1998. Family ties in Western Europe: persistent contrasts. *Population* and Development Review, 24, 203-234.
- RENDTEL, U. 2002. Attrition in Household Panels: A Survey. The Future of Social Surveys in Europe, 4.
- RICHARDSON, J. & ROBERTSON, I. 1999. Ageing and the Cost of Health Services. In: WORKING PAPER (ed.). Centre for Health Program Evaluation.
- ROBERTSON, A. 1997. Beyond Apocalyptic Demography: Towards a Moral Economy of Interdependence. *Ageing & Society*, 17, 425-446.
- RODIN, J. & MCAVAY, G. 1992. Determinants of Change in Perceived Health in a Longitudinal Study of Older Adults. *Journal of Gerontology*, 47, P373-P384.
- ROMANS, F. 2007. The transition of women and men from work to retirement. Statistics in Focus. Population and Social Conditions [Online], 97. Available: <u>http://epp.eurostat.ec.europa.eu/cache/ITY\_OFFPUB/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097-EN/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-SF-07-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-097/EN/KS-</u>
- ROSENQUIST, J. N., FOWLER, J. H. & CHRISTAKIS, N. A. 2011. Social network determinants of depression. *Mol Psychiatry*, 16, 273-281.
- ROSTGAARD, T. 2002. Caring for Children and Older People in Europe A Comparison of European Policies and Practice. *Policy Studies*, 23, 51 - 68.
- ROSTILA, M. 2007. Social capital and health in European welfare regimes: a multilevel approach. *Journal of European Social Policy*, 17, 223-239.
- ROTH, P. L. 1994. Missing Data: A Conceptual Review for Applied Psychologists. Personnel Psychology, 47, 537-560.
- ROTH, P. L. & BEVIER, C. A. 1998. Response Rates in HRM/OB Survey Research: Norms and Correlates, 1990-1994. *Journal of Management*, 24, 97-117.
- ROWE, J. W. & KAHN, R. L. 1997. Successful aging. Gerontologist, 37, 433-440.
- ROY, H. & RUSSELL, C. 2005. The Encyclopedia of Aging and the Elderly,. Medrounds Publications.
- ROYSTON, P. 2005. Multiple imputation of missing values: update. The Stata Journal, 5, 188-201.

- RUMSFELD, J. S., JONES, P. G., WHOOLEY, M. A., SULLIVAN, M. D., PITT, B., WEINTRAUB, W. S. & SPERTUS, J. A. 2005. Depression predicts mortality and hospitalization in patients with myocardial infarction complicated by heart failure. *American Heart Journal*, 150, 961-967.
- RUSPINI, E. 2004. La ricerca longitudinale, Milano, Franco Angeli.
- SACKER, A., WORTS, D. & MCDONOUGH, P. 2011. Social influences on trajectories of self-rated health: evidence from Britain, Germany, Denmark and the USA. Journal of Epidemiology and Community Health, 65, 130-136.
- SALAMON, L. M. & SOKOLOWSKI, W. 2001. Volunteering in Cross-National Perspective: Evidence From 24 Countries. Working Papers of the Johns Hopkins Comparative Nonprofit Sector Project, 40.
- SALOMON, J. A., TANDON, A. & MURRAY, C. J. L. 2004. Comparability of self rated health: cross sectional multi-country survey using anchoring vignettes. *BMJ*, 328.
- SARACENO, C. 1994. The ambivalent familism of the Italian welfare state. Social Politics, 1, 60-82.
- SAYLOR, C. 2004. The Circle of Health: A Health Definition Model. J Holist Nurs, 22, 97-115.
- SCHISTERMAN, E. F., COLE, S. R. & PLATT, R. W. 2009. Overadjustment bias and unnecessary adjustment in epidemiologic studies. *Epidemiology*, 20, 488-95.
- SCHOLES, S., COX, K. & LESSOF, C. 2008a. Methodology. In: BANKS, J., BREEZE, E., LESSOF, C. & NAZROO, J. (eds.) Living in the 21st century: older people in England. The 2006 English Longitudinal Study of Ageing (Wave 3). London: Institute for Fiscal Studies.
- SCHOLES, S., MEDINA, J., CHESHIRE, H., COX, K., HACKER, E. & LESSOF, C. 2009. Living in the 21st century: older people in England. The 2006 English Longitudinal Study of Ageing. Technical Report (Wave 3), London, National Centre for Social Research.
- SCHOLES, S., TAYLOR, R., CHESHIRE, H., COX, K. & LESSOF, C. 2008b. Retirement, health and relationships of the older population in England: The 2004 English Longitudinal Study of Ageing. Technical Report (wave 2). London: National Centre for Social Research.
- SCHRÖDER, M. 2008. Attrition. In: BÖRSCH-SUPAN, A., BRUGIAVINI, A., JÜRGES, H., KAPTEYN, A., MACKENBACH, J., SIEGRIST, J. & WEBER, G. (eds.) First Results from the Survey of Health, Ageing and Retirement in Europe (2004-2007). Starting the Longitudinal Dimension. Mannheim: Research Institute for the Economics of Aging (MEA).
- SCHULZ, R. & BEACH, S. R. 1999a. Caregiving as a Risk Factor for Mortality. JAMA: The Journal of the American Medical Association, 282, 2215-2219.
- SCHULZ, R. & BEACH, S. R. 1999b. Caregiving as a Risk Factor for Mortality: The Caregiver Health Effects Study. JAMA, 282, 2215-2219.
- SCHULZ, R. & SHERWOOD, P. R. 2008 Physical and Mental Health Effects of Family Caregiving. Am J Nurs, 108, 23-27.
- SCOTT LONG, J. 1997. Regression Models for Categorical and Limited Dependent Variables. Thousand Oaks, CA: SAGE Publications.
- SEN, A. 2002. Health: perception versus observation. BMJ, 324, 860-861.

- SHADISH, W. R., COOK, T. D. & CAMPBELL, D. T. 2002. Experimental and quasiexperimental designs for generalized causal inference, Boston, Houghton Mifflin Company.
- SHARE. The Survey of Health, Ageing, and Retirement in Europe [Online]. http://www.share-project.org. Available: http://www.share-project.org.
- SHER WOOD, P. R., GIVEN, C. W., GIVEN, B. A. & VON EYE, A. 2005. Caregiver Burden and Depressive Symptoms. *Journal of Aging and Health*, 17, 125-147.
- SHOEMAKER, P. J., EICHHOLZ, M. & SKEWES, E. A. 2002. Item Nonresponse: Distinguishing between don't Know and Refuse. *International Journal of Public Opinion Research*, 14, 193-201.
- SIEBER, S. D. 1974. Toward a Theory of Role Accumulation. American Sociological Review, 39, 567-578.
- SIEBERT, U., ROTHENBACHER, D., DANIEL, U. & BRENNER, H. 2001. Demonstration of the healthy worker survivor effect in a cohort of workers in the construction industry. *Occupational and Environmental Medicine*, 58, 774-779.
- SIEGRIST, J. & WAHRENDORF, M. 2009a. Participation in socially productive activities and quality of life in early old age: findings from SHARE. Journal of *European Social Policy*, 19, 317-326.
- SIEGRIST, J. & WAHRENDORF, M. 2009b. Quality of work, health, and retirement. *The Lancet*, 374, 1872-1873.
- SIEGRIST, J., WAHRENDORF, M., VON DEM KNESEBECK, O., JURGES, H. & BORSCH-SUPAN, A. 2007. Quality of work, well-being, and intended early retirement of older employees--baseline results from the SHARE Study. Eur J Public Health, 17, 62-68.
- SINGLETON, R. & STRAITS, B. C. 2005. Approaches to Social Research, New York, Oxford University Press.
- SIRVEN, N. & DEBRAND, T. 2008. Social participation and healthy ageing: An international comparison using SHARE data. Social Science & Medicine, 67, 2017-2026.
- SIVO, S. A., SAUNDERS, C., CHANG, Q. & JIANG, J. J. 2006. How Low Should You Go? Low Response Rates and the Validity of Inference in IS Questionnaire Research. Journal of the Association for Information Systems
- SMITH, C. & NUTBEAM, D. 1990. Assessing non-response bias: a casa study from the 1985 Welsh Heart Health Survey. *Health Education Research. Theory & Practice*, 5, 381-386.
- SMITH, D. B. & MOEN, P. 1998. Spousal Influence on Retirement: His, Her, and Their Perceptions. Journal of Marriage and Family, 60, 734-744.
- SMITH, D. H. 1994. Determinants of Voluntary Association Participation and Volunteering: A Literature Review. Nonprofit and Voluntary Sector Quarterly, 23, 243-263.
- SMITH, D. H. & SHEN, C. 2002. The Roots of Civil Society: A Model of Voluntary Association Prevalence Applied to Data on Larger Contemporary Nations. International Journal of Comparative Sociology, 43, 93-133.
- SOFI, F., CAPALBO, A., CESARI, F., ABBATE, R. & GENSINI, G. F. 2008. Physical activity during leisure time and primary prevention of coronary heart disease: an

updated meta-analysis of cohort studies. European Journal of Cardiovascular Prevention & Rehabilitation, 15, 247-257.

- ST JOHN, P. D. & MONTGOMERY, P. 2009. Do depressive symptoms predict mortality in older people? Aging & Mental Health, 13, 674-681.
- STATA CORP 2009. Stata Statistical Software: Release 11. College Station, TX: StataCorp LP.
- STEFFICK, D. E. 2000. Documentation of Affective Functioning Measures in the Health and Retirement Study. In: CENTER, S. R. (ed.). Ann Arbor, MI: University of Michigan.
- STEVENS, N. A. N. 2001. Combating loneliness: a friendship enrichment programme for older women. Ageing & Society, 21, 183-202.
- STEWART, A. L., GREENFIELD, S., HAYS, R. D., WELLS, K., ROGERS, W. H., BERRY, S. D., MCGLYNN, E. A. & WARE, J. E. 1989. Functional Status and Well-being of Patients With Chronic Conditions. JAMA: The Journal of the American Medical Association, 262, 907-913.
- STRAWBRIDGE, W. J. & WALLHAGEN, M. I. 1999. Self-Rated Health and Mortality Over Three Decades. *Research on Aging*, 21, 402-416.
- STRAWBRIDGE, W. J., WALLHAGEN, M. I. & COHEN, R. D. 2002. Successful Aging and Well-Being: Self-Rated Compared With Rowe and Kahn. Gerontologist, 42, 727-733.
- SZINOVACZ, M. E., DEVINEY, S. & DAVEY, A. 2001. Influences of Family Obligations and Relationships on Retirement: Variations by Gender, Race, and Marital Status. J Gerontol B Psychol Sci Soc Sci, 56, S20-27.
- TAYLOR, R., CONWAY, L., CALDERWOOD, L. & LESSOF, C. 2003. Methodology. In: MARMOT, M., BANKS, J., BLUNDELL, R., LESSOF, C. & NAZROO, J. (eds.) Health, wealth and lifestyles of the older population in England: The 2002 English Longitudinal Study of Ageing. London: Institute for Fiscal Studies.
- TAYLOR, R., CONWAY, L., CALDERWOOD, L., LESSOF, C., CHESHIRE, H., COX, K. & SCHOLES, S. 2007. Health, wealth and lifestyles of the older population in England: The 2002 English Longitudinal Study of Ageing. Technical Report, London, National Institute for Social Research.
- THÉVENON, O. 2011. Family Policies in OECD Countries: A Comparative Analysis. Population and Development Review, 37, 57-87.
- THOITS, P. A. & N., H. L. 2001. Volunteer work and well-being. Journal of Health and Social Behavior, 42, 115-131.
- THOMAS, C., BENZEVAL, M. & STANSFELD, S. A. 2005. Employment transitions and mental health: an analysis from the British household panel survey. *Journal* of Epidemiology and Community Health, 59:, 243-249.
- THOMPSON, L. & WALKER, A. J. 1989. Gender in Families: Women and Men in Marriage, Work, and Parenthood. Journal of Marriage and Family, 51, 845-871.
- TOMASSINI, C., GLASER, K., BROESE VAN GROENOU, M. & GRUNDY, E. 2004a. Living arrangements among older people: an overview of trends in Europe and the USA. *Population Trends*, 115, 24-34.
- TOMASSINI, C., KALOGIROU, S., GRUNDY, E., FOKKEMA, T., MARTIKAINEN, P., BROESE VAN GROENOU, M. & KARISTO, A. 2004b. Contacts between

elderly parents and their children in four European countries: current patterns and future prospects. *European Journal of Ageing*, 1, 54-63.

- TORSHEIM, T., RAVENS-SIEBERER, U., HETLAND, J., VALIMAA, R., DANIELSON, M. & OVERPECK, M. 2006. Cross-national variation of gender differences in adolescent subjective health in Europe and North America. Social Science & Medicine, 62, 815-827.
- TRIFILETTI, R. 1999. Southern European Welfare Regimes and the Worsening Position of Women. Journal of European Social Policy, 9, 49-64.
- TRIVELLATO, U. 1999. Issues in the Design and Analysis of Panel Studies: A Cursory Review. Quality & Cursory (Quantity, 33, 339-351).
- TSAI, S. P., WENDT, J. K., DONNELLY, R. P., DE JONG, G. & AHMED, F. S. 2005. Age at retirement and long term survival of an industrial population: prospective cohort study. *BMJ*, bmj.38586.448704.E0.
- TWISK, J. W. R. 2007. Applied longitudinal data analysis for epidemiology. A practical guide, Cambridge, Cambridge University Press.
- UCHINO, B. N. C., JOHN T.; KIECOLT-GLASER, JANICE K. 1996. The relationship between social support and physiological processes: A review with emphasis on underlying mechanisms and implications for health. *Psychological Bulletin*, 119, 488-531.
- UHRIG, S. C. N. 2008. The nature and causes of attrition in the British Household Panel Survey, Colchester, University of Essex.
- UMBERSON, D. & MONTEZ, J. K. 2010. Social Relationships and Health. Journal of Health and Social Behavior, 51, S54-S66.
- UN 2000. Replacement Migration: Is It a Solution to Declining and Ageing Populations?, New York, United Nations.
- UN 2001. World Population Ageing 1950-2050. In: DESA [DEPARTMENT OF ECONOMIC AND SOCIAL AFFAIRS] (ed.). New York: United Nations Population Division.
- UNESCO UIS. 2006. International standard classification of education ISCED 1997 [Online]. Available:

http://www.uis.unesco.org/TEMPLATE/pdf/isced/ISCED\_A.pdf.

- UNGER, J. B., MCAVAY, G., BRUCE, M. L., BERKMAN, L. & SEEMAN, T. 1999. Variation in the impact of social network characteristics on physical functioning in elderly persons: MacArthur Studies of Successful Aging. J Gerontol B Psychol Sci Soc Sci, 54, S245-251.
- UNIVERSITY OF CALIFORNIA BERKELEY (USA) AND MAX PLANCK INSTITUTE FOR DEMOGRAPHIC RESEARCH (GERMANY). Human Mortality Database [Online]. Available: www.mortality.org.
- UTTL, B. 2005. Measurement of Individual Differences. Psychological Science, 16, 460-467.
- VAN BUUREN, S. 2007. Multiple imputation of discrete and continuous data by fully conditional specification. *Statistical Methods in Medical Research*, 16, 219-242.
- VAN INGEN, E. & DEKKER, P. 2010. Changes in the Determinants of Volunteering: Participation and Time Investment Between 1975 and 2005 in the Netherlands. Nonprofit and Voluntary Sector Quarterly, 40, 682-702.

- VAN WILLIGEN, M. 2000. Differential Benefits of Volunteering Across the Life Course. J Gerontol B Psychol Sci Soc Sci, 55, S308-318.
- VARIOUS AUTHORS 2008a. First Results from the Survey of Health, Ageing and Retirement in Europe (2004-2007). Starting the Longitudinal Dimension, Mannheim, Research Institute for the Economics of Aging (MEA).
- VARIOUS AUTHORS 2008b. Living in the 21st century: older people in England. The 2006 English Longitudinal Study of Ageing (Wave 3), London, Institute for Fiscal Studies.
- VASCONCELOS FERREIRA, L. & FIGUEIREDO, A. 2005. Welfare Regime in the EU 15 and in the Enlarged Europe: An exploratory analysis [Online]. University of Porto, Faculty of Economy. Available: <u>http://www.fep.up.pt/investigacao/workingpapers/workingpapers.htm</u> [Accessed 24 November 2010.
- VERBRUGGE, L. M. & JETTE, A. M. 1994. The disablement process. Social Science & Medicine, 38, 1-14.
- VERDUGO, R. 2006. Workers, workers' productivity and the dependency ratio in Germany: analysis with implications for social policy. *Population Research and Policy Review*, 25, 547-565.
- VICTOR, C. 2005. The Social Context of Ageing A Textbook of Gerontology, Abingdon, Routledge.
- VIITANEN, T. K. 2007. Informal and Formal Care in Europe. IZA Discussion Paper Series.
- VON DEM KNESEBECK, O., WAHRENDORF, M., HYDE, M. & SIEGRIST, J. 2007. Socio-economic position and quality of life among older people in 10 European countries: results of the SHARE study. Ageing & Society, 27, 269-284.
- WAHRENDORF, M. & SIEGRIST, J. 2010. Are changes in productive activities of older people associated with changes in their well-being? Results of a longitudinal European study. European Journal of Ageing, 7, 59-68.
- WAHRENDORF, M., VON DEM KNESEBECK, O. & SIEGRIST, J. 2006. Social productivity and well-being of older people: baseline results from the SHARE study European Journal of Ageing, 3, 67-73.
- WALKER, A. 2001. Towards active ageing in the European Union. *Millennium Project* Workshop - Towards Active Ageing in the 21st Century. Tokio: The Japan Institute of Labour.
- WALKER, A. 2002. A strategy for active ageing. International Social Security Review, 55, 121-139.
- WALKER, A., ANDERSSON, L., LAMURA, G., BOUWMAN, B., KNIPSCHEER,
  K., PEETERS, A., POLVERINI, F., REICHERT, M., WALKER, C. &
  WEIDEKAMP-MAICHER, M. 2005. Growing Older in Europe, Maidenhead,
  Open University Press.
- WALKER, A. & MALTBY, T. 1997. Ageing Europe, Buckingham, Open University Press.
- WANG, L., ZHANG, Z., MCARDLE, J. J. & SALTHOUSE, T. A. 2009. Investigating Ceiling Effects in Longitudinal Data Analysis. *Multivariate Behavioral Research*, 43, 476-496.

- WARBURTON, D. E. R., NICOL, C. W. & BREDIN, S. S. D. 2006. Health benefits of physical activity: the evidence. *Canadian Medical Association Journal*, 174, 801-809.
- WATSON, D. 2003. Sample Attrition between Waves 1 and 5 in the European Community Household Panel. *European Sociological Review*, 19, 361-378.
- WAYMAN, J. C. 2003. Multiple Imputation for Missing Data: What is it and How Can I Use it? Annual Meeting of the American Educational Research Association. Chicago, IL.
- WENGER, G. C. 1997. Review of findings on support networks of older Europeans Journal of Cross-Cultural Gerontology, 12, 1-21.
- WESTERHOF, G. & TULLE, E. 2007. Meanings of Ageing and Old Age: Discursive contexts, social attitudes and personal identities. In: BOND, J., PEACE, S., DITTMAN-KOHLI, F. & WESTERHOF, G. (eds.) Ageing in Society: European Perspectives on Gerontology. 3rd Edition ed. London: SAGE publications.
- WESTERLUND, H., KIVIMÄKI, M., SINGH-MANOUX, A., MELCHIOR, M., FERRIE, J. E., PENTTI, J., JOKELA, M., LEINEWEBER, C., GOLDBERG, M., ZINS, M. & VAHTERA, J. 2009. Self-rated health before and after retirement in France (GAZEL): a cohort study. *The Lancet*, 374, 1889-1896.
- WESTERLUND, H., VAHTERA, J., FERRIE, J. E., SINGH-MANOUX, A., PENTTI, J., MELCHIOR, M., LEINEWEBER, C., JOKELA, M., SIEGRIST, J., GOLDBERG, M., ZINS, M. & KIVIMÄKI, M. 2010. Effect of retirement on major chronic conditions and fatigue: French GAZEL occupational cohort study. BMJ, 341.
- WHO 1980. International Classification of Impairments, Disabilities and Handicaps (ICIDH), Geneva, WHO [World Health Organisation].
- WHO 2001a. Health and aging: A discussion paper, Geneva, Switzerland, WHO [World Health Organisation].
- WHO 2001b. International Classification of Functioning, Disability and Health (ICF), Geneva, WHO.
- WHO 2001c. The World Health Report 2001, Geneva, World Health Organization.
- WHO 2002. Active Ageing: A Policy Framework, Geneva, Switzerland, WHO [World Health Organisation].
- WHO. 2006. Constitution of the World Health Organisation [Online]. Geneva, Switzerland: WHO [World Health Organisation]. Available: <u>http://www.who.int/governance/eb/who\_constitution\_en.pdf</u> [Accessed 29th September 2009.
- WIGGINS, R. D., HIGGS, P. F. D., HYDE, M. & BLANE, D. B. 2004. Quality of life in the third age: key predictors of the CASP-19 measure. Ageing & Society, 24, 693-708.
- WILSON, J. 2000. Volunteering. Annual Review of Sociology, 26, 215-240.
- WILSON, K., ELLIOTT, S. J., EYLES, J. D. & KELLER-OLAMAN, S. J. 2007. Factors affecting change over time in self-reported health. *Canadian Journal of Public Health*, 98, 154-8.
- WOLFF, J. L., STARFIELD, B. & ANDERSON, G. 2002. Prevalence, Expenditures, and Complications of Multiple Chronic Conditions in the Elderly. *Arch Intern Med*, 162, 2269-2276.

- WOLINSKY, F. D. & JOHNSON, R. J. 1992. Perceived Health Status and Mortality Among Older Men and Women. *Journal of Gerontology*, 47, S304-S312.
- YAJIMA, Y., TSUTSUI, T., NAKAJIMA, K., LI, H. Y., TAKIGAWA, T., WANG, D.
   H. & OGINO, K. 2007. The effects of caregiving resources on the incidence of depression over one year in family caregivers of disabled elderly. *Acta Medica Okayama*, 61, 71-80.
- YOUNG, F. W. & GLASGOW, N. 1998. Voluntary Social Participation and Health. Research on Aging, 20, 339-362.
- ZABEL, J. E. 1998. An Analysis of Attrition in the Panel Study of Income Dynamics and the Survey of Income and Program Participation with an Application to a Model of Labor Market Behavior. *The Journal of Human Resources*, 33, 479-506.
- ZAMARRO, G., MEIJER, E. & FERNANDES, M. 2008. Mental health and cognitive ability. In: BÖRSCH-SUPAN, A., BRUGIAVINI, A., JÜRGES, H., KAPTEYN, A., MACKENBACH, J., SIEGRIST, J. & WEBER, G. (eds.) First Results from the Survey of Health, Ageing and Retirement in Europe (2004-2007). Starting the Longitudinal Dimension. Mannheim Research Institute for the Economics of Aging (MEA).
- ZEDLEWSKI, S. R. & BUTRICA, B. 2007. Are We Taking Full Advantage of Older Adults' Potential? *Perspective on Productive Aging*, 9.
- ZUNZUNEGUI, M. V., KONÉ, A., JOHRI, M., BÉLAND, F., WOLFSON, C. & BERGMAN, H. 2004. Social networks and self-rated health in two Frenchspeaking Canadian community dwelling populations over 65. Social Science & Medicine, 58, 2069-2081.
- ZWEIFEL, P., FELDER, S. & MEIERS, M. 1999. Ageing of population and health care expenditure: a red herring? *Health Economics*, 8, 485-496.

## **APPENDIX**

Table 6A.1 Distribution of the highest educational level attained by Italian SHARE respondents by age group and sex (with and without weights); comparison with ISTAT census data

					ISTAT cen	sus data		
				50-54	55-59	60-64	65-69	70+
		lst	at	62.7	70.9	77.5	83.5	84.1
ľ	>	പ	11/	56.3	66.1	73.8	72.1	84.5
	6	RI	W	(46.6; 65.5)	(55.2; 73.9)	(66.4; 80.0)	(63.3; 79.4)	(77.2; 89.7)
	<b>D</b>	HA	TI	52.4	65.4	74.1	77.1	87.9
		S	U	(43.0; 61.7)	(58.0; 72.0)	(67.4; 79.8)	(71.2; 82.1)	(81.8; 92.2)
	-	Ist	at	26.4	20.1	15.1	10.5	9.7
7	n	(4)		33.6	21.8	17.4	17.2	10.4
E	DI	R	W	(25.0; 43.4)	(15.9; 28.2)	(12.5; 23.6)	(11.5; 24.9)	(5.9; 17.7)
Σ	IE	Η		36.3	23.7	16.8	15.4	7.7
	N	S	U	(26.6; 47.2)	(18.0; 30.5)	(12.5; 22.3)	(11.0; 21.3)	(4.3; 13.4)
		Ist	at	10.9	8.9	7.3	6.0	6.2
	H	(LL)	w	11.3	12.1	8.8	10.7	5.1
	Ð	R	w	(6.2; 19.7)	(6.4; 21.3)	(5.8; 13.3)	(5.4; 20.2)	(2.5; 10.3)
	Η	H	11	10.1	10.1	9.1	7.5	4.4
		Ś	U	(5.8; 17.2)	(6.6; 15.1)	(6.0; 13.5)	(4.4; 12.4)	(2.2; 8.4)
				100	100	100	100	100
		let	tat	70.0	78.6	83.0	883	88.7
	~	130		60.0	70.0	74.3	78.1	<u>83.7</u> 97.4
	N N	RE	W	(50.6.68.8)	(64.4:76.5)	(67 3. 78 5)	(67.6.85.9)	$(77 A \cdot 02 5)$
	L L			63.8	73 7	759	85.9	90.5
		S	U	(55.5, 71.3)	(67.7, 78.9)	(70.7:80.5)	(80.5.90.0)	(84.8.94.2)
		Ici	tat	20.5	14.8	11.5	82	7 8
Z	Σ			26.1	16.1	17.6	13.3	10.0
Z	I	RE	W	(19.4: 34.1)	(11.8; 21.7)	(13.1:22.9)	(9.0: 19.1)	(5.4:17.7)
9	Ξ	TH		24.1	15.0	17.5	9.0	7.1
2	Σ	S	U	(18.0; 31.6)	(11.2; 19.7)	(13.3; 22.7)	(6.2: 12.7)	(4.2: 12.0)
		ls	tat	9.5	6.6	4.5	3.5	3.5
]	H	G	w	13.9	13.1	8.0	8.6	2.6
	IG	R	W	(7.6; 24.0)	(8.8; 18.9)	(5.6; 11.0)	(4.3; 16.6)	(0.7; 8.8)
}	H	H	11	12.0	11.3	6.2	5.1	1.8
	L	S		(7.8; 18.3)	(7.8; 16.1)	(4.2; 9.0)	(2.9; 8.9)	(0.6; 4.2)
				100	100	100	100	100

This table compares the age-sex distribution of the education level between census respondents (ISTAT) and SHARE respondents; weighted (W) and unweighted (U) data and 95% CI. Sources: ISTAT, 2004; SHARE 2004.

				50-54	55-59	60-64	65-69
			DST	23.7	28.0	34.7	40.3
İ	>	[*]	W/-!-14-1	13.3	8.6	14.6	18.0
	0	RI	weighted	(8.9; 19.8)	(4.7; 15.1)	(9.2; 22.3)	(11.0; 28.1)
	Ĩ	HA	I I	14.8	7.5	13.9	17.4
		Ś	Unweighted	(10.1; 21.0)	(4.2; 13.1)	(9.1; 20.9)	(10.9; 26.5)
			DST	50.5	46.7	43.6	41.0
7	E E	(-)		52.2	56.1	49.6	47.1
E	E	RE	Weighted	(43.7; 51.7)	(47.3; 63.9)	(40.9; 58.3)	(36.8; 57.7)
Σ	E	НА		53.3	56.8	52.9	50.0
	2	S	Unweighted	(45.7; 60.7)	(48.7; 64.6)	(44.5; 61.2)	(39.9; 60.1)
			DST	25.8	25.2	21.7	18.6
				34.5	35.3	35.8	34.9
	<b>RE</b>		Weighted	(26.8; 42.2)	(27.5; 43.5)	(27.7; 44.7)	(25.4; 45.7)
H		HA	Linusiahaad	30.8	34.9	33.1	32.6
		$\mathbf{S}$	Unweighted	(24.2; 38.1)	(27.6; 43.0)	(25.7; 41.4)	(23.8; 42.8)
				100	100	100	100
			DOT	20.5	26.6	47.4	5( 0
				30.5	36.6	47.4	56.8
	<b>N</b>	E	Weighted	11.2	18.3	27.8	35.6
	9	AF		(7.2; 16.7)	(13.1; 25.1)	(20.6; 30.2)	(20.4; 45.9)
		HS	Unweighted	(96.105)	18.7	28.1	37.2
			L	(8.0; 19.5)	(13.0; 23.2)	(21.2; 30.3)	(28.1; 47.40
Z	Σ			36.9	40.3	35.2	29.4
ME	B	E	Weighted	34.1	(20.2, 42.0)	(27.2, 54.4)	40.1
ō	<b>B</b>	AF.		(20.2, 41.0)	(29.3, 43.9)	(57.2, 54.4)	(30.4; 30.7)
≥	Ī	HS	Unweighted	(265.414)	(31.7, 46.1)	$(37.7 \cdot 54.4)$	(32.0.51.7)
			Der	20.7	22.0	17 4	12.0, 51.7)
				54.7	<u> </u>	26.6	24.2
ļ	GH	E	Weighted	(15 6. 62 0)	(37 8. 52 2)	(195.351)	(161.340)
}	Ĭ			(43.0; 02.0)	126	250	213
Ì		BH	Unweighted	(14 0. 51 0)	(35 5: 50 0)	(193.339)	$(141\cdot 307)$
	I	1	L	100	100	100	100
1				1 100	1 100	1 100	1 100

Table 6A.2 Distribution of the highest educational level attained by Danish SHARE respondents by age and sex (with and without weights); comparisons with DST census data

This table compares the age-sex distribution of the education level between census respondents (DST) and SHARE respondents. Sources: DST, 2002; SHARE 2004.

Table 6A.3 Distribution of the highest educational level attained by French SHARE respondents by age and sex (with and without weights); comparisons with INSEE census data

				50-59	60-64	65+
			INSEE	49.9	57.7	60.6
	>	ત્મ	W. 1.4 1	31.9	50.0	61.0
	0	RI	weighted	(27.0; 36.6)	(42.1; 56.3)	(53.4; 66.8)
	L L	HA	Ilmusichted	32.4	47.8	60.1
		S	Unweighted	(27.7; 37.5)	(41.0; 54.8)	(53.6; 66.2)
	Ţ		INSEE	35.7	29.8	27.9
z	5	L.	Weighted	40.3	27.5	23.6
E	Id	R	weighted	(34.8; 45.2)	(19.4; 36.4)	(19.6; 27.5)
2	JE	H	Unweighted	40.0	28.0	23.1
	4	Ś	Onweighted	(34.6; 46.3)	(20.2; 37.3)	(19.7; 26.9)
			INSEE	14.4	12.5	11.5
	H	E	Weighted	27.8	22.5	15.4
	9	X	weighted	(23.3; 32.3)	(17.0; 28.2)	(11.7; 19.4)
ļ	H	H	Unweighted	26.4	22.6	15.6
		(n)		(22.2; 31.0)	(17.4; 28.8)	(12.0; 20.8)
				100	100	100
<u>}</u>	[		INSEE	43.4	52.2	73.8
		(+)	W/-laber 1	40.7	49.4	77.0
	6		weighted	(34.8; 45.7)	(41.0; 56.9)	(71.4; 81.1)
	<b>A</b>	H	Unweighted	41.2	48.4	75.4
ļ		S	Unweighted	(36.1; 46.6)	(40.7; 56.2)	(69.9; 80.1)
7	T		INSEE	37.6	33.5	18.9
	5	E	Weighted	35.5	27.1	15.4
N	ā	E E	weighted	(31.7; 38.5)	(21.1; 33.2)	(13.1; 17.9)
Ă	E	H	Unweighted	35.0	26.9	16.0
		02		(31.9; 38.2)	(21.4; 33.3)	(13.8; 18.5)
ļ			INSEE	18.9	14.3	7.3
	H		Weighted	23.8	23.5	7.6
	1 E	AR		(19.9; 27.6)	(15.0; 34.3)	(4.6; 12.3)
1	<b>=</b>	H	Unweighted	22.4	23.8	8.1
	<u> </u>			(18.6; 26.8)	(15.6; 34.8)	(4.9; 13.1)
				100	100	100

This table compares the age-sex distribution of the education level between census respondents (INSEE) and SHARE respondents. Sources: INSEE, 2004; SHARE 2004.

				census			
				50-54	55-59	60-64	65-69
	-		ONS	44.8	51.0	57.3	61.0
	2		Weightad	25.7	33.6	43.3	52.6
	6	SA	weighted	(22.9; 28.7)	(30.7; 36.6)	(39.9; 46.8)	(49.1; 56.1)
		E	Unweighted	24.7	32.4	42.6	51.6
			Unweighted	(21.9; 27.5)	(29.6; 35.3)	(39.2; 46.1)	(48.1; 55.0)
	-		ONS	17.8	15.1	12.5	13.0
	ŝ	T	Weishaad	30.6	28.9	21.4	21.7
	ā	S	weighted	(27.6; 33.7)	(26.1; 31.8)	(18.6; 24.3)	(19.0; 24.7)
	Ξ		t Turnet - Land	30.1	29.2	21.4	22.0
Z	2		Unweighted	(27.2; 33.1)	(26.4; 32.0)	(18.6; 24.2)	(19.2; 24.9)
ME			ONS	20.6	16.9	14.7	13.2
	E			38.8	32.6	31.6	20.1
	Ū	₹S	Weighted	(35.6; 42.1)	(29.7; 35.5)	(28.5; 34.9)	(17.5; 23.0)
	H	۲.	Thursday Internal	40.3	33.6	32.4	20.8
		-	Unweighted	(37.1; 43.5)	(30.7; 36.5)	(29.1; 35.7)	(18.0; 23.6)
		•••••	ONS	16.8	17.0	15.4	12.8
	ER		W-l-hand	4.9	5.0	3.7	5.6
THE		S	weighted	(3.7; 6.6)	(3.8; 6.6)	(2.5; 5.3)	(4.2; 7.5)
	5	3	Linusiahaad	4.8	4.8	3.6	5.6
δ			Unweighted	(3.4; 6.2)	(3.4; 6.0)	(2.3; 4.9)	(4.0; 7.2)
				100	100	100	100
			ONS	52.4	59.4	67.0	69.6
}		<u> </u>		32.8	38.6	46.7	53.0
ł	2		Weighted	(30.0: 35.7)	(35.8; 41.5)	(43.3; 50.0)	(49.7: 56.3)
	L I	ILS I		31.6	38.0	46.1	52.0
	}		Unweighted	(28.9; 34.2)	(35.2; 40.8)	(42.8; 49.3)	(48.8; 55.2)
			ONS	19.4	15.8	12.1	11.8
Į	M			33.6	29.1	22.6	19.0
[	E	<b>X</b>	Weighted	(30.8; 36.5)	(26.5; 31.8)	(19.9; 25.5)	(16.6; 21.7)
Z	E	E E		33.6	20.2 (26.6.21.9)	22.8	19.4
<b>IE</b>	Σ		Unweighted	(30.9; 36.3)	29.2 (20.0; 31.8)	(20.0; 25.5)	(16.7; 21.9)
6		1	ONS	18.8	15.2	13.3	11.8
Š	H		W/_:-1-41	24.2	20.5	19.5	15.4
I	9	<	weighted	(21.7; 26.8)	(18.3; 22.9)	(17.0; 22.2)	(13.2; 17.9)
}	E	S	I laure - Land	25.5	20.9	19.8	16.1
		Ē	Unweighted	(23.0; 28.0)	(18.6; 23.3)	(17.1; 22.4)	(13.7; 18.4)
1	<u> </u>		ONS	9.4	9.7	7.6	6.8
	<b>R</b>		Waiahtad	9.5	11.8	11.3	12.6
1	H	S	weightea	(7.9; 11.4)	(10.1; 13.8)	(9.4; 13.6)	(10.5; 14.9)
	5	E	I laure about	9.2	11.8	11.3	12.5
]				(7.6; 10.9)	(9.9; 13.6)	(9.2; 13.5)	(10.4; 14.7)
				100	100	100	100

Table 6A.4 Distribution of the highest educational level attained by English ELSA respondents by age and sex (with and without weights); comparisons with ONS

This table compares the age-sex distribution of the education level between census respondents (ONS) and ELSA respondents. Sources: ONS, 2001; ELSA 2001.

			-		-							
		, Aris Ya	Very	bad	В	ad	F	air	G	ood		ery
			INSEE	SHARE	INSEE	SHARE	INSEE	SHARE	INSEE	SHARE	INSEE	SHARE
	-	45-59	0.7	1.3	4.1	5.6	17.6	18.6	49.9	50.7	27.7	23.8
E	Mer	60-74	1.3	1.5	5.4	6.5	31.1	31.4	48.7	49.5	13.5	111
NC	-	75+	2.6	4.0	14.2	20.7	46.1	38.2	31.9	33.0	5.2	4.1
[YA]	-	45-59	0.5	1.5	3.6	4.5	19.5	18.8	50.6	553	25.0	10.0
FI	ome	60-74	1.2	1.6	5.8	6.1	33.5	30.7	47.6	50.7	110	19.9
Tot	M	75+	2.9	5.2	14.1	9.9	49.0	44.1	28.9	37.1	51	37
	i tebli io rep	eners () prodiser	ISTAT	SHARE	ISTAT	SHARE	ISTAT	SHARE	ISTAT	SHARE	ISTAT	SHARE .
		45-54	0.5	1.2	2.6	2.6	29.3	25.0	53.0	53.4	14.5	17.7
	len	55-64	0.9	1.9	5.2	5.7	43.4	27.5	42.1	52.4	8.4	12.5
×	Z	65-74	2.1	2.7	9.1	10.8	54.9	42.7	30.0	35.6	3.9	7.9
T		75+	5.4	4.1	17.4	19.6	57.9	40.7	17.2	30.7	2.1	4.8
ITA	ITALY	45-54	0.6	0.3	4.1	2.7	38.2	32.0	47.8	49.9	9.3	15.1
419	me	55-64	1.3	1.2	7.4	7.7	51.6	37.2	34.4	46.0	5.3	7.9
	Wo	65-74	2.3	1.9	14.5	14.5	58.4	47.9	22.4	31.0	2.4	4.7
		75+	6.7	4.5	23.9	23.6	55.3	47.0	12.6	23.9	1.6	1.0
				Not (	Good	_	Fair	ly good		(	Good	
			Cer	ISUS	ELS	SA	Census	EL	SA	Census	E	LSA
		50-54	11	.5	5.2	2	25.3	16	5.2	63.2		78.6
1	-	55-59	15	.3	7.4	+	27.6	18	8.7	57.1		73.8
	Mer	00-04	19	.5	8.7	7	31.3	23	.1	49.3	(	58.2
0	-	05-09	1/	.4	8.4		37.1	24	.4	45.4	(	57.2
Z		70-74	19	.6	7.4		41.1	27	.6	39.3	(	54.9
LA		757	20	.8	10.	9	42.9	31	.7	30.3		57.2
SN		50-54	12	.6	4.6	,	28.9	17	.5	58.5	-	77.9
E	H	55-59	15	.3	6.9		31.3	21	.5	53.4	1	71.6
	me	60-64	15	.4	5.9		35.7	21	.8	48.9	1	72.2
	W	65-69	17	.0	4.9		40.0	22	.9	43.0	7	72.2
		70-74	20	.5	8.9		43.5	30	.6	36.0	(	50.4
		75+	29	.2	9.4		43.8	21	1	27.0	5	0 1

## Table 6A.5 Distribution of Self-Reported Health among SHARE and ELSA respondents and other nationally representative sources, by age-group, sex and country

This table compares the age-sex distribution of the self-assessed health between INSEE "Day Life and<br/>Health" Survey and French SHARE respondents, between ISTAT "Health conditions and access to health<br/>services" survey respondents and Italian SHARE respondents and between the English 2001 census and<br/>ELSA respondents. Sources: INSEE, 2007; ISTAT, 2005; SHARE 2004; ONS 2001; ELSA 2002.

		- Dy S	U.A.					
	DEN	MARK	FRA	NCE	ITA	LY	ENG	LAND
Mobility, Arm Function & fine motor limitations	М	W	М	W	М	w	М	W
walking 100 metres	6.6	9.9	9.4	11.9	8.8	16.5	11.8	13.2
sitting for about two hours	7.5	12.6	6.7	10.0	8.0	14.3	12.5	14.9
getting up from a chair	12.7	19.0	12.2	18.7	11.9	23.3	21.8	29.0
climbing + flights of stairs	15.8	26.9	22.9	35.3	24.3	39.5	28.7	41.9
climbing 1 flight of stairs	7.9	11.3	8.7	12.1	10.7	20.0	12.2	17.4
stooping/kneeling/crouching	16.9	27.5	20.6	34.0	24.9	39.9	30.9	39.3
extending your arms	4.2	7.8	6.3	12.1	5.0	12.0	9.0	12.7
pushing large objects	5.2	12.4	10.1	19.9	7.0	15.8	12.9	22.6
lifting or carrying weights	7.9	25.76	10.0	28.2	12.3	32.7	16.7	33.8
picking up a small coin	2.0	2.7	3.0	3.3	2.8	7.1	4.4	5.8
Total (N)	756	857	1.362	1.670	1.125	.1382	5,182	6,195

Table 6.A6 Percentage of respondents reporting difficulty with each mobility item, by sex

This table shows the percentage of people in each country (Denmark, France, Italy and England) per sex who reported having problems with each of the stated activity. Sources: ELSA, 2002; SHARE, 2004. Weighted data.

Table 6.A7 Percentages of respondents declaring difficulty with ADLs, by sex an	ıd
country	

			•• J					
	DENN	1ARK	FRA	NCE	ITA	LY	ENGI	LAND
Activities of daily living (ADL)	M	W	M	W	Μ	W	Μ	W
dressing	7.4	5.4	9.8	8.0	7.2	9.9	14.2	12.2
walking across a room	9.8	2.4	1.5	1.7	2.0	3.6	2.9	3.8
bathing or showering	3.3	6.1	5.8	7.6	6.0	9.5	10.3	14.1
eating	1.7	1.9	1.5	1.7	1.1	1.7	1.7	2.0
getting in or out of bed	3.4	3.8	2.5	2.7	3.0	4.6	6.2	6.9
using the toilet	1.5	1.9	1.3	2.3	1.5	3.6	3.3	3.8
Total (N)	756	857	1,363	1,670	1,125	,1382	5,182	6,195

This table shows the percentages of older women and men reporting problems with each of the listed activities in each country (Denmark, France, Italy and England). Sources: ELSA, 2002; SHARE, 2004. Weighted data

		coun	uu y					
	DENN	MARK	FRA	NCE	ITA	LY	ENG	LAND
Instrumental ADLs	M	W	M	W	Μ	W	Μ	W
using a map	2.9	8.6	6.3	11.3	6.2	15.0	2.8	7.8
preparing a hot meal	4.5	5.5	3.7	3.5	3.1	4.0	4.0	5.0
shopping for groceries	4.2	9.9	6.5	9.7	4.1	10.0	6.7	11.9
making telephone calls	1.6	1.7	2.6	2.8	1.8	3.7	2.5	2.5
taking medications	2.2	2.2	2.1	2.8	2.2	3.2	1.5	1.4
doing work around the house	8.8	16.0	7.1	12.8	5.3	11.3	13.2	17.3
managing money	2.6	4.2	3.8	5.7	3.8	7.2	2.1	2.2
	756	957	1 262	1 670	1 1 25	1397	5 182	6 105

Table 6.A8 Percentages of respondents declaring difficulty with IADLs, by sex and

 Total (N)
 756
 857
 1,363
 1,670
 1,125
 ,1382
 5,182
 6,195

 This table shows the percentages of older women and men reporting problems with each of the listed activities in each country (Denmark, France, Italy and England). Sources: ELSA, 2002; SHARE, 2004. Weighted data

## Table 6.A9 Engagement in civic and cultural activities (%) by sex and age group –England

					Age		
i	Activities in the last year	50-59	60- 69	70-79	80+	Total (N)	Eligible Total N
	Voted*	80.4	83.3	89.0	88.9	<b>83.8</b> (3,939)	4,674
	Read a daily newspaper+	65.0	75.0	78.4	74.3	<b>71.6</b> (3,370)	4,674
	Have a hobby+	81.8	82.0	75.4	62.3	7 <b>8.9</b> (3,694)	4,674
	Holiday in UK	59.9	60.4	55.6		<b>57.5</b> (2,705)	4,674
	Holiday abroad	56.3	50.5	37.1	19.5	<b>47.5</b> (2,226)	4,674
<b>n</b>	Daytrip	76.8	69.5	58.6	45.8	<b>68.2</b> (3,177)	4,674
ž	Go to the cinema 🗸	42.5	28.0	14.9	6.6	<b>29.5</b> (1,457)	5,104
	Eat out of the house ✓	91.4	87.4	84.6	78.7	87.7 (4,483)	5,104
	Go to the theatre/opera ✓	45.8	41.6	35.4	21.4	<b>40.2</b> (2,070)	5,104
	Visit a museum ✓	41.9	40.6	31.4	16.3	<b>37.0</b> (1,902)	5,104
	TOTAL	1,893	1,585	1,156	470		
<u>├</u> ───	Voted*	77.4	85.1	83.9	80.7	81.5 (4,568)	5,600
	Read a daily newspaper+	63.6	72.2	68.8	69.0	<b>67.9</b> (3,793)	5,600
	Have a hobby+	79.8	81.2	73.3	63.6	<b>76.9</b> (4,348)	5,600
	Holiday in UK	61.1	62.9	54.0	43.9	<b>57.9</b> (3,278)	5,600
5	Holiday abroad	57.8	51.1	33.9	18.9	45.7 (2,603)	5,600
l B	Daytrip	76.7	73.2	62.0	53.5	<b>69.5</b> (3,926)	5,600
×	Go to the cinema 🗸	45.1	34.4	18.1	9.9	31.1 (2,397)	6,115
	Eat out of the house ✓	91.4	91.1	83.8	73.7	87.1 (5,367)	6,114
1	Go to the theatre/opera ✓	55.1	54.2	40.1	26.8	47.5 (3,044)	6,114
	Visit a museum ✓	42.4	43.2	30.7	14.5	36.1 (2,285)	6,114
	TOTAL	2,237	1,767	1,374	737		<u> </u>

	Denmark		Fr	France		taly	England	
	Male	Female	Male	Female	Male	Female	Male	Female
Other member(s) in	26.7	30.0	40.2	41.2	43.9	38.2	39.4	36.8
HH not in paid work	(154)	(168)	(372)	(422)	(411)	(464)	(1,463)	(1,490)
Other member(s) in	41.9	31.5	36.9	24.6	13.1	11.1	39.7	33.2
HH in paid work	(228)	(182)	(345)	(248)	(103)	(123)	(1,365)	(1,262)
Living alone	18.8	27.9	13.3	21.3	13.5	15.2	14.5	17.6
Living alone	(101)	(147)	(119)	(219)	(53)	(94)	(494)	(774)
Only 50+ Member	3.9	3.8	5.0	8.9	20.2	23.5	2.8	6.7
in the Household	(19)	(21)	(48)	(93)	(139)	(189)	(92)	(279)
With other members	8.6	6.8	4.6	4.0	9.3	11.9	3.5	5.6
who refused to answer	(43)	(37)	(43)	(41)	(69)	(126)	(112)	(231)
Total Observations (N)	545	555	927	1,021	775	996	3,526	4,036

Table 7.A1 Distribution of the economic status of the members in the household(%) by sex and country

 Table 7.A2 Participation in Formal Activities – Fully adjusted Logistic

 Regressions, by country and sex. Selected results obtained when employment status included the category 'recently retired'

		Not in paid work <2 years	In paid work	N
	Mala	0.72	0.54 **	<b>E</b> 20
Donmault	Iviale	(0.38–1.36)	(0.31-0.95)	552
Denmark	Famala	0.98	0.47 ***	E 40
	remaie	(0.53-1.81)	(0.27–0.81)	548
	Male	1.04	0.74	051
<b>D</b>	Maie	(0.68–1.61)	(0.45–1.21)	851
France	Female	0.57 **	0.57 ***	001
	remale	· (0.360.89)	(0.38–0.86)	981
		0.92	0.84	=()
T. L	Male	(0.36-2.32)	(0.40–1.77)	/03
παίγ	17	2.59 ***	0.86	000
	remale	(1.38-4.89)	(0.48–1.52)	988
	Mal	0.69 **	0.60 ***	2 200
England	Male	(0.48-0.98)	(0.46-0.78)	3,309
	Trans I.	0.73	0.70 ***	3 5 4 5
	remale	(0.52 - 1.03)	(0.57-0.88)	3,547

Sources: SHARE, 2004; ELSA, 2002. The reference category is "The person has never been in paid work or has not been in paid work for more than 2 years"

Table 9A.1 Proportion of wave 2 respondents who self-rated their health as better, about the same or worse compared to their wave 1 interview, by sex and country -SHARE respondents only

SHARE respondents only					
Denmark Wave 2		France Wave 2		Italy Wave 2	
7.1	8.0	7.1	7.1	3.5	4.6
(29)	(33)	(40)	(48)	(18)	(32)
69.3	65.6	77.5	71.3	66.9	60.5
(284)	(271)	(438)	(483)	(349)	(425)
23.6	26.4	15.4	21.6	29.6	34.9
(97)	(109)	(87)	(146)	(155)	(245)
410	413	565	677	522	702
	Den W: Male 7.1 (29) 69.3 (284) 23.6 (97) 410	DenmarkWave 2MaleFemale7.18.0(29)(33)69.365.6(284)(271)23.626.4(97)(109)410413	DenmarkFrWave 2WateMaleFemaleMale7.18.07.1(29)(33)(40)69.365.677.5(284)(271)(438)23.626.415.4(97)(109)(87)410413565	DenmarkFranceWave 2Wave 2MaleFemaleMaleFemale7.18.07.17.1(29)(33)(40)(48)69.365.677.571.3(284)(271)(438)(483)23.626.415.421.6(97)(109)(87)(146)410413565677	$\begin{tabular}{ c c c c c c c } \hline Denmark & France & In \\ \hline Wave 2 & Wave 2 & Wave \\ \hline Male & Female & Male & Female & Male \\ \hline \hline $7.1$ & $8.0$ & $7.1$ & $7.1$ & $3.5$ \\ \hline $(29)$ & $(33)$ & $(40)$ & $(48)$ & $(18)$ \\ \hline $69.3$ & $65.6$ & $77.5$ & $71.3$ & $66.9$ \\ \hline $(284)$ & $(271)$ & $(438)$ & $(483)$ & $(349)$ \\ \hline $23.6$ & $26.4$ & $15.4$ & $21.6$ & $29.6$ \\ \hline $(97)$ & $(109)$ & $(87)$ & $(146)$ & $(155)$ \\ \hline $410$ & $413$ & $565$ & $677$ & $522$ \\ \hline \end{tabular}$

Sources: SHARE, 2006. Weighted Data.